WORKPLACE RACE EQUALITY IN UK HIGHER EDUCATION:

AN EXPLORATION OF THE DIFFERENTIAL OUTCOMES FOR BLACK STAFF

Submitted to the University of Hertfordshire in partial fulfilment of the requirements of the degree of Doctorate of Business Administration (DBA).

By Herminia Rodriguez July 2021

Abstract

For more than a decade consistent attention has been drawn to the under-representation of ethnic minority staff in UK higher education. This attention has focused on the failure of institutions to represent their increasingly diverse student populations through their academic faculty, whilst statistical reports and published research continue to expose the negative workplace experiences of ethnic minority academics. In parallel, ethnic minority colleagues in professional and support roles and their lived experiences of UK higher education have remained largely in the shadows yet mirror that of their academic counterparts.

Ethnic minority staff do not share the same lived experiences and despite representing more than a fifth of all ethnic minority staff in UK institutions, black staff experience the least favourable outcomes compared to all other ethnic groups. Black staff are least likely to be represented at senior levels, are more likely to be employed on fixed-term contracts and are paid less than their peers. This dissertation provides a unique, in-depth, multi-layered exploration into the lived experiences of black staff in UK higher education and how these contribute to the differential workplace outcomes evidenced through workforce statistics. The study demonstrates that there are multi-level structural and agential factors that influence the way black staff navigate white hegemonic institutional spaces, that evade or deny talk of race or racism, creating psychological and ethnic penalties for black staff that are materially different to staff of any other ethnic group.

The research takes a qualitative, social constructionist approach by employing one-to-one semi-structured interviews with black staff in academic and professional and support roles, equality and diversity and HR practitioners, senior managers in UK institutions and sector agencies. The dissertation applies Bourdieu's concepts of habitus, field, capital, and symbolic violence to interpret participants feelings, perceptions and experiences of racism and institutional racism in UK higher education. It also offers a perspective as a practitioner-researcher and proposes recommendations to inform policy and practice in the field of organisational equality and diversity, to advance race equality in UK higher education institutions and create good diversity practice across any industry or sector.

Glossary of terms

Athena SWAN Advance HE gender equality charter
BAME Black Asian and Minority Ethnic
BBC British Broadcasting Corporation

BME Black and Minority Ethnic
BRC Biomedical Research Centre
BRU Biomedical Research Unit

CIPD Chartered Institute of Personnel and Development

CPR Critical Philosophy of Race
CRE Commission for Racial Equality

CRED Commission on Race and Ethnic Disparities

CRT Critical Race Theory
ECU Equality Challenge Unit

EDI Equality, diversity and inclusion

EHRC Equality and Human Rights Commission

HEA Higher Education Academy

HEEON Higher Education Equal Opportunities Network
HEFCE Higher Education Funding Council for England

HERAG Higher Education Race Action Group
HESA Higher Education Statistics Agency

HR Human Resources

LFHE Leadership Foundation for Higher Education

NHS National Health Service
OfS Office for Students

ONS Office of National Statistics
PSED Public Sector Equality Duty

REC Advance HE Race Equality Charter

TFEU Treaty on the Functioning of the European Union UCEA Universities and Colleges Employers Association

UCU University and College Union

UK United Kingdom
UN United Nations

UNESCO United Nations Educational, Scientific and Cultural Organisation

UUK Universities UK

US United States of America

List of tables

Table 1	Research map (adapted from Layder, 1993)	87
Table 2	Demographic profile of study participants	93
Table 3	UK national staff by age group and ethnicity	124
Table 4	UK academic staff by professorial category and ethnic group	125
Table 5	UK academic staff by mode, salary range and ethnic group	127
Table 6	UK professional and support staff by mode, salary range and ethnic group	127
Table 7	Median salaries of UK staff by activity and ethnic group	128
Table 8	Median/mean salary and pay gap by country of institution, activity and ethnicity	129
Table 9	Median/mean salary and pay gap for staff by gender and ethnicity	129
Table 10	UK academic staff by leaving status and ethnicity	132

Table of Contents

ΑB	STRAC	Т	2
GL	OSSAR	Y OF TERMS	3
LIS	T OF T	ABLES	4
AC	KNOW	LEDGEMENTS	7
1	INTE	RODUCTION	8
	1.1	Introduction	8
	1.2	RATIONALE FOR THE RESEARCH	12
	1.3	AIMS OF THE DISSERTATION	13
	1.4	AN OUTLINE OF THE DISSERTATION.	14
2	THE	EVOLUTION OF RACE, ETHNICITY, AND RACISM IN THE UK	17
	2.1	Introduction	17
	2.2	EXPLORING THE NOTION OF RACE AND ETHNICITY THROUGH HISTORY	19
	2.3	ETHNICITY AS THE NEW RACE	25
	2.4	RACE TALK — THE LANGUAGE OF RACE AND ETHNICITY	29
	2.5	RACIALISED IDENTITY AND THE IMPACT ON BELONGING	32
	2.6	POLITICISING RACE IN MODERN UK SOCIETY	37
	2.7	THE HIDDEN NATURE OF CONTEMPORARY RACISM	40
	2.8	EXPLORING INSTITUTIONAL RACISM	45
	2.9	CONCLUSION	51
3	STR	UCTURE, AGENCY AND THE ROLE OF POWER	54
	3.1	Introduction	54
	3.2.	SOCIAL STRUCTURES AND THEIR INFLUENCE ON HUMAN AGENCY	
	3.3	Using Bourdieu as a lens for exploration	
	3.3.		
	3.3.2		
	3.3.3		
	3.3.4	4 Knowledge is power	67
	3.3.5	Symbolic power as a means of domination	69
	3.4	CONCLUSION	72
4	MET	HODOLOGY AND METHODS	74
	4.1	Introduction	74
	4.2	METHODOLOGY – TAKING A SOCIAL CONSTRUCTIONIST APPROACH	
	4.3	CRITICAL RACE THEORY — THE STARTING POINT	
	4.3.		
	4.4	RESEARCH DESIGN	
	4.5	GAINING ACCESS	90
	4.6	DATA COLLECTION	92
	4.7	DATA ANALYSIS	96
	4.8	CONCLUSION	98
5	wo	RKPLACE EQUALITY AND DIVERSITY PRACTICE IN UK HIGHER EDUCATION	99
	5.1	Introduction	99
	5.2	CONTEXTUALISING EQUALITY AND DIVERSITY	100
	5.3	EQUALITY AND DIVERSITY IN UK HIGHER EDUCATION	
	5.4	THE ROLE OF THE EQUALITY AND DIVERSITY PRACTITIONER AS CHANGE AGENT	

5.5	LOCATING RACE AMID A BROADER EQUALITY AGENDA IN UK HIGHER EDUCATION	119
5.6	THE BASELINE OF ETHNIC DIVERSITY IN THE UK HIGHER EDUCATION WORKFORCE	123
5.7	Addressing race inequalities in UK higher education	134
5.8	Conclusion	139
6 тн	E OPPORTUNITIES AND CHALLENGES OF FITTING IN	143
6.1	Introduction	143
6.2	PERCEPTIONS ON WORKPLACE DIVERSITY	145
6.3	RACIALISED LANGUAGE AND TERMINOLOGY	150
6.4	THE STRUGGLES OF WORKPLACE IDENTITY AND BELONGING	160
6.5	Conclusion	173
7 TH	E LIVED EXPERIENCES OF RACE AND RACISM	178
7.1	Introduction	178
7.2	THE INVISIBILITY OF WORKPLACE RACISM	180
7.3	PERCEPTIONS AND EXPERIENCES OF WORKPLACE RACISM	193
7.4	PERCEPTIONS OF INSTITUTIONAL RACISM	201
7.5	Conclusion	212
8 CC	NCLUSION: THE DIFFERENTIAL OUTCOMES FOR BLACK STAFF	215
8.1 IN	TRODUCTION	215
8.2	IMPLICATIONS OF THE RESEARCH ON THE UK HIGHER EDUCATION SECTOR	218
8.2	.1 Understand the strategic drivers to advance race equality and how this has progressed ove 218	r time
8.2	.2 Consider staff perceptions of the manifestation of racism in the workplace	222
8.2	.3 Determine the impact of racial inequalities on black staff in UK higher education	224
8.2	.4 Consider perceptions of the existence of institutional racism and its effect within a higher	
ed	ucation context	227
8.2	.5 Explore the presence of an ethnic penalty faced by black staff in UK higher education	231
8.3	SITUATING THE SELF AS PRACTITIONER-RESEARCHER	233
8.4	Conclusion	235
REFEREI	NCES	238
APPEND	IX 1 OFFICE FOR STUDENTS (OFS) EQUALITY OBJECTIVES 2018-2022	260
APPEND	IX 2 ADVANCE HE RACE EQUALITY CHARTER PRINCIPLES	261
APPEND	IX 3 CALL FOR EXPRESSION OF INTEREST	262
APPEND	IX 4 INTERVIEW SCHEDULE FOR BLACK STAFF	263
APPEND	IX 5 INTERVIEW SCHEDULE FOR PRACTITIONERS/MANAGERS	265

Acknowledgements

This research journey would not have been possible without the support of so many people who have encouraged me along the way. This is the product of a frustrated practitioner who has borne witness to workplace inequities over a career that spans over twenty years at the point of writing this. I have hoped to bring good advice and a listening ear to those who didn't have anyone else to turn to. I hope that through my commitment to others that I have brought hope, that we've cocreated solutions, and more importantly, that through listening I have empowered those to stay focused on the game and to believe in themselves.

This is for you.

My supervisors, Dr Moira Calveley, Dr Cynthia Forson and Dr Kathlyn Wilson, have been instrumental in making this research happen. Thank you for listening, directing and for prodding me when I slowed down, and had lost motivation. You believed in me and for that I will be forever grateful. You made me push my boundaries and stay focused on bringing to light a topic that is important to us all. Hopefully, this research will contribute to the way ahead.

This research has put me into contact with so many amazing people, and I would like to extend my gratitude to those members of staff across the institutions and organisations that trusted me to listen without prejudice to their experiences and opinions. I hope that I have used your contributions in the way you had hoped and that they really tell your story. Thank you all for trusting me, your honesty and openness.

Finally, I would like to express appreciation to my wife, Erika. Thank you for putting up with me when it got challenging. Thank you for listening to me talk about this research endlessly, even when it did not make sense. You have encouraged me through the whole journey and never once thought I could not do this. I would also like to thank Meg for discussing the complexities of race inequalities over cups of tea and bacon sandwiches, and Barney, George, Alfie, and Lola who have listened intently when I needed to vocalise my thinking whilst providing me with endless furry cuddles – you helped more than you'll ever realise!

1 Introduction

1.1 Introduction

The spotlight on race has been well overdue in the UK and finally arrived in 2020 following the death of George Floyd, who was killed by a white police officer in the US (BBC, 2020). In April 2021, that police officer, Derek Chauvin, was found guilty of two counts of second- and third-degree murder and one count of second-degree manslaughter (New York Times, 2021). This horrific incident spurred the largest civil rights protests in decades drawing once again much-needed attention on racial injustice across the globe and through the Black Lives Matter campaign that would ensue.

Domestically, there were mass protests in cities across the country, including the destruction of Edward Colston's statue, a 17th century Bristol slave trader, following an anti-racism demonstration in June 2020 (Wall for The Observer, 2020). In response a school in Bristol, named and funded by Colston, was recently 'rebranded' (Wright for Daily Mail, 2020) to disassociate itself from its past. Before these protests few people would have known who Edward Colston was. Sir Winston Churchill, whose statue was daubed with 'was a racist' during the London protests in summer 2020, caused the Imperial War Museum to review Churchill's legacy. The National Maritime Museum in Greenwich are reviewing the colonial legacy of Lord Horatio Nelson, and The National Trust reported that 93 of its properties are said to have links to colonialism or slavery (Davies for Daily Mail, 2020). In September 2020, Ofcom received 24,500 complaints from the public following a performance from dance troupe, 'Diversity', on Britain's Got Talent that featured events of 2020, including the Black Lives Matter protests. ITV were later cleared of any breach of broadcasting rules following an investigation by the regulator (Waterson for The Guardian, 2020).

In July 2020, UK universities attracted negative attention for their 'tokenistic' support for Black Lives Matter (Batty for The Guardian, 2020) following an open letter signed by more than 300 academics, students, and professional and support staff, who work/have worked in UK higher education institutions, calling on sector leaders to eliminate racism in higher education (Times Higher, 2020). The open letter states that "there is no clear incentive for universities to take racial justice seriously or consequences if they do not. However, there are disadvantages to

us: black and minority ethnic staff leave, are not promoted and experience racism but these facts do not lead to the institution radically improving its practice". This comes almost a year after UK research councils revealed that white researchers are almost 59 per cent more likely to receive research funding than their minority ethnic peers (Murugesu for New Scientist, 2019). Following a recent televised documentary, the chair of Universities UK, Professor David Richardson, acknowledged that the sector is institutionally racist (Mohdin for The Guardian, 2021). This was set in the context of the student experience and failed to mention how this might also be applicable to the staff experience. This research explores the perceptions of staff across the sector and how they believe institutional racism manifests itself in our institutions.

In March 2021, the Commission for Race and Ethnic Disparities (CRED) released their much-awaited report, which has subsequently been met with public criticism (BBC, 2021; The Conversation, 2021) for its claims that ethnic minority communities could help themselves through their own agency, relabelling the slave period as the 'Caribbean experience', which has transformed how culturally African people have transformed themselves into a remodelled African Britain, as well as suggesting that misapplying the term racism has diluted its credibility, among other things. The UN have condemned this report and have called on the UK government to reject the findings from it and that the CRED should be disbanded or reconstituted to prioritise an authentic and rigorous examination of race (OHCHR, 2021). The backlash since this report's launch demonstrates that race is still a big deal in the UK. It is within this context of heightened awareness of racism and inequality, that the findings of this research are presented. This research exposes that racism and racial inequality exists in UK higher education despite the conclusions presented by CRED.

This relevant, important, and timely research focuses primarily on black staff in UK higher education and is distinctive because it takes into consideration the lived experiences of black academics and their black professional and support colleagues. Exploring race and racism in UK institutions highlights the extent of the problem black staff face. The empirical data provides a voice to the individuals who participated in this study, where so often these narratives are silenced or unheard and which conveyed the full spectrum of emotion, including anger, frustration, pain, and exhaustion. These lived experiences are critical to understanding the nature and impact of systemic racism, so that we might be better equipped to reduce disadvantage and inequality, not solely in UK universities, but within other organisations.

From a total of 391,045 staff employed in the UK higher education sector there are 51,325 staff who have declared a minority ethnic background¹ (Advance HE, 2020). From this staff cohort 21.4 percent are black, representing the second largest staff cohort from within the minority ethnic staff communities. It might be assumed that black staff would be represented across all levels of institutional life, except they are not, with only 0.6 percent of professors with a declared black ethnic background.

This research seeks to understand what lies behind these workforce data and consider the experiences of black staff in UK higher education institutions, who statistically experience poorer outcomes than any other ethnic group in the workplace. In doing so, this exploration looks to identify the opportunities available, and challenges faced by senior leaders, human resources and equality and diversity practitioners to develop strategies that can create more inclusive workplace cultures to better attract, recruit, retain and develop black staff at all levels.

The study draws upon previously published literature, desk research (Chapter Five) and qualitative research methods that enable participants' perceptions and experiences to be analysed in the context of social and organisational structures to understand whether there is a correlation between these and the workplace demographics. This is achieved by taking a multi-dimensional approach that considers a macro, meso and micro level analysis that explores influences such as history and power (Layder, 1993), across institutional structures and participant perceptions and lived experiences. The elements of history and power are relevant in terms of the way that institutions have and continue to function, considering their own structural development through time, combined with the past and continuing experiences of black staff. Together, these may influence the way that black staff are able to navigate through white-dominated spaces, systems, and structures.

I will draw upon my own experiences as an equality and diversity practitioner over the past two decades, including in UK higher education institutions, and offer a unique insight to this research study. In using my professional experience to inform this dissertation, I have researched and reflected on my own past and current management practice as I continue to lead and manage a team of diversity specialists within the public sector. In exploring this

 $^{^{\}mathrm{1}}$ This includes those who identify as black, Asian, Chinese, mixed heritage and any other ethnic background.

institutionally sensitive topic, drawing upon unique empirical data gathered during this research, this dissertation is an original contribution to our knowledge and understanding of race and racism, and advances the practice of business administration in the space of organisational race equality. I approach this important topic from a practitioner-researcher perspective, by exploring the role that equality and diversity practitioners like myself play in assisting institutions to address race equality, and I consider how organisational structures may help or hinder a practitioner's progress in effecting change. This role can be integral to advancing equality, diversity, and inclusion within institutions, however a practitioner's personal and professional context must be considered to understand the extent to which a practitioner is capable of influencing change within institutions. The influences of history and power (Layder, 1993) will be explored in the same way as other participant experiences being analysed in this research.

In parallel, these experiences will be considered through the broader application of habitus, field, capital, and symbolic violence (Bourdieu, 1977; 1984; 1990; 1991; 1998; Bourdieu and Wacquant, 1992). Like history and power, a person's background, upbringing, work, and life experience have a considerable role to play in shaping their experiences of organisational systems, structures, and processes. How these might contribute to the outcomes of black staff are highlighted in this research. This approach considers the relationality of actors involved with advancing race equality in UK higher education in terms of the interdependence, intersubjectivity, and interactivity of individual and organisational phenomena (Tatli and Özbilgin, 2009). Participants may both influence the structural and operational environments, as well as become impacted by those same structures and operations. Individually, the research participants possessed perspectives as people of colour, as practitioners or as senior managers, which are not mutually exclusive, in relation to some or all of which can be further influenced by an individual's culture, experience and field of practice.

The research participants hold a combination of these roles within the sector, making their perspectives multi-dimensional. The research also includes the perspectives of practitioners from different ethnic backgrounds, highlighting other issues associated with the advancement of race equality in the sector. A practitioner's own ethnic identity may also impact upon their own competence to address matters of race. In setting the context of this dissertation, the

following section will provide the rationale for undertaking this research and will identify the influences that prompted it.

1.2 Rationale for the research

The topic of workplace outcomes for different social groups has become a growing area of interest for me as an equality and diversity practitioner. I have been involved in the field of equality and diversity in a range of roles, and first gained an interest in social justice as a trade union representative in the 1990s, campaigning with other colleagues across the country for employment protection based on sexual orientation. As a gay woman that had at the time been outed at work, this was very important to me. Being outed at work without my knowledge left me feeling powerless. That part of my identity had been hijacked and this made me feel vulnerable, fearful and open to public scrutiny. I had not appreciated at the start of this research journey how important personal identity is. I had forgotten what it felt like to have that attribute used as a weapon against me. This aspect would emerge unexpectedly in this research through accounts of the lived experiences of participants.

Subsequently, I have worked across several industries and across all areas of equality, diversity and inclusion. Leading teams within two institutions and working independently in another that were a combination of Russell Group and Post-1992 universities. My interest became fuelled by increased activity across UK higher education around the attainment gap between ethnic minority and white students, whilst also dealing with an increasing number of workforce-related concerns that prompted reflection on the absence of attention to staff outcomes. My interactions with people of colour, not exclusive to UK higher education, that sought my advice about their workplace experiences reminded me of how I felt, and although I am not a woman of colour, the core aspect of identity resonates strongly. Whilst working in UK higher education I began to monitor workforce demographics more closely. Following years of workforce analysis, the differential outcomes for black staff in terms of their representation across those institutions, and despite institutional type and geographical location, followed similar patterns. This raised several areas for enquiry in terms of the interconnectivity of institutional culture across autonomous organisations and how this affects the representation of black staff and their experiences.

I witnessed as a practitioner the challenges faced by ethnic minority staff coming to terms with and dealing with race-related matters through informally and formally reported perceptions of bullying, harassment, and discrimination. I attempted to tackle those perceptions through the development of appropriate policy, practice and initiatives that could improve workplace culture. However, this was often set within the context of resistance from senior leadership and human resources apathy or low prioritisation against broader institutional goals. This research offers an insight into the world of institutional equality and diversity practice from my own perspective as an equality and diversity practitioner. This often-unheard lived experience will add a level of context to the challenges of tackling discrimination and advancing equality and diversity practice in UK universities because these roles, if they exist at all as specialist teams, are often under-resourced; situated within wider human resources teams, which can create conflicts of interest if challenging policy or practice; lack senior sponsorship; or lack adequate prioritisation. Together, these act as the perfect storm for glacial progress in achieving race equality.

1.3 Aims of the dissertation

The aim of the dissertation is to explore the lived experiences of black staff working in UK higher education and how those experiences create differential outcomes within the institutional workplace. Statistical evidence of differential outcomes for black staff have been reported for more than a decade through higher education sector data (Advance HE, 2020), yet there has been little improvement during this time. To explore these phenomena, the research has considered several key features to determine how the actions or omissions of institutions contribute to the perceptions and experiences of black staff, their underrepresentation across senior levels in institutions, their higher rate of attrition from institutions and their negative workplace experiences. These outcomes have all been highlighted in varying degrees by studies for more than a decade (Carter, et al, 1999; UCEA, 2003; Institute of Employment Studies, 2005; Jones, 2006; ECU, 2009, 2011; Pilkington, 2011, 2013; Philips, 2012; University and College Union, 2016, 2017; Rollock, 2011, 2012, 2019; Bhopal, 2018; EHRC, 2019).

There has been growing sectoral and academic interest towards the experiences of diverse minority ethnic groups in the higher education sector and this dissertation will seek to explore

the associations between the differential outcomes reported through workforce data and the perceptions from black staff concerning their lived experiences of institutional life. This exploration will be supported by the perceptions and opinions of agents working in the sector that are involved in advancing race equality within institutions and agencies to explore why differential outcomes for black staff persist despite equality and diversity being an explicit organisational commitment for UK higher education institutions.

To investigate the elements mentioned above the research objectives are to:

- Understand the strategic drivers to advance race equality and how this has progressed over time
- 2. Consider staff perceptions of the manifestation of racism in the workplace
- 3. Determine the impact of racial inequalities on black staff in UK higher education
- 4. Consider perceptions of the existence of institutional racism and its effect within a higher education context
- 5. Explore the presence of an ethnic penalty faced by black staff in UK higher education.

1.4 An outline of the dissertation

This dissertation has eight chapters, with this chapter providing the context of the research topic and outlining the aims and objectives of the study and structure of the dissertation.

In Chapter Two, the dissertation conceptualises race and ethnicity and the evolution of racism in the UK, by defining the terms and considering the origins of race from a historic, anthropologic, economic, and political perspective. The chapter continues to explore the evolution of language to describe background in terms of race and ethnicity and how terminology has confused and problematised discussions pertaining to race and how this has impacted individuals and organisations. Chapter Three explores how Bourdieu's concepts presents a distinct interpretation on human interactions and interdependencies within a workplace setting. These concepts will encourage greater consideration of the multi-dimensional aspects of structure and agency, and how an individual's history can impact the

way agents interpret and navigate the workplace, providing insight into potential solutions to tackle workplace racial inequalities. Chapter Four outlines the methodology and research methods employed in undertaking the study. This chapter provides an overview of the research journey and describes how access was gained to diverse staff members across a broad and geographically dispersed workforce, together with the routes taken to engage with practitioners from external agencies with an interest in UK higher education. Crucially, the research has been approached using a social constructionist perspective that interprets the data gathered through in-depth interviews and analysis of secondary sectoral quantitative workforce data.

Chapter Five discusses the evolution of workplace equality and diversity as an organisational instrument to future proof the workforce. The chapter considers how concepts have changed over time particularly within a national context, how the UK higher education sector has approached equality and diversity and the role of equality and diversity practitioners as agents for change. There is a focus on race within the broader equality and diversity agenda in UK higher education and the chapter concludes by examining the higher education workforce profile, providing an overall discussion on sectoral demographic profile, levels of pay and status and attrition when compared to white staff.

Chapter Six provides the initial analysis and interpretation from interviews with participants to explore whether staff had considered the diversity of the UK higher education sector before joining the workforce and if an ethnically diverse workplace is important to an individual's sense of identity and feeling of belonging. The chapter continues to consider the language of race within the workplace and whether terminology helps or hinders individuals and/or organisations to initiate and maintain conversations about race in the workplace. In Chapter Seven the data is analysed and interpreted further to explore the identification of workplace racism and considers the varied experiences of black staff in the context of the shape-shifting form of racism, how considering its altered state staff are able to navigate white social spaces and the coping strategies employed to function in this environment. The chapter concludes with an exploration into perceptions of the existence of institutional racism within the UK higher education sector, how this is manifested within institutions and the impact on institutional capability to deal with matters of race. Chapter Eight brings together the combined academic literature, primary data and analyses from previous chapters and

considers the research questions of this study. The chapter will propose recommendations for the UK higher education sector to address the findings in this research study to tackle workplace race inequalities across institutions and UK sector agencies.

The evolution of race, ethnicity, and racism in the UK

2.1 Introduction

This chapter will explore the historical and contemporary evolution of the concepts of race and ethnicity through an evaluation of literature considering sociological, anthropological, philosophical, psychological, and political approaches. The discussion will consider how the notion of race and ethnicity has changed over time, the influences that have made those changes occur and reflect on contemporary opinion about race. As well as charting the course of race and ethnicity through history, the chapter will also consider the evolution of racism and how this has altered over time from 'traditional' overt acts or speech to a modern form that is barely recognisable, and which involves subtle, concealed acts or omissions.

Much of the material reviewed for this chapter originates in the United States, where significant contributions have been made over the decades towards developing, and subsequently challenging, the concepts of race and racism across a variety of disciplines. Latterly there has been growing academic interest and discussion in the UK as race-related debate has advanced. This review will provide a framework to understand the contemporary position in and around our societal and individual conceptualisation of race and racism, and the prevailing change in social attitudes around these topics. To provide this foundation, the origins of the term 'race', the context of its use, and how this discourse has developed over time to shape the language of 'ethnicity' will be explored. The terms 'race' and 'ethnicity' are often used interchangeably by equality and diversity practitioners.

It is important to note that there has been a reliance on existing literature whose origins are in the United States (US), of which some dates to the 1960s, which raises some interesting considerations on the challenge this poses for this UK-based study. Firstly, in terms of the civil rights movement in the US and its genesis, which is different to the UK context of race relations, albeit the clear link between Britain's role in building the foundations for enslavement. The fundamental difference is that the British established slave colonies while Americans (US) live alongside the descendants of their slaves. The proportion of minority racial groups are much smaller in the UK in comparison to the US, however there is an added dimension of class in UK society.

There are other important differences such as the history of violence towards African Americans by white Americans, including at the hands of the police (see Chapter One). Lynchings still take place and there are geographic areas that African Americans avoid for fear of attack with no reprisal from the law, posing a threat to both psychological and physical safety.

The advancements made in US legislation through constitutional amendments in the 19th century were certainly more advanced than the UK's own race relations legislation a century later. According to Ferner et al (2005) although diversity has historically been poorly studied internationally, it has been of particular interest in the US. They propose that this is because diversity has very clear roots to the American policy agenda, such as Equal Employment Opportunity and Affirmative Action, agendas which are driven by legislative programmes requiring compliance. Additionally, Ferner et al (2005) suggest that the US legal systems do not tend to deliver equality unless they are underpinned by social mechanisms such as the inability of groups to enforce their legal rights. This is in contrast to the UK, where Trade Unions and collective bargaining are important elements of social regulation that support legal frameworks of equality and help build and enforce legal rights. Despite these socio-political differences, there are similarities in the employment context and lived experiences of people of colour in the workforce irrespective of distinct protections.

The approach to this chapter is founded in the personal stance that race as a concept has been socially constructed to categorise people according to biological criterion, predominantly based on skin colour and to reinforce a power dynamic. There is general acceptance amongst a range of academics that the notion of race has no scientific validity (Montague, 1972; Modood et al, 2002; Smith, 2002, Helms et al, 2005; Bernasconi, 2001; Kandola, 2018), and this has long been a dominant view in British social science and virtually undisputed today (Modood et al, 2002).

The legislative framework in the UK identifies race as a protected characteristic under the Equality Act 2010, however this is not new terminology and has been maintained from previous anti-discrimination legislation. The term race is one that is used often to describe an individual's national background and in practice, this can create tension, particularly when the term is used to refer to racialised minorities in the UK, rather than encapsulate all individuals' ethnic background. From a personal and practical perspective, it is common for individuals to

conflate race or ethnicity with ethnic minority rather than encapsulating any ethnic background. Anecdotally, there is no resistance to working with the term race as it has been well established within UK law, policy, and practice. However, it should be noted that the terminology is more prone to challenge amongst social scientists and in academia in general, than in other workplace environments.

In the construction of race, society has created methods upon which to oppress those with minority group status and this chapter will explore how the politicisation of race has contributed to the manifestations of racial prejudice. There will be an exploration of individual (micro-level) and organisational (meso and macro-level) racial prejudice to expose the evolution of racism from traditional or overt acts to liberal camouflaged or symbolic acts. In leading the discussion to this conclusion, the following section will investigate the notion of race and ethnicity and will be followed by an account of how history and societal advancement have shaped what we understand race and ethnicity to mean today. The subsequent sections will explore how the notion of race has advanced to broaden the spectrum of our traditional understanding, how the language around race has evolved together with notions of identity and how this can affect an individual's feeling of belonging. The latter sections of the chapter will also explore how race has become politicised over time and how geopolitical aspects have informed attitudes and public policy and will conclude to consider how racism has evolved in modern times, including an exploration of institutional racism.

2.2 Exploring the notion of race and ethnicity through history

The contemporary definition of race is given as "a group of people sharing the same culture, history, language, etc.", whereas ethnicity is defined as "the fact or state of belonging to a social group that has a common national or cultural tradition" (Oxford English Dictionary). Both these terms in practice are used interchangeably, however there are notable differences, both in their history as well as in their reception, and these aspects will be explored further. The term 'ethnic' is related to or is a characteristic of a human group with certain key features in common and is derived from the Greek *ethnos*, meaning a (non-Greek) race or people. From a sociological perspective, Winant (2000) defined race as a concept that signifies and symbolises socio-political conflicts and interests in reference to different types of human bodies. A study about ethnic minorities in Britain (Modood et al, 1997) employed a definition

of 'ethnic group' which included the physical appearance of individuals, arguing that there are important boundaries separating one ethnic group from another and supported the idea that membership can be ascribed to individuals by sociologists rather than being elected by the individuals concerned. Smith (2002) however questioned whether the definition used within the study is any more meaningful or any less racist and identified an interesting contemporary issue about the use of language, and whether too much emphasis is placed on terminology rather than the inequalities themselves.

The legal framework has provided a fixed definition of race in the UK for several decades as including colour; nationality; ethnic or national origins, and classes a racial group as 'a group of persons defined by reference to race' (Equality Act 2010). Prior to the implementation of the Equality Act 2010, a significant case that went before the House of Lords in the early 1980s debated the definition of a racial and/or ethnic group (*Mandla v Dowell-Lee* [1982]). The case itself was related to a person's religion, at a time when there was no such protection from discrimination on the grounds of religion or belief as there is today. The Court was tasked with defining which religious groups would be protected as particular racial groups, and concluded that the essential conditions for the constitution of an ethnic group were:

- a long-shared history of which the group is conscious as distinguishing it from other groups and the memory of which it keeps alive
- a cultural tradition of its own including family and social manners, often but not necessarily associated with religious observance
- a common, however distant, geographical origin
- a common language and literature.

Given that there are multiple definitions of race and ethnicity, and that these definitions bare great similarities, it is no wonder that there is so much confusion behind these terms, particularly amongst lay people, and more importantly, which term in today's society is the most appropriate to use. What we can glean from the combination of all these definitions is that race and ethnicity is broad enough to encapsulate a person's physical attributes, which may include colour and physical characteristics, language, and culture. The way our understanding has developed around these concepts, and principally constructing race as non-western, must, to some degree, contribute towards implying the exclusion of white

ethnic groups. This is despite references to colour, nationality, ethnic or national origin, which are applicable to anyone regardless of race or ethnicity. The contention surrounding the term race in practical terms sends people into a deep dark place where there is a real fear (Rollock, 2012) and where people close ranks (Cole, 2009) and this perhaps has contributed to the interchangeable use of the terminology, as well as the continuing academic debate demonstrated by the critical observations and responses (Modood et al, 2002).

The notion of race was developed by historical events, with its origins in the expansion of Europe from the 15th century and especially during the rise of the British Empire (Mason, 2000). The term race was seemingly first recorded in the English language in 1508 in a poem by William Dunbar in reference to clan or kin, and further literal references to race or racial difference were used in Shakespearean texts as a reflection of contemporary Elizabethan society (Hendricks, 2000). Up to the 18th century race was a literary work, lacking scientific attachment, and denoting a class of persons or things (Husband, 1982).

Imperial expansion across Africa, India and the Americas increased the exposure to peoples of different populations and societies. With this came an increase in accounts from narrators of the time, such as those chronicled by Samuel Purchas in the 17th century, and his interpretation of the story of Noah and the Curse of Ham, identifying the inferiority of the descendants of Ham (Africans) (Samson, 2005; Olusoga, 2017). The most striking physical characteristic of people of 'foreign' origin was their colour and this was an aspect of emphasis to underline difference. The contrast between black and white became greatly politicised through the subjugation of newly conquered peoples and the exploitation of their land and resources. To justify these acts of domination, harmful and derogatory attitudes were formed that would form the basis for the perception of the opposites between good and evil; 'white' representing 'good, purity and virginity' and 'black' the colour of 'death, evil and debasement' (Jordan, 1974).

The precursor to what we understand race to be today first originated in 18th century zoology, when Carolus Linnaeus, a white Swedish botanist, zoologist, and physician, included Homo sapiens within the taxonomy of apes and primates in his *Systemae Naturae* (1740). He took an essentialist approach to classification by identifying four main subgroups of Homo sapiens: Europaeus albus (white Europeans), Americans rebescens (Native Americans), Asiaticus fuscus (yellow Asians), and Africanus niger (black Africans) (Graves, 2001). As well as categorising

humans, he also developed behavioural characteristics that were considered essential to the category (Caspari, 2003), e.g., Homo sapiens europaeus was described as active, acute, and adventurous, whereas the descriptors used for the other subgroups lacked such positivity. This early stage of human categorisation should be set within the context of its time, when western colonisation of the Americas, India and Africa was at its peak. The bias toward white European groups provided the scientific justification for the domination of these other ethnic groups. Furthermore, enslaved Africans in North America were categorised according to seven cultural-geopolitical regions, creating a hierarchy based on ancestral groups who were thought to make the best slaves or whose ways most resembled Europeans (Kendi, 2017). Kendi (2017) argues that making hierarchies within the African kingdom can be termed ethnic racism, while making hierarchies pitting Europeans over all Africans was simply racism.

The classification of human beings would continue throughout much of the 18th and 19th centuries and science became the vehicle to confirm and reaffirm the notion of inferiority of non-western peoples (Encyclopaedia Britannica, 2021). In parallel, social Darwinism, an application of the theory of natural selection to social, political, and economic issues, began to emerge. Consequently, this concept was promoted by 19th century English philosopher Herbert Spencer, who was one of the principal proponents of evolutionary theory during the Victorian era, to suggest that white Europeans were a superior race to others, and therefore destined to rule over anyone that was not in this ethnic category (Littlefield et al, 1982).

The classification of humans into the categories of colour we are more familiar with today: black, brown, yellow, red, and white were developed by Johann Friedrich Blumenbach (Barkan, 1992; Bhopal, 2007). Blumenbach did not attach a hierarchy to his colour coding of people. The construction of race continued to evolve through political and economic expansion, and throughout the 19th century the development of polygenism, that humans have multiple origins, widened the perception of distinctions among races, and physical differences were correlated with culture and social status. Regardless of conceptual differences, monogenists (a belief that humans descended from a mating pair) and polygenists alike agreed on the natural superiority of whites and the inherent inferiority of blacks - the opposing poles of global racial hierarchy (Harrison, 1995; Kandola, 2018). With this, the ideology of race took root within the natural sciences through a sub-discipline of physical

anthropology, later becoming a proxy for the study of race, providing it with legitimacy and which academia contributed to making more respectable (Bernasconi, 2001; Caspari, 2003).

Robert Knox claimed that "Race is everything: literature, science, art - in a word, civilization, depends on it" (Knox, 1850). For the Victorians, race was a description of social distinctions rather than colour difference and by the late 19th century the English social elite now measured non-whites against the same benchmark (Malik, 1996). Racial concepts continued to develop throughout the Victorian period as something that identified with inheritance or descent (Lorimer, 1996), a concept that today might be interpreted as socio-economic background. Scientists, and later some philosophers of the time, believed there to be a link between anatomical features and mental and psychological traits (Lorimer, 1996). This latter aspect would cause significant controversy in the 20th century with a study, *The Bell Curve*, which claimed racial differences in levels of intelligence and how they are influenced by environmental and inherited factors (Hernstein and Murray, 1994).

At the beginning of the 20th century there was a notable shift in opposition towards long-held beliefs about racial inferiority and scientific racism. German-born anthropologist, Franz Boas, recognised as the father of American anthropology, would be a major contributor to modern thinking about race. During the turn of the century colour difference would not be sufficient to explain racial variety, resulting in studies of other physical characteristics, such as head and nose shape, and hair and eye colour through anthropometry and craniometry. With proponent views towards evolution, Boas' studies of skeletal anatomy showed that cranial shape and size depended on environmental factors, such as health and nutrition, in contrast to the theory of the day that head shape was a racial trait. Boas introduced culture as a primary concept of anthropology and opposed Carleton Coon's concept that suggested there were five major races of human that evolved in parallel from Homo erectus, and that these 'races' evolved at different times and rates. Moore (2009) suggests that the implication here is that the racial group that evolved the slowest would be at the bottom of the social hierarchy.

Boas has been recognised as developing the American concept of culture, which would in time purge race from social science (Visweswaran, 1998). This has been seen as an important and significant contribution by anthropologists to move away from its origins in the study of race, to the study of a variety of sub-divisions that we recognise in contemporary anthropology that includes social, cultural, linguistic, and biological or physical disciplines. Visweswaran (1998)

notes that Boas equated blood to racial inferiority and that a large black population would disappear with the sufficient infusion of 'white blood', a view that Boas is better known for disputing. He further suggests that in the attempt to rid social science of race by assigning it to biology, Boas and his students helped legitimate the scientific study of race, thereby fuelling the machine of scientific racism (Visweswaran, 1998). This is contrary to Gossett's (1997) view that Boas did more to combat race prejudice than any other person in history.

An aspect of race history that cannot be omitted is the impact of slavery on the preservation of negative attitudes and actions towards people of a very specific racial group, black Africans. Slavery became common practice since ancient Greece, Aristotle believed that slavery was natural because some people were irrational and others were not, therefore, to enslave 'barbarians' was proper (Samson, 2005). Slavery in ancient times was unrelated to skin colour and was widespread across the Mediterranean. Most of those who were enslaved were of European origin, and despite there being some African slaves at the time, slavery had not been racialised (Zuberi, 2013). British trade routes with West Africa began in the late 16th century, and to the English who set foot on those shores, the most notable characteristic of the African was their colour. Travellers of the time rarely failed to comment on this (Jordan, 1982). Following early interactions with Africans, stereotypical views about their culture, habits and character began to develop. It seems customs of the past would recur in this period, with Africans becoming viewed as the new barbarians. This is similar to the way that 17th century English would describe the Irish, as savages and incapable of being civilised, during attempts to conquer their lands and use them as forced labour (Encyclopaedia Britannica, 2021).

The slave trade developed throughout the 17th century and involved the trading of manufactured goods from Europe to the west coast of Africa in exchange for captured people provided by African traders (Olusoga, 2017). The ships, laden with their human cargo would then set off for the Americas or the Caribbean in exchange for sugar, rum, tobacco, and other luxury items back to Europe (British Library). Accounts from explorers and traders became popular reading material for the general public and in the period 1770-1860 caricatured writing occupied a crucial role in racialist thought. A notable philosopher of the time, David Hume, wrote in 1771 in Gentleman's Magazine, an influential periodical of the day, that "the Negroes were naturally inferior to Whites", and in 1788 within the same publication, the Negro [sic] was cast once again as peculiarly sexual, musical, stupid, indolent, untrustworthy,

and violent (Walvin, 1982). The individual characterisation of African slaves from the experiences of West Indian planters formulated characteristics as universal qualities that were possessed by all blacks wherever they might be, and this served to justify and defend the slave trade further amongst the masses of the era.

In addition to the characterisation of Africans in this way, their 'traders' viewed their customs and practices as uncivilised, particularly in relation to their lack of religion. By debasing and reinforcing the inferiority of black Africans through the interpretations of biblical texts, philosopher Charles de Montesquieu suggested that one could not imagine that God, who is a very wise being, would have placed a soul, especially a good soul, in a body entirely black (Samson, 2005). According to Jordan (1982) it was important for the English Christians in west Africa to differentiate between and convert the heathens and bring them into the Christian fold. It was the instrument of religious cleansing that would support the machine of slavery throughout this period and once again provide justification for the enslavement of black African peoples during this time. The slave trade was eventually abolished through an Act of Parliament in 1807, however slavery would persist in the colonies until 1838 although abolitionists continued to campaign internationally beyond this date.

History has shown that the process of racialising the human species has been weaponised to assert power over and control populations around the world for the benefit of white European interests. It is then not surprising that the social construction of race during these tumultuous times has become debated in modern times and across multiple scientific fields and the next section will explore this evolution of modified thought further.

2.3 Ethnicity as the new race

Race as a term has been tarnished through time by negative stereotyping, scientific racism, and in more modern times, the Holocaust. One U.S. study (Littlefield et al, 1982) drew attention to the contradictions amongst anthropologists in their acceptance of the concept of race, showing during the period 1932-1979 a great shift had occurred in the textbook use of the term race, from the position that races exist to a position where they do not. During the period analysed, it was clear that major historical events had occurred, such as World War II, proceeded by a significant shift in social attitudes around race, which was achieved through the US Civil Rights movement. The anthropological shift commenced following challenges

from English anthropologists, such as Hogben and Huxley, as early as the 1930s but were ignored or even ridiculed by American anthropologists until the 1960s (Littlefield et al, 1982). A similar move occurred in the field of psychology, by firstly rejecting the theoretical or scientific meaning of race, followed by critiquing racial categorisation within the science by offering an alternative, e.g., the term ethnicity (Helms et al, 2005).

To challenge the status quo and further the gains made during the U.S. Civil Rights movement of the 1960s, writers began to critique American systems with a view that race should be central to legal, educational, or social policy analysis (Cole, 2009), resulting in Critical Race Theory (CRT). CRT is a concept that is relatively new to the UK and is thought to derive from two major sources; Frantz Fanon, a French West Indian psychiatrist and political philosopher, and William Edward Burghardt Du Bois, an American sociologist, historian, and civil rights activist, and emerged from Critical Legal Studies in the 1970s. The influence of colour and diversity of perspectives from social narrators of the time around race created a space to critically debate and challenge the status quo. As such, the two major tenets of CRT are the concept of 'white supremacy' to describe the oppression based on race rather than the notion of racism; and the second is the belief in race, not social class, as the primary contradiction in society (Cole, 2009). Cole (2009) expands on this second point by articulating that white supremacy must be overthrown to establish a social democracy, however, given the advantages to capitalism of racialised capitalism, then capitalism without racism is inconceivable. It could be argued that white supremacy is not the issue, but racialised capitalism, which was fashioned by white Europeans in the formation of empires.

CRT has allowed writers to express a critical analysis of the issues that revolve around race in theoretical terms as well as in practice. CRT themes include critiques of liberalism as a means of addressing the race problem; storytelling as a means of challenging power-laden beliefs; querying where anti-discrimination legislation has failed to redress racial inequalities, etc. There is an immense body of work in this area in the U.S. however, as Cole (2009) commented, the first ever international CRT seminar took place in the UK in 2006, and there has been a growing body of work in this field in the UK since this point. Similarly, there is growing debate in the UK on the Critical Philosophy of Race (CPR), challenging the history of racial injustice and its philosophical underpinnings. Singh (2014) argues that philosophers such as Kant and Hegel, helped shape the Eurocentric view that philosophy was exclusive to the West from the

late 18th to 19th centuries. This influenced the confinement of Africa and Asia to the margins. Kant has been credited for being the first philosopher to theorise race in a substantially influential way that justified scientific debate at the time (Bernasconi, 2001; Sandford, 2018). Stone (2017) interprets Hegel's Philosophy of World History in a way that suggests that freedom can only be recognised and practiced in classical, Christian, and modern Europe, meaning that for others, freedom can only be acquired if Europeans impose their civilisations upon them.

Ethnicity, as an alternative term, has to some degree replaced race as a more acceptable term in modern times to encapsulate more than just the physical traits assumed under race, e.g., skin colour. Max Weber, the German sociologist and philosopher, provided some initial consideration on status groups circa 1911 and argued that race creates a group only when it is subjectively perceived as a common trait, especially where there would be a consequence of collective political action. According to Jenkins (1997), Weber suggested that ethnic membership did not constitute a group, it only facilitates group formation. Group membership and subjective identification is explored further by Modood et al (2002) concerning the assumption that ethnic minority groups are shaped by racism or by attitudes, behaviours, and structures of dominant groups as well as by their own heritage, collective action, and reaction to exclusion. This is a valuable perspective that suggests that if it were not for the construction of race, peoples of the world may not have been divided in the way that they are.

Expanding this further, Coates (2015) wrote that race is the child of racism, not the father. In some way this notion supports the idea that oppressed racialised affinity groups have been marked out from, or by, their oppressors (whites), who are less often inclined to recognise their own ethnic identity, and even less likely to relate to an affinity group based on this. There may be exceptions to this if a white ethnic group is marginalised, e.g., eastern Europeans, gypsies, Irish Travellers, etc. Here race can be interpreted as that which is not white, as Coates (2015) suggests that whites do not assign an ethnicity to themselves and herein lies one of the problems associated with the notion of race. It has, and continues to, shape the way this concept is understood by people of white ethnic background. Despite race applying equally to all people, the term is associated with those of non-white background, and as a result it disproportionately impacts racially minoritised people.

In the late 1990s, 'ethnicity' as a more inclusive term, would capture the diversity of difference, remove stereotypes, and at face value remove the historical biological classifications of those subjects (Modood et al, 1997). Yet would attract criticism from Smith (2002) who argued that ethnicity is equally exclusive in its social interpretation, particularly as Modood, et al (1997) suggested that physical appearance should be considered when identifying ethnic groups, which Smith (2002) believed would be racist. Despite these debates, in practice the UK and its public sector organisations consistently use systems of classification via population (UK census) and workforce monitoring that continues to use the same racial categories. Whether human subjects are referred to in relation to their race or ethnicity, Blum (2002) maintains that inequalities based on either of these descriptors continue to exist.

This section has explored the evolution in terminology from race towards ethnicity and how this has been influenced by historical events and social and academic reflection that has informed shifting attitudes. These changes have continued to inform the notion of race and how this has descended from racism, rather than racism existing as a product of racial categorisation. The complexities surrounding the historical context influence how society tackles racial inequalities and this in turn informs the language used to describe a person's background and the following section will explore the topic of race talk further.

2.4 Race talk – the language of race and ethnicity

Academic literature suggests that the terms race and ethnicity are problematic and exclusive (Modood et al, 1997; Smith, 2002) and in considering this, the terminology used might affect institutional capability to address matters of race or ethnicity within a workplace setting, and ultimately, affect an institution's ability to advance race equality. Racially minoritised individuals within the UK higher education system may be affected by their own understanding of these terms, which may impact their own sense of identity within their institution. Moreover, this sense of belonging may be further impacted by the institution's understanding, commitment, and investment towards minority ethnic staff in dealing with matters of race. Institutional commitment to and progress with race equality will be discussed in Chapter Five and language and terminology will be explored further in Chapter Seven when perceptions of racism will be considered.

In practice it has become acceptable and mainstream to use race and ethnicity interchangeably without entering into discussion about their histories, their evolution, or their difference. Some academic debates conclude that the concepts of race and of ethnicity are problematic because of their social construction (Cole, 2009; Modood et al, 2002; Smith, 2002). Despite an overall acceptance of these conclusions, the problem does not necessarily lie in their construction, but about the way the terms are used in practice. These differences will include the way an individual sees themselves within a particular ethnic group, how they compare themselves to others within that ethnic group and may be influenced by geography, upbringing, or nationality. This can become more complex according to how an individual defines themselves if they identify with multiple ethnic groups. Hirsch (2018), who identifies as mixed race, discusses how identity can become meaningless if that identity belongs to the dominant group. This perspective resonates to some extent with the work of DiAngelo (2018) and Eddo-Lodge (2018) who both comment on the lack of racial socialisation for people who identify as white. Therefore, a lack of lived experience, appreciation for, or emotional disconnect as Eddo-Lodge (2018) suggests is demonstrated by white people around the subject of race. Hirsch's (2018) reference to meaningless can therefore be interpreted as the denial, defensiveness and lack of empathy felt by some white people towards the experiences of people of colour. The challenges apparent with identity and acknowledging the sociopolitical pressures surrounding this topic create difficulties for organisations who may lack confidence with dealing with a diverse workforce and creating spaces within institutions to talk about race.

For many individuals and organisations, it has been difficult to initiate conversations relating to race due to the history, emotion and defensiveness attached to the subject matter (Eddo-Lodge, 2018; DiAngelo, 2018). It is not even the discussion *about* race and ethnicity but the perceptible lack of engagement by those in dominant ethnic groups who may grapple with unfamiliar concepts. White people frequently may not possess the vocabulary to make expressions about race, which may be as a consequence of the lack of racial socialisation. DiAngelo (2018) asserts that white people have not been conditioned to think of themselves as racial beings, and it is perhaps because of this, that conversations about this subject are such difficult ones, particularly where dominant groups participate in those conversations. Eddo-Lodge (2018) suggests that white children are taught not to see race, whereas children of colour are often taught that they must work twice as hard as white counterparts if they are to succeed.

As a society we have maintained a discursive fear around race, which Crawley's (2007) point surrounding the lack of racial socialisation for the dominant ethnic group demonstrates has become further repressed over time by the silencing of any articulations on race and ethnic minority people. DiAngelo (2018) asserts that silencing discussions around race has created an anxiety among white people within a racial discourse, where dialogues voicing experiences of racism raise anxiety levels even further. Drawing attention towards negative experiences, perceptions and opinions will act as a catalyst for fear, denial, and trepidation (DiAngelo, 2018; Eddo-Lodge, 2018; Bhopal, 2018). The subject of race is emotive and may affect the way in which different ethnic communities can share different perspectives. Feelings of guilt, shame and ignorance might prevail and cause an incendiary outcome that can cause conversations to prematurely break down and anxieties to heighten (Lingayah et al, 2018; Campbell, 2016). These anxieties might manifest themselves through defensiveness, discomfort, or trigger perceptions that privileges of the dominant racial group might be limited or removed altogether if racial inequalities were to be eradicated (DiAngelo, 2018).

Discussions around terminology are not limited to the higher education sector and this has become a subject of increasing public debate since this research study began, with an article proposing that BAME, an acronym used to describe black, Asian and minority ethnic people, should be discarded (Sandhu, 2018). A more recent article discussed racial terminology in the Civil Service and the 'dithering', observed by the authors, of people attempting to find the appropriate and correct terms to describe groups according to race (Saeed, et al, 2019). These articles highlight that there remain mixed feelings about racialised terms across diverse communities and across a variety of sectors. The act of accepting and utilising acronyms by racially marginalised people to describe themselves and others of minority ethnic background as a homogenous group is a production and reproduction by agents of colour and enables others not directly affected to perpetuate its use. Similarly, minority ethnic people may be constrained to cease using acronyms in future (Giddens, 1984) because it has become normalised within their workplace setting.

A layperson's understanding of racialised terminology can be explained to some extent by the legislative context that has informed our thinking as a society in terms of 'race relations', which can create the perception that there is a problem between people because of race. In turn those statutory limitations have influenced and driven organisational practitioners in the fields of equality and diversity and human resources managers to engage with race as a protected characteristic (Equality Act 2010). Consequently, the use of the term race can be justified from a practitioner perspective, and this has become normalised through practice. Within a higher education context, there can be greater understanding and appreciation of race and ethnicity, particularly for those who hold an academic interest in these subjects, however policy development and implementation within institutions sits generally within the corporate centre, and therefore may not be directly informed by research in this area.

This section has explored how the terminology used to talk about race can unintentionally marginalise people of colour. Racialised language and terminology is bound to have an impact on the way that people of colour see themselves and how they navigate white hegemonic workplace settings. To explore this further, the subsequent section will consider how identity and belonging are key features of this topic that provide context to the research objectives and how they may play a part in the outcomes for individuals in the workplace, specifically in relation to objective two concerning staff perceptions of the manifestation of racism and objective three in determining the impact of racial inequalities on black staff.

2.5 Racialised identity and the impact on belonging

The dictionary defines 'identity' as 'the who or what a person or thing is; a distinct impression of a single person or thing presented to or perceived by others; a set of characteristics or a description that distinguishes a person or thing from others' (Oxford English Dictionary, 2021). These definitions provide a simplistic view on an aspect of ourselves that plays a critical part in the way we see ourselves, how we present ourselves, the experiences we have as a consequence of it, and the impact it can have on our outcomes.

The way we internalise our own identity and those around us can be affected by the way we have been racially socialised. People of all ethnic backgrounds will racially socialise their children and Steinbugler (2015) suggests that children are not born with an innate interpretation or valuation of blackness or black culture and therefore black parents will emphasise facets of black culture and socialise their children to race, including raising awareness of societal racial discrimination (Shelton, 2008; Steinbugler, 2015). Steinbugler's (2015) US research discusses black identities for those who are in interracial relationships and the accounts given in this study highlight the intersections of multiple aspects of identity with ethnicity, such as gender, sexual orientation or class and the way that the participants navigate their own ethnic identity. Accounts provided insight into the challenges faced by black Americans and their relationships with white people, how their black authenticity might be questioned and the confusion about one's own sense of self where one was socialised within a predominantly white-dominated neighbourhood, education and the choices being made of whom to socialise with. This resonates with what Frantz Fanon, a West Indian psychoanalyst and social philosopher, described that the black man has two dimensions; one with his fellows and the other with the white man, explaining that a Negro [sic] behaves differently with a white man than he would with another Negro [sic] (Fanon, 1967). The importance of intersectionality cannot be ignored in terms of the experiences and outcomes for people of colour in combination with other characteristics, such as gender, age, sexual orientation, etc (Crenshaw, 1989).

Mapedzaham and Kwansah-Aidoo (2017) assert that the black body in the white space has always been constructed as a problematic difference to whiteness, and one which is inferior and othered. Blackness is therefore not merely a skin colour, but rather a social construct that

is consistently perceived in opposition to whiteness. Solanke's (2018) article raises the continued stigmatisation of black identity. Attributing this to the hypervisibility for example of young black men in the criminal justice system and the hyperinvisibility for example of black women in public life, such as the senior ranks of academia as well as the displacement of identity politics by the politics of identity. Solanke (2018) also states that the rejection of 'black' in favour of BME and BAME contributes to the ongoing stigmatisation of black people, where this stigmatisation is the precursor for the perpetuation of discrimination. The level of stigmatisation may be different for persons that have mixed ethnic heritage and Hirsch (2018) described that as social creatures, part of knowing oneself is knowing what group we belong to and that this includes not just characteristics, but also values and beliefs that we can share. This is the same for any group, however she goes on to say that it is not the muddled inheritance of a mixed-heritage person that is the problem, but the way in which people deal with multiple ethnic identities and create difficulties around identity.

The values and beliefs that Hirsch (2018) describes will also invariably involve a person's class identity, which can change over the course of a person's lifetime but can impact upon the resources made available to individuals. Class is an economic, social and cultural construct and the class position of black people will have been impacted by the limited economic opportunities made available so that their social mobility is constricted within the class structure (Moore, 2005). Moore's (2005) study focused on the importance of class and racial identities and explored the symbolic meaning of class and the nature of inter-class relations relating to urban community development activities within the black community in Philadelphia. The findings from this research identified an unacknowledged class bias among residents and particularly through tensions between middle class and residents of lower socioeconomic backgrounds who did not share their cultural orientation, with one participant commenting on the vulnerability experienced by middle class residents with neighbours with a 'ghetto' orientation. The discussion continues to highlight the potential for secondary marginalisation where black individuals who experience upward social mobility can access some of the social institutions of power and are given the responsibility by both blacks and whites of servicing, regulating and representing the race. Blacks in these positions of limited power are expected to embrace the norms and values of the dominant society. This process of regulating stigmatised individuals by members of their own racial group results in secondary marginalisation (Moore, 2005).

Belonging is a core psychological need, which forms one of the levels within Maslow's (1943) motivational theory, that is most well known as five hierarchical elements of a pyramid that includes at its base physiological needs, such as food, water, warmth. This is followed by safety, love and belonging, esteem (prestige and feelings of accomplishment) and capped by self-actualisation (achieving one's full potential). If any of these levels are left unsatisfied, an individual will not be able to reach self-fulfilment. Baumeister and Leary (1995) state that being socially accepted is a basic need for all individuals and the literature suggests that the need to develop and maintain meaningful interpersonal relationships, as well as experiencing a sense of belonging in the workplace, is vitally important (Baumeister and Leary, 1995; Mohamed et al, 2014). As important as forming and maintaining relationships is the reluctance to break social bonds and individuals are often motivated to preserve these bonds, sometimes resulting in significant personal sacrifice. Participant commentary included in Steinbugler's (2015) study touched upon the difficulties that black people experience within institutional settings when having to specifically decide who to socialise with and the backlash of socialising between competing groups, e.g., white and black, heterosexual and gay, etc.

From a workplace perspective, institutions like most large and complex organisations are driven by the need to diversify their workforce. Diversity in the workplace has a multitude of benefits and can lead to high quality group outcomes because of the broader knowledge, skills, and attributes that different individuals can bring to a group (Cox et al, 1991; McLeod et al, 1996). Although there are tangible benefits these mixed working groups can bring to an organisation and their teams as groups, there can be detrimental impacts on group individuals, particularly where individuals are dissimilar to the dominant cohort within a group. These effects can impact the level of satisfaction and commitment to the group and these outcomes were confirmed in a study (Kim et al, 2017), which found that ethnic dissimilarity can frustrate belonging and consequently, reduce group members' attachment, as well as increase levels of absenteeism and turnover.

When workers are the demographic minority, they are more likely to be seen as different, which may result in increased stereotyping, biased job performance evaluations (Wilson, 2010), and exclusion from informal workplace relations (Kanter, 1993; Ibarra, 1993, 1995).

Kanter (1993) described token subjects as experiencing feelings of isolation and ostracism and identified that token subjects are likely to encounter increased visibility; boundary heightening; and stereotyped portrayals (Kanter, 1993; Smith, 2013). Their increased visibility in the workplace can further produce anxiety regarding expectations and work performance. Those who are scarce within the workforce are more likely to attract attention (hypervisibility), stand out in a group setting and may be evaluated more critically (Wilson and Jones, 2008; Solanke, 2018). This may be amplified somewhat if an individual was born outside of the UK and being marked out through one's ethnic identity may increase feelings of isolation further. Hirsch (2018) discusses identity and recounts how meaningless the term 'black' became in a country like Nigeria which is dominated by black people.

Furthermore, individuals can experience psychological distress due to social identity threat. Social identity threat is the state of psychological discomfort when confronted by an unflattering group or individual reputation (Nelson, 2009), such as might occur if an individual is being stereotyped. A stereotype is a preconceived and oversimplified idea about a person or group (Oxford English Dictionary, 2021) where this idea is used to make assumptions. This threat can become particularly heightened where that reputation can be confirmed, through an onlooker's own confirmatory bias towards that person or not, or by one's behaviour – adding a significant emotional burden if the individual is being judged by their peers, particularly in a professional setting. An example might include a black woman challenging a decision and is therefore assumed to be stereotypically 'angry'. The impact of social identity threat can include disruptive effects on performance in the short-term, e.g., interview, examination, etc. Over time, social identity threat can prompt defensive behaviours in individuals, such as disengaging from activities or teams and can result in a self-fulfilling prophecy whereby the person begins to resemble the stereotype; therefore, living down to the expectations of others (Nelson, 2009).

While an individual might readily accept or acknowledge that they have experienced negative or unequal treatment, Pearce (2019) indicates that people are often reluctant to attribute this to racism. Ahmed (2012) suggests that a claim of racism is an institutional injury, an injury to whiteness. It can be argued that a reluctance to name racism is a way of protecting whiteness. According to Pearce (2019), people of colour often make strategic decisions not to use the language of racism, particularly if that individual already poses a problem or appears 'out of

place' in the whiteness of the institution, as to do so would further affect a person's sense of belonging within the workplace.

Chen and Hamilton (2015) argued that a reduced sense of belonging might be as a result of the subtle cues that occur within the academic work setting, which can lead to under-represented minorities questioning their sense of belonging within these domains. In these environments minority ethnic staff might become stigmatised by dominant group members and as a result experience social isolation and exclusion in formal and informal groups and networks. A person that does not feel like they belong in the workplace might employ coping strategies to address the social isolation and marginalisation they may experience (Stainback and Irvin, 2012).

This section has explored the complexities of racial identity, the positive *and* negative connotations of blackness and racial socialisation and how this may differ dependent on whether a person's socialisation has involved an emphasis on positive black cultural affirmation and the types of inter-racial relationships experienced during formative years and throughout adulthood. Identities are also impacted by social status, which may bring about conflict or bias towards others within diverse socio-economic groups, potentially affecting the way society views black communities, as well as the way that black people view themselves. This is further complicated if a person straddles multiple ethnic identities and the challenges faced in navigating from within and beyond those identities. The way individuals internalise their sense of self and how they are positioned within wider society can impact upon their sense of belonging and this will be explored further in Chapter Six.

The following section will explore how race has been politicised in modern UK society and the impact this has had on policy development and the way organisations have been enabled or constrained to deal with matters of race.

2.6 Politicising race in modern UK society

A modern political perspective around the concept of race, including the rejection of scientific racism began in 1933 with calls to rebuke 'Aryan science', however few scientists would still do so explicitly and publicly (Barkan, 1992). In the years following World War II UNESCO sponsored a large international project to disprove and discredit claims of racial science that had brought about the atrocities of the time, culminating in the *Statements on Race* from 1950 to 1967, (UNESCO, 1969).

The 20th century brought about a renewed racial politicisation through UK Government policy on immigration (Mason, 2000). Immigration can bring not just new groups of people, but immense social impact, and raises the potential for social unrest. The early part of 20th century saw new migrants being invited to the country to aid Britain's war effort during World War I. Resources were scarce and for the first time there were higher concentrations of black, Asian and Chinese communities. Tensions rose amongst white working-class communities due to post-war unemployment, the informal and formal national and governmental narratives about visible racial minorities, a heightened visibility of black workers, and housing shortages, especially in the port cities (Fryer, 1984). The first significant race riots exploded across London, Liverpool, parts of the north-east and south Wales in 1919. Here, the intersection between race and class was apparent. Added to the insecurity of limited resources were the interracial relations between non-white men and white women, which caused further strain on community relations. Not only were non-whites taking the jobs of the indigenous population, but their women too. In the aftermath the government repatriated hundreds of black people (National Archives, 2021).

Britain may have started to turn into a more multi-cultural society, but that was not reflected in Government policy. Gilroy (2008) revealed that at the suggestion of Churchill, a Conservative cabinet discussed using 'Keep Britain White' as an electoral campaign slogan as early as 1955. Despite there being great social change in 1950s Britain, there were significant undertones of race policy making following the post-war inflow from the colonies. A post-boom slump, poor housing conditions and discrimination in employment, developed into unrest once again culminating in the Notting Hill riots in London, in 1958. The racialisation of immigration was no new phenomena by the time of these riots as the Aliens Act 1905

introduced immigration controls with the main objective of controlling Jewish immigration from eastern Europe. This was followed by a rise of anti-Semitism against growing numbers of impoverished Jewish refugees arriving in London's East End. The Alien Act 1919 would form the basis of all future immigration legislation up to the introduction of the Immigration Act 1971 and would be renewed each year until then (Brown, 1995).

Despite the regular debates and updates to immigration law during the 20th century, whose restrictions would become more and more racially motivated, multiple employment initiatives that would fall outside of the legislative restrictions continued to address British labour shortages. Such an initiative was the arrival of the SS Empire Windrush with around 500 men from the West Indies to Tilbury docks on 22 June 1948. As a result of increasing immigrant numbers, racial tensions across post-war Britain would erupt. Ahead of the 1964 General Election, the Labour Party committed to legislate against racial discrimination and incitement in public places and give special help to local authorities in areas where immigrants had settled (Brown, 2018).

To tackle racial prejudice, the government needed to legislate, which it did through the Race Relations Act 1965. The legislation outlawed discrimination on the grounds of colour, race or ethnic or national origins in specified "places of public resort", such as hotels and restaurants. It did not make discrimination in employment or housing unlawful, nor did it include shops or private boarding houses. Mechanisms to monitor the progress of race-related matters were placed in the hands of Commissions with powers to act, e.g., the Commission for Racial Equality (CRE) in 1967. On the day of the Act's implementation, the press reported that Labour backbenchers wanted the legislation to go further to penalise employers who discriminated against applicants on the grounds of race and local authorities which barred people renting council homes (BBC, 2018).

According to Cabinet Papers from 1967, the Labour Party proposed to further strengthen the Race Relations Act 1965 by making discrimination on racial grounds unlawful –

- a) in respect of places and services open or available to the public at large
- b) in employment or trade union activities, except domestic employment or employers with fewer than 10 employees
- c) in the disposal of property

- d) in the provision of insurance and credit facilities
- e) in the form of discriminatory advertisements.

Following the racially motivated murder of Stephen Lawrence, the subsequent inquiry (Macpherson, 1999) resulted in further legislation through the Race Relations (Amendment) Act 2000, which introduced statutory requirements for public sector organisations. This change would introduce a general duty to positively promote race equality and 'good relations' between people of different racial groups. Other duties included assessing the impact of policy and practice on ethnic minorities and taking remedial action where necessary; monitoring the workforce, recruitment, promotion and training, grievances, disciplinaries, dismissals and other reasons for leaving; having a publicly accessible policy on race equality; and publishing monitoring data annually. In addition to this, the then Commission for Racial Equality (CRE) were given powers to issue compliance notices to a public body that did not fulfil its duties. The CRE and other commissions covering gender and disability would be disbanded and merged into an all-encompassing commission, the Equality and Human Rights Commission in 2007.

In time and with a political shift to reduce bureaucracy and introduce a legal mechanism that would facilitate a consistent application of remedies to address discrimination, harassment and victimisation across employment and goods and services, the Equality Act 2010 came into force, incorporating all previous race relations legislation. In 2012, the Public Sector Equality Duty was enforced through regulation², obliging public authorities to have in place at least one equality objective every four years. This is of stark difference to the duties set out under the Race Relations (Amendment) Act a decade earlier.

Legislative change has facilitated to some degree social attitudes in the UK towards racialised others. As such, British social attitudes should have shifted in tandem. However, in a migration survey in 2017, 45 percent of respondents agreed that there were too many immigrants, compared with 64 percent agreeing with this statement just four years earlier. Interestingly, although British people reported attaching greater importance to skills rather than country of origin in the survey, 37 percent of Britons responded that no Nigerians should be allowed compared to 10 percent responding that no Australians should be allowed (Migration

-

 $^{^{2}}$ Equality Act 2010 (Specific Duties) Regulations 2011

Observatory, 2020). Social attitudes have shifted since the 1950s however, social unrest e.g., Tottenham riots in 2011, the Syrian refugee crisis and sweeping right-wing political gains across the European Union may have influenced these results. The social climate and the race agenda has received further attention shadowing an increase in race-related hate crime and xenophobia following the UK referendum to exit the European Union in 2016 (Gabbatiss, 2017; Golec de Zavala et al, 2017; d'Ancona, 2018) and a recent social attitudes survey report confirmed that 26 per cent of people said they were 'very' or 'a little' prejudiced towards people of other races (Kelley, Khan and Sharrock, 2017).

This section has explored how Government policy has shifted social attitudes in ways that can advance race equality, as well as hinder it because of racial stereotyping and fearmongering. The following section will discuss how racial prejudice has transformed from traditional overt acts to becoming more obscure and insidious.

2.7 The hidden nature of contemporary racism

The way race has been politicised for decades in UK politics and wider society has had farreaching consequences for the way individuals are able to deal with matters of race and racism. The evolutionary political landscape of race has impacted the language society uses and the attitudes social actors possess, making this a particularly complex matter within organisational environments where lived experiences of racism will be enacted by and through social actors.

In establishing the links between the historical and political development associated with the way race and ethnicity have evolved in the UK, there have been several references to racially motivated prejudice, and it is important to understand what is meant by the term prejudice before exploring the manifestations of it within a workplace setting. Prejudice might include a preconceived opinion that is not based on reason, actual experience or a dislike, hostility, or unjust behaviour deriving from preconceived and unfounded opinions. Allport (1954) suggested that a person's prejudice is unlikely to be merely a specific attitude toward a specific group: it is likely to reflect his [sic] whole habit of thinking about the world, assuming that in general a person's beliefs and attitudes are negative. While prejudice is formed through learnt stereotypes, racism is a manifestation of prejudice and is the differential treatment enacted by an individual, group or organisation on individuals based on assumptions about a group

(Gamst et al, 2011, p.251). According to DiAngelo (2018), we are all prejudiced and therefore, we are all capable of making assumptions about the people around us. The difference lies in having prejudicial beliefs or attitudes and the way we act upon them. These manifestations or enactments can occur at institutional (law, policy, practice), societal (hate crime) and individual (racial stereotyping, microaggressions) levels in overt (traditional) or covert (modern) ways.

Essed (1991) articulated that racial discrimination includes all acts – verbal, nonverbal, and paraverbal – with intended or unintended negative or unfavourable consequences for racially or ethnically dominated groups, providing the foundation that prejudice need not be intentional and includes a range of actors at multiple levels. According to van Dijk (2002), a system of racism consists of a social and cognitive subsystem, with the social subsystem constituted by social practices of discrimination at the micro level and relationships of power abuse by dominant groups and organisations at the macro level. People of colour may also hold racially prejudicial attitudes, views and even discriminate against other people of colour, however whites hold the social and institutional power that will continue to benefit whites and uphold racism as a society-wide dynamic (DiAngelo, 2018). Would racism exist without race? For Bonilla-Silva (2015) racism creates and maintains race. Racism is the product of racial domination projects, such as colonialism, slavery, and labour migration, which produces a network of social relations at social, political, economic, and ideological levels that shape the outcomes of various groups (Bonilla-Silva, 2015). Races have been socially constructed, but they are socially real and impact upon the daily lives of many.

Colonialism, slavery, and labour migration provoke distinct reactions across the UK population. However, collective amnesia and skewed perceptions of these historical race projects means that because acts, overtly harming people of colour but which are not seen, might imply that racism no longer exists. Is the interpretation about the altered state of racism being confused with a post-racial epoch? If so, this presents a problematic discourse that disables the identification of modern racism. Bonilla-Silva (2015) suggests that new racism came into existence following the US Civil Rights movement and describes this as comprising the following factors:

- 1. Increasingly covert nature of racial discourse and practices
- 2. Avoidance of direct racial terminology

- 3. Elaboration of a racial political agenda that eschews direct racial references
- 4. Subtle character of most mechanisms to reproduce racial privilege
- 5. Rearticulation of some racial practices of the past.

Of course, overt acts of racism continue in the 21st century, albeit with less frequency and this is particularly the case in the workplace. Since the rise in populism across the United States and Europe in recent years, and domestically the UK 2016 referendum to leave the European Union, instances of anti-immigrant sentiment and racial and other hate crimes have risen sharply, where for instance 76 per cent of reported hate crimes in the UK were racially motivated (Weaver in The Guardian, 2018; Home Office, 2018). These increases in hate crime and negative attitudes toward people of colour and/or non-British people have in turn become stoked by hostile government policy in recent times and support the factors mentioned above of what may constitute contemporary racism. It is the evolution of modern racism that we must now become more aware of and find solutions to tackle in addition to becoming more assertive in addressing overt forms.

Whilst recent increases in overt racism have largely affected those seeking asylum in the UK or where religion was a factor (Home Office, 2018), scholars have suggested that overt expressions have been largely replaced by symbolic forms of racism (McConahay, 1986; Rowe, 1990; Henry and Sears, 2002; Sears and Henry, 2003; Noon, 2017), whilst others have argued that race inequality is less legitimate due to modern anti-discrimination laws and activism (Acker, 2006). Acker's (2006) point does resonate to some degree in terms of the sentiments of those who disagree in general with any forms of inequality. However, there is a chasm between cause and effect and although inequality in principle lacks legitimacy, neither law nor policy appears to effectively reduce the level of inequality of outcomes. To DiAngelo (2018) and Noon (2017) the persistent nature of racism is maintained by several aspects: colour-blind racism, where a person claims that because they do not see colour, there can be no racism; and aversive racism, which DiAngelo (2018) argues manifests itself in well-intentioned people who see themselves as educated and progressive. This is a subtle and insidious form of racism and one in which the perpetrator can hide behind a liberal, progressive façade and yet is most likely to remain unseen to others, including the target.

Scholars have provided commentary on colour blindness (Bonilla-Silva, 2002; Zamudio and Rios, 2006; Hirsch, 2018) and it is important to acknowledge the challenge for individuals and

organisations to address insidious forms of racism that can become embedded within and in the implementation of policies and practice so that race or ethnicity is now rendered invisible by this new racism. Noon (2017) suggests that symbolic, modern, or colour-blind racists are aware of their biases and do not conceal their views, since they are a socially accepted form of racism, albeit in a moderated form. Noon (2017) appears to propose that a moderated form of racism might also be accepted within organisations, particularly as our workplaces attitudinally reflect the society around us. This statement implies that our institutions could therefore be a hotbed of racist activity, particularly once organisations became more attuned to these microscopic and symbolic acts that occur in every day institutional life. Those organisations are composed of dominant and minority groups, resulting in a racism rooted in the material rewards of white privilege (Zamudio and Rios, 2006), and which motivates the persistence of racist attitudes at multiple levels, through action or inaction, and serves to reinforce colour blind racism.

At micro level, contemporary workplaces provide ample opportunity for subtle, sometimes unconscious manifestations of racial microaggressions and microinequities that include neglect, incivility, ostracism, and inequitable treatment (Fox and Stallworth, 2005). Racial microaggressions are brief and commonplace daily verbal, behavioural, or environmental indignities, whether intentional or unintentional, that communicate hostile, derogatory, or negative racial slights and insults toward people of colour (Sue et al, 2007). Equally, microinequities can be made up of apparently small events, which are often ephemeral and hard to prove; events that are covert, often unintentional, and frequently unrecognised by the perpetrator (Rowe, 1990). Webb et al (2004) suggest that these acts (being treated as inferior, denied resources, limited in their social mobility and aspirations) are not perceived as 'symbolic' forms of violence, rather their situation seems to be 'the natural order of things'. In contrast, Kandola (2018) articulates that the phrase 'microaggression' is too judgemental and accusatory, preferring the phrase 'micro-incivility', which describes more accurately the types of behaviours that people are more likely to acknowledge, accept and potentially modify. As a result, micro-incivilities are described as daily, commonplace behaviours or aspects of an environment which signal, wittingly or unwittingly, to members of out-groups that they do not belong or are not welcome (Kandola, 2018).

Other terms that describe witting or unwitting behaviours are microinvalidations and microinsults, which are other subtle forms of discrimination. Microinvalidations include behaviours that deny the importance of race, or which convey the myth of societal meritocracy. Microinsults are actions that convey insensitivity or rudeness, or directly demean a person's ethnic heritage, such as mistaking a person of colour for a cleaner for example. This latter point was featured in The Guardian headlining 'I've been mistaken for the coffee lady': experiences of black female academics to highlight the career trajectories of women of colour in UK academia (Hall, 2017). Perpetrators of microaggressions are often unaware that they engage in such communications when they interact with racially minoritised people (Sue et al, 2007, 2008; Kandola, 2018).

Within the UK higher education sector differential outcomes have been discussed in this context for some time, predominantly in relation to the outcomes of racially minoritised students when compared to white students (Gillborn, 2008; Higher Education Academy, 2008; Higher Education Academy/Equality Challenge Unit, 2011). In relation to the outcomes of staff in higher education the attention has come later and with less emphasis than has been placed on the outcomes of the student population (Pilkington, 2011; Philips, 2012). Several studies have highlighted that racially minoritised staff in UK higher education are less likely to be recruited (Pilkington, 2011), experience disadvantage through a range of microaggressions and are more likely to be isolated, ignored, racially stereotyped and micro-managed (ECU, 2011; Rollock, 2011, 2019; Pilkington, 2013; EHRC, 2019; UCU, 2016, 2017). De Cuir-Gunby and Gunby (2016) state that it is anecdotally false that as people of colour attain greater social status, more education, and higher incomes the prevalence of racial microaggressions will decrease.

The EHRC's recent report (2019) into racial harassment in UK higher education found that more than half of staff respondents did not report racial harassment. The reasons provided by respondents included having no confidence that the institution would address the matter; staff did not know how to report; staff could not judge whether the complaint was serious enough; and a difficulty in proving what had happened. Others feared the consequences of reporting, such as a negative consequence to career prospects and being seen as a troublemaker. The findings from this report provide useful and relevant themes, which are explored further in Chapters Six and Seven of this dissertation.

Identifying and reporting acts of racism can be fraught with difficulty as contemporary race talk can be strategically organised to deny the existence of racism, and as such has created a post-racial society where making accusations of racism has become taboo (Augoustinos and Every, 2010) and which conceal and legitimise racism to protect white privilege (Harries, 2014). Augoustinos and Every (2010) continue to remark that to avoid identifying racism explicitly presents racism as highly unusual and exceptional. Harries' (2014) study showed that incidents of explicit racism were downplayed or denied by respondents themselves often in response to the potential of being labelled as a 'victim', but which also played part of a greater narrative around racial tolerance in the UK. Here we see the juxtaposition between societal denial of racism at a macro level and individuals who experience acts of racism on a micro level, despite those acts being traditional or overt.

This section has explored the contemporary nature of racism where the literature suggests that it has become much more difficult for those affected by it to articulate their lived experiences as they are small and commonplace and therefore difficult to prove if seeking an organisational remedy. The following section will continue to explore the broader meso and macro manifestation of racism within an organisational context.

2.8 Exploring institutional racism

This section will continue the exploration of racism to consider an aspect that has been little researched within a higher education context - institutional racism. The term is one that was coined in the mid-twentieth century and has been resurrected at key points in the UK's racial history, most recently at the turn of the 21st century following the death of Stephen Lawrence and the resulting inquiry into the Metropolitan Police Service (Macpherson, 1999). The concept of institutional racism provides a perspective that focuses on the way that organisations, through their systems, processes and policies might have a detrimental impact upon minority ethnic people and who can be members of the workforce, customers, clients, or service users. It is important to reflect on the role organisations can play in extending the reach of individuals *through* the organisation, rather than solely *in* an organisation. In other words, that the organisation is the vehicle within which symbolically it can cause harm to ethnic minority people through its acts or omissions and that as a collective of individuals this may go largely unnoticed.

The phrase institutional racism has been in use since the 1960s principally in the United States and used to describe how white interests and attitudes saturated the key institutions in American life of the time (Gillborn, 2002). Black activists, Stokely Carmichael and Charles V Hamilton, (1967) stated that institutional racism "originates in the operation of established and respected forces in the society. It relies on the active and pervasive operation of anti-black attitudes and practices. A sense of superior group position prevails; whites are 'better' than blacks and therefore blacks should be subordinated to whites. This is a racist attitude and it permeates society on both the individual and institutional level, covertly or overtly" (in Macpherson, 1999:23).

The language used by Carmichael and Hamilton in relation to 'established' and 'respected' forces can be interpreted as pertaining to institutions, such as those within the higher education sector. Higher education itself was established in the Middle Ages in Britain and made available by and for persons of privileged background (Gillard, 2011). These early institutions with canonical roots, followed by other modern secular universities have retained an air of privilege despite the modernisation of higher education. Institutions in the context of this discussion are organisations founded for the provision of higher education. These organisations are governed by a Board or Senate with the overall operational management of an institution being held by a Vice Chancellor or equivalent, supported by a senior management team. The management structure is supported by academic and professional and support staff at various levels in the delivery of higher education provision or related services. These institutions are constituted of numerous individuals, and as such, collectively represent a microcosm of society, although the extent to which an institution wholly reflects the society around it is debatable and may indicate certain local preferences in populating those workplaces (see Chapter Five).

From the above articulation, one can envisage the potential for negative racial attitudes at an individual level within an organisation if those attitudes exist within sections of the society, however the references of racism at an institutional level have caused much debate. The notion of institutional racism in Britain was addressed in the 'Scarman Report' (1981) following the Brixton Riots of the same year. In his report, Lord Scarman rejected claims of institutional racism by stating:

It was alleged by some of those who made representations to me that Britain is an institutionally racist society. If by that it is meant that it is a society that knowingly, as a matter of policy, discriminates against black people, I reject the allegation. If, however, the suggestion being made is that practices may be adopted by public bodies as well as by private individuals which are unwittingly discriminatory against black people, then this is an allegation which deserves serious consideration, and where proved, swift remedy.

(Scarman, 1981:11)

Deconstructing Carmichael and Hamilton's (1967) notion of institutional racism and Scarman's (1981) statement that discriminatory practices might be 'unwitting', there is in both an implied acknowledgement that these discriminatory practices can occur on an individual *and* institutional level. The word 'unwitting' in Scarman's statement is later revived through the inquiry launched into the Metropolitan Police Service following the racially motivated murder of Stephen Lawrence in 1993, and following consideration about the way the case had been handled presented institutional racism as:

The collective failure of an organisation to provide an appropriate and professional service to people because of their colour, culture, or ethnic origin. It can be seen or detected in processes, attitudes and behaviour which amount to discrimination through unwitting prejudice, ignorance, thoughtlessness and racist stereotyping which disadvantage minority ethnic people.

(Macpherson, 1999:28)

Gillborn (2008) highlights that in this contemporary definition there are two important features condemning the actions of individuals through conduct, attitudes and behaviours, as well as organisations whose processes disadvantage certain groups of people. By doing so, the inquiry moved away from the assertion that racism is limited to a few 'rotten apples' within an organisation and focuses on the outcomes of actions, by identifying 'unwitting' or 'thoughtless' acts that are as equally problematic as overt acts of racism. Furthermore, Ahmed (2012) made a valid point that solely blaming individuals underestimates the scope and scale of racism, leaving us without an account of how racism is reproduced. Individual negative attitudes and behaviours can be easily assimilated over time into practice, whether their

origins were unwitting or not and whether the reproduction of those attitudes and behaviours happen consciously. In defining institutional racism, these unwitting acts with adverse outcomes reinforces Pilkington's (2011) view that a racist discourse embeds itself within institutional processes. The disparities highlighted by studies within UK higher education and the suggestion that institutional racism exists in our universities (ECU, 2011) cannot be ignored further.

The Macpherson report (1999) continued to define unwitting racism as arising from a lack of understanding, ignorance, or mistaken beliefs together with well-intentioned but patronising words or actions. It can also arise from racist stereotyping of black people as potential troublemakers. Furthermore, such attitudes can thrive in a tightly knit community, so that there can be a collective failure to detect and outlaw this class of racism (Macpherson, 1999). These types of actions are also referred to as racial microaggressions (Rowe, 1990), and were explored earlier in this chapter.

Wight (2003) explains that institutional racism is nothing other than a combination of overt and unwitting racism — both of which are firmly based at individual level. Yet if we are to consider how racism is reproduced as Ahmed (2012) suggested then it will be necessary to move beyond addressing individual manifestations of overt or subtle acts of racism, particularly where those acts have become normalised in an organisational setting, creating the notion of collective acts of racism. It may be that this inconsistency creates a challenge for organisations to deal with institutional racism. Is institutional racism an outcome or a cause of racial prejudice, or both? The conceptual and practical understanding of institutional racism (Pilkington, 2011; Souhami, 2014) shows that organisations have found it difficult to deal with the issue and challenging to interpret and operationalise the definition in organisational practice.

The Macpherson report (1999) asserted that it was not the policies of the Metropolitan Police Service that were racist, but that it was the implementation of policies and in the words and actions of officers acting together that racism became apparent. Contextualising this within higher education, Pilkington (2011) remarked in his case study of a university that the development of a policy and action plan was solely viewed as a compliance exercise. As such, there was reluctance to accept specific race-related policy and initiatives from above and below, manifesting itself in staff as well as members of the governing body perceiving the

exercise as political correctness gone mad. Writing about reform in the police service, Souhami (2014) commented that there has been little sign of new development in tackling institutional racism in the context of police reform at either conceptual or practical levels and attributed this to a lack of understanding of how it might work as a conceptual instrument for change. The inability to operationalise institutional racism within organisations might be due to the terminology used, which Solomos (1999) suggests is essential in contextualising racist discourse and practice within distinct settings and how they shape the development of racial ideas and political actions. The difficulty, as Solomos (1999) identified, was that there were no clearly defined actions within the description of institutional racism in the Macpherson (1999) report.

Wight (2003) argued that institutional racism is linked to intentionality and is an agential characteristic not a structural one and this view is contrary to Gillborn's (2008) observation that Macpherson's notion of institutional racism is devoid of intentionality. These arguments appear to be circular considering that racial prejudice or discrimination need not be intentional and can be overt or symbolic in nature (McConahay, 1986; Rowe, 1990; Essed, 1991; Henry and Sears, 2002; Sears and Henry, 2003; Noon, 2017) and be perpetrated at multiple levels (van Dijk, 2002). According to EHRC's (2019) report some behaviours, such as those described by participants in this study (explored further in Chapters Six and Seven of this dissertation), together with the responses given as part of the inquiry into racial harassment in UK higher education, might constitute direct discrimination where a person has been treated less favourably because of a protected characteristic³ and is unlawful whether it was perpetrated consciously or not and regardless of motive or intention (EHRC, 2019).

Both Gillborn's (2008) and Wight's (2003) views are compelling, however their focus is centred around intentionality albeit from opposing sides. Despite this polarity, the notion of institution is critical to ensure that organisations can understand how, as an entity, it can affect the outcomes of the workforce by the way it constructs and implements its policies, services and practices. Institutional racism is that which, covertly or overtly, resides in the policies, procedures, operations, and culture of public or private institutions – producing certain kinds of practice and at the same time being dependent upon such practice. (Wight, 2003).

_

³ As defined in section 13 of the Equality Act 2010

There has been limited research on the macro-level racism within the context of UK higher education. However, a 2011 report by the Equality Challenge Unit (now Advance HE) suggested that institutional racism existed in UK higher education and cited other studies (Carter, Fenton and Modood, 1999; Blackaby and Frank 2000; Law, Phillips and Turney, 2004; Jones, 2006) that supported this statement. The matter was not advanced further throughout the report findings, nor did it offer specific recommendations to the higher education sector on how to deal with the issue of institutional racism. More recently Sian (2019) stated in an article discussing the extent of institutional racism in British universities that racism was endemic in UK higher education and named five myths about 'liberal' universities that supported this assertion as:

Myth 1: Universities encourage inclusivity and diversity

Myth 2: Universities invest in non-white academics

Myth 3: Universities are post-racial

Myth 4: Universities desire curriculum reform

Myth 5: Universities are committed to race equality

The approach taken by Sian (2019) provides the broader context of institutional racism by exposing the issues that lay beneath the continuing racial inequalities in the UK higher education sector. She suggests that since institutions are resistant to address these five areas this confirms that UK institutions are institutionally racist and sector statistics (Chapter Five), and previous academic literature (see Chapter Two) supports this. Ahmed (2012) also asserts that institutional racism has already been identified within higher education, and even if it did exist, the sector would not use the term to describe its existence because of its implied injury to the institution, and therefore an injury to whiteness. Recently, the chair of UUK publicly acknowledged that UK universities are institutionally racist, yet despite these narratives, the recent report from CRED (2021) claimed it 'found no evidence' of institutional racism in the UK. The Commission argues that use of the term institutional racism should only be applied when deep-seated racism can be proven on a systemic level and not be used as a general catch-all for any microaggression, witting or unwitting (CRED, 2021). The report suggests that to limit the widening charge of racism, and therefore the dilution of its importance, that

assessing the intent of the perpetrator as well as the perception of the victim would be required. A perspective that contradicts Gillborn (2008) and the importance of effect rather than intentionality. The Commission also calls for further clarity and standard definitions for institutional racism, structural racism, and systemic racism.

This section has explored the notion of institutional racism and the complexities associated with the organisational responsibilities required to tackle it. The literature debating the definition as posed by Macpherson (1999) is important since it provides insight into how this can be used to inform how organisations tackle it. However, further research in this area is required to understand the extent to which the UK higher education sector is engaging with this topic from an organisational perspective.

2.9 Conclusion

This chapter has explored the construction and evolution of race, which has informed science, legislation, and government policy particularly during the post-war period in the UK. Societal attitudes towards race have been influenced by history. Whilst transformation has been evident over the past sixty years, notions of kinship and national identity have played a significant role in the pace of change to address race relations in the UK and has to some extent informed public policy decisions relating to immigration. The UK's exit from the European Union has contributed to further perpetuate nationalist and xenophobic sentiment that has degenerated the progress made.

History has played a significant part in perpetuating an understanding amongst the general population that people are from different races. These models of difference have been further developed through the racial anthropology of the late 19th century, and in later centuries this has been further supported by racialised terminology, which has become preserved by the legal framework and sets the parameters upon which anti-racists have campaigned and within which equality and diversity practitioners work.

There is a clear timeline in terms of economic, social, religious, political and scientific agendas that have all contributed to the development of and the continuation of the notion of race. Economic and political power has been the driving force in establishing oppressive regimes that have used race as the central defining feature to control populations for the creation of

wealth and enable slavery to continue for a protracted period. These regimes have been eroded by social, moral, and ethical pressure that would change social attitudes. However, the impact on society's perception of a racial hierarchy, which was created and supported during these tumultuous times, has had a significant and long-lasting effect which continues to be reproduced across different societies in modern times.

The UK legislative framework has contributed to the continuing legitimisation of race as a term, despite race as a concept no longer being scientifically valid. Theoretically, academics can continue to debate the meaning of race, yet this has not made an impact at grass roots level in the search for the eradication of race discrimination nor racial prejudice. It seems unlikely that this will change soon as remedies for race discrimination are well established within the court and employment tribunal systems in the UK. With a political and industrial drive for 'diversity' and the value of diverse perspectives in a growing global economy, it is more likely to underpin the trend of continuing to capture and manipulate information based on well-established ethnic categories, such as those collected via the UK Census (Aspinall, 2009). One could argue that politics in relation to race has existed from the moment dominant groups oppressed other groups for gain and how this system of oppression formed the racial categorisation we are used to today. The racial hierarchy that measures all other persons against white Europeans has served to preserve white privilege and white supremacy, and this is evident in published literature and higher education workforce data.

Although social attitudes have evolved, contemporary UK society continues to carry racial prejudice because it has remained unchecked and unchallenged. In trying to understand and define the nature of racial prejudice the chapter has explored how the politics of race has manifested itself within political agendas and law-setting and how it has evolved from acts that are traditionally overt to acts that are modern, liberal, and symbolic. Whilst capitalism, and increasing nationalist agenda, white privilege, class structures and racialised oppression continues, it may not be possible for the foreseeable future to move away from evaluating racialised inequalities using existing systems and by default the language that maintains it. The complexities of these phenomena will be explored further in the findings of this research (Chapters Six and Seven). The year 2020 has been filled with discussions around race and many people will have seen, listened to, read, or taken part in narratives relating to the lived experiences of ethnic minority people. There has been a noticeable increase in representation

of minority ethnic people in newspapers, magazines, television programmes and advertising. This has been a positive move, but one that has not been well received by some of the general population, such as the racist backlash following Sainsbury's Christmas advertising campaign that featured a black family (Skopeliti for The Guardian, 2020). This demonstrates that although we are making some progress, we still have some way to go in achieving racial parity in the UK.

Racialised identity and socialisation to race is important to us all. People who are racially 'othered' are constantly perceived in opposition to whiteness and this can make people of colour hypervisible, or in certain settings, hyperinvisible. Race cannot be detached from other characteristics, such as socio-economic background and which can invariably affect an individual's capability to access social and cultural capital, and this will be explored further in Chapter Three in relation to structure, agency, and the role of power.

As we have discovered throughout this chapter, racism is not solely manifested through overt acts and a new racism has evolved that is hidden and easily misrecognised. People of colour who experience this new racism, e.g., microaggressions or microincivilities, may not report these incidences because they may be too frightened to do so or lack confidence in organisational processes to tackle it (EHRC, 2019). These outcomes should present a challenge to organisations, and particularly UK higher education to deal with recommendations effectively. This is particularly pertinent for institutions in terms of their processes and policies and how they are implemented. Institutional racism is not a new concept, and the literature demonstrates that UK public sector organisations have not taken a firm grasp of what it means to check for institutional racism in a practical sense, and more importantly the steps needed to eradicate it. The debate about the definition as provided by Macpherson (1999) has not been followed up in the same way as for example, guidance, such as codes of practice (EHRC, 2014). This absence should cause some alarm considering the impact that the differential implementation of policy and practice can have on different groups of people.

The next chapter will explore social theory and its importance in terms of the topic of race and racial marginalisation. The discussion will present the work of Pierre Bourdieu as an alternative lens upon which the literature and findings within this research can be considered to elucidate meaning from these in the context of exploring the differential outcomes for black staff in UK higher education.

3 Structure, agency and the role of power

3.1 Introduction

The previous chapter introduced the notion of race and ethnicity, the language we use as a society to speak about it, the nature of racism and how it has metamorphosed amongst individuals and within organisations, and how race has been politicised throughout the ages. The consequences of this shapeshifting have directly contributed to the way that individuals and organisations deal with matters of race and has created marginalised groups who have been and continue to be, affected first-hand by it, and alienates those who do not feel confident to discuss it. The construction and reconstruction of race was made possible by and through power relations, resembling monolithic structures that seem immovable and perpetual. This social, yet structural notion remains with us today.

This chapter will consider the underlying structures that maintain the status quo within our society, which perpetuate racial inequalities. Multiple studies over the past decade have highlighted the topic of race and racism within the higher education sector (ECU, 2011: Rollock, 2011, 2019; UCU, 2016; Bhopal and Brown, 2016; UCU, 2017; Bhopal, 2018; EHRC, 2019) and consistently articulate the disadvantage experienced predominantly by academic staff within institutions. Despite the continued attention in this area, including through research undertaken by sector agencies and trade unions, there has been little change in the outcomes of racially minoritised staff in the UK higher education workforce (see Chapter Five). These publications should have prompted institutions to act, and therefore a renewed and alternative perspective on the problem would be advantageous. This alternative perspective will be explored here and will consider how the social agents within our institutions contribute at multiple levels to produce and reproduce workplace cultures, whether intentionally or not, and create exclusionary environments and unequal outcomes for black staff. Because these outcomes are a social phenomenon set within a particular context, the social relationships developed and maintained by individuals at micro level will influence those at meso level and those created and maintained at macro level. These relationships are intertwined across multiple levels and cross multiple social groups.

This chapter will begin by exploring social theory to understand the schools of thought around why social actors do what they do, the concepts of structure and agency and consider their relevance to the topic under enquiry within this dissertation. The discussion will progress by considering the work of Pierre Bourdieu, a French sociologist, anthropologist, philosopher and public intellectual, who advanced the sociology of education through several concepts that consider human agency and structures. There has been a recent proliferation within academia in the use of Bourdieu's (1977; 1984; 1990; 1998) concepts to explore human resource management topics and as such, this research will benefit from analysis using this lens to investigate the findings in an innovative way that increases our understanding of agential dynamics and the power of organisational structures and systems.

3.2. Social structures and their influence on human agency

According to Aristotle, humans are social beings. As individuals we interact with other individuals or groups on a regular basis, whether intentionally or not. For Giddens (1984), to be a human being is to be a purposive agent. Our interactions might include activities within familial networks or extended through associations with people who share similar identities, beliefs or any other characteristics that can maintain a group structure and which may be oriented towards the satisfaction of material and symbolic interest (Bourdieu, 1977). Outside of our closest networks, people interact within wider society in ways that might not necessarily be obvious to the lay person, that is pre-existent and established across multiple levels. At macro level, the social structure relates to the system of socio-economic stratification, such as class (Marx in Jones and Bradbury, 2017; Moore, 2005) or other patterned relations between large social groups, such as organisations. At a meso level, it relates to the social networks, such as friendship groups, affinity groups, etc. At a micro level, the social structures include the way that social and cultural 'norms' inform individual behaviours within that social system.

Social and cultural norms are a set of rules, attitudes and behaviours that have been developed over time by individuals and/or groups and to which we are socialised by family, friends, peers, and work colleagues. These cultural norms influence social structures through relations between individuals and/or groups to the point that the prevailing norms are set by

the dominant group. The relationship between those that conform to the pre-established cultural norms and those that do not, creates a hierarchical social stratification that favours the dominant group (Jones and Bradbury, 2017). There are several social theories that articulate society as a structure of rules, which influence and inform the behaviours of social actors. One such notion is consensus theory (Jones and Bradbury, 2017), which argues that society's cultural rules determine or structure the behaviour of its members, and regards human behaviour as learned behaviour, gained through socialisation, and suggests that people learn expected behaviours in social settings. However, these can be constrained by cultural rules that might apply, such as how a person was brought up as a child, which could be informed by multiple factors such as religion, value system, generational influence, household composition, etc. In opposition to this is conflict theory (Jones and Bradbury, 2017), which poses that society determines our behaviour by structuring or constraining it, and that the real structural determinants of behaviour are the rewards and advantages possessed unequally by different groups in society. This means that the origin and persistence of inequality lies in the domination of society's disadvantaged groups by dominant groups (Jones and Bradbury, 2017). Giddens and Sutton (2017:201) assert that the quest for power, attempts to gain social status and social inequalities lead to the formation of distinct social groups with shared interests and identities that pursue those interests against others, and therefore, conflict theory sees the potential for strife as always present.

Jones and Bradbury (2017) suggest that there are two ways that a structure of inequality can survive. Firstly, if the most disadvantaged by the structure can be prevented from seeing themselves as disadvantaged, or secondly, and even if the first is recognised, that they can be persuaded that their disadvantage is legitimate. The way this can occur is through control and manipulation of the norms and values or cultural rules, into which people are socialised. The authors argue that socialisation is likely to be an instrument of power, producing social order by means of force and domination in which the simplest way for the dominant to exercise power and maintain their advantage, is if the dominated are complicit in their own subordination (Jones and Bradbury, 2017).

Whilst consensus and conflict theories provide particular perspectives about social structures, those structures are recurrent arrangements, which influence or limit the choices and opportunities available. Structure is a concept that depicts a durable framework, defined by

sociologists such as Giddens (1984) and Bourdieu (1977) as rules and resources that are both the property of, and which give shape to, social systems. In contrast, agency is the capacity of individuals to act independently and to make their own free choices. The structure versus agency debate may be understood as an issue of socialisation against autonomy in determining whether individuals act as free agents or in a manner that is dictated by social structure. Giddens (1984:9) suggests that agency refers not to the intentions people have in doing things, but to their capability of doing things in the first place. This is further supported by Bourdieu (2001a) who asserted that far from being the conscious, free, deliberate act of an isolated 'subject', the practical construction of (and response to) symbolic power is in itself the effect of a power, durably embedded in the bodies of the dominated in the form of schemes of perception and dispositions, which sensitises an individual to certain symbolic manifestations of power (Bourdieu, 2001a:40). Interpreting this, does this amount to free will at all if as agents we replicate behaviours that are firmly entrenched?

For Giddens (1984), structure is the product of routine activities that enable and constrain agents within the social space and is reproduced through the regularised conduct of knowledgeable social agents. The rules Giddens referred to can be viewed as generalised procedures that can apply in various contexts and which allow for the methodical continuation of an established sequence, such as human resource management, legislation, etc. These rules tend to be reproduced with such consistency that they take on an objective form, and within an institutional context will provide the parameters upon which social agents will interact and coexist. The resources Giddens refers to are a form of power, which are either authoritative (in the control of people) or allocative (in the control of material objects). Giddens furthered his theory of structure and agency, and the duality of structure (structuration) by, instead of describing the capacity of human action as being constrained by powerful stable societal structures, such as religious, educational, or political institutions, or as a function of the individual expression of will, e.g., acknowledging the interaction of meaning, standards and values and power and posited a dynamic relationship between these different components of society.

Actors can be both enabled and constrained by structures and agency is not a straightforward matter of choice. In considering New Labour's social policy, Greener (2002) considered Gidden's structuration theory and the extent to which structures enable or constrain an

individual's agency and suggests that a non-reflexive agent would behave instinctively, with little control of their environment. Greener (2002) states that resource constraints may prevent actors from acquiring capital and that structural, habitual, or informational constraints may severely restrict the range of actions an agent might make, because their actions have become so ingrained. Therefore, if an actor does not, or cannot, act, they appear from a structuration perspective, to have no agency. Giddens (1979) argues that humans are knowledgeable social agents that will always have some role and/or power available to them despite constraints. Gidden's notion of the 'dialectic of control' holds that agency and power are inextricably linked and no matter how asymmetrical, the power relations are always two-way, contingent and to some degree, interdependent (Collinson, 2016).

The concepts Giddens (1984) presents are similar to and complement those introduced by Bourdieu (1977; 1984; 1990; 1998) and although the terminology is different within the Bourdieusian context, there are some noticeable distinctions where Bourdieu advanced the notion of structure through the concepts of field and habitus, which will be explored further in this chapter. Kirchberg (2007) observes that structures are not only dependent on agencydriving individuals; they are the results of social interactions, particularly those interactions that become routines with the highest degree of accountability and that Giddens' notion of routine is similar to Bourdieu's concept of social practice, where routines can create and change practices, just as practices can create and shape routines. However, Bourdieu also explains to some extent how structuration occurs by providing details of the process of structuration, such as the internalisation and reproduction of social processes. This framework enables a deeper consideration of the capacity and capability of agents within social relations through the concept of capital, which impacts on the extent an agent can create or shape practice on an organisational scale, and that incorporates multi-level interplay between agents and management structures. These never-ending cycles of structuring structures are particularly relevant to provide an interpretation of perceptions and lived experiences given by participants in this study and which will be explored further in subsequent chapters. In the meantime, the following section will explore how Bourdieu's concepts can further provide a dynamic lens upon which to analyse the empirical data of this research.

3.3 Using Bourdieu as a lens for exploration

Tatli (2011) drew attention to the lack of empirical work that operationalises Bourdieusian concepts for the investigation of workplace equality and diversity. It is useful to apply the concepts of Pierre Bourdieu (1977; 1984; 1990; 1998) and Bourdieu and Wacquant (1992), which include habitus, field, capital, and symbolic violence, to understand how the structures formed within our institutions contribute to the outcomes of racially minoritised staff and how the role of individual agents produce and reproduce these structures within the context of UK higher education. Bourdieu set out through his various theories (1977; 1984; 1990; 1998), and particularly his notion of habitus to overcome the dualism of objectivity and subjectivity.

Bourdieu's *Homo Academicus* (1984) is an ethnological and sociological analysis of the academic world in the French higher education system and is particularly relevant for this study in terms of exposing power relations and structures within the institutional social space that enables and constrains academic careers. Bourdieu's analysis and subsequent development of the concepts of field, habitus, capital, and symbolic violence will allow for the societal, organisational, and individual enablements and constraints within UK higher education to be examined and how they might shed light on why racial disadvantage persists. Social space is constructed in such a way that agents or groups are distributed in it according to their position based on two principles of differentiation, economic and cultural capital. Reality according to Bourdieu (1977) is a social concept and to exist is to exist socially and what is real is relational to those around us. Agents are distributed in the overall social space, in the first dimension, according to the overall volume of capital they possess and, in the second dimension, according to the structure of their capital, that is, the relative weight of the different species of capital, economic and cultural, in the total volume of their assets (Bourdieu, 1998:6).

This is particularly relevant within a higher education context, which contains hierarchical organisational structures, and here it can be proposed that a university professor would very likely hold more overall capital for being positioned at a higher level of the organisational structure, and therefore as a result attract a higher salary (economic) and higher cultural capital by virtue of their field (academia, subject specialism and value placed by the institution on that capital etc.). This is in comparison to a lecturer, who would hold a lower position in

the institutional hierarchy as well as potentially hold, or be perceived to hold, less cultural capital by virtue of their positioning. Interestingly, a professional and support staff member may not possess the same overall capital, despite being positioned at the same level within the institution as a professor or lecturer, which may be due to the value of symbolic capital placed upon academic staff compared with professional services staff within the institutional community (Whitchurch, 2006; Deem, 1998).

Bourdieu defines the notion of space as containing:

...the principle of a *relational* understanding of the social world. It affirms that every 'reality' it designates resides in the *mutual exteriority* of its composite elements. Apparent, directly visible beings, whether individuals or groups, exist and subsist in and through *difference*; that is, they occupy *relative positions* in a space of relations which, although invisible and always difficult to show empirically, is the most real reality...and the real principle of the behaviour of individuals and groups

(Bourdieu, 1998:31).

Applying this definition for the purpose of exploring the social dynamics of the UK higher education sector will provide a valuable perspective to further understand the enablements and constraints of the workforce, and particularly those who are marginalised and/or disadvantaged at work. The remainder of this chapter will explore Bourdieu's concepts in more detail and subsequent analysis utilising the concepts of habitus, field, capital, and symbolic violence will be applied further in chapters six and seven when considering participant accounts.

3.3.1 Social space as a field of power

Although Bourdieu related his concept of social space on a societal level, this is equally fitting of an institution, where the environment represents a microcosm of wider society. Within the confines of the institution the workforce holds an assumed mutual understanding of the environment in which staff reside, noting their differences whether as individuals or in groups according to the multiple categories of difference, such as gender, race, role, grade, perceived capital, etc. These social spaces appear as structures of differences, which are mutable structures of the distribution of forms of power or the kinds of capital that are effective and

define the rules and resources that are legitimate within that space (Bourdieu, 1977; 1991). Bourdieu refers to these spaces as *fields*; a field of forces imposed on agents and as a field of struggles within which agents confront each other with differential means and ends according to their position in the structure of the field of forces that contribute to the conservation or transformation of the structures (Bourdieu, 1998:32). Bourdieu felt that any social research should begin with defining the field within which the investigation is situated. However, the boundaries of a given field can only be understood empirically in a specific point in time because the boundaries of fields are dynamic and therefore, not fixed (Bourdieu and Wacquant, 1992).

Within the *field* power struggles and conflict take place and specific kinds of capital (economic, cultural, social, symbolic, etc.) are at stake and certain forms of habitus or dispositions are fitted for success. This helps us to understand how certain agents can be powerful in some fields and much less so in others, even though capital can sometimes become transferable between fields (Gaventa, 2003). The concept of field is particularly relevant within higher education because of its hierarchical and sometimes isolated and isolating structures that can impact inter- and intra-relational actions and practices between different groups of staff e.g., academic, professional and support staff. As suggested in a recent study, conformity to rules and roles is played out in particular ways dependent on the level and type of capital held by the agents within the field (Randle et al, 2014). Similarly, it is important to consider the field of equality and diversity practitioners and how their roles within institutions are sponsored, facilitated, or hindered in addressing racial inequalities. The social space they occupy is constructed of relationships with key institutional agents in the broader field of academia, however there are competing priorities within institutions that make it difficult to advance the equality and diversity agenda. Özbilgin and Tatli (2011) suggest that equality and diversity actors have unequal access to, and ownership of, power and resources, which constitutes a significant imbalance in the struggle for domination and legitimacy, thereby impeding any cultural or organisational change.

In defining spaces, agents play their part in creating the rules of the social game, what Bourdieu (1998) refers to as *illusio*, where agents must be invested in the game and see it as worthwhile, as well as understand the boundaries upon which agents are relationally engaged. If the agents' mind is structured according to the structures of the world in which one plays,

everything will seem obvious and the question of knowing if the game is worthwhile will not even be asked (Bourdieu, 1998). The stakes of the game will appear as an illusion for those who do not participate in the game, and each field will impose its own tacit entry fee to comply with the rules of the field, resulting in subscribing to the illusio of the field (Tatli, 2011). In defining the field, Bourdieu does not make distinctions between the agents that would occupy a field and here it is particularly apt to mention that contextualised within this investigation around the field of higher education, there are several agential qualities that may negatively impact upon an agent's access to the game, and whether agents have the requisite knowledge of the rules of the game within a given field. The struggles and confrontations Bourdieu posited are particularly relevant in understanding the lived experiences of racially minoritised staff and the extent to which institutional structures are conserved or transformed because of an agent's ethnicity. Viewing agential interaction through this lens presents an alternative view of daily institutional life within the constraints and enablements that an agent's position within the field provides, and accounts for other differences that are not job specific, e.g., personal characteristics, background and experience that will have an impact on their position-taking as agents within the field.

3.3.2 Habitus - A feel for the game

The concept of habitus has a long history and has been subject to widespread criticism, mainly because of its perceived determinism (Reay, 2004; Morrison, 2005), yet Bourdieu tried to counter this by developing the concept to demonstrate that not only was the body part of the social world, but the social world part of the body. Despite Bourdieu's focus on gender and class within his work, there is great scope to apply his concepts to matters of race within an educational context. Bourdieu (1977) posed that dispositions are the internalisation of structures, which influence the practice of individuals and/or members of a group. Therefore, exploring the habitus of staff in UK higher education will highlight individual experiences and expose how agent dispositions impact the choices of racially minoritised staff to navigate through that system and the social spaces contained within it.

Bourdieu's (1984) theory around class also helps us to understand the role of agency in shaping class identity and structure, where class is primarily conceived and communicated by consumption. The choices a person makes indicates their level of cultural knowledge as well

as an assertion about their class identity. Moore (2005) suggests that to claim a particular class identity, a person must be familiar with the cultural markers of that identity, have the resources to participate in that lifestyle and demonstrate the cultural and material resources to others in the status group. Moore (2005) emphasises that Bourdieu's (1984) model of class introduces the importance of performance as a central part of asserting class identity and that class is also about social and cultural relationships. Moore (2005) articulates that there are notable differences between black and white class structures, with black class structures relying on social distinctions, such as education and cultural practices rather than occupation and income.

Bourdieu (1977) saw power as culturally and symbolically created, and which is constantly legitimised through the interplay of structure and agency. This happens by virtue of an agent's habitus or socialised norms or tendencies that guide behaviour and thinking. Each agent, wittingly or unwittingly produces and reproduces objective meaning, and it is because subjects do not know what they are doing that what they do has more meaning than they know (Bourdieu, 1977). Habitus are systems of durable, transposable dispositions that are structured structures predisposed to function as structuring structures (ibid.) and are created through a social, rather than individual process, shifting from one context to another over time. Bourdieu later extended his view of habitus by suggesting that agents are active and knowing and endowed with a practical sense, an acquired system of preferences, of principles of vision and division, which are essentially the product of the internalisation of objective structures, which in essence is the knowledge of what needs to be done in a given situation. Bourdieu called this a 'feel for the game' (Bourdieu, 1998:25). Hay and Wincott (1998) suggested that since individuals (and groups) are knowledgeable and reflexive, they routinely monitor the consequences of their actions. In doing so, they assess the immediate and unfolding impact of their strategies in relation to earlier intentions and anticipated outcomes in light of the assessment of the conduct of those around them. As a result of this, Hay and Wincott (1998) argue that strategic actions then yield:

Direct effects upon the institutional and institutionalised contexts within which it takes
place and within which future action occurs, producing a partial transformation of the
institutional environment

Strategic learning on the part of the agents involved as they revise their perceptions
of what is feasible, possible, and desirable considering their assessments of their own
ability to realise prior goals (which complements Gidden's (1979) notion of a dialectic
of control).

Within the context of the higher education environment, a person will enter the system possessing their own specific habitus, constructed from their own background, personal characteristics, and previous experience. Depending on how this habitus has been constructed and the agent's own system of preferences, an agent's engagement into this social space already constructed by existing agents in the field, will determine the agent's positioning within that field, as well as determine future perceptions in relation to other agents around them. This pre-existing 'organisational habitus' will shape the agent's own habitus, further creating a new set of dispositions and preferences, that will impact on practice and possibly the field. This transformation may restructure existing structures; however, this may be dependent on how that transformation is legitimised by other agents in the field.

Organisational habitus is a concept that has been adapted from Bourdieu's concept of habitus and refers to the class-based dispositions, perceptions, and appreciations transmitted to individuals in a common organisational culture (Horvat and Antonio, 1999). McDonough (1998) outlined the concept of organisational habitus to describe college counselling programmes offered in a US high school setting, and how the sociocultural demographics of the school would inform the level of support provided to wealthy students compared with students of low income. Kolluri (2019) offered a more complete conception of organisational habitus that wrestles with the notion that organisations exist at the intersection of numerous, interconnected social arena considering that organisations, like the individuals that comprise them, straddle a variety of communities with diverse cultural norms and values. An agent's habitus will inform them of their own position-taking within the organisational habitus and will evolve with their immersion within this environment. An organisational habitus could be present at multiple levels, such as that which is present within sub-cultures within organisations. The enablements and constraints of organisational habitus are theoretically similar to those on an individual basis if one interprets the organisation as a collective of individual agents with diverse cultural norms and values as Kolluri (2019) suggests. The position occupied by key individuals in the social space will affect the distribution of different kinds of capital and command the representations and position-taking within this space in the struggle to conserve or transform it (Bourdieu, 1998).

In writing on the topic of habitus and the practice in Canadian public service, McDonough (2006; 2016) argues that front-line public servants' narratives reflect a public sector habitus, a socially constituted set of dispositions representing the internalisation of a dominant vision that privileges the public good over private interests. The symbolic force of the public good is embodied in, and reproduced by, public servants in their daily practice. To fully understand the pervasiveness and reproduction of cultural privilege Bourdieu argues that the norms and practices of educational institutions must first be examined, as he believed that those institutions held the most power in perpetuating one's level of status and privilege (in Payne, 2015). Considering agential and organisational habitus, such as that undertaken by McDonough (2006; 2016) will allow for a more comprehensive understanding of how higher education institutions, or the agents within them, contribute to the perpetuation of dispositions that produce or reproduce racial inequalities. This concept may be fundamental in identifying transformational solutions to address workplace race equality within UK higher education.

Agents will position themselves within the social space according to two main principles of differentiation. In doing so, agents themselves will bring a set of internalised dispositions that will inform how one navigates the social space. However, there are other fundamental considerations relating to the extent to which agents have access to Giddens' (1984) notion of resources, e.g., power. Bourdieu extended the concept of *capital* beyond the notion of material assets (economic) to capital that might be social, cultural or symbolic. However, Mander (1987) suggested that it is the differential distribution of capital that structures society, yet agents can have an impact upon the level of capital in their possession. Occupants of positions will employ structured, yet not structurally determined, strategies to defend or enhance their position in relation to these capitals. Bourdieu (1986) asserts that capital can present itself in three fundamental ways; economic, cultural and social, which can be accumulated and transferred from one area to another and be exchanged for other forms of capital. For Bourdieu (1986) capital is accumulated labour, which in its materialised or incorporated, embodied form, enables a person to appropriate social energy in the form of reified labour that has the potential to affect individual life chances.

3.3.3 The power of connections

Bourdieu and Wacquant (1992) defined social capital as the sum of the resources, actual or virtual, that accrue to an individual or a group by virtue of possessing a durable network of more or less institutionalised relationships of mutual acquaintance and recognition. Group membership provides each of its members with a collectively owned capital, which symbolically can be deemed to be a credential by virtue of group membership. The volume of social capital possessed by an agent depends on the size of the network of connections and on the volume of capital (economic, cultural, or symbolic) possessed by them and by each of these connections. Social capital is not completely independent of the economic and cultural capital possessed by an agent because the exchanges institute mutual acknowledgement that presupposes the re-acknowledgement of a minimum homogeneity, and because it attracts a multiplier effect on the capital the agent possesses in their own right (Bourdieu, 1986).

A network of connections or relationships can be for most the product of investment strategies, whether individual or collective, consciously or unconsciously aimed at producing or reproducing social relationships that are directly usable. Each member of the network is a custodian of the limits to the group, and as such the criterion for entry is at stake with each new entry and may expose the group to redefinition or alteration. Groups will favour legitimate exchanges, excluding illegitimate ones, by producing occasions, places or practices that will bring together individuals that are as homogenous as possible in order to maintain the group dynamics (ibid.) Examples of how this may manifest within the academic environment can be networks comprised of individuals who are similarly qualified, and this may be strengthened among academic staff within the same academic discipline and/or field, their academic reputation, output or even previous experience, which may include previous institutions. The criteria for 'entry' into a collective may be based on any one or multiple capitals within these categories and is not solely exclusive to academic staff as professional and support staff networks may use similar conditions for the creation and reproduction for group membership.

The possession and accrual of social capital is an important part of institutional life, particularly when the accrual of an overall volume of capital can attract gains within the structural system.

Consideration of social capital can presume that all agents are equally competent at

developing or maintaining social networks, however Bourdieu (1984; 1986) asserts that considerable time and effort are required to achieve this, which is integral to the accrual of social capital. Compton and Meier (2016) considered social capital in relation to student outcomes and concluded that stronger social networks, shared norms of reciprocity, civic participation, and trust among individuals and institutions should encourage cooperation and productive sharing of information to improve performance and outcomes. However, a social capital rich group may benefit more because social capital is not equally distributed and may lead to disparities in the outcomes of racially minoritised groups (Compton and Meier, 2016) and to mitigate some of the effects of the absence or lack of social capital among the most disadvantaged groups, a commitment to diversity management in its broadest sense is required.

Social capital, through the power of personal connections and their extended networks provides the beneficiary with opportunities within an institutional environment to gain legitimacy and credibility in ways that may not be possible on one's own. This is distinct from cultural capital, which will be explored in the subsequent section however, this attribute can be gained through the appropriate connections.

3.3.4 Knowledge is power

Cultural capital, and the means by which it is created or transferred, plays a key role in societal power relations as this provides the means for a non-economic form of domination and hierarchy (Gaventa, 2003). Cultural capital can be defined as the cultural knowledge that serves as a form of 'currency' and which can help us navigate society; in turn, this can alter the work-life opportunities available. These are non-financial assets that can help promote social mobility, whereas a lack of these assets can greatly limit social mobility (Randle et al, 2014). Bourdieu (1986) emphasised that there are three states of cultural capital: embodied, objectified, and institutionalised. In the embodied state, cultural capital consists of both the consciously acquired and passively acquired properties, e.g., language and socialisation. In the objectified state, cultural capital consists of material objects that are owned. These cultural goods can be transmitted both for economic profit and for the purpose of symbolically indicating social class. Bourdieu (1998) argued that Marx (1848) provided a false theoretical solution by affirming the real existence of class. For Bourdieu, the real class, if it has ever really

existed, is nothing but a realised class, the mobilised class, and the result of the struggle of classifications, which is properly symbolic (Bourdieu, 1998:31). In modern times stratifying agents in this way has become increasingly difficult because the boundaries of traditional class structures in the UK have blurred with post-industrial modes of production that have shifted from the production of material goods in the Marxist sense to the production of knowledge (Atkinson, 2015:26). In the institutionalised state, cultural capital is the way society measures social capital and can consist of institutional recognition, such as that afforded through academic credentials or qualifications and this can be evident in the access given to agents within specific professions, such as the legal and medical professions and within academia, where there are minimum objectified expectations required to practice in these fields.

For Bourdieu, race is an aspect of embodied cultural capital that creates certain dispositions within the individual; dispositions that define one's socially constituted nature (habitus), determining an individual's social positioning and therefore, the nature of the strategies that will be employed in social actions. If race, as embodied cultural capital, creates certain dispositions within an agent who is racially minoritised within the institution that is dominated by white agents, this environment could have a significant impact on the strategies employed by those minoritised agents in navigating that social space in terms of attaining further capital. These concepts can also explain the current subtleties of racism through the concept of symbolic violence (discussed in the subsequent section), where overt racism has become unacceptable due to the evolving societal and legislative landscape that has enabled its concealment.

Bourdieu (1998) describes power in terms of *symbolic capital*, which comes with social position and status. It can be referred to as the resources available to an individual based on honour, prestige, or recognition, and serves as value that one holds within a culture or field. One could argue that Bourdieu's concept of capital explores class without other characteristics, e.g., gender, and Huppatz (2009) made a valid point about the misrecognition on Bourdieu's part of gendered capital and how women are equally able to engage in the accumulation of different forms of capital as men, yet women can be excluded by the dominant male hierarchy. This can be particularly amplified when gender intersects with race. In classifying subjects, Bourdieu refers to age, gender, and social class, yet not to race and like gendered capital, there is scope to acknowledge racialised capital and how this might be used

for gains in specific fields by minority ethnic agents as well as translated on the contrary to become what we may recognise as white privilege.

Bourdieu defines symbolic capital as:

...any property (any form of capital whether physical, economic, cultural, or social) when it is perceived by social agents endowed with categories of perception, which cause them to know it and to recognise it, to give it value (Bourdieu, 1998:47).

Symbolic capital is earned on an individual basis and may fluctuate between members in a community. Özbilgin and Tatli (2005) observed that this conceptualisation embodies both subjective and objective properties and is formed through shared meanings of value and worth, which can also be demonstrated through organisational narratives of meritocracy, schooling, and professional experience as legitimate indicators of individual worth.

3.3.5 Symbolic power as a means of domination

Kamoche et al (2014) described that symbolic violence manifests itself in three ways: it seeks to change what is at stake through the power of pedagogy; it invokes mechanisms of social control, which are not always explicit; and it works through misrecognition. Their study examined how the dominant (senior management) team sought to control through complicity of the dominated (scientists) cohort to take on a knowledge-management system in a global organisation. This is relevant to this research study as an exploration of the notion of symbolic violence within a UK higher education context focuses on the relation of domination in a relatable context, e.g., senior management vs academics, and in what Kamoche et al (2014) described as factions within dominant classes united by habitus. It could be argued that a similar social space operates within UK institutions also, except there is an additional layer of complexity when the whole workforce is considered to include professional and support staff in institutions where there may be institutionally constructed 'class' structures at play and therefore there are multiple and multi-layered dominant-dominated groups interacting in the social space.

Using symbolic power against another implies symbolic violence and may take such forms as dismissal and judging the person inferior. This power may be dispensed without words, using physical symbols and behaviours. Bourdieu posed that symbolic power constitutes the given

through utterances, of making people see and believe, of confirming or transforming the vision of the world, action on the world and therefore the world itself and is a power that can be exercised only if it is recognised, or misrecognised, as arbitrary (Bourdieu, 1991). Symbolic violence is imperceptible, insidious, and invisible and its invisibility is an effective tool of silent domination. According to Thapor-Björket et al (2016) silence cannot be overcome by allowing the dominated to speak or for them to voice their concerns, as such acts are futile. The authors suggest that domination arising from symbolic violence is less a product of direct coercion and more an outcome of when those who are dominated stop questioning existing power relations as they perceive the world and the situation as natural, given and unchangeable.

Bourdieu's concept of *symbolic violence* was introduced to account for forms of coercion effected without physical force, which is "gentle, invisible violence, unrecognised as such, chosen as much as undergone, that of trust, obligation, personal loyalty, hospitality, gifts, debts, piety...presents itself as the most economical mode of domination because it best corresponds to the economy of the system" (Bourdieu, 1990:127). The use of symbolic violence is often so entrenched that the dominant party may not be aware that the behaviour is being perpetuated against the dominated party (Johnson et al, 2008). Symbolic violence is exerted wherever power imposes meaning and legitimacy yet conceals the power relations which impose those meanings, such as policy, practice etc. The policy and practices that assume a place of dominance are those that correspond with the interests of the dominant group and any action taken because of those policies or practices is accompanied by authority, giving it persuasive powers. Recognising the legitimacy of the dominant culture is at the same time delegitimising the dominated culture (Mander, 1987).

This symbolic power does not reside within symbolic systems. It is defined in and through a relation between those who exercise power and those who submit to it. In contextualising Bourdieu's concept of symbolic power, it is not the policy, practice, communication, etc. (symbolic system) where the power resides, but the method in which the policy, practice, communication, etc. is executed and Bourdieu further makes this point in his article relating to television (2001b) where he suggested that there were mechanisms that could wield a pernicious form of symbolic violence with tacit complicity between victims and agents where both remain unconscious of submitting or wielding to it. Symbolic violence occurs when the policies and practices, that on face value give the appearance of fairness and transparency,

are used as weapons to inflict domination and that silence groups into submission. To support this, Healy et al (2011) argued that workplace inequalities based on gender and race might be legitimated in practice through rationalisations based on different capabilities and negative stereotyping.

The interrelation of symbolic power with habitus was addressed by Bourdieu (1990) in relation to the division of labour between the sexes. Here, Bourdieu refers to the unconscious, collective creation being the basis of its durability and its transcendence to individual consciousness. On the topic of masculine domination, Bourdieu (2001a) argues that the socially constructed division between the sexes, as natural and self-evident, therefore contain full recognition of the legitimacy of the division of labour. Bourdieu states that the embodied social programme of perception is applied to all things of the world, and firstly to the body itself. Although Bourdieu writes in relation to gender and the perceptions created through its social construction, it can equally be applied to race in that despite these concepts being socially constructed they have become objectified, legitimised, and embodied through perceptions of the self and others. Bourdieu suggests that it is this (social) programme which constructs the difference between the biological sexes (and races) in conformity with the principles of the vision of the world, which is rooted in the arbitrary relationship of domination of men over women (whites over ethnic minorities) (Bourdieu, 2001a:11).

In furthering the Bourdieusian concept, Kamoche et al (2014) state that symbolic violence cannot be viewed in isolation because of its interrelation with Bourdieu's other concepts of capital, field and habitus, whereby symbolic violence is objectified through material objects, qualifications and other forms of cultural capital that affirm an individual's status and identity (Kamoche et al, 2014). Furthermore, misrecognition is key to the act of domination and Kamoche et al (2014) argued that 'cultural arbitrary' is a condition that expresses the arbitrary imposition of power by misrecognising its effects and purpose, which consequently reproduces and legitimises social inequality. The way in which dominated individuals or groups respond to symbolic violence will consequently become informed by their habitus, field, and capital, but as Kamoche et al (2014) discovered in their case study, so can peer pressure, which organisations may rely on to achieve workforce compliance.

For Bourdieu, the concept of symbolic violence informs wider theory on power and domination to explain how social hierarchies and inequalities are maintained not by physical force but rather by forms of symbolic domination. Bourdieu (2001a:1-2) also describes symbolic violence as a form of violence that is 'exerted for the most part ... through the purely symbolic channels of communication and cognition ... recognition or even feeling'. Power operates through subjective misrecognition of the meanings implicit in the action, practice and ritual, and any language (the language of the establishment) that can command attention is an *authorised language* (Bourdieu, 1977) and thus becomes legitimate. The authorised language silences other voices in the narrative through misrecognition. This can be explained by what Bourdieu refers to as the 'magical frontier between the dominant and the dominated', through which the dominated unknowingly and unwillingly, 'contribute to their own domination by tacitly accepting the limits imposed, often taking the form of bodily emotions, e.g., shame, humiliation, anxiety, or guilt' (Bourdieu, 2001a:38). It is this 'magical frontier' that will also be explored further in considering the accounts of participants interviewed as part of this research in Chapters Six and Seven.

3.4 Conclusion

This chapter has explored the significant role that social structures play in understanding human interactions within the social sphere. These structures are not necessarily physical in our general understanding of the term, such as objects, buildings, or institutions. They are also invisible and can be controlled by any one of us at any moment in time. Despite the inconspicuous nature of these structures their effect can be life changing and career limiting from a positive and indeed negative perspective.

The foundations of social theory provide a duality, that although much debated, offers the opportunity to consider the factors that influence the decisions social agents are confronted with, whether through socialisation or free will. King (2016:2000) argues that Bourdieu's attempt to move away from subjective/objective dualist social theory fails by relapsing to the objectivism that Bourdieu rejects, yet equally recognises that Bourdieu's contribution to the structure-agency debate cannot be ignored. The structures that surround social agents can be enablers as much as they can constrain, yet as Giddens (1984) theorised, human interaction is more dynamic than previously considered and therefore not mutually exclusive within that binary system.

Bourdieu advances on Giddens' structuration theory through his concepts of habitus and field, providing additional tools to explore the extent to which individuals can operate free will within the context of academia. Similar critique of Bourdieu's theories around field and capital have been reflected in Joas and Knobl's (2011:21) essay asserting a remarkable proximity to utilitarianism, despite Bourdieu's criticism of utilitarian social theorists, such as Habermas and Marx. Despite these criticisms, the work of Bourdieu offers theories that will facilitate a richer exploration of the research questions of this study. These theories provide a relevant and appropriate conceptual framework to aid the exploration of the extent to which 'power', in all its forms, operates at multiple levels across the UK higher education sector.

Considering the role structure and agency play in producing and reproducing practice and its effect on social agents within the UK higher education sector provides a powerful lens on a persistent problem. The broader exploration of Bourdieu's (1977; 1984; 1990; 1998) concepts of field, habitus, capital, and symbolic violence enables us to reframe institutional structures and systems to understand the social relations between agents within the socially constructed environment of our institutions. These societal, organisational, and individual elements represent UK higher education as a social space, which is occupied by various agents engaged in social relations across different levels of the organisational structure. This will provide a major contribution to knowledge and practice in terms of how the Bourdieusian notion of power manifests itself within institutions and how they may be operationalised to investigate equality and diversity, and more specifically for this dissertation, race.

Applying Bourdieu's concept of symbolic violence will reframe our gaze to allow an alternative analysis of how institutional structures, systems and practices might appear fair and transparent on the surface, and be misrecognised as such, yet their application affects different groups of staff in distinct and disproportionate ways. These concepts will be further applied to consider the responses of the research participants interviewed as part of this study and these findings will be presented in Chapters Six and Seven of the dissertation. The following chapter will present the philosophical approach taken in approaching the topic and the methodology and methods employed to undertake this research study.

4 Methodology and methods

4.1 Introduction

The aim of this chapter is to position this research within an ontological and epistemological perspective and describe the process utilised in undertaking this study. Throughout this chapter there will be references made to actions and considerations in the first person to personalise the activities that have brought this study to fruition.

The chapter will begin with an exploration of the methodological approach taken and how this has been informed by relevant concepts related to the topic under enquiry. The chapter will also consider the research strategy employed and will describe the methods utilised in the research in terms of the data collection and subsequent data analysis techniques and will conclude with an overview of the discussions. In anticipation of the methodological approach to this research, the personal and professional context for this study is appropriate to understand the motivations of a white, female practitioner-researcher exploring the differential outcomes of black staff in UK higher education. The remainder of this section will focus my journey to this destination; however, a more comprehensive reflexive discussion is provided in Chapter Eight (section 8.3) of this dissertation.

It is important to locate myself within this research study since who I am and what has brought me to the point of embarking on a doctoral research programme is relevant to the way I view the world. I am first-generation born in the UK to parents of working-class Spanish descent and so I bring an appreciation of the notion of being 'othered' as these were my and my family's collective experiences growing up in 1970s UK society. This habitus (Bourdieu, 1977) was my starting point in coming to terms with my identity, which had been formed by cultural norms constructed within my family unit, but which also conflicted with those in the social spaces that I would interact as a child and in my formative years where my heritage would be used as a point of difference in a derogatory sense. I came to realise that in witnessing my parent's experiences of living in a society where they were immigrants, they (to a lesser extent because of their accent) and I could at least be invisible because of our whiteness. It was not this that led me to become an equality and diversity practitioner, but my sexual orientation, and this aspect of my identity became the catalyst in assisting individuals to navigate through

complex workplace experiences that could involve bullying, harassment, and discrimination. As such, I have remained an equality and diversity practitioner working in the UK public, private, and voluntary sectors at varying levels over the past two decades, which include three different higher education institutions, and which provided me with the opportunity to begin this research study.

As a practitioner-researcher of white ethnic background the issues which have been explored in this research have not been because of any personal experience of racism per se. My interest has grown over time as my consciousness has increased in terms of the practical complexities related to race and ethnicity in organisations, and in positioning myself in close proximity to individuals within those organisations in dealing with outcomes that affect ethnic minorities in adverse ways. Over the course of this research, there has been a growing cognisance of my own ethnic background and how this has informed and shaped my view of the world and my practice. This is particularly relevant as an equality and diversity practitioner dealing with matters of race, the support and advice I have provided to people of colour around bullying, harassment, victimisation, and discrimination related to race, the development of organisational policies, procedures, and strategies, and subsequently in the way I have approached this research.

These reflections have impacted on my professional practice, particularly when dealing with the experiences of others facing race discrimination or other adverse impact because of race in the way that informs and shapes the policy and practice of the organisations within which I operate. These considerations also include the intersections of race and other characteristics (Crenshaw, 1989), such as gender, age, etc. since for example black women have different experiences and outcomes to white women or black men. It is apparent from my own professional practice that people construct their own understanding and knowledge of the world around them through their own experiences, and this may be followed by a period of reflection of those experiences for sense-making. Those experiences are further reconciled with previous thoughts, attitudes, and experience to shape future ideas and so on.

I have developed an increased consciousness of the issues affecting people of colour in society in general, as well as in the workplace. This consciousness has had a profound effect on the way in which I now view the world. I have become awakened to the structures in which I have and continue to operate in, which provide me with a distinct advantage, one that I had not

wholly appreciated in the past. I have felt guilt during this research journey as I have reflected on how I may have contributed to or supported systemic racism, not solely as a diversity practitioner, but as a manager and a colleague to others. This has also motivated me to operationalise my knowledge and contribution through this research to inform my work and organisational practice, which is transferable to any organisation or industry. I recognise that my whiteness places me closer to the centres of influence than my peers of colour may have access to, and I have a responsibility to use my privilege as an ally to those who have shared their lived experiences with me. It is for dual reasons, one of identity and the other my field of specialism, that this research topic is of personal importance, however it is critical that individuals of a white ethnic background become engaged with this subject matter within the educational sphere (Archer, 2003) as well as from the perspective of human resource management and organisational development.

Before embarking on this research, I had not appreciated the scale of experiences that for some had been built up over time, and which often would be brought together with historic psychological burden, informed, and shaped by familial experiences of race and racism, which would be part of an emotional armoury to survive in contemporary society. An awakening to these issues throughout the course of the data collection brought much frustration in attempting to resolve these issues through my own practice, and a sense of deep disappointment that I may not have noticed similar experiences as a practitioner. I have reflected on my own sense of arrogance that I would or could be the person to change workplace culture singlehandedly and yet, there is part of me that believes that had I not felt like this before embarking on this research study, this research would never have started.

The following sections of this chapter will outline in more detail the approach I have taken to undertake this inquiry on the topic of differential outcomes for black staff in UK higher education. This will include the methods I have employed to engage with participants at the heart of this study through the gathering and analysis of the empirical data that will form the basis of the findings.

4.2 Methodology – taking a social constructionist approach

This research attempts to explore social phenomena set within the specific context of UK higher education with a view to understanding how and whether the social interactions

contained within institutions, and amongst its social actors, contribute to the outcomes for black staff. Due to the nature of the enquiry within this research study, the conditions of the experience are more complex than simply what can be observed within the workplace. Berger and Luckmann (1991) asserted that society exists both as objective and subjective reality. The objective reality is created through the interaction of people within the social world, which influences actors to routinise and habitualise behaviours and actions in such a way that they become reproduced with minimal effort. As a result, these actions become normalised to such an extent that the created knowledge becomes objective. This objectivity continues to be reaffirmed in actors' interactions with others (Andrews, 2012). Before exploring the methodological approach to this research study, the rationale for this will be explained further below.

There are aspects contained within this study, such as the concept of race, where despite academic attempts to refute claims that race objectively exists, a shared understanding of race continues to persist in the social sphere where the terminology has become an ordinary and accepted part of our everyday life to such an extent that race has become objectified. The same can be attributed to the concept of racism, which lies at the heart of this research study and whether this is a contributing factor to the outcomes of black staff. The concept of racism, which can also be viewed as having been socially constructed, attracts what may appear at face value to carry a shared understanding within contemporary society, however these understandings or appreciations can vary according to life history, cultural context and experience, and this aspect is explored further in Chapters Six and Seven when considering participants' responses.

The most current UK higher education workforce data (Chapter Five of this dissertation) identified that statistically, there are differential outcomes for black staff compared to the outcomes of staff of any other ethnic background; they are least represented, particularly at the highest levels of the hierarchy and are paid less in comparison to their white peers. What is observable through the data might highlight several aspects that have contributed to this sectoral workforce profile; namely that people of colour may not be attracted to higher education; that there are discriminatory practices in place that disadvantage and exclude people of colour from being recruited to and promoted within the sector; or towards the extreme view that the data is inaccurate and therefore not a factual representation. I have

assumed for the purpose of the study that the secondary data is the most factual quantitative position in relation to the state of the higher education workforce in the UK as it is taken from a well-established and trusted source (Advance HE, 2020). As such, other than these statistical 'facts' this research explores the experiences of black staff to gain a better understanding of this workplace demographic and how this profile continues to be maintained over long periods of time, e.g., to ascertain the perceived causes of the statistical workforce profile in relation to black staff in UK higher education. In exploring the perceptions and lived experiences of participants operating within institutions and sector agencies, these accounts are set within the premise that as humans we have developed the capacity to interpret and construct reality, therefore the world of human perception is not real in an absolute sense (Patton, 2015). The data in isolation do not provide sufficient information about the conditions that lie beneath the surface in respect of the experiences of black staff and the underlying structural and operational functions within the system that inform and shape that position. Some of those underlying organisational structures and mechanisms have also become objectified, e.g., organisational policies and practices, and they too are central to the discussion contained within this research resulting in the appropriate application of social constructionism as my ontological approach to this study.

From a social constructionist perspective everyday life presents itself as a reality, which is interpreted by social actors and becomes subjectively meaningful to them as a coherent world (Berger and Luckmann, 1991) at a particular moment in time. Social constructionism insists we take a critical stance over the taken for granted ways of understanding the world and ourselves and poses that our knowledge of the world is constructed by and between social agents through daily interactions where our versions of knowledge become fabricated (Burr, 2015). As such, there are multiple perspectives on the causation of these various outcomes, and as a result this research does not intend to elicit 'facts' from the interactions with participants, nor will there be an attempt to make judgement about the accounts given by the persons who have contributed to this research. Searle (1995) suggested that judgements are subjective, since their truth or falsity cannot be settled objectively because the truth or falsity depends on certain attitudes, feelings, and points of view of the makers and the hearers of the judgement. However, there are multiple research studies and volumes (ECU, 2011; UCU, 2016, 2017; Hirsch, 2018; Eddo-Lodge, 2018; Rollock 2019) that have relied upon the accounts

of social actors depicting lived experiences that have racist outcomes within a workplace setting, and I would argue that this has become objectified over time and has contributed significantly to my interest in researching these phenomena.

In taking the view that reality is socially constructed and may attract different and conflicting viewpoints, the starting point for the research lay in the data (UCEA, 2018; Advance HE, 2020), which has suggested over several years that black staff are disadvantaged in the workplace and this has been empirically measurable by quantitative data gathering exercises at a national level. As an equality and diversity practitioner working in the field of UK higher education, I had become aware of multiple accounts that might be viewed as contributing factors to these phenomena. In addition to this, the views, attitudes, and interpretation made as a practitioner-researcher, insider-outsider (Gair, 2012) perspective all contribute to this exploration and all accounts gathered through the course of this investigation reflect the uniqueness of each participant's experiences, are equally valid and worthy of respect in understanding the phenomenon under enquiry. Therefore, this exploration seeks to draw together these multiple realities to uncover the complexities of the lived experiences of black staff set within the context of objectified workplace inequalities. In addition, these accounts will be interpreted in such a way that allows for a new contribution to knowledge about this subject as well as a contribution to practice by identifying solutions that can address the findings within this dissertation, not just within higher education but in any organisation with a diverse workforce. These solutions will be explored in the concluding chapter.

The rationale for taking this ontological and epistemological approach to this research has been based upon the consideration given to other philosophical methodologies. It would not have been possible to have taken a positivist approach, since positivism is premised on a phenomenon being real and observable and asserts that only verifiable claims based directly on experience can be considered genuine knowledge (Patton, 2015). I have previously outlined how the topics under enquiry here are more subtle and reality is individually constructed and therefore cannot be objectified. Social constructionism, which opposes positivism and empiricism as these approaches assume that the nature of the world can be revealed by observation (Burr, 2015), is therefore a more natural ontological fit. This creates a problem, insomuch that the research questions cannot verify a hypothesis that racism is the cause for the differential outcomes for black staff in UK higher education. Therefore, the

subjective perceptions and lived experiences of participants working in UK institutions and sector agencies would not amount to 'facts' in a way that could support a positivist approach. If I were utilising a more positivist stance, the means of gathering data would be compromised as the starting position might be that racism *is* the cause of differential outcomes and the enquiry would become more rigid and structured to prove or disprove the hypothesis. In addition to this, one would need to define racism, where there are multiple definitions and consequently multiple understandings, and individuals would need to relate their own outcomes within these stringent parameters.

The same consideration has been given to grounded theory, which incorporates and applies concepts such as validity, reliability, causality, and generalisability to qualitative inquiry. This approach is premised on analysing actions and processes rather than themes and structures, which would not have been ideal with this study. In terms of critical realism, this methodology posits that there is a reality out there that exists independently of our knowledge of it (Patton, 2015; McLachlan and Garcia, 2015). However, McLachlan and Garcia (2015) found the application of critical realism challenging throughout the interview stage of their research journeys in which they became aware of a more constructionist influence. Their view was that applying critical realism meant the approach was too rigid and that attempting to command the interview in an objective manner considering the subjectivity of participant accounts became a secondary consideration (ibid). Due to the nature of this research study, it was imperative that I could include myself within the participants' perceptions and lived experiences not as an objective outsider looking in but immersed within the topics being discussed to facilitate a collective generation and transmission of meaning between researcher and participant (Crotty, 1998; McLachlan and Garcia, 2015).

As a practitioner-researcher, I too construct my own reality based upon my own background, experience and attitudes towards others and the world around me. Because meaning is socially constructed, ideas, properties and understanding of phenomena are conditioned through historical processes. I therefore follow Cunliffe (2008) who states that social reality is not separate from us and that social realities and ourselves are intimately interwoven as each shape, and are shaped, by the other in everyday interactions. As I am researching the topic of race and racism through the lens of a white, female practitioner-researcher this has meant that I have had to be sensitive to assumptions I may make because of the way I view the

subject matter, which has been informed through practice, and through my interactions and experiences within a workplace setting. On reflection, these perspectives have helped *and* hindered my discussions with participants during the data gathering phase of this research, which is discussed further in section 4.6 of this chapter. The following section will provide further insight about an influential concept (Critical Race Theory) that has helped with the approach for this research and brought me closer to gaining certain capital in preparation for my interaction with people of colour within the field.

4.3 Critical race theory – the starting point

In keeping with, and as an extension to taking a social constructionist approach to the research, the study has also been influenced by Critical Race Theory (CRT), which offers a radical lens upon which to view the topic under enquiry. CRT provides the basis to make sense of, deconstruct and challenge racial inequality based on the understanding that race and racism are products of social consciousness and power and seeks to expose the way racial inequalities are maintained through systems and structures that appear normal and unremarkable (Rollock and Gillborn, 2011) and this is applied in the analysis of participant accounts in Chapters Six and Seven of this dissertation. The principles of CRT underpin the perspectives behind the research in that racism is central to the enquiry insomuch that it has become an entrenched feature of society (Delgado and Stefancic, 2000: xvi), and as a continuation of this it has become embedded within our institutions (Gillborn, 2006; Sian, 2019; EHRC, 2019) and is often taken for granted and viewed as natural. Racism can also be evidenced in the outcomes of processes and relations irrespective of intent (Gillborn, 2006; Ladson-Billings, 1998) and this too is a feature of social relations within a workplace setting.

The second principle is that white supremacy plays a significant role in maintaining the status quo within institutions and the privilege enjoyed by people of white ethnic background cannot be ignored in the context of the hierarchies of power that exist in our institutions, which are overwhelmingly in the possession of white people, particularly men. In this context, white supremacy is not what society tends to associate through extreme right-wing hate groups, but the everyday privileges enjoyed by people of white ethnic background, who will not tend to be conscious of their own ethnic identity (DiAngelo, 2018). By engaging with CRT as a theoretical approach, I must be mindful of my own white ethnic background as a practitioner-

researcher to interrogate my own racial privilege in the process of this research journey, in addition to considering my ability to unveil invisible racism in the accounts provided by participants of this study (McIntosh, 1997).

Notwithstanding the importance of white privilege and the pertinency to the topic under enquiry, the concepts of whiteness and white privilege will not be explored in their entirety within this research although these aspects will be addressed to a lesser extent in considering the organisational structures that people of colour find themselves in within UK higher education institutions. Furthermore, the intersection between race and gender is a significant area for exploration, as individuals possess multi-dimensional identities which inform perspective and experience and are likely to affect a person's outcomes (Crenshaw, 1989; Bourdieu, 2001a). Despite intersectionality playing a pivotal role on the experiences of people of colour, this aspect will not be addressed in detail within this study due in part to limited data being available that considers gender and ethnicity together, and particularly where data is disaggregated to show outcomes for specific ethnic groups. Where data is available, e.g., through ethnicity and gender pay differences, this is evidenced and discussed in Chapter Five of this dissertation.

A further strength of CRT is that it places importance on the voices and experiences of people of colour, and especially people of black ethnic background who are the focus of this research. CRT is aligned with the social constructionist perspective as it does not make assumptions that the accounts of people of colour reflect a singular truth; rather they represent a kaleidoscope of experiences that represent the experiences of people on the margins of a racist society (Rollock and Gillborn, 2011). Another viewpoint of CRT is that white people have little incentive to eradicate racism because it maintains and reinforces organisational hierarchies and power systems that serve to subjugate people of colour. It is this aspect of CRT that, as a practitioner, resonates with my own experiences in being (un)able to advance practice around race and ethnicity and this has been an area which has elicited the greatest resistance and/or emotive response from white colleagues. From experience, discussing white privilege has been problematic because white staff find it challenging to think of themselves as personally being advantaged solely by their colour or ethnic background (DiAngelo, 2018). Many times, the subject of class has crept into the conversation and/or there are complexities around the outcomes of white staff that are not UK-born, which are sometimes different to those who

are white British. These are all valid points however, these accounts tend to derail the focus away from the benefits derived because of ethnicity, and in certain cases, nationality as white people and how these are often different and advantageous compared to the outcomes enjoyed by people of colour.

My professional stance, and the manner in which I have approached this research topic is positioned firmly in the belief that race has been socially constructed; that racism exists in the UK higher education sector as it does in wider society; and that storytelling will provide an opportunity to hear the lived experiences of the participants interviewed through this dissertation.

4.3.1 Qualitative research – the vehicle to provide a voice

The catalyst for this study is the statistical data produced for the UK higher education sector (Advance HE, 2020) on an annual basis through reports that provide demographic information according to different protected characteristics⁴ for staff employed within institutions and students studying in institutions. As an equality and diversity practitioner working in UK higher education this data is critical for creating in-house reports and for benchmarking against the rest of the sector. Those data provide sufficient context to the issues affecting the workforce in relation to race, e.g., ethnic minority representation, and examples of the relevant data which have informed this research have been presented in Chapter Five of this dissertation. However, although informative, the statistical data can only identify the outcomes for staff of colour in a retrospective form and across the whole sector and lack the nuances of individual institutions (Acker, 2006). As a result of this, one must rely on generalising the issues that might affect racially minoritised staff in broad themes of representation, career development and retention without knowing the extent to which all or some of these aspects impact upon the overall demographic profile of the UK higher education workforce. Therefore, this research requires a more in-depth enquiry to understand the dynamics of perceived social relations within institutions and the impacts upon racially minoritised staff in the UK higher education sector that might influence these outcomes. To satisfy the requirement of CRT to provide a voice to people of colour, qualitative data gathering is essential to understanding

⁴ The protected characteristics under the Equality Act 2010 are age; disability; gender reassignment; marriage and civil partnership; pregnancy and maternity; race; religion or belief; sex; and sexual orientation

the experiences of black staff in UK higher education. Quantitative data highlighted a persistent problem yet is not capable of providing the context nor the richness that qualitative data is capable of offering (Sayer, 1992), especially when researching the lived experiences of participants within a highly emotive and sensitive area.

Despite the richness that qualitative data can provide, there are limitations to qualitative research in terms of the orthodox understanding of generalisability. The participants that self-selected to take part did so through various motivations and the very topic of enquiry may have attracted expressions of interest to participate from those who had a particular story to tell, which may not be wholly indicative of a generalisable experience of all black staff in UK higher education. The purpose of this research is therefore not to generalise in this way, but to further our understanding of the phenomenon (differential outcomes for black staff) by providing indicative themes and participant voice, which have been further analysed conceptually through the lens of Bourdieu, that strengthens and helps develop existing literature on this topic and which is transferable (Daly et al, 2007) to other organisational contexts where there are similar quantifiable staff outcomes identified via workforce data.

In addition to taking a social constructionist approach to understand the lived experiences and perceptions of participants in the UK higher education sector, this research also seeks to understand the social construction of various concepts, such as race, ethnicity, racism etc. as well as make sense of participants' experience of and attitudes towards race and racism in the workplace. To facilitate this, it has been necessary to view these phenomena from a particular perspective. To explore and understand the constructs of race, racism, and workplace race equality within the context of UK higher education and be in a position to explore the differential outcomes for black staff, the research takes a qualitative inductive approach. To support this stance, the study follows an interpretative phenomenological analysis (Smith, Flower and Larkin, 2009) and the empirical data has been organised thematically to answer the research questions. To engage with the experiences of others it is necessary to consider and reflect upon personal experiences, preconceptions and assumptions in interpreting participant feedback (Bryman and Bell, 2007), and process those accounts through intersubjective meaning-making (Berger and Luckmann, 1991). However, Cunliffe (2008) suggests that we are not individuals coordinating activity and coming to an understanding of

what each other thinks, but that we are 'selves-in-relation-to-others' and therefore, inseparable.

As an equality and diversity practitioner I have become exposed to the multiple perspectives and experiences of people of colour in the workplace, and all these have been equally valid in constructing and informing the design and methodology of this study. I am conscious that this professional experience has to some extent prejudiced my view about the role institutions play in contributing towards the outcomes for black staff and throughout this research journey I have reflected on how my views and perspectives might influence my interactions with participants, my interpretation of the data gathered and my practice within an organisational setting. Despite this, my appreciation of the issues when engaging with participants during the study has allowed me to gain access to participants in a way that perhaps another white researcher may not have had the privilege to secure. It is not clear whether this has been because of being viewed as an insider (employed within higher education at the time of fielding participation and conducting interviews), being known to be an equality and diversity practitioner and therefore perceived to 'understand' the issues, my reputation for activism within the sector, or whether it was a combination of all these aspects. Whatever has facilitated access, the roles I have held within higher education have enabled me to deal with the complexities of organisational equality, diversity, and inclusion in addition to addressing first-hand how the acts or omissions of organisations impact upon employees within these settings, and it is these insights that have informed my world view and professional practice, and subsequently the research design in addressing this topic.

4.4 Research design

The research design was influenced primarily by Layder's (1993) research map (Table 1, p.87), as it was a useful illustrative tool to link Bourdieu's (1977; 1984; 1990; 1998) and Gidden's (1979; 1984) theories around structure and agency and articulate the complex and interrelated social interactions of people (see Chapter Three). These concepts will be used as an interpretive tool to make sense of the data gathered from participants in the study. Using Layder's multi-level map for analysis is also a useful mechanism from a social constructionist perspective to realise that social interactions and their complexities are weaved within and across a range of layers and is particularly relevant when studying the intricacies of human

behaviour, their impact within an organisational setting and how these might affect outcomes for specific groups of people.

Layder (1993) identified four research elements as context; setting; situated activity; and self, which set out the multiple layers of social organisation by adding texture to the dualistic macro-micro perspective of social relations and sets these within a broader context of history and power, both of which inform and influence the social relationships within and across the layers. The research elements have been adapted to support this research study. The 'context' allows for a macro social organisational focus to understand how the UK higher education sector is informed by wider legislative and social policy to address its approach in embedding equality and diversity practice and how this might impact specifically on advancing race equality in institutions. Race shares a platform with other considerations in terms of the broader equality and diversity agenda and because of this, an organisation's ability to advance race equality has become more challenging. There will be numerous actors involved at multiple levels of an organisation to formulate institutional responses to legislative and policy requirements. In doing so, both 'history' and 'power' play a significant part in the extent to which an institution can respond effectively to legislation and policy development. The perceptions from a range of participants in this study will help identify the extent to which institutions are indeed committed to embedding the principles of equality and diversity and more specifically in the advancement of race equality.

The 'setting' within this research focuses on the intermediate meso-level social organisation and seeks to consider how institutions address the concepts of race and ethnicity, within a broader equality and diversity agenda, and the way institutions approach the management of diverse workforces. This is particularly relevant as UK institutions are set within a backdrop of a white hegemonic structure. The research seeks to understand the views of equality and diversity practitioners within institutions and in agencies with an interest in the sector and consider the opportunities and challenges as practitioners in the field of equality and diversity in driving or informing the agenda. Additionally, the research seeks to establish the shared meanings and understanding of the aspects of this research that relate to race, ethnicity, and racism and whether those meanings and interpretations are indeed shared at all. It can easily be assumed that there is a societal understanding of the concepts of race and ethnicity, and this study wishes to explore whether in fact this is replicated within the workplace. There is

an added consideration that the environment under analysis is one of higher learning and perhaps there is a greater assumption that actors within these settings should be, or are, better informed.

The element of 'situated activity' focuses on the micro-level social activity of actors and set within the context of this research will consider the identities of participants and their perceptions and experiences of working within UK higher education. This is particularly relevant for minority ethnic staff working in institutions where this staff cohort might be isolated by virtue of their ethnic background. In considering situated activity, the research hopes to identify the strategies employed by staff of colour in navigating through organisational structures and shedding light on the opportunities and challenges that this journey brings in respect to their working lives in UK institutions.

Layder (1993:74) described that the 'self' and 'situated activity' are separable elements although the two cannot easily be separated from the social situations in which they are routinely embedded. The research map will assist in considering and analysing the complexities of social interactions through the vertical macro- to micro-organisational layers combined with the horizontal time-space continuum of history through to power, enriching the interpretation of empirical data captured from participants. The research map compliments the theories around structure and agency (Bourdieu, 1977; Giddens, 1984) and will allow greater consideration about the structures of power within institutions and understand better how they operate at macro through to micro level. Furthermore, these layers are overlaid and influenced by history, which is considered in Chapter Three of this dissertation, and considers the imbalance of power that exists within organisations. Although Layder's (1993) research map provides a multi-layer framework for analysis, the map is dynamic and multi-dimensional that allows for richer consideration of complex organisations where agents may act within and across multiple layers of their institution, and in certain circumstances, the wider higher education sector.

In terms of the aspects described above in relation to race, ethnicity and racism it is the manifestation of these aspects that this research seeks to consider and interpret as lived experiences and how they impact upon the self as a participant in this study, as an employee within their respective institution, the way in which these perceptions affect day-to-day workplace activity, the relationships between themselves and other colleagues and their

attitude and behaviour towards the wider organisation. In addition to this, people of colour are to a greater or lesser extent represented across all layers and levels within our institutions and consideration of positioning, identity, belonging and the challenges and opportunities that these aspects bring for black staff at different levels will be explored further within Chapters Six and Seven and Layder's map can facilitate this analysis further.

Table 1: Research map (adapted from Layder, 1993)

	Research element	Research focus and Objectives	Key methods	Theoretical consideration		
	Context	Focus: The emergence of equality and diversity within organisations and their impact on race equality Objective: Explore to what extent the evolution of equality and diversity in UK higher education has advanced workplace race equality	Systematic literature review and interviews with staff and relevant stakeholders with an interest in UK higher education	Historical and contemporary theories of race, ethnicity and racism in the UK and US Critical race theory Organisational social interactions and power		
лу	Setting	Focus: Race and ethnicity in the context of white hegemonic organisational structures Objective: Explore the experiences of black staff in the context of differential outcomes in UK higher education institutions	Systematic literature review Secondary empirical data – workforce demographics Semi-structured interviews with participants	Historical and contemporary theories of organisational equality and diversity in the US and UK Critical race theories	er	
History	Situated activity	Focus: Meanings and understandings of social interactions in an organisational setting and the impact on outcomes for specific groups. Objective: Explore experiences of black staff in UK higher education and how they situate themselves within the organisational context	Systematic literature review Historic and current literature on the experiences of people of colour in the workplace Secondary empirical data – workforce demographics Semi-structured interviews with participants	Critical race theories Bourdieu (habitus, field, capital, symbolic power and symbolic violence) Giddens (structure and agency) Race and racism	Power	
	Self	Focus: Experiences and perceptions of black staff in the study and on own experience and practice as E&D practitioner Objective: Explore the ways in which organisational structures and social interactions impact on working environments	Semi-structured interviews	Critical race theories Bourdieu (habitus, field, capital, symbolic power and symbolic violence) Giddens (structure and agency) Race and racism Identity and belonging		

4.5 Gaining access

The first phase of my data gathering involved an electronic communication (Appendix 3) sent via two higher education networks requesting expressions of interest. At the time, I was employed as Head of Equality, Diversity and Inclusion in a post-1992 higher education institution and because of this I was a member of those networks and had access to, and was able to share messages through, its distribution list of academics and practitioners who had self-subscribed to these networks. I used my position in the field, which gave me legitimacy and credibility, as a way of 'infiltrating' the rest of the sector to gain access to personnel that might be difficult to engage through my own connections (Patton, 2015). Participation was sought from staff who identified as possessing black ethnic background, employed at any level and who might be either an academic or professional or support staff member. In this communication expressions of interest were also sought from human resources and equality and diversity practitioners and senior managers of any ethnic background. Participation was sought from all types of institutions and across all areas of the UK with access to these networks. Interestingly, despite outlining within the invitation that the focus of the study was to explore the differential outcomes for black staff in UK higher education and that participation was sought from this cohort, there were a number of expressions of interest received from staff of diverse ethnic backgrounds and in hindsight a definition of the meaning of 'black' should have been made clearer. The networks utilised for the purpose of promotion of this study were the Higher Education Equal Opportunities Network (HEEON) and the Higher Education Race Action Group (HERAG). Both networks have self-selecting membership and are accessible to academics, students and practitioners across the UK higher education sector and coordinated through Advance HE.

Initially, the intention was to interview fifty participants following the receipt of approximately seventy expressions of interest from various colleagues across different institutions in England, Scotland, and Wales. Subsequent attempts at contacting these colleagues, some unsuccessfully, resulted in a total of thirty-one participants being selected on a first come-first-served basis, and interviews began in May 2016. The thirty-one participants interviewed represented various mission statement institutions, e.g., Russell Group, Post-1992, non-affiliated institutions, research institutes, and included practitioners

from external agencies with an interest in higher education. The geographical locations of participants were selected to represent a cross section of different institutional settings with diverse workplace and local population demographics, which resulted in participants representing institutions that were, except for one institution, predominantly based in or near large urban centres. Both the University and College Union (UCU) and Unison, as the two most prominent trade unions representing staff in UK higher education, were contacted to take part however both declined to do so. A small selection of Vice Chancellors were approached once it became apparent that there was no representation at this level amongst those that expressed an interest. As a result of this exercise, one Vice Chancellor agreed to take part in the study. The demographic profile of participants interviewed for this research study can be found at Table 2 (p.93).

Black staff have represented varying levels across academic and professional and support roles, including those who hold an institutional position where they are either a senior manager, human resources or equality and diversity practitioner. Participants identified as various ethnic backgrounds including black, Asian, and mixed heritage although black staff accounted for most participants interviewed as part of the study, e.g., staff from a black African, black Caribbean or any other black ethnic background. Despite it not being the original intention of this research study, on reflection it has been a useful exercise to compare the experiences of staff from diverse minority ethnic backgrounds in the context of UK higher education and how their perceptions and experiences might have been influenced by their ethnic background.

In recognition of the limited ethnic diversity within roles such as senior leaders, human resources or equality and diversity practitioners, expressions of interest were sought from people with any ethnic background to gain an understanding of the opportunities and challenges of advancing race equality in higher education from these perspectives. Interestingly, there was an overall under-representation of staff representing human resources, equality and diversity practitioners and senior managers although the shortage was most acute for senior leaders and human resources professionals. This came as a great surprise considering the number of practitioners involved and engaged across the Higher Education Equal Opportunities Network (HEEON) and Higher Education Race Action Group (HERAG) networks in which the call for participation was promoted. On reflection, it is

possible that due to the topic under investigation that few practitioners may have been prepared to expose the issues affecting their institutions in relation to race, and particularly to another serving practitioner, subjecting their own and colleague's practices under scrutiny. However, despite these challenges, there was no overall shortage of participants who were forthcoming through snowballing for expressions of interest via these practitioner networks. The following section describes the process employed to collect and analyse the data gathered from participants in the study.

4.6 Data collection

This section will explore the considerations given to the methods employed in accessing participants for the research and provide an overview of the themes of the questions that would be used to gather data and subsequently in the analysis of participant feedback. Interviewing is a key tool for a qualitative researcher and in terms of gathering data this is an invaluable method of researching in social sciences (Bryman and Cassell, 2006). The purpose of qualitative interviews is to interpret descriptions of individual's lived experiences and perceptions (Bryman and Bell, 2011). Critique of qualitative data gathering methods has focused on the notion that the interview is not a static event, but an active dynamic process where the interviewer and participant co-construct meaning throughout the interview encounter (Holstein and Gubrium, 1997; Bryman and Cassell, 2006; McLachlan and Garcia, 2015).

Interview schedules (Appendix 4 and 5) were constructed based on the objectives of the study, which was informed by the preliminary literature review undertaken at that point in time and data was gathered using semi-structured one-to-one interviews in person, by phone or via Skype. There was a specific schedule of questions for those who identified as black for the purpose of the study and included those who identified as Asian and mixed race. There was a separate schedule of questions for human resources and equality and diversity practitioners and senior leaders, as there were many crossovers between this specific group in terms of their understanding of race and the opportunities and challenges of advancing race equality across higher education.

The use of a semi-structured interview technique was considered the most appropriate method to extract information around the lived experiences of individuals and would be

undertaken on a one-to-one basis. Interviews were conducted using an informal conversational approach (Patton, 2015) so that I could quickly form a rapport with participants and utilise the time that was available to me to gather as much information as possible. This was considered preferable to using focus groups to question multiple participants due to the personal nature of the questions, e.g., experiences of racism, which I did not feel would be possible to gather had the questioning been carried out within a group setting. Carrying out semi-structured interviews allows a reciprocal relationship between researcher and participant in a way that allows for a fluid mutual negotiation of meaning and power and can allow, not just the collection of data, but the interaction and engagement with participants in real time (Galletta, 2013) using diverse methods of questioning to clarify, probe and to check for understanding on both sides. Not being completely prescriptive with the interview schedule provided a guide to the conversation that allowed participants to move within and around relevant, and sometimes quite abstract, topics at will. Using semistructured questions allowed me to return to aspects within the conversation that would be specific to the participant and/or to elicit further explanation in a more dynamic way (McLachlan and Garcia, 2015).

The interviews themselves were conducted in a variety of places and were arranged to mutually suit both parties. Some interviews took place within my own private office within my institution or in the participants' institutions within a confidential setting. One interview took place in a café in central London at the request of the participant, which on reflection may not have been the most appropriate setting to conduct an interview due to the background noise of customers, however the setting created an informal environment where we could talk openly about some complex organisational issues. It was important to provide participants with the opportunity to select the venue where we would have a discussion as it was paramount that the participant could feel safe, comfortable and that the environment could provide neutrality for the discussion (Patton, 2015). This would be particularly useful to participants who felt strongly about being away from their usual workplace environment. All interviews, including those conducted over the phone or via Skype were recorded to ensure that an accurate record of the conversation could be produced, and which would facilitate transcription and analysis later. Interviews varied in length, the shortest taking approximately 90 minutes and the longest taking three hours.

An aspect that had not been anticipated with this research was the proportion of self-selecting individuals that expressed an interest to participate who were female. Participants were interviewed in the order that they expressed their interest to take part in the study. It could be interpreted that the higher proportion of women that wished to participate reflects the multiple complex issues that affect women of colour in UK higher education. The same could be observed in the equality and diversity practitioner cohort, who were all women. It is unfortunate that this study has been unable to interrogate this specific cohort of women further to understand their experiences in greater detail in a way that is able to probe the intersection of gender and race and how this might manifest itself within an institutional setting. Despite this, it would be beneficial to explore this aspect further in future.

On reflection it may have been useful to gather information relating to participant nationality status, particularly as the secondary data discussed in Chapter Five mentions the workforce profile according to UK and non-UK national status of staff, although some of the participants talked about their national origins indirectly during interviews.

In summary, the demographics of the participant cohort is as follows:

- 35.5 per cent are employed in Russell Group institutions
- 80.6 per cent are women
- 67.7 per cent are from a black ethnic background
- 48.5 per cent are employed in academic roles
- 41.9 per cent are aged 45-54

The table below provides a demographic outline for the participants interviewed in this study.

Table 2: Demographic profile of study participants

Participant	Institution type	Role	Ethnicity	Gender	Age group
Participant 1	Russell Group	Equality & Diversity	Black	Female	45-54
	2	Practitioner			05.04
Participant 2	Post-1992	Academic	Black	Male	25-34
Participant 3	Post-1992	Professional	Black	Male	55-64
Participant 4	Russell Group	Academic	Asian	Female	35-44
Participant 5	Post-1992	Senior	Mixed race	Female	45-54
.		Professional	51 1		25.44
Participant 6	Research	Academic	Black	Male	35-44
Dantisia aut 7	institute	Coming	Disale	Famala.	FF C4
Participant 7	Russell Group	Senior	Black	Female	55-64
Doutisins at 0	Deat 1002	Professional	Othor	Famala	45 54
Participant 8	Post-1992	Academic	Other	Female	45-54
Participant 9	Post-1992	Academic	Asian	Female	45-54
Participant 10	Post-1992	Professional	Black	Female	45-54
Participant 11	Post-1992	Professional	Black	Female	16-25
Participant 12	Post-1992	Academic	Asian	Female	35-44
Participant 13	Post-1992	Senior Professional	Mixed race	Male	45-54
Participant 14	Russell Group	Academic	Asian	Female	35-44
Participant 15	Post-1992	HR Practitioner	Black	Female	25-34
Participant 16	Post-1992	Academic	Black	Female	35-44
Participant 17	Post-1992	Academic	Black	Female	45-54
Participant 18	Post-1992	Senior	Black	Female	45-54
. apa = 0	. 550 2552	Professional			
Participant 19	Sector Agency	Equality &	White	Female	35-44
	0.1	Diversity			
		Practitioner			
Participant 20	Post-1992	Academic	Black	Female	45-54
Participant 21	Post-1992	Academic	Black	Female	55-64
Participant 22	Post-1992	Academic	Black	Female	35-44
Participant 23	Sector Agency	Equality &	White	Female	35-44
•	,	Diversity			
		Practitioner			
Participant 24	Russell Group	Professional	Black	Female	25-34
Participant 25	Russell Group	Professional	Black	Female	45-54
Participant 26	Russell Group	Academic	Black	Male	45-54
Participant 27	Russell Group	Academic	Black	Female	35-44
Participant 28	Russell Group	Academic	Black	Female	35-44
Participant 29	Russell Group	Senior Leader	White	Male	45-54
Participant 30	Post-1992	Equality &	Black	Female	45-54
-		Diversity			
		Practitioner			
Participant 31	Russell Group	Equality &	Black	Female	25-34
		Diversity			
		Practitioner			1

4.7 Data analysis

All the interviews were recorded using a portable digital voice recorder to ensure that the whole conversation between myself and the participants could be retained for transcription. As Flick (2013) notes, a transcript is the result of transcribing performed by a single person or several persons and for the purpose of this research the interviews were transcribed verbatim in person and by utilising a professional transcriber for some of the interviews due to time constraints. I utilised an analytical framework approach to identify thematic issues (Patton, 2015) by reviewing all transcripts manually to identify content of interest, which were then copied into a separate spreadsheet that identified the most common themes. The spreadsheet focused on several themes that were reoccurring throughout all the interviews and which would inform the content of the dissertation, particularly in terms of identifying the most important areas for inclusion that could address the research questions, desk research, and supporting literature review. Although these themes featured most in the interviews with participants some of the areas did not feature within the final dissertation and this was predominantly due to the scope of the research study, which focused on race and ethnicity, identity and belonging, racism, institutional racism, and challenges as a practitioner.

These themes were as follows:

- Race and ethnicity
- Identity and belonging
- Racism
- Institutional racism
- Racial hierarchy
- Institutional commitment*
- Networking*
- Challenges as a practitioner*
- Future considerations*
- Intersectionality

A number of these themes, namely those marked with an asterix (*) elicited few and mixed responses, which were related to individual institutions. Some participants offered their

opinions on what their institution or the overall sector could do to address race equality. Using Layder's (1993) research map this could have provided perspectives relating to context but there was insufficient content to elicit a meaningful discussion around these topics and did not provide responses that answered the research questions. In terms of intersectionality, I was unable to gather sufficient data that could provide good quality insights on this topic as this was not included in the research schedule but did arise through general conversation and to probe further when the topic of intersecting identities arose.

I had asked all participants from an ethnic minority background whether they were members of their institutional ethnic minority staff network to understand how engaged staff are with affinity groups and whether they offer the support needed in the workplace, despite this broader topic not being included within the research questions. Most participants answered in the negative. Although this is a significant finding, I decided to exclude this topic from the overall findings because it did not answer the research questions. However, the responses gathered from participants around network membership could have provided insight around the aspect of 'self' using Layder's (1993) research map, and the literature review did not include a discussion around support mechanisms used by ethnic minority staff within the workplace in this instance.

A qualitative inductive analysis is a method to condense raw textual data into summary format to establish links between the research objectives and the finding derived from the empirical data (Thomas, 2016; Patton, 2015), which was gathered from participants through the semi-structured interviews. On establishing these themes an inductive analytical strategy was employed to make sense of the data collected and to interpret broad issues that arose through the interviews and identify the appropriate themes that warranted inclusion within the dissertation. Quotes were subsequently added into the spreadsheet to create a narrative from each participant covering each broad subject heading where this existed. Adopting a reflexive approach to this research allowed me to examine my own preconceptions about how I interact and view the social world or being aware of how I arrived at a particular perspective (Johnson and Duberley, 2000; McAuley, Duberley and Johnson, 2007). For Bourdieu (1977; 1984; 1990) reflexivity entails systematic reflection by the social scientist to make the unconscious conscious and the tacit explicit to reveal their formative social location and corresponding habitus that can influence any account (Johnson and Duberley, 2003). To

ensure that I would not become influenced by my own habitus in the selection of data and in my interpretation of it, I approached the analytical function of the study in an objective manner by including all data that was related to the themes detailed above, regardless of the data. There was much data that could not be included within this dissertation and therefore some distinction needed to be made to ensure that the most relevant participant lived experiences and perceptions could be included that could answer the research questions detailed in Chapter One.

4.8 Conclusion

This chapter has discussed the research methodology employed during this study and has provided a framework outlining the research design and methods utilised in achieving the aims and objectives of the research. This study was undertaken to explore the differential outcomes for black staff in UK higher education institutions and understand the perceptions and lived experiences of black staff to identify the contributing factors for those outcomes.

By using a qualitative approach to this research, it has been possible to focus on the meaning participants attach to social phenomena, understand the world around them and reflect on why they may see the world around them as they do. As practitioner-researcher I too have used reflexive practice to better understand the concepts explored within this research and apply my learning to my professional practice. There have been some obstacles to overcome in gaining these insights, such as the length of some interviews and the time-consuming nature of manually analysing the empirical data gathered to identify themes. However, on reflection these obstacles have been far outweighed by the opportunities afforded by that personal data analysis, which has generated a multitude of perspectives that I feel may not have been gathered in such detail. Using an interpretive methodology has allowed me to make meanings from the themes that have arisen from the interviews to meet the aims and objectives of this research study and answer the research questions set out in Chapter One.

The subsequent chapter will explore the desk research that considers the strategic drivers in UK higher education to advance race equality and how this has developed over time, including the evolution of equal opportunities to diversity management and the role of change agents. Later, the chapter will contextualise these concepts to UK higher education and there will be a closer look at the workforce demographics according to ethnicity.

5 Workplace equality and diversity practice in UK higher education

5.1 Introduction

The previous chapter provided the philosophical position taken to approach this research study as one that is constructionist. The topics being explored relate to UK higher education employees, and as such, each participant contributed to the interviews by sharing their personal stories and perspectives about their workplace, their lived experiences and provided accounts of their personal and working lives that were *their* lived realities at the time. As practitioner-researcher my role was not to elicit facts, but to provide a platform for their voices to be heard and to offer an interpretation to these accounts within the context of my own lived experiences and understanding of the world, the supporting literature and desk research, which informed the topics under discussion.

This chapter will address the first research question, namely, to understand the strategic drivers to advance institutional race equality and how this has progressed over time. In answering this question this chapter outlines the historic and contemporary development of equality and diversity theory and practice in the UK and contextualises this within a higher education setting, i.e., the context in which the participants are operating. The way in which these concepts have been socially constructed over time have been influenced by and reflect societal fluctuations within national constructs, as well as beyond our state borders. These concepts have been incrementally shaped by legislation together with policy and business practice and continue to develop within our modern organisations to suit employers and stakeholders alike.

The evolution from equal opportunities towards diversity management has been demonstrated through a shift in organisational approach and engagement with the agenda, and the chapter will explore the differences in these perspectives and how that might advance or inhibit change. At a macro level, social, political, legal, business, and ethical considerations have shaped the equality and diversity landscape, and at meso level, it is the organisational understanding of these factors that will influence the approach taken by senior leaders and other actors in embedding practice. The theoretical and practical implications of organisational thinking around equality and diversity rely on institutional change agents and

there will be an exploration of the issues that arise for this body of advocates, including questioning whether HR and equality and diversity practitioners are as active around the topic of race as they are with other protected characteristics⁵.

The chapter will also consider how the UK higher education sector has embraced equality, diversity and inclusion, how this has evolved over time and will consider the rationale behind focusing on specific areas of equality and diversity whilst attending to all protected characteristics. Assisting change across UK higher education are sector-specific agencies with their own agenda and priorities that address equality and diversity for the benefit of both individual higher education institutions and promoting good practice across the sector. These agential priorities may shift attention away from areas that require further focus in terms of individual institutional priorities and may take an opposing direction of travel to the needs of the individual institution.

Finally, there will be an examination of the current workforce demographics in relation to the focus of this research; ethnicity, to provide a framework to the remainder of this study, which will focus specifically on race. There will be a description of the ethnic demography of our UK higher education institutions to understand the challenges the sector faces in diversifying its workforce to reflect and meet the demands of an ever-increasing diverse student body. The data presented will highlight the overall workforce profile according to ethnicity, and will consider pay differentials and retention, all of which will have an impact on the overall representation of ethnic minority staff in UK higher education.

The following section will begin by defining the terminology often used in practice and charting the evolution of workplace equality and diversity, how this has been shaped by international influences, and exploring the distinctions between these concepts and their impact on different organisations.

5.2 Contextualising equality and diversity

Equality and diversity are terms that have been coined in the UK to champion aspects of society that encompass equality of opportunity, multiculturalism, human rights, and anti-discrimination. The terms are often used interchangeably in practice but are distinct in the

_

⁵ Protected characteristics are those which are specifically protected under the Equality Act 2010 and include age, disability, gender reassignment, marriage and civil partnership, pregnancy and maternity, race, religion or belief, sex and sexual orientation

theories that underpin them and how they are applied in an organisational setting. These terms also chart the evolution of people-centred policy and practice in organisations through the latter half of the 20th and early 21st centuries.

The contemporary meaning of equality can be defined as the state of being equal, especially in status, rights, or opportunities (Oxford English Dictionary, 2021). The concept of equality has shifted through modern times as a tool for political and social reform and has cut across economic, political, and social spheres. Equality has been attached to the socialist critique of poverty, supported the idea of social justice, and shifted from its attachment to class, status and wealth to the relationship between different groups and individuals in society (Institute for Public Policy Research, 1997).

In relation to the individual, the concept of equality is not new and yet, the irony is that early application was gendered. It was in the US that the first statement was made according to the status of men. The phrase 'all men are created equal' was included by Thomas Jefferson in the U.S. Declaration of Independence in 1776, and that phrase was not only significant during the American revolutionary period, which led to subsequent probing into the validity of slavery but is also thought to have continuing importance in modern American society (Green, 1976). In the struggle for civil rights, prominent activists would use the Declaration of Independence as a means of arguing the case for equality for women and African Americans (Library of Congress, 2021). During the French Revolution, equality, along with freedom and fraternity, became the basis for the Déclaration des droits de l'homme et du citoyen of 1789 (Stanford Encyclopaedia of Philosophy, 2021). Furthermore, the concept of equal treatment would become a principle of the Treaty of Rome in 1957 and would establish the application of the principle that men and women should receive equal pay for equal work (Article 157, TFEU 2009). These principles would permeate broader civil and social rights thereafter. A modern perspective on the concept of equality might result in the acknowledgment that people are not in fact equal and as a result cannot be treated equally, however we can consider ourselves to command equal dignity and respect and this is now an accepted and assumed Western standard.

In 1961 the Kennedy administration introduced the concept of equal opportunities to the United States via affirmative action to ensure equal treatment in employment on the grounds of specific characteristics and would set the scene for subsequent developments promoting

equality of opportunity. The contemporary framework for equal rights in the UK can also be traced back to the 1960s, where advances were made in direct response to the civil rights movement in the US. The UK journey has subsequently been shaped by developments in anti-discrimination legislation and research.

The 1990s saw a shift from moral and social justice arguments for equal opportunities to an emphasis on a more deregulated business self-interest (Riley et al. 2013; Dickens, 1994; Liff and Dickens, 2000) paving the way for a renewed focus on organisational diversity. Equal opportunities is a term used in the UK, with roots set firmly in compliance with antidiscrimination legislation and is often expressed as the moral or social justice case for diversity (Department for Business Innovation and Skills, 2013). This concept was especially promoted in public sector organisations due to higher levels of scrutiny imposed through legislation, e.g., Equality Act 2010 and public duties. This approach has influenced organisations to focus on the collective pluralistic needs or outcomes of groups, rather than concentrate on individuals. According to Dickens (1999) a state intervention is central to an equality agenda because the market tends to produce discrimination, not equality. This approach can dilute the issues faced by individuals and can mask the inequalities arising through the intersections of multiple characteristics. A focus on equality has provided an organisational environment that occupies itself with understanding the barriers that different groups might experience (Healy, Kirton and Noon, 2011) and provides an important step for organisations to implement a system of measures that can evidence the multiple barriers employees, and service users, can face.

The focus on equal opportunities was also undermined by the negative publicity created in the late 1980s amid outcries of 'political correctness gone wrong' (Shultz, 1993), which had the effect of discouraging people from engaging with equal opportunities and which created a culture of fear surrounding appropriate terminology. Political correctness was a term coined during the 1980/90s that was defined as "the avoidance of forms of expression or action that are perceived to exclude, marginalise or insult groups of people who are socially disadvantaged or discriminated against" (Oxford English Dictionary, 2021). It is concerned with language, policies, and actions however, it eventually became synonymous as a term to describe an ideological straitjacket (Philpot, 1999; Crawley, 2007). A survey commissioned by the then Commission for Racial Equality revealed that political correctness was thought a

contributing factor to the decline of Britishness. Respondents commented that political correctness was seen to be anti-British because it was perceived to undermine the democratic idea of freedom of speech (Ethnos, 2006⁶).

No respondents took issue with the idea that language is a powerful tool to achieve social change, yet almost all respondents were angry that they felt they could no longer express what they took to be legitimate concerns and criticisms in relation to ethnic minorities (Ethnos, 2006). The prevailing political climate around language and freedom of speech concerning matters of race influenced the way in which equality could be tackled by organisations, and levels of resistance translated from society into the workforce. Dickens (1999) argues that the state as regulator standing back would result in equality being edged out only to be picked up within management prerogative. A period of less enforcement of affirmative action would ensue, and with subsequent political administration changes in the late 1980s, a path had been forged that would let go of past measures to redress the imbalance of past employment discrimination and herald the beginnings of diversity management (Kelly and Dobbin, 1998).

Schwabenland and Tomlinson (2015) describe the anxiety felt among practitioners whilst operating in the equality and diversity field within UK voluntary sector organisations, demonstrating unease in operationalising diversity practice, signalling that political correctness as a perceived 'straitjacket' has not disappeared. In higher education, a recent article (The Week, 2020) described how German university academics felt that a climate of political correctness in universities and colleges is stifling academic debate with a survey showing that four in five academics believe that far-right groups, such as the Alternative for Germany (AfD) party should be able to air their views without fear of reprisals.

A report published in the late 1980s highlighted the need to develop organisational culture and policy for the changing demography of the U.S. workforce to consider the growing numbers of women, black and minority ethnic and ageing employees. The report, *Workforce* 2000, (Johnston and Packer, 1987) warned that the American workforce would over the

⁶ The definition used in the survey described it as: "Carefully chosen language can encourage, promote or establish certain social outcomes and relationships, or the idea that language that excludes or insults a particular group in society should be resisted, or the belief that the resulting changes from these two efforts can benefit society".

subsequent decade grow more slowly, become older, more female, and increasingly disadvantaged, adding that only 15% of new entrants to the workforce would be made up of native white males. This report would become the catalyst for considering the impact of a more diverse workforce and how American industries could take advantage of this. At the same time, the prospect of workforce diversity would create an industry dedicating itself to organisational diversity that has continued to gather momentum ever since. The engagement with diversity management in the UK developed as the business benefits were established in the United States due to the shifting composition of the labour market and supply and demand of labour (Gilbert and Ivancevich, 2000). This slowly provided a shift away from the concept of equal opportunities in UK organisations towards the creation of a business case for attracting and managing a diverse workforce.

Healy, Kirton and Noon (2011) identified that diversity is now part of the discourse of modern business and public organisations, although there may be different organisational drivers. The business case for diversity has been criticised as it can undermine the notion of equal opportunity. Even within an organisation that appreciates the business case for action, any approach taken would be selective and there is no guarantee that the business case would concur with the needs of disadvantaged groups of employees (Dickens, 1999). Similarly, Noon (2007) argues that the business case for diversity is associated with management rhetoric that focuses on individualism, identifying personal traits as more important than social group characteristics and ignores and suppresses the legitimacy of social justice arguments.

Diversity as a concept has been presented as being an understanding and acknowledgement of difference (Kandola and Fullerton, 1998), where diversity is multi-dimensional, taking account of different personal characteristics that are visible, as well as more hidden features, e.g., religion. Mazur (2010) states that these dimensions are not isolated, they interact with and influence one another, and emerge or are displayed differently in different contexts, environments, and circumstances. There is substantial literature posing the advantages, including better problem solving and higher productivity, and sometimes disadvantages, such as cultural clashes and bias, of a diverse workforce (Kandola and Fullerton, 1998; Trenerry and Paradies, 2012; Department for Business Innovation and Skills, 2013; Bassett-Jones, 2005; Mazur, 2010; Riley, Metcalf and Forth, 2013; Özbilgin, et al. 2016; Smulovitz, et al. 2019). There are also considerations required in terms of diversity management and Ivancevich and

Gilbert (2000) identified that a 'gameplan' that includes diversity thinking, discussion and analysis would require patience, full participation and carefully conducted research that would entail expanding the discussion beyond race, ethnicity, and gender.

From a Human Resource Management perspective, there are two approaches that organisations can take in terms of the relationship between themselves and employees which is pertinent to diversity management. The unitarist approach is the view that the organisation as a unitary structure is based on harmony and trust between employer and employee. The emphasis is placed on their common values and objectives, which unite the parties in a common enterprise. This approach suggests there is no inevitable conflict because parties share common interests in the organisation (Van Buren, 2020). Conversely, a pluralistic approach sees the organisation as comprising different groups with common and divergent aims and objectives. These differences require employees to exercise voice to defend their interests because there are inevitable conflicts between employer and employee (Geare, et al, 2014)

Taking into consideration the differences inherent within an organisational workforce, diversity management has been defined as a management philosophy of recognising and valuing heterogeneity in organisations with a view to improve organisational performance (Özbilgin and Tatli, 2011). Mazur (2010) argues that diversity is a subjective phenomenon created by actors based on their own different social identities who categorise others to be similar or dissimilar to themselves. Ng (2008) contends that it can be problematic to approach diversity management effectively if there is insufficient senior commitment and lack of acknowledgement within the organisation that people are more attracted to and prefer to associate with people who are similar to themselves. Organisational leaders and decisionmakers will also exercise discretion as to whether and how to manage diversity as their decisions will be affected by their own commitment to the agenda and their values and beliefs (Bassett-Jones, 2005). Ng (2008) argues that organisations that voluntarily pursue diversity management do so where business objectives coincide with group-based needs, e.g., women or ethnic minorities. In support of this, Syed and Özbilgin (2009) argue that unless diversity is addressed through a comprehensive cultural transformation at multiple levels, diversity management in organisations would be cosmetic and uneven. One can easily observe through the Black Lives Matter movement in 2020 how there have been global calls for organisations

to tackle systemic race inequalities (McCulloch for Personnel Today, 2020). Actions taken by organisations, if any, following the death of George Floyd in May 2020, may be prompted by organisational and reputational risk associated market forces and there has been criticism that businesses should engage beyond adding a hashtag to their external communications campaigns (Davies for The Guardian, 2020).

The discussion thus far has focused on the broader organisational scope of equality, diversity, and inclusion however, this has been distinct within the public sector environment where the driver for change has centred around the legislative requirements implemented by means of public duties. To support this, Greene and Kirton (2009) suggest that the public sector has arguably a better record on equality and diversity because of its proximity to UK equality legislation and given the dual role of the state as employer and legislator. There may also be additional pressure to be better because the public sector is heavily unionised (Greene and Kirton, 2009). The first public sector duty, relating to race, was implemented because of the Macpherson report (1999), the disability equality duty came into force in 2006, followed by the gender equality duty in 2007. In terms of race, the duty amended the Race Relations Act 1976, imposing certain obligations on UK local authorities specifically to make appropriate arrangements with a view to securing that their various functions were carried out with due regard to (a) eliminate unlawful racial discrimination; and (b) promote equality of opportunity and good relations between persons of different racial groups (McColgan, 2015). The power of this duty brought about a significant change in the way local authorities considered their policies and practices, and as a result the Equality Framework for Local Government came into existence to provide closer scrutiny of the main functions within local authorities⁷ (Local Government Association, 2018).

The contemporary Public Sector Equality Duty (PSED) is contained within s.149, Equality Act 2010, replacing the earlier duties in April 2011 and is applicable to England, Scotland, and Wales. The PSED would become operational across the wider public sector, including UK higher education. The PSED as it stands today is specific to public authorities. Public authorities are named within Schedule 19 of the Equality Act 2010, and in broad terms an organisation that falls within the definition is one that exercises public functions (EHRC, 2014).

-

⁷ Knowing your communities; Leadership, partnership and organisational commitment; Involving your communities; Responsive services and customer care; and A skilled and committed workforce

In doing so, the PSED now asks for attention to be given to all nine protected characteristics and those public authorities are required to have due regard to the need to:

- Eliminate unlawful discrimination, harassment and victimisation and other conduct prohibited by the Act.
- Advance equality of opportunity between people who share a protected characteristic and those who do not.
- Foster good relations between people who share a protected characteristic and those who do not.

There is an expectation on public sector organisations to fulfil their PSED requirements by collecting and analysing workforce data across the different protected characteristics. This is with the aim of understanding whether organisations are, for example, recruiting employees who are disadvantaged or under-represented; promoting people fairly whatever their protected characteristic; checking that women's and men's pay is comparable in similar or equivalent jobs or because the work they undertake is of equal value in relation to factors such as effort, skills and decision-making; and making progress towards the aims set out in their equality policy if they have one (EHRC, 2014).

At face value this approach is broadly consistent with Ivancevich and Gilbert's (2000) point that attention should be given to diversity more broadly, however as Syed and Özbilgin (2009) observed, diversity management may not be fit for UK purposes as it is currently articulated, nor does it achieve equitable employment outcomes for diverse employees. There is little empirical evidence that current diversity management practices are adequate to foster race equality at work and Janssens and Zanoni (2014) assert that these practices are ineffective because they focus directly on individuals' cognition rather than address the structural dimensions of privilege, domination, and disadvantage. Greene and Kirton (2009) also make a valid point in stating that diversity management is more difficult within a public sector environment linked to the complexities of the notion of 'customer' and in terms of the resource constraint and budget rationalisation, which is affected by government policy and can deprioritise organisational equality and diversity efforts. To support Syed and Özbilgin (2009) further in terms of their observation, the broadening of the PSED has in practice resulted in the direct focus enabled by its predecessor duty towards specific protected characteristics to become lost. The Black Lives Matter movement has exposed that there has

been insufficient progress on race and an even more focused approach is required to address persistent racial inequalities that exists within the employment sphere.

In 2015, NHS England introduced a Workforce Race Equality Standard citing that a motivated, included, and valued workforce helps deliver high quality patient care, increased patient satisfaction and better patient safety and to ensure employees from black and minority ethnic backgrounds have equal access to career opportunities and receive fair treatment in the workplace. Thereby acknowledging that amidst a comprehensive approach to equality and diversity, a large and complex public sector organisation such as the NHS, felt it was relevant and appropriate to add specific focus to one protected characteristic (NHS England, 2018). In hindsight this is ironic considering the unequal outcomes of ethnic minority people during the COVID-19 pandemic, where disproportionate numbers of ethnic minority people were likely to contract and/or die of the virus. A report by Public Health England (2020) stated that discussions with stakeholders showed that COVID-19 did not create health inequalities, but rather the pandemic exposed and exacerbated longstanding inequalities affecting ethnic minority groups in the UK (Public Health England, 2020).

The attempts made by NHS England to implement the Workforce Race Equality Standard provides an example of a hybrid approach (McGrandle, 2017), involving diversity management *and* moral imperative, to ensure that equitable employment outcomes are addressed. However, this approach is one that has been driven by the desire to tackle race equality by NHS England rather than be mandated through legislation. Following Public Health England's report (2020), it will remain to be seen whether NHS England will revise their race equality standard in future.

In UK higher education, a study of academic recruitment and retention issues in English universities suggested that the diversity debate did not have a high profile amongst academics (Deem, 2007), and may account for the position in the sector today. An earlier study provided a similar outlook to the limited progress made against race equality and ethnic diversity within the public library services in the UK (Durrani, 2002) due largely to the absence of an effective champion. This highlights that there is a disconnect between the theoretical concepts outlined by Ivancevich and Gilbert (2000) given that the diversity management agenda focuses primarily on individual worker attributes, whereas an equality approach considers equality of opportunity, as well as equality of outcome. This has certain implications for the

practical basis upon which the PSED was introduced to the UK public sector to address the acknowledged inequalities relating to particular personal characteristics, as well as the need to diversify public sector organisations to reflect the communities they serve. This aspect will be explored further in this chapter by considering how race as a protected characteristic has been addressed in UK higher education. Like NHS England and their race equality standard, the UK higher education sector does have its own Race Equality Charter, which will be discussed in further detail in section 5.5 of this chapter.

Oswick (2011) explored the patterns and trends of diversity-related work and identified that diversity remains a popular academic discourse, which has been the case in relation to diversity management (Özbilgin and Tatli, 2011; Kandola and Fullerton, 1998; Syed and Özbilgin, 2009; Tatli, 2010). Tatli (2010) observed in her exploration of the diversity management field that there has been a polarisation in the approach to diversity management, between group-based discrimination in employment and the performance related outcomes of diversity. The shift from equal opportunities towards diversity management as a backlash against affirmative action programmes in the US identified by Kelly and Dobbin (1998) has been mirrored to some extent within the UK, especially in terms of the popular misconception surrounding positive action and positive discrimination within employment (Noon, 2010).

The following section will consider how the UK higher education sector has approached equality and diversity and the way it has embedded these principles through individual institutions and sector-specific agencies.

5.3 Equality and diversity in UK higher education

The UK higher education sector is not unique to other public sector organisations in the way it has developed and implemented diversity-related policy and practice. The organisational context differs slightly in that UK higher education institutions have developed into more neoliberal establishments where they are income generating, and therefore more able to reinvest in ways that may not be possible if solely funded by the public purse. However, in his critique of the apparent unquestioned neoliberal context of contemporary organisations, Monbiot (The Guardian, 2016) argues that 'the organisation of labour and collective bargaining by Trade Unions are portrayed as market distortions that impede the formation of

a natural hierarchy of winners and losers. Inequality is recast as virtuous: a reward for utility and a generator of wealth, which trickles down to enrich everyone. Efforts to create a more equal society are both counterproductive and morally corrosive. The market ensures that everyone gets what they deserve'. He goes on to say that this 'neoliberal philosophy' has been accepted and internalised by society, arguing that '... those who fall behind become defined and self-defined as losers'. This is a compelling argument when one considers the inequality that exists for black staff within higher education institutions (see workforce data in Chapter 5). To develop Monbiot's debate further, he is describing society as a collection of neoliberal selves. The neoliberal self is a person who weighs up the pros and cons of a situation from the perspective of 'how does/will it affect me?'. According to McGuigan (2014) when writing about the neoliberal self, he suggested that inequality is no bad thing, since there must be winners and losers of any genuine competition. An interesting consideration in institutions where it would appear that much of the inequality of opportunity is one dimensional.

However, engaging with the notion of winners and losers from an organisational diversity perspective would require a consideration of positionality within Bourdieu's field, which was discussed in Chapter Four of this dissertation. Bourdieu's (1986; 1998) argument is that different forms of capital position people within the field and these positions are relational vis a vis others. If you take a capital portfolio approach, 'loss' is only in relation to the symbolic nature of that capital and therefore, the winner/loser arguments stem from a privileged position. There have always been losers, however in the context of race, the discomfort is that the historical winners (white people) may now be the potential losers as organisations focus their attention on tackling racial inequalities. In the US this fear resulted in 'reverse discrimination' challenges to affirmative action such as the Supreme Court case, *Regents of the University of California v Bakke* [1978], where this argument stems from white privilege. Positioning this from an organisational perspective, HR professionals and diversity practitioners would be uncomfortable in acknowledging 'losers' as this might also imply white HR/diversity practitioner perspectives where the status quo may be under threat.

There are multiple and complex challenges to consider in prioritising various policies and initiatives that litter the organisational setting to keep up with ever-changing approaches and practices. The organisational characteristics of higher education institutions, particularly those with strong traditions of professional autonomy and freedom in decision-making, can

make this environment challenging to successfully implement policies and strategies to address race within the broader equality and diversity context (ECU, 2011). Despite this, the general assumption is that the public sector is a more inclusive and diverse working environment and is supported to some extent by Greene and Kirton's (2009) assertion that the public sector has a better record at equality and diversity. In advancing equality and diversity for marginalised groups within complex organisations there will naturally be winners and losers, and this is likely to cause disquiet and unrest amongst the losers, especially if they come from the dominant racial group.

In the late 1980s a review of equal opportunities policies was conducted in higher education to establish how it compared to other public sector organisations (Williams et al, 1989). That study found that few higher education institutions had any policies in place. By the late 1990s, a survey found that nearly all universities had equal opportunities policies in place (Commission on University Career Opportunity, 1997), however the report expressed concern that those policies needed to be actively pursued, and there were several organisations where policies were not being implemented. Hoque and Noon (2004) referred to these types of policies as 'empty shells', where an equal opportunities policy may have been adopted by an organisation but not enacted through action plans or enhanced opportunities for disadvantaged groups, such as targeted recruitment.

Hoque and Noon (2004) also observed that there was no greater likelihood of a public sector organisation having an equal opportunities policy in place, but where they did do so, it was less likely for that policy to be an 'empty shell'. This was potentially due to public sector organisations being unionised, having specialists in place, e.g., human resources and equality and diversity officers, and with the potential for increased public scrutiny. Also, the authors found they could predict the incidence of equal opportunities policies according to workforce characteristics. In the context of race for example, workplaces where more than 10 percent of its workforce was from a minority ethnic background were more likely to have a policy related to ethnicity than those with fewer than ten percent (Hoque and Noon, 2004), demonstrating that an ethnically diverse workplace can to some extent be the catalyst for the promotion and implementation of inclusive policy and practice.

There has been a significant shift in the way that higher education institutions have increased the visibility of equality and diversity and this direction of travel has been assisted through

heightened prominence by government funded and subscription-based agencies that support UK higher education institutions. The growing focus on the importance of equality, diversity, and inclusion in the provision of education to the public, as well as the need to reflect the communities that UK universities serve through its workforce has become embedded in the strategic vision of these agencies as well as many higher education institutions. For example, student diversity has been largely driven by Government policy to increase the participation of the black and minority ethnic student population, especially in low participation communities, through the setting of national targets⁸ (Turner, 2019; Connell-Smith and Hubble, 2018). Diversifying the student population has largely been a successful enterprise by universities over the years, yet Bhopal (2018) reminds us that the outcomes of those minority ethnic students may not be as successful as their white counterparts.

The Higher Education Funding Council for England (HEFCE), since superseded by the Office for Students (OfS) in 2018, funded and regulated universities and colleges in England. They recognised that a diverse and inclusive organisational culture made for a more effective and productive workforce (HEFCE, 2018). HEFCE also had in place its own Equality Objectives and annually reported against its Equality and Diversity Scheme. Its 2016 report stated that the rationale for enhancing equality and diversity in higher education is based on two tenets; a business case to embed equality of opportunity and foster diversity that creates conditions for an excellent higher education system; and a social justice case where the agency is uniquely placed to support the higher education sector. Their approach to equality was to guide, encourage and test that equality is being considered across the sector, whilst recognising that each institution is responsible for its own compliance (HEFCE, 2016).

The articulation of HEFCE's rationale for equality and diversity summarises the way that higher education institutions approach the agenda in practice. It is one that considers the business benefits of diversity, together with the legal rather than social justice case, to reduce barriers and improve the rate of engagement from a range of stakeholders. Neither HEFCE nor OfS incentivise nor enforce higher education institutions to proactively address race equality, diversity and inclusion in ways that are similar to the constraints in place for the Athena SWAN (gender) awards, which inhibited institutions being shortlisted for research

_

⁸ Set in 2015 to increase the number of black and minority ethnic students entering higher education by 20 percent by 2020

funding. This is discussed later in this chapter. The OfS also have equality objectives in place to 2022 (Appendix 1) which, not surprisingly, have a primary focus on prospective and existing students within higher education. There is little focus on their workforce and less of an indication within the document of how OfS will work to ensure these objectives are achieved.

Another prominent higher education-focused agency is the Equality Challenge Unit (ECU), which merged with other higher education agencies to become Advance HE in 2018. The ECU was funded by HEFCE from 2002-2015 to focus on sector-specific work related to equality, diversity, and inclusion. From 2015 it changed its funding stream to rely on a subscription model that required individual higher education institutions to pay an annual fee on a sliding scale based upon the institution's income. The ECU's mission was to work in partnership with universities, colleges, and other sector bodies to advance equality, promote diversity in their staff and student bodies and build inclusion into all aspects of higher and further education (ECU, 2018).

The ECU provided practical solutions through guidance, research, statistical reports, training, and consultancy to support the higher education sector in their work to advance equality, diversity, and inclusion. In addition to these services, the ECU also coordinated the gender equality (Athena SWAN) and Race Equality Charter (REC) awards, to address the cultural and systemic issues that can create barriers for people with these characteristics. At the time of writing, there are 164 Athena SWAN members holding 962 awards between them (July 2021) and 80 REC members and 17 award holders (July 2021). Both charters have key principles in which institutions commit to become a member (Advance HE, 2021).

These award initiatives are not mandatory, except in 2011 the UK Chief Medical Officer announced that the National Institute for Health Research (NIHR) would only expect to shortlist medical schools for Biomedical Research Centre (BRC) and Unit (BRU) funding if the associate academic school holds a Silver Athena SWAN award. This was later expanded to include Patient Safety Research Centre funding in 2012 (Advance HE, 2020). The link between NIHR funding and Athena SWAN has recently been removed (Oloyede et al, 2021). There has been no such incentive with the REC, which may account for the far fewer institutional members and even fewer award holders. Research undertaken on behalf of the University and College Union (Bhopal and Pitkin, 2018) investigated the impact of the Race Equality Charter mark across UK higher education and highlighted that the charter enabled institutions

to focus more on race, however there was still more to do that could address ethnic minority underrepresentation in institutions and called for more senior accountability in championing race equality.

There are other agencies with an interest in UK higher education that promote equality, diversity and inclusion in the sector; Universities UK (UUK), whose primary function is policy, advocacy, analysis and research; the Higher Education Academy (HEA), whose purpose is to champion teaching excellence; and the Leadership Foundation for Higher Education (LFHE), whose function involves developing and improving the management, governance and leadership skills of existing and future leaders of higher education. The LFHE have several programmes that support the equality and diversity agenda in the sector. Their focus is to address the under-representation of women and minority ethnic leaders through mentoring programmes. The Bell Review (2017) recommended that the number of core agencies taking subscriptions from institutions (particularly in England) should reduce from nine to six over the succeeding two years. This meant that from August 2018, a new agency, Advance HE, brought together the functions of the ECU, the HEA and the LFHE.

In addition to these sector agencies, it is important to mention the role that the Trade Unions have in promoting and advancing equality, diversity, and inclusion. There are two recognised Trade Unions within UK higher education; University and College Union (UCU) that supports members employed on academic contracts; and Unison, which supports members employed on professional and support contracts in universities. UCU claims to put equality at the heart of its activities on behalf of its members, yet only some branches or local associations have equality officers to assist members if they experience difficulties (UCU, 2018). However, the Trade Union does have comprehensive resources and guidance available online, as well as member networks for particular interest groups. UCU also conducts research on issues affecting the academic workforce. Unison also promotes equality and diversity and have resources and guidance available together with local and national interest group networks.

This section has identified the range of internal and external actors and agencies across the UK higher education sector with an interest in the design and implementation of change using equality and diversity concepts as a catalyst. The following section will consider the role of equality and diversity practitioner as change agent and explores the opportunities and challenges available to this practitioner cohort.

5.4 The role of the equality and diversity practitioner as change agent

The theoretical and practical implications of embedding organisational equality and diversity have received modest academic attention despite acknowledgment that internal change agents are required in the design and implementation of diversity management policies and initiatives (Tatli and Özbilgin, 2009; Tatli et al, 2015). According to Tatli et al (2015), equality and diversity officers play a key role in negotiating change and it is of significant academic and policy importance that the roles of these agents are understood in initiating and promoting change. On this basis, it could be argued that sector agencies could play a more proactive part to progress equality and diversity in the higher education sector by mobilising a team of diversity specialists to work with institutions.

The environment within which equality and diversity practitioners operate can be extremely complex. More than two decades ago, Agócs (1997) provided important context by highlighting the difficulties change makers experience whilst dealing with individual and organisational resistance. This resistance can take multiple forms and manifest itself across multiple layers, including behaviours that decision-makers employ to actively deny, reject, refuse to implement, repress, or dismantle change proposals and initiatives. The institutionally embodied form of resistance is expressed through organisational structures and processes of legitimation, decision-making, and resource allocation and this can have a significant impact on the operational effectiveness of the ability, capability and credibility of equality and diversity practitioners in the field in which they operate. If that practitioner is female or of ethnic minority background, they would find themselves working within the very structures and conditions that create and perpetuate their disadvantage, and therefore the playing field is never level (Agócs, 1997). Agócs (1997) described a typology of the forms of institutionalised resistance as:

- 1. Denial of the need for change:
 - a. Attacks on the credibility of the change message
 - b. Attacks on the messengers and their credibility
- 2. Refusal to accept responsibility for dealing with the change issue
- 3. Refusal to implement change that has been agreed to
- 4. Repression: action to dismantle change that has been initiated.

Wiggins-Romesburg and Githens (2018) view resistance to diversity as a dynamic interplay of individual and collective behaviours, with individual resistance rooted in unconscious motivation and organisational resistance rooted in the collective behaviour of individuals. Interestingly, there are several similarities in Agócs' (1997) typology of resistance and DiAngelo's (2018) explanation around the way white fragility can manifest itself when an individual is faced with an accusation of racism, particularly in terms of attacking the messenger, denial and undermining the importance of racism.

In Healy and Okielome's (2007) study of the UK NHS, they provided a framework to locate equality actors into a four-fold classification:

- Management sponsored diversity actors (in particular equality/diversity officers) employed to
 ensure at least the minimum compliance with the law but may seek to employ a more radical
 agenda of culture change and management sponsored networks: for example, Black
 networks.
- 2. Trade union sponsored actors: for example, Black workers' conferences, women's conferences, self-organising groups.
- 3. National identity networks, independent of management and trade unions, which focus on country-of-origin identity.
- 4. Community networks, which are community alliances.

This section of the chapter will solely focus on management-sponsored diversity actors in the first classification. Equality and diversity practitioners' roles can complement or even replace some of the further classes identified above, e.g., Trade Unions (Healy and Okielome, 2007; Tatli and Alasia, 2011). Where these other actors are placed within or have alliances with an organisation, equality and diversity practitioners will at some point engage with those stakeholders in their role.

Organisations may also identify a senior champion to drive the organisational equality and diversity strategy (Ng, 2008), however Healy and Okielome (2007) warn that high profile equality roles might be seen as the public face of strategic action, while at the same time may also disguise strategic inaction. At an operational level, an important factor for effective change relies on the organisational placement, and the level of seniority, of the practitioner as this may help to legitimise the role and the change agenda more broadly (Tatli and Özbilgin,

2009; Tatli and Alasia, 2011; Tatli, et al, 2015). Due to the organisationally disruptive nature of equality and diversity roles, power is required to fundamentally challenge the status quo (Healy and Okielome, 2007) and it is power that can enable change to happen through the constructive confrontation of entrenched individual and institutional views, attitudes, and behaviours.

Tatli et al (2015) considered the opportunities and constraints experienced by equality and diversity officers through situatedness, relationality and praxis. The study found that practitioners in this field experienced lack of resources, which might indicate an absence of senior-level commitment. The equality and diversity officers' perceptions also included lack of prestige, lack of seniority and lack of authority, which made it difficult for officers to be seen as credible and influential agents for change within their institutions (Tatli et al, 2015). It has been acknowledged that equality and diversity practitioners suffer from greater jobrelated stress than their peers (Kandola, et al, 1991). This may be due to their persistent attempts to navigate often complex organisations to lead strategy development, design, delivery, implementation and monitoring of organisational change (Tatli and Özbilgin, 2009), whilst experiencing resistance throughout the journey to achieve organisational change; and persistence attracts resistance. Ahmed (2012) states that the more resistance, the more persistence is required. Furthermore, practitioners must steer themselves between the contradictions that arise between the sponsorship provided by senior management and the potential risk from a lack of will from others in the implementation of change, particularly if that involvement does not form part of their usual role (Acker, 2000).

Equality and diversity practitioners require capability to address the full agenda of equality and diversity, which of course is complex. This involves having the requisite knowledge and skills to apply a narrative that the organisation finds palatable, but also the tenacity required to persist when there is a lack of support (Ahmed, 2012). This can be exacerbated by the absence of any specific qualification criteria for fulfilling an equality and diversity role, which may result in organisations appointing practitioners with little/no previous experience in the area. Practitioners may therefore be inclined to accept the hegemonic norms and values of their organisation without challenge, which in turn will adversely impact the practitioners' capacity to be effective (Tatli, 2011). Tatli (2011) continues to remark that the perception of an 'unqualified' practitioner may impair the legitimacy of the diversity management field and

practitioners within that field at large making it much more challenging for practitioners to gain status within their institution, enabling access to senior leaders, and which can secure buy in from internal stakeholders.

The academic discussion surrounding the field of equality and diversity practitioners has advanced to consider further aspects of power, structure and agency to analyse the opportunities and challenges arising for these practitioners as change agents through the lens of Bourdieu (Özbilgin and Tatli, 2011; Tatli, 2011; Tatli and Alasia, 2011; Tatli and Özbilgin, 2009; Tatli, et al, 2015), and was considered in Chapter 3 through Bourdieu's concepts of agency, field, habitus, capital and symbolic violence (Bourdieu, 1977, 1984, 1990, 1998).

This section has considered some of the challenges affecting equality and diversity practitioners, despite a lack of sufficient research into this area. However, it does provide an insight into the challenges that might impact the effective design and implementation of equality and diversity strategies and policies within organisations. The practitioners' role can also be impacted by the organisation's own priorities, given that the agenda is so broad it is difficult to apply an equal amount of effort to all groups. As a result of this, there has been a focus on specific characteristics, e.g., gender or race etc, not only in research, but in practice. Acker (2006) critiqued that often research is focused on a specific characteristic, and rarely considers the intersections of identities, such as gender, race, and class. Focusing on one category obscures and oversimplifies other interpenetrating realities, and in terms of race, this encapsulates multiple social realities, which are inflected by gender and class differences (Acker, 2006; 2012). This has been demonstrated by some of the initiatives implemented across a range of organisations aimed at increasing the representation of women on Boards, in male-dominated industries and more specifically to higher education, the implementation of Athena SWAN and REC awards. The following section will consider the rationale for focusing on race within the broader equality and diversity context within higher education and how the sector is advancing equality in this area. The literature review will highlight the evolution of equality and diversity within higher education and track those considerations through to the environment within contemporary institutions.

5.5 Locating race amid a broader equality agenda in UK higher education

The *Dearing Report* (The National Committee of Inquiry into Higher Education, 1997) recommended that higher education institutions should identify and remove barriers that inhibit recruitment and progression for social groups and monitor and publish their progress towards greater equality of opportunity. The report identified that people from minority ethnic groups were at a structural disadvantage in the academic labour force, and experienced harassment and negative stereotyping as well as discrimination in recruitment, selection, and promotion processes (JNCHES and ECU, 2003). The report's recommendations, which included the development of race equality policies; increasing the representation of ethnic minorities; and improving the quality of statistical data, became embedded within the *Equal Opportunities in Employment in Higher Education: 'Framework for Partnership'* agreement between the Universities and Colleges Employers Association (UCEA) and the national higher education Trade Unions. This agreement was initially implemented in 2000, however was revised in 2003 and renamed the Partnership for Equality: Action for Higher Education (JNCHES and ECU, 2003).

Concurrently, a study undertaken later on that decade (Carter et al, 1999) began to highlight that a third of institutions did not have a race equality policy in place; that three quarters of institutions routinely monitored recruitment applications by ethnicity; that more than 25 percent of minority ethnic academics had personally experienced discrimination in recruitment; that 55 percent of British ethnic minority staff believed that there was discrimination in employment in higher education; and that ethnic minority staff and research students expressed resentment at being stereotyped because of their ethnicity (Carter et al, 1999).

The recommendations made in UCEA's (2000) report was interpreted by the University of Leeds (Turney, Law and Phillips, 2002) soon after its publication, as 'institutional racism', by identifying weaknesses in the foundations that higher education structures are built upon; the policies, processes, culture, and attitudes that underpin the workings of any higher education institution. Further qualitative studies undertaken into the experiences of black and minority ethnic staff in higher education have found that there is a lack of patronage and support, (Jones, 2006); therefore, having the effect of denying cultural capital to staff of

colour (Chapter Four discussed capital). Later studies identified multiple barriers, harassment and disadvantage experienced by black and minority ethnic staff in UK higher education (ECU, 2011: Rollock, 2011,2019; UCU, 2016, 2017; EHRC, 2019), which were discussed in Chapter Two.

Career progression, recruitment, attitudes, organisational culture, and meeting the needs of ethnic minority staff in higher education are all crucial elements to creating an inclusive working environment. Deem (2007) highlighted that from an organisational perspective the concept of an equitable higher education institution, where there is no discrimination "does not sit easily with managerial, business-focused approaches to the running of higher education" (Deem, 2007: 619), due to competing priorities within higher education governance models. Deem continues to point out that for many senior academic staff, a meritocratic system of higher education has been one where their own careers have been developed, which may result in some staff being less willing to challenge the system that has supported them (Deem, 2007).

A study identified hidden inequalities for staff in higher education (Institute of Employment Studies, 2005), which included a pay differential between minority ethnic staff and white staff; a higher proportion of white staff involved in academic and research areas than black and minority ethnic staff; and higher proportions of black and minority ethnic staff with shorter length of service than white respondents to the study, indicating that qualitative data would be beneficial. There were also high levels of mistrust from black and minority ethnic respondents in relation to the collection of ethnicity data in comparison to white respondents. A further study looking at the recruitment and retention of academic staff in higher education (Metcalf et al, 2005) failed to report in any detail about the experiences of black and minority ethnic staff. This sizeable report had little to offer in relation to the levels of satisfaction of black and minority ethnic staff in higher education across all aspects of recruitment and retention, including recruitment practices, promotion, or indeed career development. Considerations of ethnicity within the report have been contained in the sections dedicated to equal opportunities and the section dedicated to discrimination offers no more than a minor focus on pay differentials.

A briefing from the Centre on Dynamics of Ethnicity (CoDE) at the University of Manchester (2014) showed that black and minority ethnic people of working age in England and Wales

were more likely to have degree level qualifications than white British people. The groups with the highest proportion of people with degree level qualifications were Chinese (43%), Indian (42%) and Black African (40%) (CoDE, 2014). Despite this, they are less likely to gain access to higher education as an employee as the statistics later in this chapter indicate. Further research will be required to establish how the higher education sector attracts and recruits prospective black and minority ethnic candidates, and whether this is a contributing factor to the under-representation of minority ethnic staff at all levels and across all roles. Added to the suggestions of systemic racism in the higher education sector, it may also be useful to understand the reasons why minority ethnic employees leave their institutions as this may not always be articulated at the time of departure through exit interviews.

There has been a plethora of press activity over the past decade in relation to the under-representation of black and minority ethnic academics and the levels of racism within the sector, with headlines such as "14,000 British professors – but only 50 are black" (Shepherd for The Guardian, 2011) and "Race discrimination in academia 'has not improved' over past 20 years" (Gibney for Times Higher, 2013), "More than half of BME university lecturers and staff have suffered racial abuse at work" (Garner for The Independent, 2014), "UK universities condemned for failure to tackle racism" (The Guardian, 2019), demonstrating that the issues experienced by people of colour within institutions is persistent and systemic and shines a public spotlight on the sector's inability, or perhaps unwillingness, to address this problem as a recent article headlined "Cambridge may drop BAME mentoring of white academics" suggested The Guardian (2020).

In the background there have been some developments to address race as a separate agenda within the broader context of equality. Once again highlighting the need to focus on specific protected characteristics to reduce employee disadvantage. If the higher education sector had solely focused on diversity as an approach, these disadvantages may not have been identified because of the focus that diversity management has on individualism. As mentioned previously, Advance HE (previously ECU) coordinates the Race Equality Charter (REC) mark, which was rolled out to the higher education sector following a pilot exercise in 2015, involving 21 institutions. The REC is underpinned by five fundamental principles, which are detailed in Appendix 2.

The REC requires institutions to carry out a self-assessment of their organisation across a range of functions that consider both the staff and student populations using quantitative and qualitative data. The analysis covers the self-assessment process; the institution and local context; staff profile; academic and professional and support staff recruitment, progression, and development; the student pipeline; and teaching and learning. The process of analysis and submission is very similar to that employed with the Athena SWAN award scheme, however the REC is only applicable on an institutional basis, whereas Athena SWAN awards can also be sought at departmental level once the institution has gained an award.

The REC requires institutions to pay regard to race, whilst still focusing on a broader equality and diversity agenda, and institutions are expected to consider the intersections of other characteristics together with race in terms of their staff and student populations. This is a positive step although this is contingent on the institution identifying the relevant intersections and having the data available to undertake the analysis. According to ECU's *Race Equality Charter Awards Handbook* (2016) there is an expectation that intersectionality is considered increasingly as progress is made with the institution's race equality work. For first-time bronze applicants, this consideration might be aspirational, for example to begin building in mechanisms to understand and explore intersectionality.

Despite focused attempts to address race equality in UK higher education institutions using REC, there is still much work required. A recent review of the impact of REC (Oloyede et al, 2021) has identified recommendations for improvements to the scheme. In 2019, the Equality and Human Rights Commission launched an official inquiry into racial harassment in higher education. Their inquiry looked at the different types of racial harassment experienced by students and staff, the routes for reporting racial harassment and how effectively those reports are dealt with. The report (EHRC, 2019) highlighted how institutional systems and processes affect the outcomes of minority ethnic staff and students affected by racial harassment working and learning in UK higher education institutions and made several recommendations that cover three themes that arose from the research. These surround protections, transparency, and scrutiny, by increasing university transparency on how they are tackling harassment; effective redress and how available, accessible and effective routes to redress are; and changing university culture, where university leaders understand harassment and governing boards carry out due diligence and where appropriate, take action

on tackling harassment in line with the PSED by taking more responsibility for preventing and addressing harassment. The report makes ten recommendations across these three themes to higher education institutions, external agencies, and higher education funding councils. In the wake of these recommendations, there has been a noticeable increase in the number of seminars (webinars because of the Coronavirus lockdown in 2020) on the subject of race in the higher education sector and this will no doubt be as a direct result to the events of 2020, with a view to discussing and operationalising the issues raised.

This section has considered the advances in approach taken across the UK higher education sector that have had, and continue to have, an impact on race equality but which indicate the level of progress achieved in the last two decades. The following section will explore and reflect on the UK higher education workforce demographics with a focus on ethnicity across several aspects including overall representation at different levels, salary bands, pay reporting and exit from UK higher education to identify the unequal outcomes for black staff working in institutions.

5.6 The baseline of ethnic diversity in the UK higher education workforce

The first real attempt at drawing together the findings of existing research and to establish a baseline for the experiences of the black and minority ethnic workforce in higher education by a sector agency came with a literature review (ECU, 2009). This resource aimed to provide an outline that did not solely rely on data, which has been reported through the Higher Education Statistics Agency (HESA) for more than a decade, but that reported on staff experience. Higher education institutions are expected to follow prescribed data collection requirements for the Higher Education Statistics Agency (HESA) to gather demographic information relating to UK higher education staff and students across different protected characteristics (HESA, 2020). The diversity categories used by HESA are informed by those used in the UK population census, which has become the benchmark across the public sector and beyond.

The findings from this report showed a low representation of black and minority ethnic staff (8.6 percent academic and 6.9 percent professional and support staff) of which overall only 6.1 percent of staff were UK nationals as opposed to 22.3 percent who were non-UK nationals (based on 2006/7 HESA data). There has been little shift in demographics from an earlier

ethnic analysis undertaken using 1996/7 HESA data (Carter et al, 1999), which showed that black and minority ethnic academic staff accounted for just 5.5 percent of the total academic staff population.

The ECU (now Advance HE) found that the higher the grade, the lower the black and minority ethnic staff representation and mirrors the findings in the Carter, et al (1999) analysis that showed higher concentrations of black and minority ethnic academics at lecturer level. A study of staff perceptions of equality policies in higher education institutions also showed that black and minority ethnic staff tended to be found mainly in lower-level support roles or temporary posts (Deem and Morley, 2006). The report supported that overall black and minority ethnic staff received lower levels of pay on average than white staff and were less likely to benefit from permanent or open-ended contracts (ECU, 2009).

The most recent census statistics (ONS, 2011) show that the black and ethnic minority population represented 14% of the total population in England and Wales. The working age employment rate in the UK (2018) showed that 77.2 percent of the white population was in work compared to 67.3 percent of the black African/Caribbean population (Joseph Rowntree Foundation, 2019) and that 9.7 percent of the black African/Caribbean population aged over 16 years were unemployed compared to just 3.3 percent of the white population (ONS, 2019). According to the last available Annual Population Survey, 6 percent of the black African/Caribbean population identified as being employed in the 'Managers, directors and senior officials' category compared to 11 percent of the white population in the UK (ONS, 2018). These data have been further supported by a comprehensive Governmentcommissioned review undertaken in 2017, which highlighted numerous concerns around the underemployment, overqualification and economically disadvantaged state of ethnic minorities in Britain (McGregor-Smith, 2017). According to the Race Disparity Audit (revised March 2018), relative to the UK population overall, people living in households headed by someone in the Asian, black, or other ethnic groups were disproportionately likely to be on low income. 16 percent of black households were in persistent poverty (Cabinet Office, 2018). The report also found that black households were more likely to be single parent households, limiting the number of potential earners and in terms of actual household income, black households and those from other ethnic groups were most likely to have a gross weekly income from all sources of less than £400 (Cabinet Office, 2018). The Chartered Institute of Personnel and Development (CIPD) also reported on the barriers experienced by ethnic minority workers in achieving career progression and noted that black employees (29 percent) were more likely than white British employees (11 percent) to say they had experienced discrimination in the workplace (CIPD, 2017).

From a broader UK public sector perspective, the Race Disparity Unit (Cabinet Office, 2019) released a summary discussing how ethnically diverse public sector workforces are, the changes in diversity over time and whether diversity targets had been set. The main findings were that NHS medical staff (44.4 percent) are the most ethnically diverse workforce and that firefighters (4.1 percent) were the least ethnically diverse. NHS medical staff have the highest percentage of leaders from ethnic minorities, accounting for two in five consultants. However, most consultants (30.7 percent) are from an Asian background compared to 2.9 percent who identify as black. Contrasted with 16% of the working age population in England and Wales in 2017 that were from an ethnic minority background, of the twelve workforces included in this report, four had a percentage of ethnic minority staff that was higher than this. There had been little change in the ethnic diversity of public sector workforces between 2014 and 2018 (Cabinet Office, 2019).

In the private sector, the Parker Review (2017) made several recommendations including that each FTSE 100 Board should have at least one director of colour by 2021 and that each FTSE 250 Board should have at least one director of colour by 2024. In its update in February 2020, the Parker Review Committee reported that 172 directors of colour held 178 director positions across the FTSE 350 companies. This amounted to 6.8 percent of all FTSE 350 directors, including directors where ethnicity is unknown. The report highlights that 59 percent of companies did not meet the target of having at least a director of colour on their Board, and there was a particular lack of ethnic diversity observed among FTSE 250 companies (Parker Review, 2020).

In terms of the UK higher education workforce, the most recent data available through the staff statistical report for 2018/19 (Advance HE, 2020) shows that the representation of black and minority ethnic staff has marginally shifted, and of the total workforce in UK higher education that declared an ethnic background⁹, black and minority ethnic staff accounted for

-

⁹ Note that Advance HE statistical data only reports against declared ethnicity and does not provide figure as a proportion of the whole sector workforce, which may be less than stated

14.5 percent (including non-UK nationals). In England alone, this was slightly higher at 15.7 percent of the total English higher education workforce. Among the total UK national staff, 10.3 percent (11.6 percent in England) were from a black and minority ethnic background, as opposed to 30.7 percent (31.5 percent in England) who were non-UK nationals in the UK workforce¹⁰. A UCEA report (2019) concerning the UK higher education workforce stated that the ethnic diversity of the academic workforce is in fact masked by the significantly increased ethnic diversity of non-EU international staff members (UCEA, 2019). This observation has been demonstrated in these most recent workforce data as UK ethnic minority academic staff represented 7.2 percent, whereas non-UK academic staff represented 9.6 per cent. There has not been similar commentary around the ethnic diversity of professional and support staff in UK institutions. However, these data show an overall under-representation of ethnic minority staff as 9.1 percent who are UK staff compared to only 3.1 per cent that are non-UK staff. Considering that this employee cohort would be recruited from surrounding populations to UK institutions, theoretically it should be more likely to draw higher proportions of ethnic minority candidates for these roles.

In 2018/19 there were higher proportions of UK national black and minority ethnic staff (23.7 percent) who were employed on fixed-term contracts than white staff (19.4 percent) across all contracts and black and minority ethnic academic staff are almost twice as likely to be employed on a fixed-term contract (31.4 percent) than their professional and support staff colleagues (18 percent). The representation of UK national black and minority ethnic academic staff has risen by just over two percentage points since 2003/4 from 4.8 percent to 7.2 percent in 2018/19. The representation of black and minority ethnic UK national staff in professional and support roles in our institutions has seen a slightly greater shift, with a rise of just over four percentage points since 2003/4, from 4.8 percent to 9.1 percent in 2018/19.

Of the academic staff that are recorded as employed in category SOC1 (Managers, directors, and senior officials) in UK higher education 5.2 percent (n=25) are from a black and minority ethnic background. Of these there are no academic staff with a declared black ethnic background. In terms of UK national professional and support staff employed at this level, 6.6 percent are from a black and minority ethnic background (n=705). This represents 3.6 percent

 $^{^{10}}$ Note: it is not possible to make comparisons between English regions as this data is not available from Advance HE

of total black and minority ethnic staff compared with 5.9 percent of total white staff employed at this level (Advance HE, 2020). An article featured in The Guardian headlined with 'British universities employ no black academics in top roles, figures show' (Adams, 2017) highlighting that figures recorded zero black academics in the elite staff category of 'managers, directors and senior officials' for a third year in a row. From these latest figures, that headline has remained static.

In relation to the ethnic diversity of subject areas across UK higher education, there is a slight difference according to SET (science, engineering, and technology) and non-SET disciplines, with 11.7 percent of UK national black and ethnic minority academics in SET compared to 9 percent in non-SET subject areas. For SET disciplines, archaeology is the least ethnically diverse subject area and overall Asian staff make up most ethnic minority UK academic staff, whereas UK black academic staff make up only 1.4 percent of this cohort. Within SET, the highest proportion of UK black academic staff are located within nursing and allied health professions (25.8 percent) and clinical medicine (20 percent). In non-SET subject areas, classics is the least ethnically diverse and the highest proportion of UK black academic staff are represented in business and management studies (32.4 percent) and law (9.2 percent) (Advance HE, 2020).

Table 3: UK national staff by age group and ethnicity (Advance HE, 2020^{11})

BAME White All staff												
	BAME	BAME					All staff					
	No.	↓%	→%	No.	↓%	→%	No.	↓%				
UK nationals												
<u><</u> 25	2,875	8.5	14.7	16,720	5.7	85.3	19,595	6.0				
26-30	3,740	11.1	12.0	27,520	9.4	88.0	31,265	9.6				
31-35	4,515	13.4	11.7	33,975	11.6	88.3	38,495	11.8				
36-40	4,930	14.7	11.7	37,140	12.7	88.3	42,070	12.9				
41-45	4,455	13.2	11.3	34,910	11.9	88.7	39,365	12.1				
46-50	4,430	13.2	10.1	39,430	13.5	89.9	43,860	13.5				
51-55	4,020	12.0	9.0	40,775	14.0	91.0	44,800	13.7				
56-60	2,765	8.2	7.6	33,795	11.6	92.4	36,560	11.2				
61-65	1,400	4.2	6.8	19,120	6.5	93.2	20,520	6.3				
<u>≥</u> 66	515	1.5	5.5	8,780	3.0	94.5	9,295	2.9				
All UK nationals	33,645	100.0	10.3	292,175	100.0	89.7	325,820	100.0				

Analysing the age profile of ethnic minority staff in institutions may provide some context to the overall lack of ethnic diversity at higher levels. Table 3 shows that black and ethnic minority staff are represented in higher levels compared to white staff up to the age of 45

-

 $^{^{11}}$ Taken from Equality in Higher Education: Staff Statistical Report 2020, Advance HE

years. Although this may not wholly account for the lack of ethnic minority representation at higher levels of academia, there are specific hierarchical structures in terms of progression for academic staff, which may to some degree be associated with age.

From the total UK national staff at professorial level, 9.1 percent are from a black and minority ethnic background. This equates to 9.7 percent of the total black and minority ethnic UK academic staff population compared to 11.2 percent of the total UK national white staff at this level. The headlines become clear from the disaggregation of these figures, where specific ethnic groups can be explored further. For example, in 2018/19 from the total black UK national academics, 4.5 percent are at professorial level, equating to just 100 academic members of staff. There are a further 45 black professors who are non-UK nationals. When considering the intersection of ethnicity and gender, there are a total of 490 female ethnic minority professors in the UK, 355 of these are UK nationals. Of the total female ethnic minority professors, only 35 are from a black ethnic background, of which 30 are UK nationals. This compares to 1,575 ethnic minority male professors, of which 1,035 are UK nationals. From this cohort, 105 are from a black ethnic background of which 70 black male professors are UK nationals.

Table 4: UK academic staff by professorial category and ethnic group (Advance HE, 2020)¹²

		22 / 1 2		5 /	5	, ,	. ,			
		Professors			Non-professo	ors		All staff		
		No.	↓%	→%	No.	↓%	→%	No.	↓%	
W	White	13,845	90.9	11.2	109,485	89.4	88.8	123,335	89.6	
BAME	BAME total	1,390	9.1	9.7	12,995	10.6	90.3	14,390	10.4	
Α	Asian	555	3.6	9.5	5,300	4.3	90.5	5,855	4.3	
В	Black	100	0.6	4.5	2,105	1.7	95.5	2,205	1.6	
С	Chinese	335	2.2	16.7	1,660	1.4	83.3	1,995	1.4	
М	Mixed	205	1.3	7.9	2,385	1.9	92.1	2,585	1.9	
0	Other	200	1.3	11.5	1,545	1.3	88.5	1,745	1.3	
All	All UK staff	15,240	100.0	11.1	122,485	100.0	88.9	137,720	100.0	
W	White	3,915	85.3	9.3	38,225	67.6	90.7	42,140	68.9	
BAME	BAME total	675	14.7	3.6	18,340	32.4	96.4	19,020	31.1	
Α	Asian	270	5.9	3.9	6,690	11.8	96.1	6,965	11.4	
В	Black	45	0.9	2.2	1,875	3.3	97.8	1,915	3.1	
С	Chinese	200	4.3	3.6	5,290	9.4	96.4	5,490	9.0	
М	Mixed	50	1.1	2.6	1,840	3.3	97.4	1,890	3.1	
0	Other	115	2.5	4.2	2,645	4.7	95.8	2,755	4.5	
All	All non-UK staff	4,590	100.0	7.5	56,565	100.0	92.5	61,155	100.0	

To put the statistics into perspective: UK national black staff account for 21.4 percent of the total black and minority ethnic UK national workforce in UK higher education (Advance HE, 2020). UK national black staff are the second largest ethnic group from the UK national workforce that declared their ethnicity, yet black academics are less likely than any other

_

 $^{^{\}rm 12}$ Taken from Equality in Higher Education: Staff Statistical Report 2020, Advance HE

ethnic group to be represented at professorial level, making up just 0.6 percent of UK national professors. In terms of black staff that are employed in professional and support roles, the largest concentration of UK national black staff (34.1 percent) are employed in administrative and secretarial occupations. UK national black professional and support staff (17.3 percent) are more likely to be represented in SOC9 (elementary occupations) than any other ethnic group (Advance HE, 2020). The data has shown that the sector still has some way to go in ensuring that there is an even distribution of black staff across all levels, particularly in senior levels of academia.

In response to the under-representation of racially minoritised staff within senior levels of UK institutions there has been a growing body of academic work relating to the challenges experienced by minority ethnic staff at higher levels and the difficulties experienced once at these levels (Arday, 2018; Rollock, 2019). A report for the LFHE found that many black and minority ethnic academics felt that they met the criteria for promotion and progression yet had been overlooked in favour of others with equal or lesser qualifications (Bhopal and Brown, 2016). These experiences were also reflected from respondents in the call for evidence to the McGregor-Smith review (2017) indicating that this is not a problem unique to UK higher education. All the women interviewed for the LFHE research stated that their ethnicity played a part in the way they were perceived in their role as a senior leader, whereas men were less likely to think so. In addition to this, many of those interviewed felt that there needed to be more formal, transparent processes in place to support black and minority ethnic academics, that included formal mentoring and an understanding of the key issues affecting this academic cohort in seeking leadership roles (Bhopal and Brown, 2016).

In support of previous research that found that black and minority ethnic staff were more likely to receive lower pay then their white colleagues (ECU, 2009), Table 5 reveals that over a decade later lower proportions of minority ethnic staff are employed within the highest salary range (>£50,000) however, this is particularly pronounced for black academic staff than for staff from any other ethnic background. This outcome was especially marked for black UK national academic staff working part-time, where only 12.3 percent were within this salary range.

Table 5: UK academic staff by mode, salary range and ethnic group (Advance HE, 2020)¹³

			White			BAME total			Black		
			No.	↓%	→%	No.	↓%	→%	No.	↓%	→%
	Full-time										
	<£30k	Under £30,000	2,410	3.1	88.6	310	3.1	11.4	45	3.0	1.6
	£30-50k	£30,000-£50,000	36,010	46.8	87.7	5,070	47.9	12.3	890	63.1	2.2
	>£50k	Over £50,000	38,440	50.0	89.8	4,380	49.0	10.2	480	33.9	1.1
	All	All salary ranges	76,860	100.0	88.7	9,760	100.0	11.3	1,410	100.0	1.6
<u>s</u>	Part-time										
Sug	<£30k	Under £30,000	4,690	10.1	89.2	565	10.5	10.8	75	9.6	1.5
aţį	£30-50k	£30,000-£50,000	30,085	64.7	90.9	3,020	64.8	9.1	620	78.1	1.9
5	>£50k	Over £50,000	11,700	25.2	91.8	1,040	24.7	8.2	100	12.3	0.8
5	All	All salary ranges	46,475	100.0	90.9	4,630	100.0	9.1	795	100.0	1.6
	All modes										
	<£30k	Under £30,000	7,100	5.8	89.0	880	5.8	11.0	120	5.4	1.5
	£30-50k	£30,000-£50,000	66,095	53.6	89.1	8,090	54.1	10.9	1,510	68.4	2.0
	>£50k	Over £50,000	50,140	40.7	90.2	5,420	40.1	9.8	575	26.2	1.0
	All	All salary ranges	123,335	100.0	89.6	14,390	100.0	10.4	2,205	100.0	1.6

For staff employed on professional and support contracts, there was a marginally higher proportion of black and minority ethnic staff in the salary range between £30,000 to £50,000 than white staff, and once again, the lowest proportion of black professional and support staff in the highest salary range than staff from any other ethnic group. Similar to the outcomes of part-time academic staff, a significantly lower proportion (1.2 percent) of black part-time professional and support staff were within the highest salary range compared to their white colleagues.

Table 6: UK professional and support staff by mode, salary range and ethnic group (Advance HE, 2020)¹⁴

			White			BAME total			Black		
			No.	↓%	→%	No.	↓%	→%	No.	↓%	→%
	Full-time										
	<£30k	Under £30,000	54,975	47.3	89.5	6,465	47.1	10.5	1,735	48.2	2.8
	£30-50k	£30,000-£50,000	47,085	40.5	88.2	6,295	41.3	11.8	1,590	44.2	3.0
	>£50k	Over £50,000	14,195	12.2	91.8	1,270	11.6	8.2	275	7.7	1.8
	All	All salary ranges	116,250	100.0	89.2	14,030	100.0	10.8	3,600	100.0	2.8
<u>s</u>	Part-time										
nationa	<£30k	Under £30,000	36,305	69.0	90.2	3,965	70.2	9.8	1,145	81.7	2.8
aţi	£30-50k	£30,000-£50,000	13,905	26.4	92.5	1,125	25.6	7.5	240	17.1	1.6
	>£50k	Over £50,000	2,375	4.5	94.7	135	4.2	5.3	15	1.2	0.7
ž	All	All salary ranges	52,590	100.0	91.0	5,225	100.0	9.0	1,400	100.0	2.4
	All modes	3									
	<£30k	Under £30,000	91,280	54.1	89.7	10,430	54.1	10.3	2,880	57.6	2.8
	£30-50k	£30,000-£50,000	60,990	36.1	89.2	7,420	36.6	10.8	1,830	36.6	2.7
	>£50k	Over £50,000	16,570	9.8	92.2	1,405	9.3	7.8	290	5.8	1.6
	All	All salary ranges	168,840	100.0	89.8	19,255	100.0	10.2	5,000	100.0	2.7

The median salaries for academic and professional and support staff show that there are differential outcomes for different ethnic groups compared to white staff. For both staff cohorts, Table 7 demonstrates that on average black staff are paid less than their peers (£33,199), however it is black professional and support staff that are paid the least median salary when compared to their peers.

 $^{^{13}}$ Taken from Equality in Higher Education: Staff Statistical Report 2020, Advance HE

¹⁴ Ibid

Table 7: Median salaries of UK staff by activity and ethnic group (Advance HE, 2020¹⁵)

		staff	Professional and support staff	All staff
		£	£	£
W	White	47,263	28,660	36,261
	ME BAME total	45,892	28,705	36,261
	Asian	45,902	29,323	35,459
В	Black	43,267	27,830	33,199
	Chinese	50,132	31,752	42,701
M	Mixed	43,267	27,863	35,211
0	Other	47,263	30,117	40,322
All	All UK staff	46,824	28,660	36,261

In relation to whether there is a pay differential between white and black and minority ethnic staff in UK higher education Table 8 shows that the gap has reduced to just 2 percent in 2018/19. However, this overall position is influenced by the negative pay gaps evident for Scotland, Wales, and Northern Ireland, which may be as a result of some ethnic minority staff employed in institutions within those jurisdictions being employed at higher levels and commanding higher salaries. The ethnicity pay gap is largest (14.9 percent) in London institutions despite there being a larger concentration of minority ethnic staff in the capital than anywhere else in the UK. Unfortunately, there is no data available to ascertain pay differentials according to specific ethnic groups. Comparing mean and median pay gaps, it is professional and support staff that suffer a larger pay differential than their academic colleagues and this has further influenced the overall UK pay gap figure. It is possible that the structured academic grades operated across the higher education sector may influence the pay gaps evident for academic staff.

-

 $^{^{15}}$ Taken from Equality in Higher Education: Staff Statistical Report 2020, Advance HE

Table 8: Median/mean salary and pay gap for UK staff by country of institution, activity and ethnicity $(Advance\ HE,\ 2020)^{16}$

			Median salar	Y	Mean salary				
		White	BAME	Pay Gap (%)	White	BAME	Pay Gap (%)		
		£	£	%	£	£	%		
Academ	ic staff								
ENG	England	47,263	45,892	2.9	50,829	49,726	2.2		
LON	London	48,677	45,504	6.5	54,793	50,671	7.5		
EWL	England (excluding London)	47,139	45,892	2.6	49,830	49,225	1.2		
NIRE	Northern Ireland	50,132	50,132	0.0	52,458	54,731	-4.3		
SCOT	Scotland	48,677	48,677	0.0	51,632	50,427	2.3		
WAL	Wales	42,036	48,677	-15.8	47,839	51,964	-8.6		
UK	UK	47,263	45,892	2.9	50,756	49,867	1.8		
Professi	onal and support staff								
ENG	England	29,335	28,850	1.7	32,442	31,313	3.5		
LON	London	37,412	33,359	10.8	41,214	35,437	14.0		
EWL	England (excluding London)	27,512	25,482	7.4	31,000	28,265	8.8		
NIRE	Northern Ireland	26,243	22,017	16.1	30,269	26,931	11.0		
SCOT	Scotland	27,025	26,243	2.9	30,772	29,094	5.5		
WAL	Wales	27,025	24,771	8.3	30,175	28,116	6.8		
UK	UK	28,660	28,705	-0.2	32,081	31,185	2.8		
All staff									
ENG	England	37,181	36,009	3.2	40,308	38,967	3.3		
LON	London	43,327	37,412	13.7	48,220	41,033	14.9		
EWL	England (excluding London)	35,211	34,189	2.9	38,722	37,632	2.8		
NIRE	Northern Ireland	34,866	43,267	-24.1	38,542	45,545	-18.2		
SCOT	Scotland	34,189	38,460	-12.5	38,721	41,345	-6.8		
WAL	Wales	35,211	39,609	-12.5	37,837	42,666	-12.8		
UK	UK	36,261	36,261	0.0	39,964	39,174	2.0		

Although it has not been possible to provide more detailed analysis according to specific ethnic groups and their mean and median pay gaps, it is possible to provide data concerning the median salaries for staff across academic and professional and support roles by ethnic groups. It is clear from Table 6 that there is a significant median gap between white and black staff, and in particular academic staff, from the information provided.

Table 9: Median/mean salary and pay gap for staff by gender and ethnicity (Advance HE, 2020)¹⁷

		N	ledian sala	Mean salary			
		Female	Male	Gender	Female	Male	Gender
				Pay Gap			Pay Gap
				(→%)			(→%)
		£	£	%	£	£	%
BAME	BAME total	34,189	38,434	11.0	36,002	41,559	13.4
W	White	34,394	39,609	13.2	37,283	44,687	16.6
BAME/white Pay Gap (↓%)		0.6	3.0	N/A	3.4	7.0	N/A

In the absence of pay gap information across different ethnic groups, Table 9 provides information from an intersectional perspective and shows that there is a small median salary gap (0.6 percent) between white and ethnic minority women. However, there is a 3 percent

 $^{^{16}}$ Taken from Equality in Higher Education: Staff Statistical Report 2020, Advance HE

¹⁷ Ibid

gap between white and ethnic minority men. In terms of the mean salary, there is a gap between white and ethnic minority women of 3.4 percent, and there is a larger gap reported between white and ethnic minority men of 7 percent. This indicates that there is an ethnic penalty related to salary outcomes and this is compounded by gender, particularly as overall there is a 13.4 percent mean gender pay gap between ethnic minority men and women and an even larger mean gender pay gap of 19.5 percent between ethnic minority women and white men. Heath and Cheung (2006) define ethnic penalty as the sources of disadvantage that leads an ethnic group to fare less well in the labour market than the similarly qualified dominant ethnic group. To support this, a UCEA report (2019) stated that ethnic minority staff in UK higher education earn less than their white counterparts and that there are significant differences between broad ethnic minority categories with black staff systematically earning less than all other groups (UCEA, 2019).

Thus far, these data present a profile of under-representation of minority ethnic staff across the UK higher education sector and especially if one considers that the profile of the UK ethnic minority population would have expanded since the UK census in 2011. This is further exacerbated by the proportion of those staff employed on fixed-term contracts and receiving lower pay overall than other colleagues. Despite black staff representing a significant cohort among minority ethnic staff, black staff are less likely to hold professorial positions than any other ethnic group and are less likely to be positioned within the highest salary range (>£50,000 per annum) than any other ethnic group.

Table 10: UK academic staff by leaving status and ethnicity (Advance HE, 2020)¹⁸

		BAME			White			All leave	rs
		No.	↓%	→%	No.	↓%	→%	No.	↓%
UK employi	ment								
HEI	UK other HEI	300	27.0	11.7	2,280	25.4	88.3	2,580	25.6
OEI	UK other education institution	40	3.7	13.0	270	3.0	87.0	310	3.1
RES	UK research institute	25	2.3	13.6	165	1.8	86.4	190	1.9
STU	UK students	75	6.9	16.6	390	4.3	83.4	465	4.6
MED	UK NHS/medical or dental practice	155	13.8	28.6	385	4.3	71.4	540	5.3
PUB	UK public sector	35	3.1	8.2	390	4.3	91.8	425	4.2
PRIV	UK private sector	80	7.2	10.8	660	7.4	89.2	740	7.4
SELF	UK self-employed	40	3.5	8.3	435	4.8	91.7	470	4.7
VOL	UK voluntary sector	5	0.4	9.3	40	0.4	90.7	45	0.4
All	All UK employment	760	68.0	13.1	5,015	55.9	86.9	5,770	57.2
Non-UK em	ployment								
HEI	Non-UK HEI	25	2.1	12.4	160	1.8	87.6	185	1.8
OEI	Non-UK other education institution	5	0.4	14.7	30	0.3	85.3	35	0.3
RES	Non-UK research institute	10	0.7	14.5	45	0.5	85.5	55	0.5
STU	Non-UK student	0	0.1		5	0.1		5	0.1
MED	Non-UK health service	0	0.0		0	0.0		0	0.0
PUB	Non-UK public sector	5	0.4		10	0.1		15	0.1
PRIV	Non-UK private sector	5	0.4	12.1	30	0.3	87.9	35	0.3
SELF	Non-UK self-employed	0	0.0		5	0.0		5	0.0
VOL	Non-UK voluntary sector	0	0.0		5	0.1		5	0.0
All	All non-UK employment	45	4.1	13.7	290	3.2	86.3	335	3.3
No longer i	n employment								
NREG	Not in regular employment	245	21.9	10.6	2,065	23.0	89.4	2,310	22.9
RETI	Retired	60	5.4	3.9	1,490	16.6	96.1	1,550	15.4
DECE	Deceased	5	0.6	6.0	110	1.2	94.0	115	1.2
All	All no longer in employment	310	27.9	7.8	3,665	40.9	92.2	3,975	39.4
All Destinat	ions								
All	All leavers	1,115	100.0	11.1	8,965	100.0	88.9	10,080	100.0

In terms of the data related to staff retention, the data can only indicate a snapshot at a fixed point in time and is unable to provide the full picture of the factors that contribute to the sector workforce profile. What is clear is that the growth of black and minority ethnic staff over more than a decade is insufficient against the rate of attrition shown in Table 10. This is especially marked among academic members of staff, showing that a higher proportion of black and minority ethnic academic staff left UK higher education than their white colleagues, including to go to another non-UK higher education institution. Research undertaken by ECU suggested that black and minority ethnic academics were likely to leave UK higher education due to their research being in black history or ethnic diversity and did not feel that this research area was particularly valued in UK higher education institutions (ECU, 2015).

5.7 Addressing race inequalities in UK higher education

The current dichotomies between the moral, legal and business case for equality and diversity and those present between race and ethnicity are not helpful in researching racial inequalities

134

¹⁸ Taken from Equality in Higher Education: Staff Statistical Report 2020, Advance HE

for UK higher education. The frame in which they are set limits the capacity for higher education institutions, and those charged with effecting change within them, to become fully cognisant for why workplace racial inequalities persist. The concept of equal opportunities provides the prospect of focusing on race solely within a broader equalities agenda, however at a macro and meso level it relies on numerous facilitators to advance outcomes for black staff. Namely, these are legislation; policy; and senior management commitment to maintain a sustained effort in reducing disadvantages in the workplace. In practice, institutional priorities are not always aligned to the needs of individuals and Sian (2017) argues that this can result in a lack of engagement with race equality or in addressing racism, particularly where it affects the workforce.

The Equality Act 2010 in isolation offers a framework for employers to ensure that they do not discriminate or that they take steps to eliminate discrimination, harassment, and victimisation. The legislation also provides a remedy for employees where they believe this has happened. However, legislation does not eliminate workplace racial inequalities in UK higher education institutions, continuing to be evidenced since the legislation came into force (ECU, 2011; Rollock, 2011; Pilkington, 2011, 2013; Philips, 2012; University and College Union, 2016, 2017; EHRC, 2019). The legislation is unable to address dual discrimination, where there might be intersecting characteristics, e.g., race and gender. Within the context of UK higher education, institutions are obliged to meet the requirements of the PSED and although not addressed in this research, further exploration is required to gauge institutional interpretation of the PSED and to understand the extent to which institutions are meeting their responsibilities to race. This is particularly relevant considering that the EHRC¹⁹ assessed the publication of information and equality objectives shortly after the due dates for initial publication in 2012. The reports showed that university performance was mixed and was either average or slightly below average, suggesting that there was considerable scope for improvement, particularly in terms of publishing more information on staff and potential service users, assessing impact and recognising and addressing gaps where they are apparent (EHRC, 2012; 2013). Monitoring for compliance should be supported by the sector agencies

¹⁹ Equality and Human Rights Commission

to ensure good practice and to address the issues raised through these assessments across institutions in future.

There are various issues relating to policy implementation (Hoque and Noon, 2004; Pilkington, 2011), particularly where there is inconsistent awareness and knowledge of relevant policy and/or the policy is not implemented fully or effectively. Conversely, there can be detrimental and unfair impact from the hyper-formalisation of procedures, which can result in what Noon et al (2013) described as 'circumvention by compliance' through the robotic, defensive, and sometimes malicious implementation, that may undermine the original purpose of the procedure and cause further inequality. Added to the difficulties of the practical application of legislative frameworks in organisations, the approaches used through diversity management concentrate primarily on diversifying the workplace and assumes principles of meritocracy, where a person's background is believed not to preclude success. For this reason, Syed and Özbilgin (2009) argue that diversity management is unable to achieve equitable outcomes for diverse employees and unless a comprehensive approach is taken to reform multi-level processes and practices, employment opportunities and outcomes will continue to persist.

In its attempts to address inequalities across the UK higher education sector, including how some of its practices might be contributing to these outcomes, there has been an amplified implementation of unconscious bias training across the sector. This intervention has been increasingly endorsed across other sectors and industries and featured as a recommended tool to tackle race inequalities (McGregor-Smith, 2017). Unconscious bias training is being delivered across the higher education sector together with other learning and development interventions that address equality and diversity with varying degrees of success (Swan, 2009). There is no evidence to demonstrate whether senior leaders are included in this activity across institutions. Noon (2017; 2018) has critiqued the 'latest fashion' in self-awareness stating these measures would likely have little impact on the pernicious nature of everyday racism in the workplace.

Noon (2017; 2018) also pointed out that there is no guarantee that knowing about bias will eradicate it and the no-blame approach used within unconscious bias training is reliant on an individual firstly acknowledging their bias, and secondly, being willing to modify their behaviour. Within institutions themselves there are various methods of delivery for equality

and diversity learning interventions that include both face-to-face and an increasing reliance on e-learning, which is more cost effective. There is no evidence to support or contradict the role that equality and diversity e-learning interventions contribute to challenging individual attitudes and behaviours. However, anecdotally, after years of these interventions being in place in most, if not all, UK institutions there is a likelihood that Noon's (2017; 2108) critique of unconscious bias training can be applied to broader equality and diversity training. In a recent move by the Civil Service, unconscious bias training has been phased out from its programme of equality and diversity learning interventions because of the lack of evidence that it can make a difference to organisational equality outcomes (Coughlan for BBC, 2020; Cabinet Office, 2020; Personnel Today, 2020; CRED, 2021). More recently, the Commission on Race and Ethnic Disparities report recommended that organisations discontinue the delivery of unconscious bias training (CRED, 2021). However, further research is required to understand the extent to which equality and diversity learning interventions are being offered across UK institutions, whether those interventions are mandatory, and to understand the level of take up. Certainly, an evaluation is required to ascertain whether current learning interventions are having the required effect on reducing discrimination or disadvantage towards groups of people based on race.

Following on from the discussion in Chapter Four, academics have used, and continue to explore the use of Critical Race Theory (CRT) as a more radical approach to address anti-racism within the UK context. CRT has been used to scrutinise contemporary institutional life (Gillborn, 2005; 2006, Rollock, 2011). Historically, CRT's two main principles are firstly aimed to address 'white supremacy' to describe the oppression based on race rather than the concept of racism; and the second that 'race' not social class was the primary contradiction in society (Cole, 2009). These principles have advanced and expanded over time and now include the notion that racism is normal, not aberrant and is embedded within the fabric of society; that racism reinforces and advances the interests of whites and helps to maintain the status quo (interest convergence); that race is socially constructed; and that although CRT is primarily focused on race, the intersections with multiple aspects of identity are important in understanding the extent of inequalities (Ladson-Billings, 1998; Gillborn, 2006; Delgado and Stefancic, 2007, Rollock and Gillborn, 2011). CRT employs storytelling and counter-storytelling

by integrating experiential knowledge from people of colour to expose the way racism is omnipresent in society.

Akin to what is recognised as a business case for diversity, Bell (1980) developed the concept of *interest convergence* to suggest that white people will support racial justice only when they understand and see that it is in their own interest to do so, when there is a "convergence" between the interests of white people *and* to progress racial justice. Bell asserted that the U.S. Supreme Court ended the longstanding policy in 1954 of "separate but equal" in *Brown v Board of Education* since it presented to the world that they supported civil and human rights. From this, it is possible to assert that diversity management fits the concept introduced by Bell as organisations voluntarily pursue increasing organisational ethnic diversity only when business objectives coincide.

The approach used through CRT is useful in unpicking the complexities of racial inequalities using the premise that racism exists across all aspects of society and that it is commonplace and most often hidden in plain sight. The use of storytelling offers an instrument to expose the perspectives of lived experience, to highlight the effects of everyday situations that are legitimised by dominant groups. However, as both Ladson-Billings (1998) and Gillborn (2006) have expressed, a more radical perspective and even more radical solutions are required to address racism in education. These radical solutions must involve addressing the white hegemonic systems that sustain and extend workplace racial inequalities and will necessitate a renewed perspective that is able to keep pace with the evolving nature of racism in UK society and within the UK higher education sector.

The renewed calls for more radical action lack a clear perspective of how the issue of 'white supremacy' or white privilege can be addressed in our higher education institutions with a specific focus on staff. In 2019, The Guardian investigated their evidence of widespread racism in UK universities (Batty, 2019) and suggested that there was an 'absolute resistance' to tackling race discrimination. Bhopal (2018) suggests that the academy works to protect its own image of white privilege by excluding minority ethnic staff from positions of power and that whiteness within the academy operates in the form of white academics having access to a 'network of knowns', which allows for access strategies to be established to progress. Anecdotally, the same can be assumed for the position of professional and technical staff in institutions, although there is no research focusing on this specific staff cohort in UK

universities. The solution Bhopal (2018) proposes that institutions outline how they address inequalities and for a specific recognition of the value of diversity, is insufficient if white privilege within institutions is not acknowledged as a contributing factor to the differential outcomes for black staff in UK higher education. Acker (2006) asserted that people in dominant groups generally see inequality existing somewhere else, and not where they are, and this is exacerbated by the entanglement of privilege with gendered and racialised identity that makes privilege difficult to unsettle.

It is the denial and/or avoidance of white privilege that maintain the order of things within higher education institutions and white staff are unlikely to be willing to relinquish their positions of power or privilege. In support of this, Bhopal (2018) believes that as long as white identity and white privilege are not threatened, white groups are supportive of diversity and inclusion. In other words, universities can promote themselves as being fair if their white privilege remains intact and unthreatened. Additionally, class is likely to be an issue within the institutional context, and which tends to be hidden by talk of management, leadership, or supervision among managers (Acker, 2006). Recently, UUK proposed a series of recommendations for UK universities in response to the EHRC's inquiry into racial harassment that included increasing staff and student understanding of racism, racial harassment, microaggressions and white privilege through training developed from an anti-racist perspective (UUK, 2020). It would be interesting to explore whether the predominantly white senior leadership teams within institutions are engaged in discussions about white privilege and the role they play in maintaining the status quo. UUK have stated they will carry out a review of the impact of their guidance to the sector and identify areas for further improvement by summer 2022.

5.8 Conclusion

The concept of equality has existed in society for some time, and it was not until the latter part of the 20th century that it became to mean what we understand it to be today. Its origins can be traced back to the US Civil Rights movement that was able to influence societal change and bring about remedies through affirmative action. It was these initiatives, with aims to reduce disadvantage on the grounds of race, which would bring about change throughout the 1960s and 1970s. This change resulted in a backlash throughout the 1980s in a move towards

a more neo-liberal political environment that sought to reduce bureaucracy and lessen regulation. This ideology crossed the Atlantic to the UK and brought about a move in organisations from an equal opportunities approach, with its roots in compliance with anti-discrimination legislation, to a voluntaristic diversity management approach.

Diversity management was embraced in the UK since it brought a greater focus on the business benefits of managing people as individuals and this is a concept that senior management could relate to and/or is easier to realise greater organisational buy-in. To some extent this was successful, however critics have raised concerns that a diversity management approach alone is unable to achieve equitable employment outcomes for diverse employees, making it a plausible suggestion that organisations approach the business benefits and employee outcomes in a hybrid model. The research suggests that neither one approach nor the other is effective in contemporary organisations.

As discussed in this chapter, taking a dual approach, and using the principles of equality together with diversity management, theoretically takes a balanced stance in creating diverse workplaces as well as giving due regard to equality of opportunity. However, this is limited to just these areas and does not ensure equality of outcome (Kaler, 2000), which could be achieved by refocusing on the role agents play within organisations at multiple levels to influence the efficacy of policy and strategy that addresses not just equality and diversity, but equality of outcome on the grounds of race. Within these existing frameworks, the emphasis is placed on either groups of people (equality) or on individuals from different groups (diversity), however from either perspective there is a lack of focus on the role an organisation plays in constructing inequalities and reproducing disadvantage.

To some degree the UK public sector has adopted equality *and* diversity approaches in its efforts to address disadvantage as well as diversify its organisations. The UK public sector has not traditionally adopted the individualistic diversity management approach seen more commonly within the private sector and this may be due primarily to the increased regulatory environment within the sector. Public sector organisations are required to evidence how they pay due regard to the way they tackle disadvantage, advance equality of opportunity and foster good relations. UK higher education institutions must equally comply with these requirements, yet workplace inequalities continue to persist. There have been a range of activities to focus institutions' attention on equality and diversity in the last decade,

specifically to tackle gender and race equality, as well as maintaining attention across the broader spectrum of organisational equality and diversity.

These activities cannot take place without internal actors, and it is human resources and equality and diversity practitioners, as well as other external change agents that support the sector in their approach to equality, diversity, and inclusion. The provision of equality and diversity practitioners is uneven across UK higher education institutions, and this is further exacerbated by practitioners' need to navigate a challenging internal environment, where there is individual and organisational resistance. Equality and diversity practitioners may lack capital, agency, and capability to effect change in their institutions particularly if they lack senior level support and this may be even more challenging if the practitioner belongs to a marginalised group themselves as they operate within the structures that create and perpetuate their disadvantage.

In recent years there has been a renewed voluntaristic focus on race equality, and this has been further encouraged by the introduction of the REC, however the initiative lacks organisational incentive and take up is low. Statistics show that there is clearly a need to focus specifically on race with continued evidence of low workforce representation of faculty of colour. This is particularly evident at senior levels across academic and professional and support staff alike.

There are deeper inequalities beyond workforce representation, where further research might offer an insight to the recruitment practices of higher education institutions to understand the recruitment success rates according to the ethnic background of applicants. Existing black and minority ethnic staff are more likely to be employed on fixed-term contracts, receive lower levels of pay and are more likely to leave their institution than their white colleagues. The concern here is that the rate of attrition surpasses the level of growth, necessitating urgent action in the UK higher education sector to address this. The following chapter will explore the extent to which the perception of racism might be a contributing factor to the outcomes of black staff in UK higher education.

The literature continues to report that there are ongoing persistent race inequities across the sector and although these can be seen for ethnic minority staff, it is particularly marked among black staff. These studies expose that the approaches taken within institutions to

address equality and diversity for minority ethnic groups are ineffective. As such, the discourse on workplace race inequities requires reformation and refocus, so that in practice CRT is placed at the heart of institutional strategy and policy making. Considering organisational structure and agency would enable institutions to consider how power informs the structures within the organisation that influence agents in the production and reproduction of structures within a workplace setting. The concepts of structure, agency, and forms of power (explored in Chapter Four) are applied to the interpretation of the empirical data in Chapters Six and Seven.

6 The opportunities and challenges of fitting in

6.1 Introduction

The previous chapter considered the evolution of equality and diversity practice and how this has become a feature of contemporary organisational strategy and practice over the past fifty years in the UK. The field of equality and diversity is cyclical in nature, in that it was primarily constructed to tackle inequalities for specific groups and over time has broadened to consider the needs and outcomes of multiple groups, and then returned to focusing on particular groups once again, with race and race equality remaining a leading concern throughout this period.

This research has offered multiple complex topics for investigation, which require the findings to be explored over two chapters. The findings have been analysed by drawing together the previously published literature set out in earlier chapters alongside the quantified workplace context outlined in Chapter Five, together with the emerging themes from participant interviews that are relevant for this chapter and are explained further below. The analysis of participant accounts will also take into consideration the macro, meso, and micro-level aspects depicted in Layder's (1993) research map whilst reflecting on the horizontal spectrum between history and power drawing on the concepts of Bourdieu (1977; 1984; 1990; 1998) to gain a better understanding of the complexity of institutional workplace relations with a view to identifying solutions for sectoral leaders and practitioners in tackling workplace race inequalities in the future.

The aim of this research study is to explore the lived experiences of staff in UK higher education concerning the topic of race, with the objective to understand how these experiences create differential outcomes for black staff in the institutional workplace. To meet this aim, this chapter will consider how the social space in which black staff operate affects their sense of identity and belonging, the coping strategies employed to deal with this, and how the use of racialised language and terminology impacts on their working environment. The themes discussed in this chapter identified perceptions and experiences that addressed notions of race and ethnicity and terminology used in institutions that can either empower or disenfranchise staff. This chapter will also explore the views and beliefs

around identity and how these inform and influence an overall sense of belonging to their peer group and immediate colleagues and whether this sense of belonging is different to their institution.

In the first instance and to set the scene around the topic of equality and diversity, this chapter will begin by providing an account of the overall perception from participants in terms of the importance of diversity in the UK higher education sector and whether this was a consideration at the point of entry into higher education as an employee. This will be followed by an analysis of participants' understanding of and explanations around race and ethnicity, feelings towards terminology used to describe matters of race, the way institutions monitor ethnic identity across their workforces and will explore whether these organisational approaches have an impact upon a sense of institutional belonging on the participants.

Language and terminology about race and references made about racial identities have become an increasingly debated subject in recent times (DiAngelo, 2018; Eddo-Lodge, 2018) and this has become especially topical during the Black Lives Matter campaigning during 2020 (BBC, 2020). This has rightly caused outrage among black communities in the US and UK and in response, emotions have run high across different communities (Wall for The Observer, 2020; Edwards for BBC Wales, 2020). In parallel, these recent events have raised the profile once more of societal racism and feelings of guilt, shame and ignorance have become exposed through social media channels highlighting the polarity of views from rage on one extreme to 'all lives matter' at the opposite end of the spectrum. These reactions confirm the assertions made around conversations that can prematurely break down because of opposing views or a lack of understanding about the issues, which may cause anxiety, heighten tension and defensiveness and cause discomfort (Campbell, 2016; Lingayah, Khan and McIntosh, 2018; DiAngelo, 2018).

In discussing the notions of race and ethnicity this chapter will explore the perceptions of participants in terms of racial hierarchies, which might be visible within institutions and how this might manifest itself in the academy. The next section will provide a view of the landscape in terms of how participants feel about the concept of diversity and whether this was an important factor when initially considering entry to the UK higher education workforce and whether their feelings about this have changed during their employment in the sector.

6.2 Perceptions on workplace diversity

This section begins by setting the scene at a macro level within UK higher education and the sector's relationship with conceptualising and realising equality and diversity. In terms of the sector's journey in developing and implementing diversity-related policy and practice it is not unique in its challenges to prioritise policies and initiatives in the attempt to keep up with evolving practices. The organisational characteristics of higher education institutions can make this a challenging environment in which to successfully implement policies and strategies to address race equality (ECU, 2011). Despite these challenges, the UK public sector does promote equality and diversity in a proactive way and one which features representing the society it serves as a driver in promoting this agenda.

Anecdotally, people might believe the public sector to be very diverse and in pockets of the public sector this is indeed the case (Cabinet Office, 2019). The public sector may be ethnically diverse in pockets, but not within the higher levels of those institutions that make up this sector (ibid). It is often assumed that employees of ethnic minority heritage will, because of their background, have an expectation for, and an interest in workforce diversity. To explore these further, ethnic minority participants were asked whether this was an important consideration before applying to work in their institution. Being able to understand the motivations of institutional staff prior to entry to the UK higher education workforce was important to establish whether the concepts of equality and diversity played an important role in their decision making, their perception of whether they would have a fruitful career in this sector and whether as a person of colour they would feel valued as part of a diverse workforce.

The responses were initially surprising and quite unexpected, mainly due to my own expectation that diversity *would* play an important part of imagining oneself within institutional life, particularly if an employee belonged to an under-represented or marginalised group. I made this assumption because I imagined that a person that belonged to a marginalised group would feel it was important to see themselves within a diverse organisation, and this was purely based on the importance that *I* place on this organisational attribute.

Some accounts from participants that had previously studied in UK higher education acknowledged that from their own student experience in the UK, there lacked academic staff diversity, and that based on this they were able to manage their own expectations around institutional diversity. For most participants this was not the case, and regardless of role, e.g., academic or professional and support, they stated that organisational diversity *did not* play a part in the decision-making process for joining their current and/or initial institution. In fact, there was an expectation that the institution *would be* diverse and therefore, not an area to dwell upon further. An explanation of this was well captured in the following comment:

I don't think it was in terms of whether or not I would feel comfortable...I was young and foolish, looking for a job basically, so...no, I didn't think about it at all.

Participant 5, mixed-race female senior professional staff, Post-1992

When asked to elaborate on whether she thought about diversity presently, the participant provided a different answer with justification why she felt this way:

...I do think about it more, yes. I think it worries me more because of how this institution seems to replicate itself in pockets on all sorts of levels...and I do observe some staff, I sense sometimes there is strain and stress to do with the fact that they don't fit in. I know that myself actually. I don't quite fit in to...the norms, perceived as the norm in that particular department.

Participant 5, mixed-race female senior professional staff, Post-1992

Approximately 95 percent of participants had stated that they did not consider the diversity of the institution they had joined prior to taking up post. Like Participant 5, several other participants had mentioned that finding a job was more important than limiting oneself in the job market and it was only once in post, it was the lack of diversity in higher education that acted as the catalyst for considering, and for some becoming involved in, equality and diversity in the future. For some, being part of the minority meant that diversity was constantly at the forefront of their thinking in terms of how best to navigate the system. One participant stated that she had not considered diversity during her initial job search however, would do in future, except not in terms of ethnic diversity:

I would say that if a department was all white...that wouldn't put me off at all because I've been in lots of situations and that's not an issue. If a department was all male, then that would be an

issue. I wouldn't not accept a job if I applied for it, I think I would perhaps feel...possibly more inhibited.

Participant 4, mixed-race female academic, Russell Group

In some ways, this comment can be interpreted in quite a positive light that for this participant the ethnic composition of her department would not be the main concern, but that gender would be. This may reflect the way that Participant 4 identifies and whether she sees herself in terms of gender first and ethnicity as a secondary consideration or if she perceives that her gender inhibits her more within the academic space than her ethnic background. Participant 4's comments might identify issues relating to women navigating a male-dominated social space (Bourdieu, 1990; 2001a) and how this impacts on what Bourdieu described as the division between the sexes and more specifically, the distribution of labour according to these perceptions of a woman's role in the workplace. Unfortunately, this aspect of identity was not explored during this interview and therefore it is not possible to explore this aspect of intersectional identity further in this instance.

It appears that Participant 4 has considered her identity in terms of her working life and that she is cognisant of the structures (Bourdieu, 1977) within which she operates and is utilising her agency (Giddens, 1984) in her decision-making in terms of considering the gender profile of a prospective department. However, this participant did state that she would not decline a position if it were to arise even if she were the only woman in the department and would clearly need to consider how she would navigate that space.

Encountering this matter early in the interview process did inform the way subsequent interviews were carried out to elicit a more in-depth account of the identity hierarchies held by participants. Overall, Participant 4's view was not shared across participants from a minority ethnic background, and it was especially not reflected among those with a black/black British background. To elaborate on the point of identity further, one participant shared a specific experience where she was asked to comment on an issue from a black woman's perspective. Her account demonstrated a powerful point where she felt it was impossible to be acknowledged in the context of her own identity in the academic space as a black woman:

...I just need to feel as if it's possible...because...it's very possible to be a white woman. But to be a woman of colour within the academic space within higher education...that was the first time that someone had asked me that expressly and I just burst into tears.

Participant 27, black female academic, Russell Group

The account from Participant 27 demonstrates that UK institutions should provide marginalised groups with a platform to be heard. Not just in terms of the experiences of all people of colour within the academic space, but as women of colour navigating a white hegemonic, and in some disciplines and academic hierarchy, male-dominated social space. Here is where CRT and the use of storytelling to reveal and explore the lived experiences of people working in UK higher education institutions is a critical mechanism for institutions wishing to advance race equality. In this case, being asked for her opinion as a black woman was extremely powerful.

Coincidentally, a much higher proportion of female participants (80 percent) had self-selected to take part in this study. This was unexpected and where possible aspects related to the intersections between race and gender were explored throughout the course of investigation. The study has not become occupied with the complexities of multiple identities and how the outcomes of participants may have been affected by the intersection of race *and* gender or any other characteristics and is acknowledged as a significant area for further exploration. There were several comments relating to this aspect of workplace and life experience and some of these will be addressed in Chapter Seven surrounding perceptions of racism in UK higher education institutions.

There were a few participants that had commented on their experiences and perceptions of diversity as students, as well as working within especially diverse student populations within UK higher education. Bhopal (2018) suggests that diversifying the student population has largely been a successful enterprise by universities over the years, yet she reminds us that the outcomes of those minority ethnic students may not be as successful as their white counterparts, the following comment captured these sentiments particularly well:

I was somewhat prepared I suppose because having done undergraduate studies I noticed that none of the staff teaching me, bar one, were of the same ethnicity as myself, so I went into the same environment knowing that, you know, there's some barriers that you're going to have to

cross here and knowing that I was going to be kind of isolated in my experience because there weren't very many other people like me.

Participant 16, black female academic, Post-1992

Like Participant 4 earlier, Participant 16 articulated how she was able to use her acquired institutional habitus and 'feel for the game' (Bourdieu, 1998) albeit from a student perspective, to inform her expectations of life as an academic within her institution. Participant 16 used her agency to prepare her for the challenges of isolation that her experiences as a student has highlighted. The conversation with this participant raised the dichotomy that some institutions have become adept at attracting and creating ethnically diverse student populations, however this has not been replicated within the workforce, regardless of an institution's geographical location. This frustration was felt by many participants, particularly in terms of the ethnic diversity of staff at higher levels of institutions and what might perpetuate the persistent problem of the ethnic minority staff underrepresentation. One participant neatly suggested that:

I think traditionally it's been easy to say OK, we don't have BME people in senior positions in HE because they don't go to university, no they're not getting the firsts and 2:1's, they're not entering academia. That's just not the case anymore. They're just not being promoted...

Participant 23, white female practitioner, external agency

The seemingly obvious issue being identified by the participant is the lack of emphasis on addressing ethnic minority representation within the higher education workforce. The suggestion within the comment is the connection between outcomes of ethnic minority undergraduate students with the under-representation of ethnic minority staff in the workforce. As a result of this, there may be ensuing impact on the proportion of ethnic minority candidates applying for roles within the higher education workforce, particularly in academia. However, there may be a connection between students' observations of a lack of ethnic diversity academic staff as per Participant 16's comments. This assumes there is a dearth of ethnic minority candidates for academic roles in UK higher education institutions, which is not possible to ascertain without analysing each institution's recruitment data.

There are persistent issues around the attainment gap between white and ethnic minority students, which was raised in Chapter Two, and which may negatively impact on ethnic

minority students continuing their higher education. Participant 23 has raised some pertinent concerns. At face value, the former point (ethnic minority students under-achieving) is being proposed as an excuse to uphold the latter situation (insufficient high calibre candidates applying for jobs). With a continued organisational, and societal struggle (Lingayah, Khan and McIntosh, 2018) to address matters of race, further attention is required to understand what lies beneath that reticence to drive this agenda forward. The lack of workforce ethnic diversity is not a new problem and studies (UCU, 2016, 2017; Bhopal, 2018; Rollock, 2019) on workforce representation have focused primarily on the demographics of academic staff, however systemic under-representation of minority ethnic staff can still be observed across professional and support staff, particularly at senior levels (Advance HE, 2020).

It is possible, as a recent report on racial harassment in the UK higher education sector suggests (EHRC, 2019), that an institution's inability to start and maintain conversations about race within its workforce is a contributing factor to the problem and the next section will continue to explore some of the related aspects that may also contribute to creating barriers to advancing ethnic representation as well as tackle racial equality in our institutions.

6.3 Racialised language and terminology

To make sense of the assumptions we might make about people's understanding about race and ethnicity, and to gauge the extent to which the terms acted as a help or a hindrance within their institutions or agencies, participants were asked what they understood these terms to mean. Academic literature suggests that the terms used are problematic and exclusive (Modood et al, 1997; Smith, 2002) and part of this study has sought to understand whether the terminology used affects institutional capability to address matters of race or ethnicity within a workplace setting and whether it affects an institution's ability to advance race equality. There were also questions to be explored around the extent to which ethnic minority staff within UK higher education are affected by racialised terms and whether it impacts upon their own sense of identity within their institution. Moreover, whether this sense of belonging is further impacted by the institution's understanding, commitment, and investment towards staff of colour in dealing with matters of race. This latter aspect will be explored further in Chapter Seven when considering perceptions of racism and the ways in which institutions are tackling matters of race.

In practice, the language of race and ethnicity can be problematic, in that it has become acceptable and mainstream to use both terms interchangeably (Sandhu, 2018; Saeed et al, 2019) without entering into discussion about their histories, their evolution, or their difference. From discussions with participants, including those from a minority ethnic background, some found it difficult to distinguish between the two terms and this was particularly the case where the participant was born outside the UK. Interestingly, those born and brought up outside the UK had a different perception of their own ethnic identity, such as not having been previously marginalised because they represented the majority population in their home country as opposed to being 'othered' as a minority in the UK (Ahmed, 2012; Hirsch, 2018; Olusoga, 2017; Eddo-Lodge; 2018; Bhopal, 2018). As this participant vocalised, it was evident that she was faced with developing new strategies to navigate within a society where her visible difference could impact her career aspirations as this was not something she was familiar with in her native African country:

...diversity was not something that I thought of but...when I first came [to] this country...by this time people had told me so many stories about how the chances of me getting into higher education in this country are so slim... And they based it on nothing else but the colour of my skin. It was quite confusing because I wasn't used to it, it wasn't something I came with... And so, all this whole confusion of being in England and, all of a sudden, I have to talk race, it's really just...something new.

Participant 21, black female academic, Post-1992

This participant's experience resonates with Hirsch's (2018) comments around identity in Nigeria and how ethnic background will mean less if you are part of the ethnic majority. This account offers a different perspective to those participants that either came to the UK as children or were born in the UK and have been socialised to accept and expect being or becoming racially marginalised in UK society, particularly where personal histories have been passed down to participants through older generations of relatives. As such, the perceptions and understanding of these terms may be affected by an individual's UK history, identity, and upbringing, which could provide individuals with a greater understanding of racial inequalities, and consequently, equipped to deal with systems and structures. It could be argued that this provides black British citizens with embodied cultural capital (Bourdieu, 1998) that will create certain dispositions that will define an individual's habitus (Bourdieu, 1977) in

a positive way that would inform strategies to navigate systems and structures to gain further capital, or in a negative way, where an individual could begin to resent or mistrust the systems around them and begin to withdraw from their social space resulting in further isolation (Wilson and Jones, 2008; Solanke, 2018).

Our institutions are made up of a variety of personnel from diverse backgrounds that may never have been exposed to or been affected by matters of race and levels of understanding may have been created and/or exacerbated by a lack of awareness raising on matters of race at an institutional level. Yet further clarification is required to understand whether people of colour are affected by the terminology and whether the academic debate has impacted on institutional practice. Anecdotally, there is certainly a disconnect between academic theoretical expertise and organisational practice around race and ethnicity within institutions, e.g., equality and diversity practitioners within institutions do not tend to work together with academic subject matter experts to devise bespoke solutions for their institutions, hence this research study. This may be further impaired by the positioning of equality and diversity practitioners within institutions, as they are often employed within Human Resources departments and therefore may find themselves detached from academic research of this type unless specific effort is made to engage with academic material as well as industry or field-specific considerations. The latter material is the most likely that equality and diversity practitioners would turn to in the first instance.

Participants were asked a variety of questions related to experiences of working in UK higher education as well as opinions about issues that might shed light on the real or perceived barriers to achieving race equality in the sector. In respect of terminology used to talk about race and ethnicity, which might include policy, legislation or data collection through workforce monitoring, the findings indicate that a significant number of participants generally have no objection to the use of terms such as race and ethnicity. There was a core of participants who expressed that use of these terms are problematic in the way that they isolate and exclude individuals and groups, as well as create a false impression about those who might fall within or outside the definition of these terms. Examples of such comments include:

If you try to use a language that is going to be acceptable to people outside the box, people inside the box will feel patronised. Similarly, if you use a word or a term that's associated with people in the box, where they feel comfortable, then it alienates those on the outside.

Participant 1, black female practitioner, Russell Group

...we're stuck with the word race as part of a social landscape, as part of a political landscape but I think in terms of an experiential landscape it misses the trick. It means that..it's almost as if race can be the excuse rather than the denominator by which we delineate individual experience.

Participant 17, black female academic, Post-1992

This feedback somewhat supports the academic debates that conclude that the concepts of race and of ethnicity are problematic because of their social construction (Cole, 2009; Modood, Berthoud and Nazroo, 2002; Smith, 2002). Despite an overall acceptance that this is the case, the problem does not necessarily lie in their construction, but about the way the terms are used in practice. Generally, there was acquiescence to the terms when used to advance discussions about racialised discrimination and harassment; provide an understanding of the context; their interchangeable use dependent on the audience; and where their use was valid to address issues that were important within institutions. Whichever way the terms are used it was felt that it was impossible to appease everyone inside or outside of the sector, or even people of different ethnic backgrounds.

This view was supported by several participants. Corroborating the literature discussed in Chapter Two, participants felt that the term race was loaded with negative historical connotations and that the term was perceived to be related only to people of colour rather than to everyone:

...I do think there are problems with both of those terms, and I think really the better thing is to exclude race because of its history, and to...lean more towards a self-defined ethnicity, which we accept as problematic because it is based on experience, it's based on capital, it's based on all sorts of things but let it speak for itself and accept that in all its heterogeneous ways.

Participant 16, black female academic, Post-1992

The challenges apparent with identity and acknowledging the socio-political pressures surrounding this topic create difficulties for large and complex organisations that are predominantly white that may lack confidence with dealing with a diverse workforce and creating spaces within institutions to talk about race. As mentioned in Chapter Two, race sends people into a deep dark place where there is a real fear and discomfort when talking about race and racism (Dirlik, 2008; Rollock, 2012) and where people close ranks (Cole, 2009). This challenge was described by a participant about her experiences working with higher education institutions:

...there can be a tendency to conflate race with black...race itself is bound up in a historical situation...I think people run away from it...

Participant 19, white female practitioner, external agency

This view was further supported by another practitioner participant based within a university who shared her experience of providing safe spaces for staff to discuss matters of race:

...for some individuals, they have never been given the opportunity...to discuss it and drill down about what it is and what it means because they think the finger...of...guilt – I don't know what it is – whatever they presume will happen if you start discussing this word, I can see it in their face, in their eyes.

Participant 30, black female practitioner, Post-1992

Here, Participant 30 touched on an important aspect that generic equality and diversity 'training' does not address, perhaps due to the significant scope of the agenda that it attempts to tackle. However, there is more to be said around facilitating discussions with staff around the topic of race, and this is particularly pertinent for white staff who may not have ever been socialised to this (DiAngelo, 2018). Also, creating spaces for ethnic minority staff using the principles of CRT to share their lived experiences or to provide anonymised accounts to expose institutional life from the perspective of a person of colour can bring to life issues of race and racism to a wider audience for discussion, greater understanding and the first steps to creating a more inclusive and anti-racist movement within institutions.

In relation to the social construction of race, multiple participants expressed their deep disapproval to acronyms that are used to describe people of colour. It was acknowledged that an acronym widely used to describe minority ethnic cohorts, e.g., BME (Black and Minority

Ethnic) also exists outside the higher education sector as well as other variations of this, such as BAME (Black, Asian and Minority Ethnic). The subject of racialised terminology and the use of acronyms has been debated over the past few years with calls to discard them altogether (Sandhu, 2018; Saeed et al, 2019).

Participant 30 continued by providing her perspective on the acronym 'BME', which is widely used in the UK higher education sector:

Oh, I can't stand that. Sorry! [Laughs] What does it mean? Again, it's denying me authentication because I'm just part of a group that does not have any distinguishing features, you've just made me invisible basically. I know some people say 'well it serves a purpose' but the more you use it, the more it has less of a human content... It's not real, they're not people, it's like people, individuals use the phrase 'migrants', 'refugee', after a while there's no human connection with that, it's just a phrase and a word and then you can say what you like because you don't think you're hurting anybody because it's like a soulless entity...

Participant 30, black female practitioner, Post-1992

The tendency to use abbreviations such as these has caused a normalisation of language relating to race as shorthand referring to people from minority ethnic heritage as a homogenous group and can be interpreted in this context as symbolic power and implies symbolic violence (Bourdieu, 1998). It is terminology that has been legitimised through practice by dominant groups to refer to those that are in a dominated group. Bourdieu (1977) expresses that any language of the establishment that can command attention becomes an authorised language and therefore becomes legitimate. Consequently, power can operate through the subjective misrecognition of the terms BME or BAME. Misrecognition is key to the act of domination and Kamoche et al (2014) argued that 'cultural arbitrary' is a condition that expresses the arbitrary imposition of power by misrecognising its effects and purpose, which reproduces and legitimises the use of this terminology. Another participant openly challenged the use of the term by stating:

BME and BAME, which also makes me want to regurgitate yesterday's dinner....I just think, 'do you know what? I'm not an acronym, and I didn't give you permission to call me an acronym and that's not OK' and it bothers me that...it's an acronym that's just flung around with little consideration... I have no issue with either race or ethnicity...the problem with both of them is

that they are seen to only be pertained to people of colour. So, white people are seen not to have a racial identity...

Participant 27 black female academic, Russell Group

Paradoxically, some participants had appeared to come to terms with the use of these acronyms to describe themselves and other people of colour. During interviews many participants used the acronym 'BME', which further legitimises the use of this terminology within their institutional setting, acceding to symbolic power and symbolic violence (Bourdieu, 1990; 1998). The legitimisation through habitus in the use of such terminology has created a structure into which people of all ethnicities have become socialised. Giddens (1984) stated that rules are reproduced consistently so that they take on an objective form. In this sense, socialisation to these rules can be used as an instrument of power by the dominant group to maintain their advantage, which can be guaranteed if the dominated group are complicit in their subordination (Jones and Bradbury, 2017). Randle, et al (2014) similarly describe the practice-acceptance-internalisation-practice cycle that results in structures being reproduced and reflected through conformity with those norms and their reproduction. Similar to the point made by Participant 30 in relation to creating space to discuss race within institutions, there is no reason not to include discussions about racialised terminology and its impact on dominated groups in the social space.

Using this phraseology might be a manifestation of how people of colour have become institutionalised, or to use the phrase from Bourdieu, have become influenced by the structured structures that are structuring structures (Bourdieu, 1977; 1990). The act of accepting and utilising these terms to describe oneself and others of minority ethnic background as a homogenous group is a production and reproduction by social agents of colour and enable those who are not affected directly by the terminology to continue to use it. Similarly, people of colour may be constrained to stop using these terms for lack of agency (Giddens, 1984). These participants have taken on the institutional language which has become normalised, and upon which has through production and reproduction, become embedded through practice. Perhaps this appropriation of language by people of colour is being used as a mechanism to facilitate their navigation through the system which uses language, and specific terminology, to other, homogenise and dehumanise ethnic minority staff within institutions. One could interpret the use of these acronyms as a symbolically

violent act that reduces people of colour to a homogenous cohort that lacks specific and nuanced identities and deflects attention away from the differential experiences and outcomes of individuals of diverse ethnic backgrounds that lie within this grouping. Legitimising these terms through practice also sustains a position where white people can continue to see themselves outside of a racial or ethnic context.

Bourdieu (1977) articulates this in a way that suggests that certain dispositions existing in a group setting are exercised amongst the group in practice and are therefore reinforced and confirmed and subsequently become the foundation of collective belief (Randle et al, 2014). Translated in terms of the language used among higher education colleagues, through practice, duplication, and reinforcement, it has become acceptable by the majority to use acronyms such as BME, BAME, etc. when discussing people of colour and matters of race. To take a stance that is contrary to that which has been established though collective practice, for example, if a person of colour were to state openly that they were unhappy about being referred to as a BME person, could potentially place that person of colour in an unsafe situation in terms of their position within the field of academia and within their group of peers. As a result, the language has become customary and therefore established within the structured structure of the institution itself.

In accepting and reinforcing this position, and by replicating the use of racialised terminology people of colour are, unwittingly, replicating the symbolic violence (Bourdieu, 1998). It is evident from participants' responses that staff of an ethnic minority background, who may or may not approve of the terminology, accept their use through practice. This may be through choice, or through lack of agency. Perhaps this is also an attempt to gain social capital through coming to terms with the system within which they must navigate and therefore Giddens' (1984) concept of a dialectic of control is pertinent here whereby a subordinate agent can influence the dominant forces to accrue other capitals. This impacts upon their habitus, and to make their place in their field, might ultimately lead to gaining greater capital (social, cultural, and symbolic power) if they are perceived to integrate with institutional norms and practices.

There were further complexities surrounding terminology used by institutions in attempting to categorise staff in terms of race or ethnicity. Some participants with minority ethnic heritage did not have an objection to the use of ethnic categories such as black/black British

etc. Although one participant shared her exasperation about the way in which political correctness has hindered talk around race because it has contributed to people becoming afraid (Philpot, 1999; Crawley, 2007) by the language used:

I think people are scared of it. I think because things have got very PC to the extent that it's too PC sometimes, you know when you heard [sic] nonsense – people can't use the word 'blackboard', oh please give me a break...

Participant 25, black female professional, Russell Group

Additionally, some participants expressed frustration and mistrust about the way institutions interrogate staff about race or ethnicity through diversity monitoring, despite organisations, such as higher education institutions needing to use workforce data to meet their Public Sector Equality Duty (ECU, 2009; EHRC, 2014). This mistrust is in part caused by the absence of a communicated rationale to collect such data when the institution might appear to do nothing with it. The frustration has been borne from the categorisation of staff into distinct racial groups and that the forms and systems being used had not caught up with the complexities of people, particularly where there were conflicts around identity, such as being born in Britain with non-British heritage. These frustrations were expressed in the following comments:

...I'm black British, sometimes you don't see that...I'm not from Nigeria, I was born here...

Participant 11, black female professional, Post-1992

I put African Caribbean. I am not a colour, you know, my ethnic origin is African Caribbean and I always write that. If there's not a box I just put 'other'. So, when we're collecting information for HEFCE etc., we have a lot more blanks than we used to do, that has risen, the people putting 'other' or 'not wishing to answer that question' has risen. And I think it's because they're not confident to say, 'this is...', or they don't know what they are, how to record themselves.

...there's a gap in terms of how we identify, how we define ourselves, so people of colour, because of your experience, I'm just going to be general now, because of their experience, might not wish to define themselves, as like myself, African Caribbean, because African has negative connotations for them and the Caribbean might have negative connotations for them, but they don't see themselves as black British either...

Participant 30, black female practitioner, Post-1992

Another participant shared a perspective around the language used to describe ethnic background and that being able to identify as black can be an empowering and powerful notion to possess. This reclamation of blackness felt like this participant was using her embodied cultural capital (Bourdieu, 1998) to inform the strategies she employs in being able to navigate within her field and in the social space of her institution:

A friend of mine once said to me that she doesn't like being called black. She is a black person. And I said 'hmm, that's really interesting because I never felt deep about being called black. But now that I see the divide between black and white, I don't mind being called black because now, for me, black is almost like power in itself.

Participant 22, black female academic, Post-1992

The characterisation of ethnic identity can be viewed as a further example of symbolic power (Bourdieu, 1998). For example, higher education institutions follow data collection requirements set by external agencies, such as the HESA to collect information relating to UK higher education staff. In turn the data categories are based on the UK population census. The UK census has become the benchmark across the public sector and beyond and therefore has been legitimised at the very highest level of society, who are most likely to be white and male (Parker Review, 2020; McKinsey, 2020). Furthermore, the symbolic power wielded using racialised terminology delegitimises the identities of the dominated group(s) by legitimising the dominant culture (Mander, 1987).

When participants were asked to suggest alternative terms, participants offered 'people of colour', 'global majority', 'visual minorities', 'culture' and 'heritage', yet some recognised that practically, it might be difficult for an institution to monitor effectively if staff were given the option of self-defining their ethnicity in this broad way. This was a surprisingly pragmatic view, as I had expected there would be a more vehement rejection of a symbolically powerful structure in terms of the systems used to assist organisations to benchmark themselves against other like organisations and/or against the UK population. In hindsight and as an interpretation of the responses on this topic, participants' comments appeared to assume a feeling that the system is perhaps too big to tackle and that there were more pressing matters, such as differential outcomes and experiences of racism that needed to be addressed first and foremost. At the very least, the data and their categorisation, was deemed

to be a helpful method to identify matters related to workforce demographics and outcomes for particular ethnic groups.

This section has considered the views and perceptions of participants in terms of the way that society and its institutions marginalise people of colour using terminology. In speaking to varied participants, it was apparent that there were mixed feelings about the terms used, whether in a positive or negative sense and yet through practice, many participants supported these marginalising structures by using the very terminology that they were opposed to. This may reflect participants choosing which battles to fight, and this in comparison to other concerns, is insignificant. However, this may also be an indication of how society has normalised language and practice around race to the extent to which people of colour can no longer see that they are complicit in the systems of oppression that surround them.

Importantly, the problem that language and terminology brings is inextricably linked to the identity of racially minoritised staff in UK higher education. It is in this next section that the topic of identity and belonging is explored further to understand in more detail the experiences of participants within their institutions.

6.4 The struggles of workplace identity and belonging

This section will consider the perspectives from ethnic minority participants in terms of their own identity and whether they feel they can be their authentic selves within the workplace. Racial identity is complex and can be influenced by multiple factors including the way a person has been brought up, whether an individual has been socialised with positive reinforcement about their own ethnic background (Steinbugler, 2015) and whether their parents and wider familial network have discussed matters of race or racism (Shelton, 2008). The way a person perceives themselves can also be affected by experiences in adulthood and the interactions they continue to have through their personal relationships and work experiences (Steinbugler, 2015).

Participants discussed how to varying degrees they had internalised the institutional structures in which they operate, demonstrating certain agential dispositions, which have influenced their decisions and practice in the workplace, e.g., choosing to speak or dress in a particular way. Bourdieu (1977) suggested that agents actively and knowingly implement

certain measures in a given situation as a result of reflecting on or anticipating the consequences of their own or others' actions. As a result of this internalisation, some participants have structured structures around identity, which create structuring structures in terms of the way in which they and other people of colour should present themselves and act within the institution. This is an important game to play for participants to gain advantageous positioning within their field and amongst their peers.

Initially, identity had not been an explicit topic that had been included within the interview schedule with participants, however the matter was raised in the first interview around the subject of 'blackness'. This participant was curious to understand whether others felt the same way and whether specific aspects of black identity, whatever this equated to, was left behind when at work. The participant's enquiry and comments about this was sufficient grounds for including a question on this topic with subsequent interviews and a range of interesting interpretations and perspectives were forthcoming thereafter. The participant's interpretation of blackness in this interview was as follows:

I would perceive [blackness] as being the way that I will speak with other black people, not at work...even if I had my hair relaxed and it was long and straight and I...didn't wear anything ethnic, which I don't anyway, but that to me would be people thinking...probably an African person would see that as leaving their blackness at the door, and not coming in their attire. Not wearing anything with any ethnicity. Leaving their blackness and coming in with a suit and tie, suit and trousers, whatever.

Participant 1, black female practitioner, Russell Group

Participant 1 spoke about blackness in ways that were more physical, and therefore subjectively demonstrable according to taste and this would differ for each individual. To some extent there is an element of stereotyping that an African person would wear 'ethnic' attire as a matter of course. The comment relating to the way that she would converse with other black people echoes Fanon's (1967) statement that black people will interact differently with other black people than with whites and this theme is repeated in Participant 7's account below and can be interpreted as people of colour feeling more authentic in these environments. As the interviews progressed it was clear that 'blackness' meant different things to different people, revolving around language, the use of different dialects, clothing styles, accent and customs:

...when I'm home I speak a little Jamaican patois, [or] if I'm joking with somebody here, I might do that. If somebody gets me really angry it comes out. So, my dialect or whatever, so that's something that I would leave at the door because obviously this is my normal voice, and I don't speak any different at home. But, at home I'll break into something else. So, you learn within the structures and the politics of certain things. And I line manage about ten people, so that's again a different scenario, so what not to say, what to do…because it will get you into trouble.

Participant 7, black female senior professional, Russell Group

Here, the participant was clearly aware of her workplace environment, insomuch that there are certain cultural expectations that she perceives as the norms and values to be complied with whilst at work. Participant 7 is active and knowing and employs this knowledge in practice by adapting her behaviour and playing the game to fit in with the objectified structures of her field and habitus Bourdieu (1977, 1998).

For the following participant, he felt very clear that there is nobody stopping a person of colour maintaining their identity. However, there was a time and a place for everything and that there are certain values to uphold within a workplace setting to function 'without fuss':

It's not about colour, it's about social identity. ...when in Rome do what [the] Romans do. Don't come to Rome and tell them that they've got to be Muslim or black or whatever you want them to be. That's nonsense. If you want to work, if you want to live in England, you don't have to leave your music behind, your food behind, the way you interact socially, your dominoes behind and the rest of it. Even your clothes and your thing. In a working environment you have to accept that...you've got to leave certain parts of that behind as well, so that this social group can operate and function without fuss. Now, that means you can't come in here, in my opinion, I might be wrong, but in my view, you've got to accept certain values. You can eat, breathe, live, sing, you can do what you like. We'll embrace it. But don't bring the negative values in here.

Participant 13, mixed-race male senior professional, Post-1992

Participant 13 did not elaborate on his meaning of negative values; however, his comments did resonate with the literature relating to the negative perceptions about blackness (Mapedzaham and Kwansah-Aidoo, 2017; Solanke, 2018). Recalling our conversation, the participant felt that he had achieved or rather overachieved the expectations made of him, because he has worked hard and progressed from growing up on a sink estate in London to a

director level role in a university. The participant also articulates his views about others needing to leave parts of their identity behind at work to operate 'without fuss', which demonstrates that he has internalised the structures of the institution, which he then projects on to other members of the same social group (Bourdieu, 1977). According to his perception of UK society's expectations of black people and perhaps his own lived experiences, he felt that the black community needed to complain less and do more, stating:

...stop being fixated with the unfairness of the general situation. We are not proactive, we are not outspoken, except when there's a protest and anger. We don't support ourselves. We don't support our children. We don't support our community.

Participant 13, mixed-race male senior professional, Post-1992

Participant 16 aired her negative view around the term blackness, expressing that it is not a term used amongst her friendship group and described its use as unhelpful, which once again was reflected in the literature about negative perceptions relating to being black (as above):

...I suppose what they're talking about though is leaving their identity, everything that is subsumed within the notion of a black identity. It's not language I use. It's not language that is frequently used around me or even with people that I associate [with] that are of the same ethnicity as myself. They tend to refer to themselves actually by their heritage and maybe that's why I lean towards that. ...obviously there's issues of power, structural power here, but you don't hear white people coming in and saying I'm bringing my whiteness or I'm leaving my whiteness. We don't hear that, and...that is around what is the dominant presence and the dominant privilege. I don't think that using terms such as blackness helps us.

Participant 16, black female academic, Post-1992

The point made about power and structure is a relevant one since blackness can be interpreted as a concept, which is socially constructed (Mapedzaham and Kwansah-Aidoo, 2017) in opposition to whiteness. However, the distinction between the two concepts is that whiteness has become an objective structure through power and domination. The rejection of an objective form of the notion of blackness is clearly an aspect of this participant's habitus, where cultural heritage and the nuances of these provide greater scope for subjective expression and therefore, not constrained within fixed parameters of *what* and *how* a black person should be.

Perhaps to a certain extent, Participant 16 has internalised the negative aspects of blackness as fitting the stereotypical caricature of black people and therefore not wishing to engage with the positive attributes, which for some participants included aspects of language and personal presentation. The literature covering racial socialisation (Shelton, 2008) by black parents referred to messages and strategies employed to teach their children about black culture, racial pride and black achievements and this may account for the reason why this participant views this particular term so negatively.

Another participant shared how she presents herself at work and the level of consciousness required to 'pass' as an integrated member of staff within the institution:

There are times obviously because of the meetings that I attend, I will obviously just be the only individual who's black at that meeting. So, yeah, I have to – not say play down – but, you know, there's my mannerism and how I speak and whatever, but obviously when I'm with my colleagues it's a different thing, it's more laid back...So, yeah, I'm two people basically. ...I'm always conscious....I am the black [name] with the black family, the black child...but when I'm in the workplace, I am a little bit of black, in fact 5% black, but 95% of the time I'm this other person. ...I think I couldn't possibly as a black woman come into work, say for example with my head wrapped, you know, with a turban or...I just couldn't, because I just think they're going to look at me and think typical...I couldn't possibly come into work in this environment and act like a Jamaican say for example, [spoken with a Jamaican accent] 'yeah, you alright hon', you know, could you imagine at the meeting '[spoken with a Jamaican accent] yeah, man, me hear what you're saying'! I just couldn't do that! I feel bitter but, at the same time, I say to myself, I've got a daughter, and...it might not be in my lifetime, but maybe my daughter might be one of the ones that's going to benefit from what I'm going through...

Participant 17, black female professional, Post-1992

The views provided around black identity, whether positively interpreted and assumed or not, present a further dimension to the complexities of institutional identity for people of colour. Although some of us may assume a workplace persona at some point in our working lives, it appears that this might be more burdensome on minority ethnic colleagues if they need to be careful about the clothes they wear, the way they style their hair (and this is particularly an issue for black women) and the way they talk to others, including their accent as well as inflection. These aspects of hidden behaviours will have a significant impact on, and has the

power to influence, a person's success (Hirsch, 2018) in achieving a sense of belonging within the institution if they stand out too much.

There is too the added complexity for those of mixed heritage and the ability to straddle more than one ethnic and cultural background. One participant shared his interpretation of race and ethnicity and the difficulties that have arisen with terminology and his ability to relate to, and be accepted or embraced by, both sides of his own ethnic identity:

Race [and] ethnicity are very watered-down words you know. I mean we're so used to them that they don't really mean anything. But if you talk about sub-terms, such as in my instance my mother is white and my father is black, so there are sub-terms...like mixed race, half-caste, whatever you want to call [sic]. These are the two main ones. They are deeply offensive, you know, and even black people offend you when you say as a mixed-race person, you know they treat us blacks like this and they say, 'well you're not black' you know. They see progress, a lot of them, the poorly educated ones see progress as betrayal...if you're in a management role, supervision role...then you'd have more rebellion, more open hostility, obstruction from your own people than you would from other races because it's this Uncle Tom thing, you know, 'oh you're alright because you're mixed-race'.

Participant 13, mixed-race male senior professional, Post-1992

The participant alludes to conflicts he faces with other people of colour, where he perceives that he is being 'accused' of colluding with white people in his senior role within the institution. There appears to be a frustration about his own feelings of belonging insomuch that he acknowledges his mixed heritage yet feels rejected by people of colour due to the historical conflict that lies between blackness and whiteness. The participant signals the capitals he has accrued (Bourdieu, 1998), which elevate his status within the institution. Comparing himself to other people of colour in the workplace where power struggles and conflict arise where game playing (Bourdieu, 1977; Randle, et al, 2014) is required to fit in to the white hegemonic structure of the institution as well as appease the expectations of others who share the same or similar ethnic background. These complex issues associated with multiple ethnic identities and experiences are not explored in detail within this study, however, the workplace opportunities and challenges of mixed heritage employees would benefit from further investigation in future.

As discussed in the literature, social acceptance is a basic need for all individuals and is essential to developing and maintaining meaningful relationships and to derive a sense of workplace belonging (Baumeister and Leary, 1995; Leiter, 2012; Mohamed, et al, 2014). The reluctance to break social bonds causes individuals to preserve attachments with others that can result in substantial personal sacrifice. In order to understand how individuals managed this part of institutional life participants were asked to comment on their level of belonging within their institution.

A high proportion of participants responded negatively to a sense of institutional belonging, often citing a sense of marginalisation and isolation, although this was more likely to be the case where the staff member had worked for the institution for a long time, causing feelings of detachment towards the rest of the institution. Feelings of detachment towards the social space will result in significant adverse impact that may constrain an individual's ability to navigate their field (Bourdieu, 1998). This will ultimately affect their habitus and acquisition of capital (Bourdieu, 1986) where social interactions become more limited through marginalisation and increased isolation. From a positive perspective, many participants responded that their feelings of belonging were strongest within their immediate teams, and feelings of detachment were more intense when participants had negative workplace experiences and where trust had been lost with co-workers.

I feel a sense of belonging to my team. We have a very strong team ethos. We look out for each other...my experience of the current senior management is that they don't give a damn who you are...so they're not intimately involved.

Participant 5, other ethnic background female professional, Post-1992

...I feel a sense of belonging in my department but not outside it...

Participant 10, black female professional, Post-1992

It is not clear from these comments whether the lack of belonging is attributable to ethnic identity alone or whether it is due to the organisational structure of the institution. It is not uncommon in higher education for teams to work in silos, and this may intensify negative feelings that creates a chasm between immediate team and the rest of the institution. The siloed nature of institutional teams will certainly have a substantial negative effect where an individual's field is dependent on the relationships within their team, and this may be

particularly intensified from a scholastic perspective if their career in academia is dependent on research output, which includes partnerships with other academic staff within the same institution.

One participant reflected on her generally positive feelings of belonging to her team as opposed to the rest of the institution in the following way:

My sense of belonging really, it's within the team that I teach with...Because we have common respect for each other, we contribute to what we are doing, so I belong to that little group...I just feel that with my small group, they acknowledge my skills, they appreciate my contribution, they clearly can see that...I am a person who is critical, who can think, who can actually contribute to student life and their learning experiences, as it were. Whereas, in the bigger picture of the organisation itself...I'm just there to do a task, I'm only recognised when something has gone wrong with the task.

Participant 21, black female academic, Post-1992

Feelings of partial belonging might be as a result of the subtle cues that occur within the academic environment described by Chen and Hamilton (2015), which can lead to under-represented minorities questioning their sense of belonging within these domains. In these environments minority ethnic staff might become stigmatised by dominant group members, resulting in social isolation and exclusion in formal and informal groups and networks.

The theme of detachment, marginalisation and isolation (Stainback and Irvin, 2012) was reflected by many participants who felt that they were often the only person of colour within their work area, and at meetings etc. and therefore felt they would be treated as the spokesperson for their entire race (Solanke, 2018). This creates a significant burden for some who fear that their performance is used to judge not just themselves, but others like them. The following reflection provides an insight into the strategies used by the participant drawing on her agency to protect herself against the feelings of social and emotional isolation:

It's a strange space and it's a space that contributes to a growing sense of unbelonging [sic]. I am the only member of staff that does not have an office in the department. So, there is that sense of...social isolation as well as a kind of unconscious sense of isolation. ...I treat it in a way that allows me to compartmentalise how I feel when I get in. What it means is that there's a kind of an emotionless part of me that walks in through the door. So, if that's what they mean by leaving blackness at the door, maybe that's it. It's not about blackness for me, it's about

my...it's my vulnerability that I leave behind because if I bring it with me, it will be destroyed. So those sensitive inner parts of me, I keep them very protected. So, the version that they get of me is the 2D me, not the 3D.

Participant 16, black female academic, Post-1992

The participant offers a perspective on her social space, which according to Bourdieu (1977; 1991) are a system of mutable structures that distribute different forms of power or capital, and which define the rules and resources that become legitimised within that space. It is unclear whether the lack of an office has been imposed. It appears that Participant 16 has either accepted the structures that have been presented, and may indicate her lack of capital (Bourdieu, 1998) in that social space, which could be a combination of social and cultural capital, or that she lacks agency to challenge the situation she finds herself in. Despite this, the participant is clearly aware that she is limiting who she is in the workplace, and this is an active part she is playing. Therefore, demonstrating agency as a form of protest. This may have more detrimental impact on her career and workplace experiences as she has begun to self-isolate as a form of self-preservation, yet without fully considering the effect this might have on her ability to acquire further capital through networking with peers in that social space.

Furthermore, Participant 16 raises an important issue in terms of expectations placed upon people of colour in the workplace. This brings to life the points made on hypervisibility (Kanter, 1993; Ibarra, 1993, 1995, Solanke, 2018) that place minority ethnic staff at a higher risk of being othered, increasingly scrutinised and subject to exclusion or even self-exclusion, reducing employee engagement and impairing employee authenticity and productivity. One participant felt she was over-scrutinised once she had reached a higher level within the institution:

...If I'm at a certain level I'm invisible, but the minute that I'm a bit further up I get challenged...and I found that I was being challenged a lot more than my counterparts.

Participant 15, black female professional, Post-1992

In the literature, Solanke (2018) spoke about the hyperinvisibility that the absence of people of colour within senior levels of academia creates, which as Participant 15 recounts transformed to hypervisibility once she had crossed a particular structural threshold within

the hierarchy. An increased scrutiny will inevitably produce anxiety and may attract boundary heightening and an escalation in the critical evaluation of performance (Kanter, 1993; Smith 2013). It is possible that the hypervisibility that Participant 15 speaks of may be related to her sex, her ethnic background or the intersection (Crenshaw, 1989) of both. Bourdieu (2001a) spoke about symbolic power and the notion of women participating in social spaces that have been traditionally reserved for men. Therefore, this participant's experiences could be interpreted as over-scrutiny of her leadership attributes and whether these meet the cultural arbitrary of the domination of men over women in this domain. One can add to this the participant's ethnicity where there are statistically few black women in leadership positions in UK higher education. There were a number of participants that provided accounts of their workplace experiences where this was the case:

...you feel like you've got to work harder than...your non-ethnic [sic] peers to achieve the same status... or opportunity...and I've had to do that all my professional life. That's been my driver and that's why I've achieved, but at the same token sometimes you go, 'well why do I have to work so long?' Why am I having to bend over the extra mile? ...if I do have a day where, 'you know what, I'm going to leave at five or leave at half six'. Why do I feel guilty? You know, when other people clearly don't.

Participant 13, mixed race male senior professional, Post-1992

The comments made by Participant 13 were echoed by several participants and this perception appeared not only in accounts about their own experience but is information shared inter-generationally to prepare the next generations. There was no evidence that this experience was specific to any gender, as similar accounts were reflected equally by women, however the impact on black men may be heightened because of associated negative stereotypes. These accounts echo the literature (Shelton, 2008, Steinbugler, 2015) around strategies parents may employ in racially socialising their children about life in whitedominated workplaces. This was reflected by one participant who spoke about her own experience in the institution and the advice given to her son:

I would say as a black person, I've had to personally work damned hard. Harder, than most people have had to work. It's the same with my son. He's had to work jolly hard to be the best of the best because, as I've always said to him, 'you're young, you're black and you're male, people will look at you first'.

You know, and he gets it...I said, 'you don't have to be rude...you don't have to do that because you've lost your value then...you don't have to be nasty, be yourself, but always be aware that you have to work harder than everybody else because there's a stereotype – young, six foot two, black male, oh God...before you've opened your mouth'.

Participant 25, black female professional, Russell Group

The topic of stereotyping arose in many of the discussions, either as a direct reference, such as that from Participant 25 or when discussing assumptions made by others in the workplace. The experiences shared were in all cases negative or adverse in their impact. The assumptions being made were based on their ethnic identity and took various forms, including microinvalidations, such as being misrecognised, etc. In the following statement, the participant spoke about arriving early for a student conduct panel, which she was invited to attend as a new panel member:

...I was told...'can you wait out here and we'll call you in when we're ready for you'... ...I thought, did she think that I was the student that did the plagiarism?

Participant 14, Asian female senior academic, Russell Group

My name is very unusual. People are very surprised when they see me. I remember once working...in one of the departments...and for months...I was speaking to this consultant on the phone and he came in the office one day for something, never met each other or anything, and there was a young lady who was English sitting there, she was the temp and then there was me and he came in, he went straightaway to her and said, 'are you [name]?' and she said, 'no, this is [name]' and the look on his face was like...and he actually...said, 'you're [name]?!'. I said 'yes'.

My response to him was 'no I don't have two horns and I do not smell of sulphur' – that's what I said to him! That has happened blatantly...and for me that's really sad because when I'm dealing with anybody I don't even care what they look like or what they sound like...but I think also the preconceived ideas of you could be ghetto. No, seriously, you could be ghetto, you're a black chick, you could be ghetto. The stereotypes are there and it's very, very subtle.

Participant 25, black female professional, Russell Group

These examples not only highlighted the prevalence of racial stereotyping (Gamst, et al, 2011) but also provide an insight into the seemingly innocuous interactions described in the literature as microaggressions (ECU, 2011) or micro-incivilities (Kandola, 2018). As a response

to this, individuals can experience social identity threat, that induces the state of psychological discomfort when confronted by an unflattering group or individual reputation (Nelson, 2009), such as those assumptions reflected in the above comments. This threat can become particularly heightened where that reputation can be confirmed by one's behaviour – adding a significant psychological burden if the individual is being judged by their peers. The impact of social identity threat can include disruptive effects on performance in the short-term, e.g., interview, examination, etc. Over time, this can prompt defensive behaviours, such as disengaging from activities or teams and can result in a self-fulfilling prophecy whereby the person begins to resemble the stereotype (Nelson, 2009).

It might be assumed that stereotypes made about ethnic minorities originate from others that do not share the same ethnic identity. However, there were some unexpected comments made by ethnic minority participants towards others of the same ethnic background that reflected certain attitudes towards British-born ethnic minorities:

...with progression for black people or ethnic minorities I think it opens up a new dimension where people may not see themselves as 'I'm not going to get that opportunity because people are going to be racist to me', so there's that mental block in the minds of maybe black people... Most of the people don't have internal locus of control, where they blame everything on the outside. Everything is external to them rather than people making the first move...because I tell people if you're born here and I have come all the way from Ghana and I'm trying to get myself opportunities, and you're here telling me that people are putting blocks in your way, then it doesn't work. I will say yes, and then holding people back in the system will be those that are born here, because the black people that I know that are born here, the system holds them back. But black people, let's say immigrants, the system doesn't hold them back, because where we come from, let's say Ghana or Nigeria, you're not given things, so I find over here there are so much [sic] opportunities. ...my auntie's children, they were always saying there are no opportunities. I'm saying there are opportunities, because I look at...the big picture from way down they think the Government owes them something... It's about attitude.

Participant 2, black male academic, Post-1992

Participant 2 raises interesting points that require further exploration. His observations about British-born black people, although a generalisation, may signal social aspects that the participant, who was born and brought up in Ghana, may not have experienced. Some

participants spoke about racial socialisation (Shelton, 2008, Steinbugler, 2015) and equipping their children with the tools to navigate within a white-dominated society and contained within this may be discussions around race and racism that might be encountered. Socialising around the expectations of racism, may place individuals in a state that they could mistrust institutions on race-related matters and therefore internalise the structures that constrain certain groups of people.

Participant 2 does not appear to interpret the structures (Bourdieu, 1977, 1991) as inhibiting, rather he is using his agency (Giddens, 1984; Bourdieu, 1977; 1984), which has been informed by his habitus, to seek out opportunities to gain various capitals (Bourdieu, 1977; 1998). Interestingly, the workforce data presented in Chapter Five of this dissertation shows that non-UK black academic staff (3.1 percent) have an overall higher representation within institutions than their UK black counterparts (1.6 percent) and are proportionally more likely (0.9 percent) to hold professorial positions than their UK-black academic counterparts (0.6 percent).

Another participant reflected on his experiences of attitudes during the recruitment process and hiring prospective candidates where potential issues might arise:

...there is this core mistrust with employing black people and it doesn't matter if you're a racist or not. Everybody, even black people who are managers, who employ other people, they may have candidates who are black, they're worried. What if this guy's a difficult person you know? How am I going to handle that because you know they can get pretty militant, and they can get the unions and they get race relations on to us? We don't want that.

Participant 13, mixed-race male senior professional, Post-1992

The participant's reflections capture how complex and powerful the social construction of race is and how these problematic concepts continue to be internalised by people of colour (Mapedzaham and Kwansah-Aidoo, 2017; Hirsch, 2018). Stereotyping and other negative attitudes about black people are not solely held within dominant groups and this may be intensified further by class or socio-economic background and/or taste, which Bourdieu (1984) refers to as distinction and where differences in power become inscribed into symbolic or cultural differences. What has become clear following the analysis and interpretation of such a rich data source is that there are certain rules that apply to presentation, conduct, and

navigation within an institution. All these actions will have an impact and a consequence that may feed further negative preconceptions and/or place an individual in a position where they need to adjust their mental or physical state to cope with the ongoing threat.

This analysis offers an alternative view on the diversity versus equal opportunities debate where there is conflict between an organisation's will to diversify the workplace, yet there are significant issues to contend with to ensure recruitment processes are conducted in a fair, transparent and unbiased way. The account from Participant 13 highlights that bias, whether intentional or not, can be a product of different lived experiences and that perpetrators can be from diverse backgrounds. It is imperative that institutions are cognisant of this in their efforts to increase diversity across their workforce and manage that diversity throughout the employee lifecycle.

6.5 Conclusion

This chapter has reflected on the feelings, perceptions and experiences of participants that took part in this study by exposing aspects of fundamental institutional life in a way that is not often discussed; our language, our identity, and our need to connect. These are perhaps aspects that most of us take for granted. The findings reveal the participants' vulnerability, which appear hidden in plain sight.

The analysis within this chapter has focused predominantly on the research elements of situated activity and the self (Layder, 1993). In terms of these elements, the exploration has considered the social interactions of participants from their accounts to make sense of, interpret and understand the lived experiences of black staff in UK higher education. This has been possible by referring to and considering the secondary statistical workforce data, the relevant literature around structure and agency, social theory, racial socialisation, identity and by comparing and contrasting the experiences of participants against these theories. At the level of the self, the analysis of the primary data collected shows that experiences are diverse, and outcomes may be affected by place of birth and racial socialisation, gender, role and level within the institutional hierarchy. The richness and value that the primary data provides because of this participant diversity cannot be ignored and this chapter has made an insightful contribution to our knowledge and understanding around concepts that have not been discussed previously within a UK higher education context, particularly when

considering the lived experiences of professional and support staff in the sector, where accounts are largely absent.

Participant accounts have contradicted my own initial assumptions that people of colour make conscious decisions to join public sector organisations because of their diversity. This is particularly noticeable in terms of academic faculty, and particularly those that studied in UK institutions, who managed their expectations regarding their institutions knowing that academic staff are not ethnically well represented. For both academic and professional staff, there was general acknowledgement that senior levels were the most barren for ethnic minority representation. Most participants supported the notion that it was this lack of ethnic diversity that became the catalyst for further interest and involvement in equality, diversity and inclusion work within their institution, some becoming actively involved in institutional projects or within trade unions.

In relation to racialised terminology, there was good quality discussion surrounding the complexities around the notions of race and ethnicity, their distinction, their histories and the challenges associated with the way language and terminology is employed. Overall, participants' views were consistent with the problems identified within relevant literature (DiAngelo, 2018, Eddo-Lodge, 2018; EHRC, 2019) and there was almost unanimous agreement that the mention of race within an institutional setting instilled fear. As a result, institutions were more likely to avoid the topic and did not provide space for staff to discuss matters related to race and in some cases, it was felt that this was fuelled by the pressure to maintain political correctness. There was, however, an overall acceptance that the terms race and ethnicity were used interchangeably, which according to the literature (ECU, 2009; EHRC, 2014) has been normalised over time through the methods employed by institutions through diversity monitoring exercises.

In addition to this, the broader understanding of this terminology can be explained to some extent by the legislative context that has informed our thinking as a society in terms of 'race relations'. To a great extent, practitioners in the fields of equality and diversity and human resources, have been influenced by 'race' as a protected characteristic (Equality Act 2010) and thus driven organisations to engage with this term. Consequently, the term race can be justified from a practitioner perspective, and this has become normalised as a result. Within

a higher education context, there is no guarantee that there will be a greater understanding and/or appreciation of the broader issues surrounding race and ethnicity.

The role of policy development and implementation within institutions sits generally within the corporate centre, and therefore may not be informed by research directly. It is also relevant to note that despite a relatively healthy proportion of black staff employed in administrative or secretarial occupations in UK institutions (Advance HE, 2020), anecdotally there is limited ethnic diversity in the corporate departments within institutions (see Chapter Five). The lack of diverse decision makers within the corporate centre will also mean that diverse perspectives may not be considered with the development of new policy and strategy despite good practice requiring the completion of equality analysis²⁰.

With all UK institutions involved in monitoring workforce diversity, some participants reported feeling suspicious about the collection of ethnicity data. These perceptions are often fuelled by a lack of transparency on the rationale for data collection considering that ethnic minority representation has remained static and differential outcomes persist. That being the case, some questioned the point of data collection if it was making no difference. Participants acknowledged that diversity monitoring had not kept up with contemporary expressions of individual identity but recognised that making the system over-complicated might be unhelpful in making a real difference.

An aspect of questioning that attracted intense disapproval from some participants was on the topic of terminology, specifically through the acronyms BME and BAME. The use of these terms was seen by participants as making individuals invisible, disregarding their different histories, background and experiences and dehumanising people of colour in our institutions. The irony is that people of colour have become complicit in supporting this symbolic violence (Bourdieu, 1990; Jones and Bradbury, 2017; Randle, et al, 2014). The extent to which some participants used these terms without considering the impact of their use was unexpected and highlighted a personal assumption that more participants would reject them.

At the core of this chapter is identity and unsurprisingly, this is an important factor for participants within the context of their institutional life. This topic raised a number of complex issues that included perceptions, attitudes and interpretations of 'blackness' and how this is

-

²⁰ Also known as Equality Impact Assessments

understood and manifested in the workplace and at home. Some participants related blackness as self-expression be that through language, mannerisms and style of presentation. Participants commented that they could be authentic in the company of other people of colour (Fanon, 1967) and this can be expected within the context of institutional structures that produce and reproduce white, middle class cultural norms and expectations within the workplace. However, some rejected the notion of blackness because it stripped away the nuances and diversity of black people, making this problematic. Literature (Mapedzaham and Kwansah-Aidoo, 2017) suggests that blackness in opposition to whiteness is problematic and reflecting on this and participant commentary this concept echoes the historic ideals developed of good-bad binary that white is good, and black is bad (Jordan, 1974; Olusoga, 2017; Hirsch, 2018). Many participants spoke about how they scan their social field and make decisions so that they appear to fit in (Bourdieu, 1977, 1998). Whereas others internalised the institutional structures, objectifying them so that they inform their expectations of other people of colour and establishing new rules and structuring structures that impact on the field and habitus.

Another complexity to the discussion around identity, were the perspectives of those with mixed heritage and the difficulties encountered with self-identity, their sense of belonging to any particular social group and how they were perceived by others, including majority and minority ethnic groups. The unexpected accounts from a mixed heritage participant about his experiences of conflict within and between different ethnic groups appears to make operating within the structures of the institution challenging. Not only are strategies required to work within a white-dominated space, but other strategies are required to work with minority ethnic staff, without feeling accepted by either group (Bourdieu, 1977; Randle, et al, 2014). As such, an individual's accrual of capitals (Bourdieu, 1998) within an institutional setting can also create conflict between whiteness and blackness. This highlighted the excessive strain on self-identity and the human need to belong considering the challenges that will arise from both/all sides of that individual's identity, including prejudice.

Certainly, the topics discussed emphasise how important self-identity and perceptions of identity in the wider institution are to the decisions people of colour must make to navigate and operate within the given structures. These structures are objective to participants because they consistently spoke about the expectations around how one should act, dress,

speak and this understanding was shared across this cohort. This shared understanding is evidence of an objective structure that is in operation regardless of the type of institution (Russel Group or Post-1992) in which they worked, and therefore this indicates the production and reproduction of an institutional habitus (Horvat and Antonio, 1999), which could be argued is sector wide. Consequently, these structures create differential outcomes and an ethnic penalty for black staff because they must create strategies to operate within this environment. If black staff are unable to effectively play the game (Bourdieu, 1977) and decide to move to another institution, it is most likely that the culture would be the same and would impact negatively in the pursuit of progression within their field, and this is apparent from recent research relating to UK black female professors (Rollock, 2019).

The absence of institutional spaces and opportunities to discuss matters of race and ethnicity means that their differences and histories are not explored and remain neglected. Accounts from equality and diversity practitioners in the field support that this would increase institutional confidence to tackle race and racism. Some participants born outside the UK explained that their first exposure to race has been in the UK, which drew attention to their own ethnic difference, and this has caused those participants to feel confused and unsettled about their career prospects and made them feel othered in a way that had not been present before.

In conclusion, this chapter has explored the environment where minority ethnic staff operate and how their identity and sense of belonging are fundamental in facilitating their institutional journeys. The next chapter will advance the investigation further to make sense of participants' experiences and accounts of racism in the UK higher education sector, how racism is manifested within contemporary institutions and the way participants develop strategies to continue to operate within these workplace environments.

7 The lived experiences of race and racism

7.1 Introduction

The previous chapter considered the impact of the social space in higher education institutions and how the structures, which have become objectified through production and reproduction by social actors have impacted upon black staff. The structures have been created by, and for, dominant ethnic groups requiring black staff to develop strategies for their effective navigation through the institutional structures. This requires sense-making, interpretation and reasoning within their habitus and constant deliberation and judgement to map out actions that do not attract adverse attention, heightened scrutiny and/or further isolation.

In this second findings chapter primary data have been analysed by drawing together previously published literature. The analysis also considers the macro, meso and micro level aspects of Layder's (1993) research map whilst considering history and power, which play a significant part in the production and reproduction of systems and structures across these levels through social actors in the social space. In considering participant accounts, the analysis and interpretation of the empirical data will draw on the concepts of Bourdieu (1977; 1984; 1990; 1998).

The aim of this research is to explore the lived experiences of staff in UK higher education on the topic of race, with the objective to understand how these experiences create differential outcomes for black staff in the institutional workplace. This chapter will advance the exploration of the previous chapter by focusing on how participants understand racism as a concept, if and how they are able to identify racism in the workplace, and understand the strategies developed by individuals to navigate the institutional structures and systems effectively. This chapter will specifically address the research objectives: 2) to consider staff perceptions of the manifestation of racism in the workplace; 3) determine the impact of racial inequalities on black staff in UK higher education; 4) consider perceptions of the existence of institutional racism and its effects within a higher education context; and 5) explore the presence of an ethnic penalty faced by black staff in UK higher education.

Institutional racism, as discussed in Chapter Two, is a deeply controversial subject and there has been very little exposure to this topic from a UK higher education perspective despite the shared public sector interests. A recent official inquiry (EHRC, 2019) into racial harassment in higher education looked at the different types of racial harassment experienced by students and staff, the routes for reporting racial harassment and how effectively those reports are dealt with. The report highlighted how institutional systems and processes affect the outcomes of minority ethnic staff and students affected by racial harassment and made recommendations to higher education institutions, external agencies and higher education funding councils. This report is particularly relevant in this research as it forms part of the multi-level analysis that provides insight to the sectoral context relating to racial harassment that describes the setting within institutions where racial harassment is present, provides situated activity around the meanings and understandings of social interactions that relate to racial harassment and considers accounts of participants (Layder, 1993). This chapter will consider these and other aspects that relate to racial discrimination, institutional racism, and participants' experiences in this research study.

The topic of race has the power to drive people and organisations away (DiAngelo, 2018, Eddo-Lodge, 2018). It is not the discussion about race and ethnicity but the perceptible lack of engagement with the topic by the majority ethnic group grappling with concepts that are often alien to them. DiAngelo (2018) suggests that white people have not been conditioned to think of themselves as racial beings and it is perhaps because of this, that conversations about this subject often lack the engagement of dominant ethnic groups within organisations. Crawley (2007) asserts that the social environment has encompassed a one-dimensional view of race, which has become repressed by the straitjacketing or silencing of articulations on race and ethnic minority people. The silencing of race talk has created an anxiety among white people within a racial discourse, which is unable to escape a dialogue that includes voicing experiences of racism by those who experience it. Consequently, when voices are heard they express and draw attention towards negative experiences, perceptions and opinions about race that act as a catalyst for fear, denial and trepidation (DiAngelo, 2018; Eddo-Lodge, 2018; Bhopal, 2018). That fear, denial and trepidation does not solely lie in the domain of white people as the remainder of this chapter will reveal and these concerns are both complex and multi-dimensional and have been evidenced through the narratives as a consequence of the Black Lives Matter movement in 2020 (Wright for Daily Mail, 2020; Wall for The Observer, 2020; Davies for The Daily Mail, 2020; Waterson for The Guardian, 2020; Skopeliti for The Guardian, 2020; Edwards for BBC Wales, 2020).

The analysis and discussion will begin by considering how racism is understood and explore examples of the way it manifests itself within contemporary UK higher education. This will be followed by a deconstruction of the manifestation of racially prejudiced attitudes and behaviours including considering the comments and lived experiences of those participants within the study who have in multiple and complex ways discussed their understanding of it, and how they have come to terms with these experiences at work. Relevant concepts and theories will be applied to make sense of these workplace experiences to help identify how individuals and organisations might develop and implement solutions as a sector and locally within institutions.

7.2 The invisibility of workplace racism

The narratives that surround the historic and political development of race and ethnicity in the UK (discussed in Chapter Two) often include references to racially motivated prejudice. Allport (1954) suggested that a person's prejudice is unlikely to be merely a specific attitude toward a specific group. While prejudice is formed through learnt stereotypes, racism is a manifestation of prejudice (Gamst, et al, 2011) and according to DiAngelo (2018), we are all prejudiced. The difference lies in the way we express or act upon it. These manifestations or enactments can occur at institutional, societal and individual levels in overt (traditional) or covert (modern) ways (McConahay, 1986; Rowe, 1990; Essed, 1991; Henry and Sears, 2002; Sears and Henry, 2003; Noon, 2017). Although overt acts of racism tend to occur less frequently, they do still happen, and one participant in this research provided a brutal example about her experiences delivering training to staff across the UK higher education sector:

I see all types of racism...and I've seen it from the very explicit someone having 'nigger' written next to their name on a sign in sheet and this was in 2009...and it was the same individual, and I have no idea who the individual was, but it was the same individual who wrote 'Paki' on the sign in sheet next to another person's name.

Participant 19, white female practitioner, sector agency

Another participant provided the following insight around the subtle changing face of racism:

...I've had it in my face, people would shout stuff in my face. They don't shout stuff in your face now. They will just either shun you or they don't include you.

Participant 1, black female practitioner, Russell Group

Participant 1 describes how the more subtle, covert mode of racism can be misrecognised because these acts or omissions can be made up of small events that are hard to prove (Rowe, 1990). The following quote provides an insight into the subtleties of racism where this perspective suggests that actors may unwittingly engage in racist acts:

I think it's more unconscious as well. I don't think people are aware of it for the most part. I think it's the way, it's the assumptions we make, the criteria we use for the lack of support that we give, even though we think we give support, we don't. I think it's a very, very complex and deep-rooted problem and most of it is well below the surface...I think subtle is right but I'm not even sure that it is covert. I don't think people are covering something up. I don't even think they realise they're doing it...they think they're very liberal and open and tolerant and are, you know, but taken together the whole set of assumptions and ways that we do things actually result in it being extremely difficult...kind of unwelcoming and closed off...

Participant 29, white male senior manager, Russell Group

Participant 29 reveals a thought-provoking argument about the covert nature of racism, suggesting that acts or omissions are *not* 'covered up' and this may be as result of individual and institutional behaviours that have become normalised over time through routine practice (Bourdieu, 1977). The field here is also relevant because the participant describes the perception that colleagues are 'liberal', 'open' and 'tolerant', providing a description of perhaps how UK higher education as a social space wishes to be perceived. DiAngelo (2018) explains that aversive racism is a manifestation of racism that well-intentioned people are most likely to exhibit and exists subconsciously whilst allowing an individual to maintain a positive self-image. Therefore, aversive racism becomes legitimised because of perceived attributes that social actors have in this space, which contribute to the conservation of institutional structures and white solidarity (Bourdieu, 1998). As an observation, Participant 29 is from a white ethnic background, and this may inform his perception that the

manifestation of these behaviours might be unwitting, yet this account could be interpreted as a manifestation of white fragility (DiAngelo, 2018).

Herein lies the challenge that faces UK institutions to understand the unconscious (or otherwise) series of behaviours that may constitute forms of racism, particularly when senior levels of the institutional hierarchy are constituted primarily of white males and therefore may not recognise behaviours to be racist. These behaviours can include neglect, incivility, ostracism, and inequitable treatment (Fox and Stallworth, 2005). As asserted by Sue, et al (2007) racial microaggressions are brief and commonplace daily verbal, behavioural, or environmental indignities, that communicate hostile, derogatory, or negative racial slights and insults toward people of colour. Webb, et al (2004) suggest that being treated as inferior, denied resources, limiting social mobility and aspirations are not perceived as 'symbolic' forms of violence, rather their situation seems to be 'the natural order of things'. If the natural order of things is a culture that has formed amongst dominant groups within an organisation, then it could be understood that these forms of violence are created on a micro level and constituted on a macro level becoming normalised by being produced and reproduced through practice (Bourdieu, 1977). Participants in this study talked about experiences that had occurred during their time in higher education, and that often due to their subtle nature, they felt unable to link these experiences to their ethnic background, despite having a "gut feeling" about the root cause:

...it's hard to explain, but it's definitely a gut feeling...it is about race, but to pinpoint and say it is racism is hard, but it is definitely about race...from my experience, I just remember a line manager that I had who promoted other colleagues, or she will inform them of development opportunities, but I was always kept in the dark, yeah? It's a gut feeling ...it's hard to explain, but it's definitely a gut feeling. And racism is not just black/white, it can be Asians and black. It's very difficult. I have to really ask the question 'is that a racist...or is it somebody who just doesn't understand or somebody who's paranoid or scared that you're going to take their job and so that was their reaction?' So, it's very difficult.

Participant 17, black female professional, Post-1992

Racism is subtle and nuanced. So, what you find is white staff being given access to opportunities, access to networks, access to support, access to supervision, whatever, that black staff won't necessarily have similar access to. So, we don't have the networks. What that

means, we have to be brokered into the networks and unless someone is prepared to open those doors for us, we can't put the foot in the door because the door will be shut on the foot.

Participant 16, black female academic, Post-1992

...it's not about somebody being brutally, like, face to face racist towards you, but it's just using their power of privilege to continually disadvantage you. They will still smile with you, they will talk with you, they will exchange little messages with you, but you are never part of their life in that sense, not as individuals, not as a collective, not as anything. It's all very under the cover.

Participant 21, black female academic, Post-1992

It is possible that the situation Participant 17 is describing may be the manifestation of (un)conscious bias, however it is often impossible to know or understand the motivations of others in their decision making or their willingness to support or sponsor another with workplace opportunities. There is another angle to this account in terms of this participant's agency and their ability to challenge the fact that others were being promoted or supported, rather than remain a passive observer.

The quotes above highlight that participants have a feel for the game within their respective institutions and understand the habitus in which they operate (Bourdieu, 1977, 1998), despite what they believe to be racism manifested through subtle acts or omissions. Yet knowingly, they continue to operate within the field, which they recognise as being burdened by these conflicts, challenges and struggles (ibid) with and between different social actors, effecting the development of social connections, and resources that can be gained through them (Bourdieu and Wacquant, 1992). The perceptions of these three participants also describe how symbolic power is being utilised by social actors in these institutions without words and which are imperceptible, insidious and invisible (Thapor-Björket, et al, 2016). This symbolic violence explains how social hierarchies and inequalities are maintained by forms of symbolic domination, where the dominated unknowingly and unwillingly contribute to their own domination by accepting the limits imposed (Bourdieu, 1990, 2001). The participants are therefore constrained by the structures they operate within and would be faced with choices of disrupting the status quo through challenge or to leave the institution, both of which may be impossible to do because of the consequences.

Another participant provided his opinion on how staff demographics would indicate race-related issues. Although initially suggesting that overt racism had disappeared, he recounted an experience of inappropriate and overt language used by a senior staff member whilst working at a previous institution:

I've worked in [Russell Group institution], [local government], I've worked in so many different universities and you just walk into a room, and you go well hang on, I know what to expect when I walk into a room. I've never worked here before, but I know when I walk in there it's going to be that I'm the only black guy. Does that mean that there are no other black people who can do these jobs? So, the statistics don't lie, but the overt racism has gone. I was in a meeting with another [institution], my boss at the time was a female mixed-race lady...and we went into a meeting with the college secretary, one of the most powerful people in [Russell Group institution] and his board room, we sat either side of him at the board table and we explained...a difficult problem that we've got...and he sat there, and he looked, stared for a moment and he went, "wow, this is a real nigger in a pile". [My boss] looked at me and I looked at [my boss] and we looked down and...and we looked at him and he didn't even know he'd offended us.

Now, he was like ex-military, you know...very upper class, aristocrat sort of thing and we walked out there, and I said, 'did you hear what he said?' She said, 'he didn't mean it.' Yeah, but in every other way he's a gentleman, he's not really...and we both walked away thinking how could you say that to someone? Did he walk away after the meeting and think, 'oh shit, I shouldn't have said that.' But he didn't look like that... And that's...relatively trivial...I mean... because the real sense is that, how does it affect your life, how does it affect your work, how does it affect your opportunity? And that racism (huff) I've seen it probably far too many times to mention...

Participant 13, mixed race male senior professional, Post-1992

This participant's reference to statistics relates here to workforce demographics and that he interprets this as evidence that higher education has a difficulty with ethnic diversity despite overt racism largely disappearing within institutions. This is a pertinent point because the absence of overt racism and organisational ethnic diversity are not mutually exclusive. From this account it is clear that overt acts of racism do still occur, even within the most prestigious institutions. This example reflects the normalisation of clearly offensive language, as casual racism, and used in such a way that completely disregarded the audience and demonstrates

a microaggression (Sue et al, 2007) or microincivility (Kandola, 2018). Examples such as that described by Participant 13 could easily be explained away by institutions as unconscious bias, particularly if one is to consider Participant 29's comments about the unwitting nature of people's behaviour. Participant 13 did not provide details of the age of the college secretary that used this derogatory term, but it could be assumed that this man was white and of more advanced years to be in the position that he was in. As such, power plays a significant role in the reason why this participant and his boss felt that they could not challenge his use of language for fear of reprisals.

Bourdieu (1977, 1991) expresses that the social space is made up of mutable structures of the distribution of power and this helps to unravel the issues within this scenario, insomuch that the senior person alluded to by Participant 13 used his position within the institution to define the rules of engagement, which have been legitimated through his position in the hierarchy within that space. This act was not only one of overt racism, but also symbolic violence due to the validation given to the language used by a person in a powerful position within the institution, becoming authorised language (Bourdieu, 2001a). Interestingly, his colleague was very quick to excuse his behaviour, and this may indicate that she possessed sufficient knowledge and reflexion about the social space to inform their decision to take no further action for fear of negative repercussions. Interpreted in an alternative light, the choice not to act in the moment perpetuated the dispositions that reproduce racial inequalities. However, the distribution of symbolic capital in this scenario was balanced against the participant and his colleague to be able to transform the social space (Bourdieu, 1998).

The evidence emerging in the higher education sector in recent years (ECU, 2011; Pilkington, 2013; University and College Union, 2016; Rollock, 2019) shows that institutions and their minority ethnic staff are not immune from negative workplace experiences. Despite recommendations made as a consequence of research by sector agencies and trade unions (see Chapter Five), there have been few initiatives in higher education to address these findings. The social space within institutions may be a contributing factor in masking racism and from the example given above, there are complex dynamics that maintain the status quo even where racism is explicit. There are other issues to consider, which includes whether individuals can identify less obvious or hidden racism and whether people of colour feel confident enough to call out racism. One participant offered their perspective on the matter:

I think people are mortally offended if you might suggest that anything they do, say or think could be considered racist, and I think that it's difficult to even confront ourselves what our own unconscious biases can be with regard to race.

Participant 19, white female practitioner, sector agency

It appears from this reflection that confronting racism is problematic, and this is further frustrated by our inability to identify acts of racism especially those that are small and covert within an organisational setting. Analysing this further, there are links here to the comments made by Participant 29 who suggests that social actors in the institutional space perceive themselves to be liberal, open and tolerant, making accusations of racism so offensive. This social identity threat is a response to everything the dominant group believes themselves to be (Nelson, 2009) and is further supported by DiAngelo (2018) and her observations around aversive racism among white progressives. However, since they are the dominant group, they are able to use their symbolic power and white fragility (DiAngelo, 2018) to deny these actions, whether unconsciously or unwittingly, and maintain the structures within the institutional social space (Bourdieu, 1998).

Within this institutional habitus it is difficult for dominated groups to transform the social space. Yet, Stainback and Irvin (2012) suggested that 'token' workers are more likely to name a negative workplace experience as racial discrimination. To understand this aspect further, ethnic minority participants were asked whether they would be likely to report that they had experienced a negative workplace experience perceived to relate to their ethnic background. Contrary to Stainback and Irvin's (2012) theory most participants felt that it would not be in their interests to report any incidents that might be related to race or that they felt uncomfortable with using such 'harsh' terms.

...I couldn't do that...I couldn't be so blatant...you want to have a good relationship...you wouldn't want to spoil it, so you just keep quiet...

Participant 10, black female professional, Post-1992

[I wouldn't challenge it] ...because I'd be seen as a troublemaker...and to get on in life you need not to be a troublemaker...it's career suicide.

Participant 18, black female senior professional, Post-1992

...As much as I want to talk race, I'm aware of the people I'm dealing with and I know that if I mention the word race, it's going to be such a big thing, even without saying anything, I will be accusing someone of being a racist, so I'm quite aware of being politically correct, which I hate.

Participant 22, black female academic, Post-1992

These examples provide powerful testimony of the difficulties associated with discussing racism in these social spaces. Their comments support the literature that there is immense psychological burden to maintain positive workplace relationships and therefore, being socially accepted (Baumeister and Leary, 1995) in their field and habitus. The reluctance to speak about racism assists the conservation of the structures that have become internalised, which inhibits race talk (Bourdieu, 1977, 1998). Layder (1993) expressed that the self and situated activity were not elements that could be easily disassociated and with these participants' accounts it is easy to understand why this may be the case. These individuals are invested in their fields, social spaces and the resources that might be gained from these social interactions. However, the structures that have become produced and reproduced act as inhibitors to their agency because the language of race and racism is not authorised language within the institutional setting (Bourdieu, 1977) and is used to silence the voices of people of colour through symbolic violence (Bourdieu, 2001b). The institutional structures are so powerful that people of colour will not or cannot feel as though they can raise negative workplace experiences because they are being constantly invalidated through denial (Augustinos and Every, 2010; DiAngelo, 2018) and misrecognition (Kamoche et al, 2014) and this perpetuates the myth that racism does not exist within UK higher education.

In the main, participants felt that it was often difficult to prove that incidents might be racially motivated and that it would be career damaging to pursue a claim of race discrimination or race-related bullying and harassment. One academic stated very clearly why she would not attempt to 'rock the boat' and these sentiments were shared by other participants:

...we're not rocking the boat enough. But many of us are scared. We're scared because it's our jobs, it's our livelihoods. You know, when you've vested your career in something to rock that boat means you've got to be prepared to face some consequences. I think because of the current employment climate that we're in, rocking of the boat is less of an option for many people...and you know what it's like as well, you know higher education is a bit incestuous. People know each other, people talk to each other, so if you're going to rock that boat hard

enough you better know that you're going to stack shelves in Sainsbury's or something. Higher education is out of it for you. So, unless you're going to go into your own consultancy, or you're going to go abroad or what have you, it's career change time if you rock that boat. So, I think it's a survival tactic to put up and shut up.

Participant 16, black female academic, Post-1992

One participant, who is a local Trade Union representative within her institution spoke of her unease and frustration in dealing with prospective cases that may have a racial element to them:

Even in the union, as I said, you know, we talk about racism and we talk about what you should and shouldn't do, often I'm very careful. Now, within the union we have a section where if a member of staff comes and says, 'right, I've been racially abused blah blah' or whatever, we refer straight to [the] solicitors and they will look at the case and make a decision.

Participant 17, black female professional, Post-1992

When asked to talk about how that made her feel the participant added the following:

[Sighs] There's a bit of anger because it's like it's a no-win situation. Well...personally, I just think you'd be targeted, you'll be known as that's the individual who raised that race case, or whatever. If...you went down that line, you're doomed anyway, that's it, finish. I've seen many colleagues have come to me and raised an issue, they don't go as close to say it's racism, but they touch on it and the majority of the times they've left the university, because life is not easy after that. Because the management gang up, you see.

Participant 17, black female professional, Post-1992

These accounts support the literature asserting that people of colour are reluctant to attribute negative workplace experiences to racism and that individuals will employ strategies to avoid the subject (Pearce, 2019) since making accusations of racism has become taboo (Augoustinos and Every, 2010). Furthermore, it is possible to interpret these avoidance tactics as examples of how new racism (Bonilla-Silva, 2015) is manifesting itself in UK higher education institutions.

Those academic members of staff who felt they could not report or complain about negative workplace experiences, perceived that their journey through higher education would be much easier if it were not tainted by issues relating to their identity. The 'incestuous' nature

of the sector can be interpreted to demonstrate how minority ethnic academics might become silenced if they wished to pursue a long and fruitful career in their chosen field. To contextualise this concern, more than seven out of ten staff who said they had experienced racial harassment said they experienced micro-aggressive acts and demeaning behaviour, the most common perpetrator of racial harassment of academic staff being other academic colleagues (EHRC, 2019) and so it is no wonder that academic staff feel that by reporting racism would in some way jeopardise their career. The sentiments voiced by professional and support staff participants of that study reported that the main perpetrators of racial harassment included other professional and support staff, line managers and students (EHRC, 2019).

From the perspective of equality and diversity practitioners in this study there were some significant concerns raised around the type of advice or guidance that could be provided to staff experiencing negative workplace incidents. There were several comments that highlighted many of the practical implications of advising and/or supporting prospective claims of racism. These included:

...it's very subtle now, so it's very difficult for people to pinpoint it and hence why not many people come forward to say that it is racism...you even second guess yourself to thinking that it's racism. On occasions I've had conversations with people where I've said to them, they've automatically jumped on the 'race card' and I've said, 'no, you really need to sit back and think, because if you keep wearing out the 'race card' it won't have its credibility anymore', and some people have taken that as offence thinking that I don't believe in them. I've said that I do believe that something has happened but don't necessarily have to say that it's that...

Participant 1, black female practitioner, Russell Group

My interpretation of this account is that Participant 1 is cognisant to the problems that might arise from claiming racism within her workplace setting and this may be reflective of the participant's own ethnic background and habitus. Therefore, her own experiences and knowledge may inform the strategies she employs to navigate the system, and as a result, advise others to follow. This is an example of the way structures are reproduced, and although this advice comes from a well-intentioned position, they may maintain the existing structural inhibitors that silence people of colour. The remark concerning the 'race card' is noteworthy because this alludes to the myth that people of colour are prepared to complain of racism all

of the time, which as discussed above has no basis (Pearce, 2019) and has been further supported by participants' comments included in this chapter, representing most participants in this study.

There were other frustrations in dealing with race matters in institutions, particularly in terms of institutional commitment, and included 'distraction by data' and other tactics employed to delay actions, such as gathering further information or the need to produce reports to various institutional stakeholders. One practitioner captured this situation well and used the analogy of a sandwich to describe and recognise the lack of progress and apathy to tackle race equality in her institution:

A bit like a sandwich, OK, bottom layer doesn't care — I'm getting paid — whatever. Middle layer, oh OK, I can put lots of seasoning, lots of flavour, etc., a bit of policy, a bit of change management, etc., hmm, yeah and season it. And then the top layer can be I can have it toasted or sprinkled, but that's if the top layer wants to be involved in the sandwich. You need a top on the sandwich in order to drive it through or to be eaten, kind of, if you like your sandwich with two layers on it. But sometimes what happens is the top layer doesn't commit...So it's OK as moving around and sprinkling and putting gherkins on, whether we like them or whatever, or not, but unless the top layer commits to make a full sandwich you're always going to be in that middle ground...Unless somebody closes down and says, 'this is what we're going to do', you're always going to have that...'oh let's add some more topping' and that's why I figure the reports are, 'let's do another report', 'let's do some fact finding', 'oh, have we got the figures for that?', 'let's do some more number crunching and some more number crunching', 'let's go ask the students', 'let's set up a people of colour focus group and ask them again and again and again and again.

Participant 30, black female practitioner, Post-1992

What Participant 30 describes is reminisce of DiAngelo's (2018) description around white people's discomfort to talk about race and racism and that strategies employed by dominant groups include deflection and may manifest itself through the lack of senior commitment discussed by Tatli, et al (2015). The situations described by this participant completely resonate with my own experiences as an equality and diversity practitioner and are unfortunately commonplace.

From a practical perspective, the challenges of advancing race equality in higher education institutions should not be limited to the actions of lower or middle managers and equality and diversity practitioners. One senior manager spoke candidly about his perception of higher education and the near-impossible task of sanctioning behaviours that are not overt:

It's funny, well it's not really funny, there's a sort of assumption that universities are open, tolerant places where free and open debate takes place...but actually I don't find them like that at all. I find them highly sort of privileged and elitist and exclusive...But it will probably take...some kind of a scandal really to make it an issue. The thing is that most of the time it's not overt...so it's difficult to sanction stuff, which is subterranean, so it's much more about changing culture and changing attitudes.

Participant 29, white male senior manager, Russell Group

Comparing the previous comments made by Participant 30, in terms of the difficulties with engaging management in providing commitment and direction to tackle race, with the comment made by Participant 29, can be interpreted as the challenge of sanctioning subterranean attitudes and behaviours is too great a hurdle to surmount. The latter's feelings of waiting for a scandal can almost be inferred as pushing race matters to the periphery in anticipation that at some point in time a matter would arise that would necessitate urgent and direct action to avoid outrage. A similar view was shared by an equality and diversity practitioner:

The sector is trying, but I'm not sure that there is a firm enough will to do it and I keep saying that I think we probably need a Stephen Lawrence event. I know it sounds horrible, but I think you need something of that magnitude that is going to wake people up and realise we can't carry on the way we are.

Participant 1, black female practitioner, Russell Group

There is something very troubling about these perspectives; from the uppermost levels of the institutional hierarchy there appears to be a lack of responsibility and accountability for addressing long-standing and persistent issues in higher education. From a cynical perspective this could be interpreted as the white saviour entering centre stage when the situation has become untenable on one end of the spectrum, set against a clear frustration among equality and diversity practitioners in the way they see their institutions performing in terms of their

commitment to, and action to change institutional culture and practice – a cry for help. From accounts it appears that there is a lack of collaborative approach to tackling racism; one that is strategically driven by the institution at the highest levels and implemented through managers, with the support of staff groups and equality and diversity practitioners and cascaded across the whole university community.

It is possible that in the move from a collective equal opportunities (pluralist) approach (Geare, et at. 2014) towards the more divisive unitarist approach of diversity management has created an organisational environment, which assigns responsibility on an individual basis rather than taking a collective stance of zero tolerance to racism, which many organisations may espouse within their bullying and harassment policies. A unitarist approach which emphasises common values and objectives (Van Buren, 2020), may create fear for those who claim that there might be conflict and therefore will be less likely to 'rock the boat'. Ironically, society now finds itself in a similar position to where it was following the death of Stephen Lawrence in 1993 and the last few months have been largely occupied with narratives around racism and the Black Lives Matter movement. Time will tell whether the protests of 2020 and the organisational commitments made to tackle race equality during this time will yield the advances promised.

From a CRT perspective, it can be assumed that as a microcosm of society, racism is embedded within UK higher education and racial inequalities are maintained through systems and structures that appear normal and unremarkable (Rollock and Gillborn, 2011). Secondly, from the accounts provided in this section it is clear that white supremacy plays a significant role in maintaining the systems and structures and constrain people of colour to speak up or act against racial inequalities or racism. Amid these barriers, staff of colour continue to experience varying levels of microaggressions, and it is this aspect that will be investigated further in the next section. In exploring the perceptions and experiences of racism within our institutions we will also consider the impact on people of colour as a result of those inappropriate behaviours, and the different coping strategies employed considering the self-imposed and structural barriers to report such experiences.

7.3 Perceptions and experiences of workplace racism

In the previous section we explored some of the challenges that affect minority ethnic staff in UK higher education institutions to identify modern workplace racism. Most participants when asked whether they had experienced racism in their institution answered that they had not. Yet, in almost all interviews those same participants proceeded to talk about wideranging situations and incidents describing diverse microaggressions. This provides evidence of the extent of misunderstanding and misrecognition of modern racism at work, including by those who are the targets of inappropriate behaviours or attitudes. This finding was completely unexpected and contrary to my own assumption that people of colour would be able to identify racism when it happened, would take action to address it, and would report it. This offers a different perspective on the multiple and complex differential outcomes that affect black staff; not only are these outcomes exhibited statistically in terms of black staff representation and pay inequality across our institutions (discussed in Chapter Five), but their lived experiences in the workplace are being exposed through this dissertation as well as previously published literature. Particularly within senior levels, black staff are statistically less likely to be paid at the same rates as their white colleagues and may also be disproportionately disadvantaged by workplace policies. This latter point will be explored further in the chapter.

The literature highlights that racially minoritised staff in UK higher education are less likely to be recruited (Pilkington, 2011), are more likely to experience microaggressions and be isolated, ignored, racially stereotyped and micro-managed (ECU, 2011; Rollock, 2011; Pilkington, 2013; UCU, 2016, 2017). As people of colour attain greater social status, more education and higher incomes the prevalence of racial microaggressions does not lessen (De Cuir-Gunby and Gunby, 2016). People of colour also experience microinvalidations and microinsults, which include behaviours that deny the importance of race, or which convey the myth of societal meritocracy. Microinsults are actions that convey insensitivity or rudeness, or directly demean a person's ethnic heritage. Perpetrators of microaggressions are often unaware of their behaviour (Sue et al, 2007; 2008) and an example of this is provided by Participant 13 earlier in this chapter. These examples have all been experienced in varying degrees by many of the participants interviewed for this study.

The previous section discussed the difficulties ethnic minority participants encounter within the institutional social space around matters of race and racism, particularly in explicitly naming racism for fear of upsetting their colleagues and becoming marked as a troublemaker, which could be career limiting. Participants talked about their attempts to downplay their experiences (Harries, 2014) and this presents racism as highly unusual and exceptional (Augoustinos and Every, 2010). In exploring with participants whether accusations of racism are taboo and if experiences of racism were hidden for fear of victim status, it became clear from minority ethnic participants that this is the case. This was exacerbated by an inability to recognise their experiences as racially charged or motivated. Many participants who had grown up in the UK, recounted stories of their youth and of being on the receiving end of overt racist abuse, one participant stating that at least they knew where they stood with such comments. Another participant shared a childhood experience that has informed how she now deals with negative race-related experiences:

I think it's probably from my childhood, if I went home and said, 'mummy, the neighbours are calling me a gollywog', and she'll say, 'well, what did you do about it?' 'Well, I cried.' 'Why are you crying - are you a gollywog?' 'No'. Well...my mum was like, 'don't talk to me about stuff like that because that's not what you are'. So...if I accept or identify or talk about it, it means I'm showing a weakness, that it's actually affecting me, so it shouldn't affect me, so I won't talk about it.

Participant 18, black female professional, Post-1992

This participant's comments felt full of pain and resentment that to protect herself and not to be perceived as weak by others that she must psychologically detach herself from her own lived experiences. Detachment coping mechanisms such as that mentioned above appeared a common coping strategy mentioned by black participants in this study and are characterised by not wishing to discuss issues they are facing and diminishing their experiences of racial microaggressions (De Cuir-Gunby and Gunby, 2016). Another participant expressed her reticence in using the terms racism or discrimination and found it hard to reconcile why a person would want to be identified in some way to their ethnic background, demonstrating that there are multiple and complex reasons that may affect a person's willingness to acknowledge negative workplace experiences especially when it was linked to this aspect of identity:

I wouldn't call it racism. I don't want to use the word discrimination either or maybe non-identity, maybe, not being, like so you don't exist sometimes. I do not know what term to use. I find racism a very strong term to use. The word racism, it is always so negative, why would people want to be recognised by your skin colour or your ethnicity...

Participant 9, Asian female academic, Post-1992

The feedback provided by Participant 9 and other participants cited in the previous section is a frequent finding in that while an individual might readily accept or acknowledge that they have experienced negative or unequal treatment, people are often reluctant to attribute this to racism (Pearce, 2019). To support this, some participants had expressed that they did not wish to isolate themselves further by making these types of accusations and once again, demonstrates that participants use their agency but are constrained in deciding which strategies will enable them to navigate their institutional terrain (Giddens, 1984; Bourdieu, 1977).

In exploring the different lived experiences of microaggressions among the minority ethnic participants, one participant talked about her experiences of attending committees within her university and the strategies she employs to ensure that she is heard in that forum:

I've had to before sit in front of the chair [of the Committee] to make sure he doesn't ignore me...I've also had the opposite where I'm trying to say something...and I'm there putting my hand up like I'm at school...

Participant 7, black female senior professional, Russell Group

Taken in isolation, this instance may not indicate that there is any sinister behaviour occurring within this setting. The participant had felt frustrated at being ignored over time and therefore, as a result needed to find a solution that would elevate her position, even in a literal sense, to become more visible and therefore, more included within that group. Viewing this as an isolated incident may also be part of the problem ethnic minority staff face in being able to articulate negative workplace experiences because of the very nature of microaggressions (Rowe, 1990; Sue et al, 2007). A similar account was provided in a recent study of the experiences of UK black female professors, where an account from a participant stated that she had her hand up in a meeting for half an hour before being 'allowed' to speak by a senior white male member of staff (Rollock, 2019).

The experiences above could be interpreted as examples of symbolic violence (Bourdieu, 1998; 2001a) because these actions are not specifically overt in terms of a physical act. The power brandished by the Chair of the committee in restricting contributions forces the participant to keep her hand up for a prolonged period of time, in full view of her peers, enforcing a state of powerlessness, and symbolically less valuable. There is a gendered power dynamic within this scenario, which compounded with the participant's ethnicity creates a situation where the participant is required to develop strategies that can contend with the symbolic power wielded to support the dominant structures (Bourdieu, 2001a). This act could also be interpreted as one that reinforces behaviours found within a school classroom setting, by infantilising the individual in a way that might not be imposed on other (white) colleagues. Not giving permission to speak in a way that might be available to white colleagues may normalise this behaviour across the peer group over time. In turn, this might impact the participant's ability to develop or access further opportunities to build capital within her organisation. Bourdieu (1991) stated that capital partly defines the degree to which a person is accepted, which is relevant here, making the receiver complicit in accepting the rules of engagement, which have been set by the dominant group. The participant has taken some remedial action to symbolically show that she is not accepting the behaviour by moving to a more prominent position, however in doing so the participant is taking on an extra burden to prepare for these meetings, not just on content, but psychologically to contend with the culture within this environment. This same participant went on to add that:

...when I think about it, I've been in meetings and because of the profile of the committees that I sit on sometimes I'm the only black person there, and I'm very tired and I don't want to say anything and I thought but you can't leave the meeting and you haven't said a word because then you'll be stereotyped as that black woman that doesn't have nothing [sic] to say, and I've got to find something to say...I feel I have to have a voice.

Participant 7, black female senior professional, Russell Group

In this account participant 7 indicates how visible she becomes being the only person of colour in the room and with a perception of raised expectations that she is not only representing her part of the business within the institution but is expected to represent her whole race (Solanke, 2018) and therefore feels she must perform for fear of adverse peer assessment. Something that DiAngelo (2018) argues is not expected of white people. Kanter (1993)

described token subjects experiencing feelings of isolation and ostracism. Their increased visibility in the workplace can further produce anxiety regarding what is expected from them and their work performance. As well as increased visibility, Kanter (1993) identified that token subjects are also likely to encounter boundary heightening and stereotyped portrayals (Kanter, 1993; Smith, 2013). Consequently, experiences like these will intensify feelings of anxiety and isolation that Kanter (1993) identified. An example given by one participant demonstrated that a person is often unable to know how boundaries have been heightened and, in this situation, it was not until another colleague brought the issue to their attention that the participant could consider the reason she was unable to gain promotion, which resonates somewhat with the point made by Healy et al (2011) in relation to what they argued were the 'shifting sands' of rationalisation for the lack of women's progression which are reflected in the gendered and racialised order of organisations:

[an Italian colleague] turned [a]round one day to me and he's the first one who mentioned it, he said, 'actually, don't you realise these people are racist?' I never thought of it like that and that's how either honest I am or naïve I am. And when I broke it down, I thought actually yes, why can't I get promotion, I've been top of my banding since October 2011. I've had a job review, if people really wanted to fight for me, they would...I've done above and beyond in my job, why can't I get rewarded for that? I don't want to hear, 'oh well you don't meet the criteria', OK, make me reach the criteria because I've put myself above and beyond...

Participant 25, black female professional, Russell Group

Another participant provided an example of ways in which her visibility is increased because of passive aggressive techniques employed by others to highlight an issue, in this case by copying in a senior colleague to emails:

...I think what really upset me was that she actually put the head of department, copied them there, so that I look like I didn't know what I was doing.

Participant 21, black female academic, Post-1992

Like most participants who described examples of microaggressions, there was very infrequent reporting, either to call out the behaviour that was deemed inappropriate with the perpetrator or through more formal mechanisms. Examples of the reasons for not wishing to speak about racism or reporting incidents have already been provided in the previous section,

and those reasons are understandable due to the perceived scale of the consequences that could be faced. However, one of the unexpected consequences of not reporting is that it could further serve to entrench the unconscious nature of symbolically violent acts and would perpetuate the cycle of microaggressions that could become normalised within the institution (Johnson et al, 2008; Randle et al, 2014). On raising this point with participants, there was a general acknowledgment that this was possible, however perceptions strongly indicated that the stakes were just too high to risk reporting. Moreover, racial microaggressions can leave people of colour feeling subordinate to their white counterparts, which may result in a passive and potentially destructive form of coping where their plight remains hidden.

The EHRC's (2019) report into racial harassment in UK higher education found that more than half of staff respondents did not report racial harassment. Respondents reported having no confidence that the institution would address the matter, staff did not know how to report, staff could not judge whether the complaint was serious enough and found it difficult to prove what had happened. Others also feared the consequences of reporting, such as a negative consequence to career prospects and being seen as a troublemaker (EHRC, 2019). Worryingly, these actions pass largely unnoticed despite the perception that an experience might be related to race, and which are difficult to deal with by line managers and HR professionals because of their 'subterranean' nature.

From the thirty-one participants interviewed, only one participant shared an experience which described a serious overt racist incident in his institution. The perpetrators were a group of students and the acts included imitating the participant inappropriately by 'blacking up' during a social event. The participant found the incident extremely upsetting and due to complaints made by third parties the matter caused a significant confrontation within the institution, and which required an official investigation. Although the incident and aftermath were deeply shocking, the participant decided to stay in the institution and address the issues that had arisen as a consequence of the formal complaint. The participant wished to be involved in developing solutions that would advance race equality:

...I wanted to stay here and actually be part of the process to look at racism, possible racist practices and basically see if I can be involved in highlighting and implementing any changes...

Participant 26, black male academic, Russell Group

The participant is socialised and established in his field as an academic within his institution. He had felt that because of this he had gained various forms of capital throughout his career, and as such felt that there was a role to play in addressing the issues that had developed in the workplace. It appeared that the participant has internalised the external structures, together with the expectations for operating in that environment, however, he found it difficult to truly progress within the institution because of the structures that are in place. The participant feels he has agency in this situation and understands his habitus to meaningfully contribute. Despite feeling sufficiently empowered to act in response to the incident, he expressed that his overall experience in the institution resulted in a lack of overall capital within his department because he could not gain progression as he had hoped. His feelings around the complaint provided him with a different perspective on how he could become useful in supporting the university and use this opportunity to gain further agency and capital, as he felt that there was scope for the co-creation of strategies to address race equality.

This situation places the participant in an interesting position; a person who has been the target of racism assisting the institution as an agent of change. Additionally, this participant had commented on his identity as a Russell Group academic, which can be perceived as carrying higher symbolic status in UK higher education, and that he was now at odds with the way he has traditionally seen himself within an institutional environment, and which had been until the incident, devoid of ethnicity. This has been contrary to how he feels others perceive him:

...through the majority of my career, I've always wanted to be colour neutral, to be seen for my science background, not to be separated off into a particular group of my colour because that's what I've always sort of strived for. Of course, people I work with, the powers that be, haven't been able to make that distinction...over the years I've always partly sort of ignored my identity in order to fit in within this environment. There's a part of this what's happening within the School this year, I've been asked why have I always been so careful in the things that I do – well I've always tried to do things correctly, what I do in my job, that I don't stand out too much. I stand out anyway but psychologically, mentally, I've always tried to not stand out. I think I've always strived to follow the rules, to do things correctly, to work within structures, not be a maverick and maybe I should have been more of a maverick. Stand out a lot more and actually pushed and jumped over barriers more aggressively...I leave the so-called blackness behind when I come into my working environment.

The participant's comments appear to suggest that he has been left with no other alternative but to face up to a part of his identity that he has psychologically suppressed throughout most of his career. His reference to trying not to stand out can be likened to the advice given by Participant 25 to her son (Chapter Six), that as a black male he will attract negative racial stereotypes. The incident was the catalyst for this shift in self-perception and now feels that he should use his full identity as a *black* scientist to advance matters of race within the institution. However, in now acknowledging his blackness he felt that this could negatively impact upon his ability to function as an academic as well as take certain actions that could promote ethnic diversity. He expressed that this would be a difficult decision for him to make considering that there are few academics of colour in his institution, yet he now felt compelled to stand up and be counted as a black academic, although this is the way he has always been perceived by others throughout his time in the institution.

This example demonstrates the adverse consequences of being forced to tackle racism, either directly experienced on a one-to-one basis, or in the case of Participant 26 an indirect target of racism where the act in question involved a negative portrayal of him to others. An assumption might be made that all people of colour feel that they can be their authentic self in the workplace, whatever that might involve. However, this participant has by all accounts attempted to create another self, an alter ego, to ensure that he is able to navigate the system with the least friction or challenge. The participant has therefore been confronted with the need to re-establish his identity physically, psychologically, and professionally to forge a new beginning within his field, his team, and the wider institution.

The experiences of Participant 26 were the most extreme and overt example of racism identified through discussions with participants in this study. There is however a shared outcome from the examples given and despite the majority illustrating different forms of concealed racism and microaggressions. These accounts have all highlighted how the systems and processes that are in place to deal with staff and student conduct, grievance and disciplinary procedures, create challenges for those needing to engage with them in order to deal with complaints of racial discrimination, bullying and harassment in the workplace. It is these systems that will be explored further in the following section to understand whether institutional policies, processes and practices help or hinder the capability of organisations to

deal effectively with workplace racism of any sort, the way those systems and processes are applied and how they impact on ethnic minority staff in UK higher education institutions.

This section has explored the perceptions and experiences from participants of racism within their respective institutions. These accounts provide further evidence that not only do overt acts of racism continue within the UK higher education sector, but there are multiple trivial acts or omissions that amount to microaggressions. The following section will explore another form of racism, but which is perpetrated not by individuals per se, but through the macro structures within institutions.

7.4 Perceptions of institutional racism

In considering the complexities of institutional racism and the continuing reluctance within the UK higher education sector to address it in practice the research findings will begin to explore perceptions of institutional racism within UK higher education from a diverse range of participants. These findings will provide an insight into the ways that institutional racism is perceived to manifest itself within the UK higher education sector and the impact it may have on the outcomes for black staff in our institutions.

The literature has revealed that, since the last iteration of institutional racism in the UK through the Macpherson report (1999), there has been limited research in this area, particularly with a focus on higher education. However, those academics that have contributed to the debate (Wight, 2003; Gillborn, 2008, Pilkington, 2011; Ahmed, 2012; Sian, 2019) have commented on the way the definition attempts to condemn the conduct, attitudes and behaviours of individual social actors as well as placing the organisation central to the debate where processes implemented by an organisation may have disadvantageous effects on ethnic minorities. Placing higher accountability on organisations ensures that the reproduction of racism is considered since racist discourse can become embedded within institutional processes through practice (Pilkington, 2011).

Macpherson (1999) defined unwitting racism as arising from a lack of understanding, ignorance or mistaken beliefs together with well-intentioned but patronising words or actions. It can also arise from racist stereotyping of black people as potential troublemakers. Chapter Six and earlier sections of this chapter outlined participants' lived experiences in their

institutions where they described such situations. Macpherson (1999) added that such attitudes can thrive in a tightly knit community, such as a university, where there can be a collective failure to detect and outlaw this class of racism.

For the purpose of this study, the definition used within Macpherson (1999) has been used to ensure that participants were clear about the terminology used within that definition to facilitate discussion. This definition is as follows:

The collective failure of an organisation to provide an appropriate and professional service to people because of their colour, culture, or ethnic origin. It can be seen or detected in processes, attitudes and behaviour which amount to discrimination through unwitting prejudice, ignorance, thoughtlessness and racist stereotyping which disadvantage minority ethnic people (Macpherson, 1999:28).

To understand whether institutional racism was perceived to be present in UK higher education institutions participants were asked to read the abstract above during their interview and asked to provide their perspective on this in terms of their current and/or previous institution, and where relevant, the higher education sector as a whole. Participants took the time to read through the definition and reflect on whether they believed that their institutional practice (or sector where appropriate) corresponded with the definition given and to describe how it manifested itself. The following example set the scene well in the context of higher education:

Well, if we break down the quote, the collective failure of an organisation I'll change to collective failure of the sector to provide appropriate professional services to people. We know, I mean the degree attainment gap is just a classic, that's the failure of the sector to provide an appropriate professional service to people based on their ethnicity or race. Detected in processes, attitudes and behaviour. I think if you look at all of the existing research and current research coming out on the way that BME academics and students feel it's often about the attitudes of their institution and the behaviour of colleagues. Unfair processes, goal posts moving, particularly in promotions. I'd say it fits this definition entirely. I suppose the bit that's up for debate is whether it's unwitting. Is it unwitting prejudice, ignorance, thoughtlessness? If it is, you can change unwitting prejudice, you can certainly change thoughtlessness and ignorance because you can educate people. I suppose there's always the debate about conscious/unconscious bias.

Another participant who is a senior leader within his institution commented on the difficulties the terminology presented to institutions, particularly in terms of understanding its meaning and translating this in a way that could be operationalised. Despite saying this, he agreed that institutional racism was present in UK universities and would not be able to defend an accusation of this:

Well, I think that being much clearer about the terms would actually help; I mean I will probably get onto all of this but my feeling in universities has been that, essentially universities are institutionally racist, institutionally sexist as well. It's certainly all of the universities that I've operated in and...I take it and assume that it's the case in all UK universities. And...you just do not see an ethnically diverse population amongst the senior management teams of UK universities... I still think...if somebody asked me to defend universities against the charge of being institutionally racist, I couldn't do it. I'd find that very difficult. I wouldn't be able to get up and defend that view publicly very easily in a sort of water-tight way. I could put my arguments that's all. And I don't want to be unfair to people either...maybe I'm wrong, that's possible, and also would it create the wrong impression, I don't know. Maybe it's better to think of other ways of putting this than them being as blunt as that? On the other hand, I might be wrong, maybe that is the only way of doing it.

Participant 29, white male senior manager, Russell Group

The comment made by Participant 29 reflected Solomos (1999) who identified that there were no clearly defined actions within the description of institutional racism in Macpherson (1999) and echoes Souhami (2014), who observed that there lacked new development to tackle institutional racism in the police service at conceptual or practical level, attributing this to a lack of understanding of how it might work as an instrument of change. There have since been a number of scholars that have offered further insight to the contrary in this area (Gillborn, 2008; Ahmed, 2012). Despite conceptual and practical misconceptions around the notion of institutional racism there is scope to assume that this is a systemic problem in UK higher education following the remarks from this senior-level participant. Disappointingly, when asked to expand on what actions could be taken to address institutional racial inequalities, Participant 29 stated that money could be used as an incentive because people

could not be forced to do things, therefore alternative motivation would be required. He continued by saying the following:

...it's very difficult to apply sanctions that would affect you academically for something that is not an academic matter. So, what else do you do, you know? We don't really have many sticks to beat people with and it's not always a good idea to do that anyway. The way to give it real teeth is to say if you get this race equality mark, or if you don't have it, there are certain things you can't apply for or you don't have access to, and you'd have to think really carefully about what that is. But it could either be money that's already there that presently have access to that they would be denied or something new that you put in place that is there for that purpose.

Participant 29, white male senior manager, Russell Group

If this is in fact the case, and institutional racism is systemic in UK higher education institutions, then understanding why institutions have not raised consciousness around the topic is necessary. The Metropolitan Police Service was at the centre of the Stephen Lawrence inquiry and subjected to the recommendations made through the Macpherson report (1999). However, the nature of the public sector and its service user population should have sparked a review of all public services as a result. The comments made by Participant 23 relating to the differential outcomes for black student attainment is a primary example and one that has been well documented over the last decade in UK higher education (Connell-Smith and Hubble, 2018; Bhopal, 2018; Turner, 2019).

The literature review in Chapter Two identifies that race and racism is a difficult topic to address and a reticence to tackle racism appears to include the rejection of the notion of institutional racism, reflected through the absence of activity in this area. Through denial, institutions need not address it. The development of theoretical understanding by scholars and antiracist activists do not appear to have informed the sector's ability to address institutional racism from within and the topic has been largely omitted in discussions around race at an institutional level. Most recently there has been a movement to decolonise the curriculum in institutions, such as the School of Oriental and African Studies (SOAS), University of Oxford, University College London, etc, nevertheless this has been an endeavour that has been student-led and student-facing rather than in relation to the workforce, which appears to be silent. For example, there has not been a parallel ethnic diversification of the workforce to support this move and so this exercise might be viewed cynically by critics. One participant

offered their view on the difficulties associated with the UK higher education sector exposing itself to the possibility that its systems and processes might be inequitable:

...I think there is a sense, especially when we're coming in talking about institutional racism because it's not a 'you think this way, you are prejudiced' etc. it is built into the structures and the processes. What we're asking people to do is challenge the very thing they do. They've decided they want to be in academia, they've gone through the system, they understand the system, they've kind of been 'Stockholm Syndrome'd' into the system and it puts you in a very vulnerable position to go, 'is this system right and is this system effective?' So, it is through our access activities, through our admissions activities, it is the way we learn and teach. It is the fact that we do have a dominant white curriculum that we...and even pedagogically, it's a very western wealthy way of teaching...so collective failure of an organisation to provide appropriate and professional service because of their colour, culture or ethnic origin...we've created a sector...that is based on the needs and wants of the white middle class to upper class society...it is a collective failure but it's a collective failure because it's a collective ignorance. I think you can see that in things...from the outright discrimination to the fear and the worry and the concern. I also think thoughtlessness is a really good term there for both the sector and the institution generally, and thoughtlessness can sometimes...come across as quite a negative term, but again it's just that if you don't know and if you haven't thought about it, then it's not going to translate. So, I think that when you've got an environment that is flat out, and it is institutionally racist, you're not welcoming, you're not supporting your students to be as engaged or to reach the potential that they have, and I think you're also doing a disadvantage to the majority students because you're not giving them that broadening experience that kind of enrich[es] their learning. You're not actually encouraging to do what we ask our students to do, which is think critically and experience...and be open to new things.

Participant 19, white female practitioner, Sector agency

Macpherson (1999) emphasised that it was not the policies of the Metropolitan Police Service that were racist, but that it was in the implementation of policies and the actions of officers acting together that racism became apparent. Pilkington (2011) observed in his case study that a university was reluctant to accept a race-related policy because it was perceived as political correctness gone mad. Although Pilkington (2011) referred to a race equality policy, the account given by the following participant highlights that it is the manner that

organisational policies are implemented that can conceal the nature of the complaint, particularly if the matter was related to race:

I know it's an instance of a [white] student's abuse of a [black] member of staff. Now...what was interesting was the institution's reaction towards it. It tried to go down a conciliation route instead and understand that the student is a good student...and so it is a dignity at work issue, which seemed to...be a certain type of thoughtlessness, which I would argue was institutionally racist. So, this wasn't a dignity at work issue it was a racism issue...the point is it still responded very badly to any insistence that it still has a bad policy. The thing is if it happens again, even though they've tried to change the policy, there is no guarantee that they're going to treat it any better than the last time, which is kind of sticking your fingers in your ears and saying, 'la, la, la' a lot. It's a refusal to apologise, it's a problem, rather than an isolated incident. It could be a continual problem...I think it's in the institution's best interest to not acknowledge it, you know, otherwise they're going to be known as that institutionally racist institution. Rather than that institution that solved the racist problem. So, it's in their interest to minimise...or push to the periphery any instances of racism.

Participant 6, Black male academic, Research institute

One participant, who is an HR practitioner, provided insight from a perspective of proximity to the development and implementation of institutional policy. Her feeling was that by default she would not accept that institutional racism existed, but acknowledged that policy could be implemented according to 'local practice' and that this would need to be addressed if this were the case:

Maybe it's my role and obviously if I was in another role maybe people would feel more comfortable, but because I'm in HR, I think that just by default I just wouldn't [agree that institutional racism existed in the institution]. Looking at our policies, making sure that our policies are compliant with the law, because the law is there to protect us. Making sure that there are local practices to make sure that that is stamped out because the policies and procedures are there for a reason to ensure fairness. Now, if everybody has got their own local thing going on that's where you get this, so it's to make sure that everybody knows that...the university's rules and regulations, this is there to protect the university from challenge, so we need to follow that and we need to follow it consistently.

Participant 15, black female HR practitioner, Post-1992

The participant viewed this issue as one of compliance, which may highlight a misrecognition of institutional racism at an organisational level and how policy might be interpreted by managers and staff, including those within Human Resources. This is a fundamental problem if institutions are to eradicate institutional racism. These comments may imply that institutional policies are not detailed enough or leave too much room for interpretation by those who are tasked with implementation. Alternatively, Noon et al (2013) argued that the hyper-formalisation of policy can result in undermining the original purpose of the procedure, which can result in the 'circumvention of compliance' through robotic, defensive and at times, malicious use of procedures as a means to an end. Reflecting on this conversation, one could question how it would be possible for Human Resources to become aware of a policy being implemented in subtly different ways that could result in differential outcomes for a person of colour, particularly where participants are less likely to complain about racism in this study and within the literature (Chapter Two). This indicates that Human Resources teams must do more to ensure that policies and practices are applied properly, consistently and without bias, as well as guarding against what Noon et al (2013) stated was the fetishisation of procedures that are perceived as good, fair, thorough and necessary. One participant commented that on a conceptual level her institution did understand the meaning of institutional racism, and suggested that the lack of acknowledgement may be attributable to a denial of racism:

I think they have the intellect to understand what it means on a conceptual level but possibly that they think it's no longer an issue or it doesn't affect them because they think that...doesn't really affect them because they're not prejudiced or racist.

Participant 4, Asian female academic, Russell Group

Another participant stated that she felt that the lack of institutional understanding may derive from the lack of ethnic and cultural sensitivity among senior managers:

...I don't think that as an institution [institution name] has enough, potentially enough experience with understanding people of colour and what their needs and what their issues might be to be able to potentially avoid that...I think it's not just whether you're white or not but also depending on your experience...I think most of the people at senior management that I know, but I mean I could be completely wrong, but I think they don't even have experience. So, they are white but also, they don't necessarily have any experience that involves working with other kinds of group cultures.

The account given by Participant 14 concurs with the comments made by Participant 29 (page 190) around assumptions senior leaders make, which result in unwelcoming environments. The suggested lack of conceptual and practical understanding ties in with Wight's (2003) point in terms around lack of clarity of cause and effect, and structure and agency from the definitions of institutional racism given in both statements from Scarman (1981) and Macpherson (1999). His point being that in both, although they refer to structural racism (institutional) the cause is linked to what he describes as methodological individualism, such as the unitarist approach of diversity management, which emerges and is reproduced within a structural context, however the behaviours portrayed in these definitions are purely agential. Wight (2003) suggests that Scarman's definition is at the level of agents, that Macpherson's definition is at the level of habitus (Bourdieu, 1977), and the structural level will be considered within this chapter. On this latter aspect, he suggests that overt racism is a form of practice that intends racist outcomes and unwitting racism is that which has racist outcomes but where none were intended. To contextualise this and explore these concepts further, one participant gave a very personal account of ongoing issues she was experiencing in gaining support from her department for her race-related doctoral research and began to consider this experience through the lens of institutional racism:

...firstly, there's a collective failure of an organisation to provide an appropriate and professional service. So, going back to my situation, the appropriateness and professional-ness [sic] of my whole PhD application. Utilising the systems that this organisation has in place, or the department, at least has in place, which are poor...because the systems are crystal clear and keep changing every flipping year, those systems then can be changed. Now again, whether it's overtly or covertly is another matter, but they definitely changed to disadvantage. There were four people on that panel. I go to the meeting with the panel and the whole panel should be there. The whole panel's not there, it's just the Dean and the Head of School scolding me. Basically. They don't do that for everybody. 'We would do this for anybody.' That's a lie, because if that's the case, all four members of the panel would be there and...all four would have that conversation with me. That was not the case. Now, if the panel can be changed at will, where is the consistency? Because of my colour, culture and ethnic origin, I would say all three. Again, because of my area of enquiry was completely and totally dismissed...when I did go back to the second round of funding for that meeting, my manager and senior manager said, 'I know it's

difficult [name]'...so one level of management is doing...something different to the other level of management and at no point has the senior management gone to the management level for me and told them that what they have done was incorrect. All blame was put on to me and at no point, even now, has my manager or his manager ever had any question asked about the advice that I followed...That's institutional racism right there.

Participant 20, black female academic, Post-1992

From the above scenario it is not possible to ascertain intentionality despite there being a specific policy in place stating that a panel would be convened, and in theory should be applied equally to all individuals. The outcomes for Participant 20 reflect how informal processes dominated the formal procedure and resulted in distortion and unfairness yet was positioned to the participant as a legitimate decision (Noon et al, 2013). This account described that the policy was not implemented as published. Applying Wight's (2003) theory here might indicate that the panel's unwitting racism, manifested through microaggressions, had racist outcomes. However, this is problematic since there were two individuals making a decision at this panel and it could be argued that it would be too easy for the individual panel members to claim lack of intentionality despite racist outcomes. Taking this stance allows the individuals making the decisions in this instance and the institution not having to be accountable for this participant's outcome leaving no case to answer if the participant had chosen to challenge the decision.

This is a good example of how policies might be manipulated in a thoughtless manner and without considering that outcomes may disadvantage people of colour because of the way they are implemented. From a structural perspective, the policy is an objective structure and if it were applied equally to all staff in the same way, it would be fair and transparent. However, the way in which the policy has been implemented, which from this account was not as written, has been used in a form of symbolic violence (Bourdieu, 2001a) against the participant with disadvantageous effect.

In addition to this, the decision makers in this instance were the Dean and the Head of School, which provides an added element of objective structure, that is the institutional hierarchy. This is a manifestation within the social space, where the participant is socialised to accept the hierarchy within the habitus and field in which she operates, the distribution of power or capitals which define the rules and resources (Bourdieu, 1977; 1991). These decision makers

thought it was appropriate to continue with the panel meeting without meeting the requirements of the policy and must have felt empowered to do so on behalf of the institution. Their actions demonstrate that they are using their symbolic capital (Bourdieu, 1998) to produce, and potentially reproduce, structures in a way that suit their needs without fear of reprisals.

It is plainly important for an organisation to understand how as an entity it can affect the outcomes of the workforce by the way it constructs and implements its policies, services and practices. Institutional racism is that which, covertly or overtly, resides in the policies, procedures, operations and culture producing certain kinds of practice (Wight, 2003). Wight (2003) appears to contradict his point relating to human agency by stating that it requires, as a condition of possibility for action, materials that make action possible and what makes those actions possible is the structural context that is inhabited and inherited from the past, concurring with Bourdieu's (1977) notion of habitus. This also corresponds well with the notion of playing the game to gain capital. There is a complexity here that constructs and reconstructs workplace culture and practice and participants in the game are not necessarily able to see that practice might be racist, but through practice they acquiesce to and grant permission to others to replicate behaviours. With this in mind, there is scope to establish what is meant by 'institution' in terms of accountability and this should be aligned in the same way as accountability for the PSED, which lies with the institutional governing bodies and senior leaders.

The comments made by participants communicate a perception that often the institutional culture and subsequent behaviours, however unwitting, continue to disadvantage ethnic minority people. Bourdieu (1977) stated that each agent, wittingly or unwittingly produces and reproduces objective meaning and because subjects do not know what they are doing, what they do has more meaning than they know. Observations from participants' perceptions of the existence of institutional racism show that on considering Macpherson's (1999) definition, a significant majority of participants agreed that it was present in multiple forms within their institutions. There were no overt accusations from participants that their institutions were intentionally racist, however there were many suggestions that institutions applied policies and practices in ways that could and did affect the outcomes of people of colour in an inequitable manner:

...I suppose you know just looking at this definition in its raw form and the first thing is you know it's a collective thing OK. So, racism cannot thrive, in my opinion, if just one person is doing it. There have to be a number of other people that assent to...that way of thinking. So, what we have in the institution is a whole group of people if you like, a group in the widest sense that ascribe to these kinds of practices...So, that's a collective kind of thing so everybody, whether they are outwardly or inwardly endorsing that and the fact that people see these things happening...discrimination by commission or omission, so you're either doing it or you're seeing it and you're not necessarily saying anything about it because either it doesn't affect you or you're too scared for yourself.

Participant 16, black female academic, post-1992

On the point of unwitting production or reproduction of behaviours, one participant discussed her personal experience of the grievance process in her institution and how the system is developed to deter claims of racism:

You know it's going on...you must be friends with the person doing it or going in the pub and chatting about it, you know, you're just as guilty as the person doing it...if not worse because you're hiding it. You see what I mean? So, people have got to stop this nonsense about, 'oh well I didn't know', like I said, 'I didn't realise [name] why you were off sick'. Was it you? Own it. And people don't want to own anything anymore.

Out of sight, out of mind...or HR don't want to deal with it because it's too much paperwork. Like in my case, let's pay her off, she'll go. Well, I ain't [sic] going nowhere! Why should I leave a perfectly well-paid job with a team which 99% I actually enjoy working with just so that people can think that they could treat me like crap — it's not happening my friend. And that's why...I had to get an employment lawyer. Shouldn't have to do it.

I think sometimes the grievance procedure's there just to put you off doing anything about it. Because it's a bit like the Police investigating the Police, they're all friends, they're all HR, they're all management. So, you know, God, and then you fill out the form and then there's something else, I have a colleague who's Indian who actually left, and she went the grievance route, she still didn't get any help anyway. In the end, she left. But then I thought hell, no, I want every day you look at me in my face, you know don't play with [name]. And I got myself a black lawyer too, because you have to be taught it's not the one. If you could come and you say OK, these are the reasons why, but you did covert nastiness behind my back and...then smiling in my face. Really? OK.

When asked to provide an explanation of why she thought these practices were allowed to continue the participant offered the following around her perception of what information is reported to senior leadership about the way institutions operate:

...because of course the top person is in their glass ceiling, so they don't know what's going on – you understand me? But perhaps if the top knew what was going on they'd be thinking...because the top is supposed to be the figurehead and representing everybody else, there is a saying that I've heard, 'if the head of the stream is dirty the whole stream is dirty' – they need to be held accountable and it shows that if your team at the bottom are not doing their jobs and covertly hiding stuff, you're just as guilty too – I'm sorry! Because when you go out there and you're promoting your place and you're saying 'oh we're this fantastic organisation' or so on...you're going under falsehoods because actually they're not.

Participant 23, black female professional, Russell Group

The negative perceptions from participants about this topic demonstrate that there is real concern about how higher education institutions provide an appropriate and professional service to people of colour. The accounts given by participants provide an insight into the operational life within institutions that have not been offered previously, particularly in the manner policies, services and practices are implemented in a way that might disadvantage black staff. These accounts and their interpretation propose that much more attention is required to tackle bias and address negative attitudes and behaviours that can be meted out through institutional policy and practice.

7.5 Conclusion

This chapter has discussed the complex nature of racism in contemporary UK higher education. The empirical data shows that racism is not always what society traditionally understands it to be – overt, and there is still evidence here that overt acts of racism continue in institutions. Despite this, racism has evolved to become more concealed within acts or omissions and are far more commonplace within the institutional social space. Racism has become so hidden that it is difficult to recognise and therefore, as participants have identified, it is more complicated to prove.

This chapter sought to understand whether the changing face of racism has made it challenging for individuals and institutions to tackle and the testimonies provided by participants in this study have confirmed that it is extremely challenging, both on a personal and professional level. Participants suggest that race in our institutions is still too difficult a subject to handle and that there is a lack of capability to articulate effectively what it looks like from an individual and collective perspective. This is regardless of the university's mission statement and geographical location and exposes an inability in UK higher education to confront the effects of racism, whether overt or otherwise, across the workforce. Participants have highlighted how the new racism (Bonilla-Silva, 2015) this chapter has discussed is hidden in discreet acts, microaggressions, microinvalidations and microinsults (Rowe, 1990; ECU, 2011; Rollock, 2011, 2012; Pilkington, 2013; UCU, 2016, 2017). The actions are so small and inconsequential that ethnically minoritised staff are unable to always recognise them, yet cumulatively their impact is immense. If the targets of such acts are misrecognising these, there is no wonder that white staff, especially those who can act as change agents within institutions, are unable and/or unwilling to acknowledge that they remain a problem. As DiAngelo (2018) states, if racism cannot be seen, then it does not exist.

This is not to say that misrecognition is a valid excuse, it is however a factor in understanding why racism continues in UK higher education institutions. If a person of colour has experienced disadvantage at work and felt that this might be related to their ethnic background, participants were resistant to share those experiences for fear of the consequences. Not only did those participants become silenced by their own fear, but they began to deny that it was linked to their ethnic background altogether (Augoustinos and Every, 2010). This exposes a weakness in institutional culture that is unwilling, and therefore, unable to move beyond recognising that individuals are capable of racial prejudice and therefore contrary to the self-image of a liberal, well-meaning environment such as higher education (DiAngelo, 2018). The structures that surround race and racism are so embedded that they have become objectified. They have become produced and reproduced by the dominant groups within institutions through the use of symbolic power and symbolic violence (Bourdieu, 1998). Thus, invalidating the lived experiences of black staff, which deflect accusations of racism and which form part of the institutional habitus. As such, these

structures have been internalised by black staff, who lack social and cultural capital (Bourdieu, 1998; Randle et al, 2014) to enable transformation of the social space.

Racism is not solely an agential problem and this chapter touched upon the way in which institutional operations can have differential impact upon ethnic minority employees. This can apply to the policies, services and practices the institution has in place and which can be used by employers and employees alike as weapons to silence and subdue, such as disciplinary, grievance, promotion, etc. and which affect the outcomes of black staff more acutely than staff of any other ethnic background. In exploring the concept of institutional racism, most participants were quick to suggest that they perceived their institution to be institutionally racist and the sentiment was shared by staff at all levels, including those participants that represented sector agencies, confirming UUK's recent acknowledgement to this effect (Mohdin for The Guardian, 2021). This is a damning indictment and did not only reflect how institutions operated in relation to their workforce, but also in the way that institutions provide adequate and professional services to diverse student populations.

The operational issues surrounding race also impact equality and diversity practitioners' ability to become effective change agents within their institutions and how they are often frustrated in their efforts to take appropriate actions that can advance the race equality agenda. Fears that playing the 'race card' will undermine future claims gives the impression that we have acknowledged that racism exists in our institutions. However, there appears to be an unspoken code that dictates that complaints or challenges should be used sparingly, and practitioners should consider whether they are inadvertently silencing people for fear that race fatigue will spoil it for everyone else. As mentioned earlier in this chapter, this coded terminology creates a structure that symbolically dominates those in less powerful positions within the social hierarchy (Bourdieu, 2001a).

The final chapter of this dissertation will bring together the relevant literature and findings from this research study to consider how UK higher education institutions can address matters of race and racism. In doing so there will be an opportunity to explore how sector agencies, institutions and their minority ethnic staff, human resource and equality and diversity practitioners can work together to co-create solutions that will bring about a culture change in the workforce.

8 Conclusion: The differential outcomes for black staff

8.1 Introduction

The higher education sector is trusted by UK society to develop the critical thinking of our future workforce. This research has demonstrated that it does not apply the same level of criticality on the topic of race or racism on itself as a sector – there is a disconnect between the literature and institutional practice. In failing to do so, the sector absents itself from acknowledging the role it plays in maintaining racism, and this may in part be due to the benefits derived from conserving the position of white superiority and white domination. The latter being reflected in the ethnic profile of senior leadership across institutions (Advance HE, 2020).

My unique perspective as a practitioner-researcher has enabled me to apply my insideroutside knowledge, experience and expertise in the field of equality, diversity and inclusion to this in-depth qualitative research, not only in terms of tackling institutional race equality, but also in appreciating the institutional space as a social space of diverse interactions. I have seen first-hand that institutions have not been sufficiently focused on tackling racial inequalities. With concerted endeavours to address attraction, recruitment, and retention, particularly regarding black staff, and including sustained efforts to create and maintain inclusive work environments, this will have an impact on the representation of ethnic minority staff in UK higher education and draw upon the lost pool of talent. These activities should also include setting organisational measures to drive change at organisational, school and departmental levels.

As a part of the public sector, UK higher education has not done enough to be transparent to employee, student and public scrutiny and it is apparent that institutions are not consistent in this regard. From experience, HR functions are often not willing to participate with equality and diversity practitioners to undertake policy review, particularly in areas that might expose inconsistencies or disadvantages for different groups thus requiring difficult conversations, explanations, and actions. This often makes it difficult for equality and diversity practitioners to collaborate as equals with HR practitioners on these projects, despite equality and diversity practitioners often being employed within HR teams. We as practitioners, are seldom trusted

as critical friends to the organisation, therefore being treated with caution or suspicion – we are the outsider-within. This is not helped by the lack of status, seniority, and senior buy-in, therefore overall lacking in social and cultural capital within the academy. No wonder the field of equality and diversity is regarded as a high-stress environment and one that causes such high rates of burn out.

The exploration of identity within this research was not initially factored into the research objectives. However, this became a core feature for the findings discussed earlier in this dissertation, and which appeared to be fundamental to the outcomes for black staff. Reflecting on participants' accounts in relation to concepts, such as blackness, identified that not only has this been problematised within the academy, but that reconciling their blackness is a significant psychological burden carried by black staff, whether positively or negatively, in the academic social space. I interpret this not only as a differential outcome for black staff, but as a psychological ethnic penalty that is different to the material and economic disadvantage experienced (Heath and Cheung, 2006), and which are different to the experiences of staff with other ethnic backgrounds.

It is clear from the ethnic profile discussed in Chapter Five that black staff do indeed experience differential outcomes in the workplace, through under-representation across academia and at senior levels, which may be due to multiple factors, some of which could not be explored through this research, including attraction strategies, recruitment success rates for different groups and levels of avoidable attrition. However, the lived experiences shared in this research provides valuable insight to the contributing factors to the lack of representation of black staff, not only within specific academic disciplines but also across the sector and its hierarchy. Participant narratives emphasise the lack of opportunities, racial stereotyping, the failure of institutions to effectively tackle racial bullying, harassment and discrimination, which I interpret as institutional racism. It has been a positive step that UUK also believe this to be the case (Mohdin for The Guardian, 2021).

This research has considered varied literature on the topics of race and racism, including those that have focused on UK higher education (ECU, 2011; Pilkington, 2011; Ahmed, 2012; UCU, 2016, 2017; Bhopal, 2018; Rollock, 2011, 2012, 2019; EHRC, 2019). Racism is not a new phenomenon in the sector and attention on racial inequalities has been constant and a growing concern. As an important and relevant research topic, this dissertation makes a

distinct contribution to this growing body of work, not just for its inclusion of participants' lived experiences, but in the way these have been interpreted and analysed to consider the factors that may have contributed to the prolonged silence, stigma and fear to challenge the status quo. Considering accounts through the lens of structure, agency, habitus, field, capitals and symbolic violence, this research demonstrates that there are multi-level, multi-dimensional (Layder, 1993), complex and insidious contributing factors to the perpetuity of racism within our institutions. These accounts paint a bleak picture, not just from ethnic minority staff, but also those identifying with the dominant ethnic group. These voices reflect disquiet on the topic of race, which sadly is not unique to higher education as the literature explored in this research attests. Much of that literature is devoid of individual voices and it is these narratives that will be most valuable to the sector in facilitating change.

The aim of this dissertation was to explore the lived experiences of black staff in UK higher education and how those experiences create differential outcomes within the institutional workplace. Following consideration of published literature, the desk research into the UK higher education sector and participants' lived experiences, this research concludes that black staff working in UK higher education experience differential outcomes in employment. In facilitating the journey to this inference, the research objectives were to:

- Understand the strategic drivers to advance race equality and how this has progressed over time
- 2. Consider staff perceptions of the manifestation of racism in the workplace
- 3. Determine the impact of racial inequalities on black staff in UK higher education
- 4. Consider perceptions of the existence of institutional racism and its effect within a higher education context
- 5. Explore the presence of an ethnic penalty faced by black staff in UK higher education.

This dissertation draws on literature from varied disciplines that consider the multi-layered concepts, which contribute to this topic and provide a comprehensive understanding to achieve the research objectives detailed above. The following section will consider the implications of this research on the higher education sector and provide commentary to how the research objectives have been addressed through this dissertation to meet the overall aim of this research.

8.2 Implications of the research on the UK higher education sector

The dissertation reflects the interpretation of the narratives from participants to explore the lived experiences and differential outcomes for black staff in UK higher education. In analysing the empirical data and interrogating the supporting literature, various themes surfaced that shed light on the complexities faced by black staff within institutions. A notable finding is that the outcomes experienced by academic staff bore a remarkable resemblance to those of professional and support staff, notwithstanding their distinct career structures and responsibilities. Participant accounts and perceptions also show no difference between institutions, e.g., Russell Group versus Post-1992, nor is there a distinction in the experiences of staff according to geographical location; a higher proportion of minority ethnic staff or working in an institution where the local population is ethnically diverse did not preclude negative workplace experiences because of ethnicity.

The following section will revisit each of the research questions posed in this dissertation to provide a synopsis of the findings. In summarising, there will be recommended actions to be taken by individuals, senior leaders and their institutions, relevant sector agencies, HR and equality and diversity practitioners.

8.2.1 Understand the strategic drivers to advance race equality and how this has progressed over time

As noted in the literature (ECU, 2011; UCU, 2016, 2107; UCEA, 2019; EHRC, 2019) and recent media attention (Adams, 2017; Hall, 2017; Batty, 2019; Howard, 2020) surrounding race-related issues in UK higher education, there is a sector-wide imperative to address racial disadvantage because of the prominent role higher education plays in UK society. It appears that this external pressure has made more impact over the years than the legislative obligations placed on public sector organisations, such as universities. This might indicate that either senior leaders in universities (and relevant practitioners) do not fully understand their obligations against the PSED in relation to race, or institutions do not fear being challenged for inaction and in some cases non-compliance (EHRC, 2012, 2013, 2019). This may also indicate the lack of will to enforce action on the part of the EHRC against universities that do not comply with the PSED.

In practice organisational risk aversion has become embedded within institutions, demonstrated through reticence to focus resources on race for fear of attracting negative attention. As a result of this, initiatives designed to address racial disadvantage are met with resistance or couched in a broader diversity context that can benefit others and/or the organisation in other ways (Bell, 1980). Participant feedback also suggests that institutions may prefer damage limitation when issues arise, indicating reactive institutional practices. Alternatively, the resistance to tackle race in isolation can be interpreted as deflection by dominant groups (DiAngelo, 2018) because drawing attention to race might attract complaints of negative workplace experiences from people of colour, which would require some form of action. From experience, interest convergence can be a strategy employed by equality and diversity practitioners to incorporate matters of race by stealth to engage dominant groups with the subject, however this is practitioner dependent and therefore not consistent practice across the sector.

More recently, the strategic and operational drivers have become propelled by sector initiatives, such as the Race Equality Charter (REC). From experience, the REC can rouse suspicion and anxiety at senior management levels because it requires institutions to probe into workforce data and the experiences of ethnic minority staff and students, which must be submitted to peers for assessment. This may also account for the limited take up of this charter, and Advance HE must make more effort to raise the confidence of senior leadership teams to raise participation rates across the sector. These issues were also identified in a recent review of the REC (Oloyede et al, 2021).

Unfortunately, the REC does not have the same level of kudos within the sector as Athena SWAN, which although has been in existence for far longer, had until recently the added incentive of restricting research funding for institutions without a bronze level award. Advance HE must accelerate plans to demonstrate that race equality has equal standing to gender (see recommendation two), and recommendations made by Oloyede et al (2021) to include intersectional (gender/ethnicity) considerations at Athena SWAN silver level could instigate change. Oloyede et al (2021) identified that black staff have tended to benefit least from the impacts of REC and white women have largely benefited from Athena SWAN. It therefore remains to be seen if this recommendation will reap the anticipated outcomes.

Another important feature of the discussion concerning race in higher education surrounds the way institutions collect and utilise workforce data. There were mixed reactions to the categorisation of race and ethnicity and frustration and mistrust around the way data is used within institutions. Comments included allowing staff to self-identify enabling individuals to be authentic in the workplace rather than categorise oneself in the nearest classification. Because staff are sometimes unable to identify with the categories on offer, some staff declare that they are from 'any other ethnic background', which may not be helpful to institutions from a monitoring point of view. Agencies and institutions should consider adapting the questions they ask about ethnicity that take into account the ways individuals identify in terms of their cultural heritage rather than against colour-coded categories. Approximately 6.5 percent of the university workforce does not declare their ethnic background at all, and a lack of declaration is more likely to be the case in non-UK nationals (Advance HE, 2020).

It is to the higher education sector's benefit that Advance HE invests its time and resources into producing a comprehensive set of workforce data that annually reports across all protected characteristics. This should be maintained and where possible continue to be developed to support institutions to benchmark their own workforce profile against the rest of the sector. Considering that higher education holds some of the most reliable data on student and staff demographics than any sector, it fails to use this trend data to effectively tackle ethnic minority under-representation. Institutions must be encouraged to use these data to inform their own race equality action plans to effect change.

From a practitioner perspective, this could be easily addressed by institutions investing time through regular internal campaigns to raise awareness of diversity data collection, how the data support institutional objectives, and who is able to access the information. In addition, what is *not* reported on is as significant, and the annual statistical reports produced by Advance HE only includes declared ethnicity and does not include the proportion of the workforce where employees choose not to declare or where there is no record.

As a result of these findings the following actions are recommended:

1. Sector agencies and institutions must work together to support the professionalisation of equality and diversity practitioners in a way that HR

- practitioners are supported through relevant qualifications, e.g., CIPD. A community of practice should be set up to support the continuing professional development for practitioners in this field.
- 2. Sector agencies must put in place a team of equality and diversity specialists that can effectively work with institutions on the topic of race, to share good practice, set and maintain consistent standards within the sector and who are empowered to challenge institutional practice, naming and shaming where relevant.
- 3. Advance HE should develop and deliver briefing sessions in collaboration with institutional practitioners to equip senior leadership teams with the necessary skills to champion and advance race equality in UK higher education.
- 4. Advance HE must accelerate progress in engaging institutions with REC and realise equal status to the Athena SWAN award.
- 5. That Advance HE delivers a programme of activity that systematically equips equality and diversity practitioners with the requisite skills, knowledge and capability to effect change in the sector in relation to race.
- 6. Advance HE must revise the Athena SWAN charter by requiring institutions and departments to consider as standard the intersection between gender *and* race and consider the differential outcomes faced by women of colour.
- 7. Institutional HR functions must review the collection and analysis of workforce data, including the demographic monitoring of the Board of Governors or Senate. Institutions must provide clear explanation around the reason this information is important to the institution, together with a rationalisation for the categorisation of ethnicity to increase confidence to declare. There should be greater clarity for staff around how they might wish to identify to make data gathering and analysis more worthwhile.
- 8. Equality and diversity practitioners (individuals and teams) should be located in more strategic positions within institutions because of the scope and importance of their work. This will require direct reporting to the Vice Chancellor's/Provost's Office providing independence from HR functions. The work of the Equality, Diversity and Inclusion (EDI) teams should be directed through the creation of comprehensive EDI strategies that have been agreed by the whole institution, including the Board of Governors or Senate.

8.2.2 Consider staff perceptions of the manifestation of racism in the workplace

Many participants in this study spoke candidly about their experiences of racism through microaggressions, yet the majority either denied that it had happened or were unable or unwilling to make the link between those experiences and their ethnic background. These feelings of denial and misrecognition appear to have been coupled with internalised feelings of shame, guilt and fear (Pearce, 2019; Ahmed, 2012) and rather than take action, participants explained how they employed coping strategies (Stainback and Irvin, 2012; Kim et al, 2017). More alarming were the responses from senior leaders and equality and diversity practitioners who felt unable through misrecognition, denial, institutional distraction tactics, or reluctance, to support a claim of racism for fear of being perceived to overplay the 'race card'. These comments provide evidence that talk of racism is taboo, which creates barriers for black staff and institutional equality and diversity practitioners to deal with matters of race.

DiAngelo (2018) argues that aversive racism exists under the surface of consciousness, where individuals will enact racism that will enable themselves to maintain a positive self-image. As a sector, consideration should be given to the blind spots that lie within institutional cultures that denies the existence of racism, which may be further impaired by the perceived liberal, educated, and progressive nature of the sector. Our incapability to deal with difficult conversations adds a further barrier in offering the transparency and openness required to inform contemporary solutions. The UK higher education sector must demonstrate that the responsibility and accountability in developing these solutions is held by us all and transcends all ethnic identities. If real change is to happen, then those within dominant groups have the greatest role to play in creating radical strategies where actions are owned by the most powerful and lead the direction of travel to eradicate the differential outcomes experienced by people of colour.

In understanding the extent to which black staff were able to identify racism and whether they felt confident in addressing matters of race, this research discovered that besides overt acts of racism, a significant proportion of participants did not feel confident in pursuing a complaint because of the insidious nature of acts or omissions experienced (Bourdieu, 1990; Johnson et al, 2008). Ironically, almost all minority ethnic participants provided examples of

racial microaggressions but did not recognise those as warranting the same consideration as overt acts (Rowe, 1990) and in essence these participants had downplayed their experiences of racism (Harries, 2014), which may signify a reluctance to be perceived as a 'victim' or for sheer exhaustion. This can be interpreted as reluctant or constrained agency for participants who recounted these experiences, essentially to maintain a degree of dignity, control, and self-care. Disappointingly, many participants commented that they had become immune to microinsults, microinequities and microaggressions because to deal with every incident would for some mean a daily ordeal, and this would be unsustainable and psychologically damaging, although many recognised on exploring their experiences further that their lived experiences were already adversely affecting their mental health and wellbeing due to their cumulative effect. This further supports the notion that black staff experience a psychological penalty, that is fundamentally different to the ethnic penalty described by Heath and Cheung (2006) and is a new contribution to knowledge relating to the experiences of minority ethnic employees in UK higher education.

Similar findings were reported recently in relation to racial harassment in higher education (EHRC, 2019). As a result of these responses, it is critical that institutions and their HR and equality and diversity practitioners remain vigilant because of the implications around an employers' duty of care for employee health and wellbeing in the workplace. It can also be interpreted from participants that institutions are not operating sufficient campaigns internally that raise awareness of what bullying, harassment and discrimination, including microaggressions, are and how they manifest in the workplace.

As a result of these findings, the following actions are recommended:

9. Institutional governing boards and senior leadership teams must be publicly accountable for the monitoring and measurement of reports of racism and any targets that are implemented to measure progress. Institutions must develop at least one Equality Objective that is related to race. It is essential that the outcomes of such monitoring exercises are reported annually, communicated across the institutional community for transparency, and progress included within the statutory public reporting requirements of the PSED.

- 10. Senior leaders must have a performance objective that addresses matters of equality, diversity and inclusion and more specifically linked to the elimination of discrimination, harassment and victimisation. These objectives should be aligned with any institutional objectives to eliminate race discrimination and harassment and progress reported to the Board of Governors on a regular basis.
- 11. Senior leaders and HR teams must take decisive action around the remedies available to ethnic minority staff in reporting race discrimination or harassment. HR teams should convene expert panels in race-related matters who are able to make unbiased, independent decisions around the facts of each case so that minority ethnic staff have more confidence in institutional processes and systems.
- 12. Institutional processes are required to facilitate the recording of racial microaggressions, whether anonymously to gauge the extent of incidents and/or in a more formal way that allows staff to seek a remedy through internal processes. As well as monitoring incidents, diversity data must be gathered to understand the demographic profile of staff making such reports so that an accurate picture can be formed.

8.2.3 Determine the impact of racial inequalities on black staff in UK higher education

The literature asserts that race can be a difficult subject to navigate and discuss (DiAngelo, 2018; Eddo-Lodge, 2018; EHRC, 2019), particularly for those who do not regularly engage with the topic. Despite academic debate arguing that the language of race is problematic, participants did not feel that terms such as race and ethnicity were of particular concern if the terms could be used to advance the race equality agenda. Some participants, including practitioners, commented on how often race is conflated with people of colour, and that limiting conversations about the topic meant that white people do not engage, contributing to the heightened sensitivities around the subject (DiAngelo, 2018).

In terms of language used to describe race-related topics, some participants raised an objection to the acronyms often used to describe people of black and minority ethnic background, e.g., BME, BAME. Participants that commented thought that the abbreviations dehumanised people of colour. This feedback was thought-provoking because these terms have been normalised to describe ethnic minority people for some time. Prior to beginning

this research, I had not considered the impact of this terminology in the context of my professional activity and now refrain from using these terms and encourage others to do the same.

Continuing with the notion of language and how it may empower or constrain (DiAngelo, 2018; Eddo-Lodge, 2018) views around blackness were understood by most, but reviled by some. This contentious term facilitated a discussion with black staff about dialect, appearance, mannerisms and other personal attributes bringing to life the challenges faced within the academic environment. One where some participants felt hyper-aware of how they were portraying themselves and hyper-anxious about how they were perceived, which could attract negative attention, therefore manoeuvring around the social space would entail modifying aspects of their behaviour to fit in (Steinbugler, 2015; Solanke, 2018). The accounts from many participants portrayed a despondency in coming to terms with not being authentic in the workplace and this can be interpreted as blackness being problematic in white spaces (Mapedzaham and Kwansah-Aidoo (2017), and which can be interpreted as symbolic violence (Bourdieu, 1990).

An important and unexpected empirical finding from this research revealed the importance of identity on the lived experiences of black staff in UK higher education. The subject provided further perspectives on how different groups of black staff perceived other black staff and this was particularly pronounced through comments made by those who were born outside the UK concerning those born in the UK, and where participants were of a mixed-race background (Hirsch, 2018), e.g., stereotyping groups that were different from themselves (Steinbugler, 2015). This was despite a strong sense from minority ethnic participants that they felt stereotyped in one form or another and that this had a negative impact on their workplace experiences (Kanter, 1993, Ibarra, 1993, 1995). The evidence gained through this research has found that stereotypical attitudes are not solely experienced by people of colour at the hands of dominant ethnic groups, but also from within the same/similar ethnic groups.

The participants' notion of identity was closely linked with their sense of belonging to the department and institution. For many being valued by their immediate team strengthened their sense of belonging within it, however this was not mirrored in relation to the institution, with many reporting feelings of invisibility (Solanke, 2018). This was further intensified by the lack of ethnic diversity within the rest of the institution thereby increasing feelings of isolation

and paradoxically heightened visibility because of their difference (Solanke, 2018). Participants perceived that with this attention came a heightened expectation that the individual would be required to outperform or become accountable for every other person of colour in the institution, and this supports Solanke's (2018) work on hypervisibility and hyperinvisibility. This caused increasing levels of anxiety, particularly in those participants employed in senior roles within their institutions who felt under more scrutiny than their peers (Wilson and Jones, 2008; Solanke, 2018). Consequently, it appears that identity is not only bound in the self, but in the field and the habitus (Bourdieu, 1977), causing individuals to participate in a game where the rules have been long established and entrenched by the white hegemonic culture that exists within UK higher education. Participation is not voluntary if one wishes to progress in their chosen field. Deconstructing the participants' lived experiences by applying Bourdieu's notion of power through habitus, field, capital and symbolic violence (1977, 1984, 1990, 1998) provided a novel insight to describe how the institutional systems and structures can enable and constrain across multiple levels and is a meaningful contribution to knowledge and practice. Thus, enabling practitioners to view their institutional processes and practices in an alternative way to better understand the disproportionate impact of these systems and structures on ethnic minority staff.

Identity and belonging are inextricably linked to levels of confidence people of colour, and particularly black staff, feel in recognising racism when it occurs and taking action to challenge it. However, herein lies part of the problem and which is the residual impact of some of the perceptions and experiences described by participants. Black staff are constantly reminded that they are the minority within a minority by virtue of there being such few black staff in academic and professional and support staff roles, and this is particularly acute in senior levels of our institutions. Not seeing oneself represented at multiple levels undermines identity and belonging within a white-dominated environment, which has seen an increase in research and media attention (Times Higher, 2013; The Independent, 2014; The Guardian, 2019, 2020; EHRC, 2019). Despite this, the sector is resistant to act. Consequently, the collective inaction continues to fuel perceptions among black staff that the threat of destroying professional relationships or being labelled a troublemaker is far worse than enduring the impact of racism (Baumeister and Leary, 1995).

As a result of these findings the following actions are recommended:

- 13. Advance HE to design and deliver a mandatory sector-wide programme encouraging 'active bystanders²¹' to create more inclusive workplace cultures and uphold a zero-tolerance stance to bullying, harassment and discrimination.
- 14. Institutional policy makers to review and implement, if not already in place, a mechanism which allows staff to report anonymously and provide additional external support, such as an Employee Assistance Programme (EAP), where individuals are able to gain specific independent advice. HR functions must use the data gathered from EAP usage to determine appropriate actions to tackle bullying, harassment and discrimination.
- 15. Institutions to develop and implement focused learning interventions on the topic of race covering the history, language, locating white people within the wider race agenda, white privilege and white fragility. These should be mandatory and delivered to all staff, beginning with senior leaders and governing boards.
- 16. Institutions to facilitate sessions for ethnic minority staff to talk openly about their lived experiences at work. These conversations should be facilitated by at least one senior leader with the support from HR, equality and diversity practitioners, and staff networks so that racism and racial microaggressions can be discussed without repercussions and to normalise conversations about racism. The outcomes of these sessions should inform institutional strategies and policies.
- 8.2.4 Consider perceptions of the existence of institutional racism and its effect within a higher education context

Establishing the existence of institutional racism in the context of UK higher education is a new contribution to knowledge, and especially where participants have been asked explicitly whether they perceive this to exist within their institutions or within the sector. To reduce doubt or misunderstanding on the topic of institutional racism, participants were presented with the definition as described by Macpherson (1999).

The collective failure of an organisation to provide an appropriate and professional service to people because of their colour, culture, or ethnic origin. It can be seen or detected in processes, attitudes and behaviour which amount to discrimination through

 $^{^{21}}$ Active bystander strategies can be employed to safely intervene or challenge inappropriate behaviour

unwitting prejudice, ignorance, thoughtlessness and racist stereotyping which disadvantage minority ethnic people.

(Macpherson, 1999:28)

Most participants (approximately 95 per cent) responded that institutional racism did exist, and that it manifested itself in a multitude of ways. This finding provides a significant insight to the perceptions of staff and stakeholders by revealing the role UK higher education institutions play in affecting the outcomes of black staff and other people of colour that are employees and/or students.

The topic of institutional racism has attracted much debate since the term was resurrected in Macpherson (1999), focusing primarily on the dichotomy within the definition in terms of structure, described as the collective failure, and agency posited through actions such as unwitting prejudice, ignorance, thoughtlessness and racist stereotyping (Wight, 2003; Gillborn, 2008). These terms are most commonly associated with agential acts or omissions, and therefore has contributed to confusion in operationalising the concept within relevant organisations (Souhami, 2014) and through claims of no defined actions to address institutional racism (Solomos, 1999). Gillborn (2008) highlighted the dual responsibility and accountability within the contemporary definition used in Macpherson (1999), which extends beyond the actions of individuals through conduct, attitude and behaviours to include the organisational processes which disadvantage people of colour. Ahmed (2012) argued that solely blaming individuals for their actions underestimates how racism is reproduced.

In exploring the perceptions around institutional racism in UK higher education and its effect on institutional capability to tackle racial inequalities, these findings offer the opportunity to consider how operationally institutions function in terms of the implementation of policies, processes and practices that disadvantage people of colour. To date, there has been little activity within the sector to explicitly address this concept and this may be as a consequence of misunderstanding institutional racism when it refers in part to the collective nature of operations yet focuses on individual acts or refusing to look for or accept that institutional racism may be present. This may be indicative of shared organisational habitus (McDonough, 1998; Horvat and Antonio, 1999), since the denial of institutional racism is sector wide and can be interpreted as symbolic violence (Bourdieu, 1990; 2001b) since the denial creates the

illusion that institutional workplaces are fair and equitable to ethnic minority staff despite this research supporting the contrary. The discussion has identified how the acts of individuals can be produced and reproduced to become part of the cultural fabric of an institution, and although its policies and practices might seem benign in their articulation and appear at face value to be equally applicable to all, the way in which they are implemented may be inequitable.

In contrast to the literature, participants were quick to dissect the definition in terms of the structural and agential considerations and found no difficulty in understanding and contextualising how aspects of their own institution's functions corresponded to these features. Examples of such operationalisation included the student attainment gap, the negative workplace experiences of black and minority ethnic staff, access, admissions, learning and teaching, curriculum content and pedagogy, unfair process, the moving of goal posts particularly within the promotions process, etc. There was no shortage of examples that participants could see affected the outcomes of staff *and* students. A conclusion that can be drawn from these examples is that the apparent confusion or lack of defined actions to address institutional racism postulated by Souhami (2014) and Solomos (1999) is not reflected through participant accounts.

In considering the concept of institutional racism from the perspective of 'institutional' symbolic violence (Bourdieu, 1990) this can allow for a reconstructed definition of Macpherson (1999) so that it is simpler to operationalise within organisations. As such, I would propose an adapted form of the definition in Macpherson (1999) to the following:

The collective failure of an organisation to provide an appropriate and professional service, equal opportunity or equal outcomes to people because of their colour, culture or ethnic origin. It can manifest through attitudes, behaviours, unwitting prejudice, ignorance, thoughtlessness and racist stereotyping and can be seen or detected in the implementation of processes, practices and services on behalf of the organisation.

This revised definition acknowledges that every individual actor within an organisation can hold or demonstrate attitudes, whether unwitting or not, and can implement a series of actions, or indeed omissions, on behalf of the organisation and regardless of level of seniority, that may disadvantage people of colour. It is this recognition that will place the organisation

as a collective responsible and accountable for the outcomes of ethnic minority employees and service users at the hands of its workforce. This presents a model that is not dissimilar to the notion of vicarious liability already present in the UK legislative framework.

Policy implementation is only one aspect of institutional action to address differential outcomes and it is also the way that it tackles racism, or the potential for racial disadvantage. Accounts by some participants raised concerns that it would take a serious incident to be the catalyst for change, rather than taking a proactive approach for setting a more inclusive workplace culture. This indicates that the lack of ethnic diversity among senior decision-making groups in institutions is having an operationally detrimental impact, mainly while leaders do not have the cultural sensitivities required to address matters of race and racism within institutions. It is apparent from participants' responses and the management profile of institutions (Advance HE, 2020) that people of colour are not given a voice to inform institutional policy and strategy in relation to race equality.

Participants felt there was a general lack of engagement and denial across the sector that institutional racism was a problem. If institutional racism occurs through unwitting prejudice, thoughtlessness and racist stereotyping, these matters could be addressed through education. An HR practitioner highlighted the difficulty of delivering institutional policy and monitoring implementation at local level, which raised a noteworthy concern around an institution's ability to ensure that policies were being implemented in a fair, transparent and consistent way. In my own professional opinion, I have not yet worked in any organisation where there is systematic monitoring to ensure that policy is implemented in a fair and consistent manner, yet it is in this area that institutions could make the most significant impact by working with minority ethnic staff in the review and development of institutional policy as critical friends.

As a result of these findings the following actions are recommended:

17. That sector agencies, together with UK higher education institutions review and implement a programme of systematic appraisal of internal policies and practices that impact staff, e.g., recruitment, promotions, probation, disciplinary, grievance, bullying

and harassment, etc. This should include equality analysis²² to ensure that due regard is given to social groups during the development, review and revision of policy, services or practice. This exercise must include data analysis regarding policy usage and collaboration with equality and diversity practitioners, staff networks and trade unions to support the review process and consultation.

- 18. That Advance HE creates specific guidance for managers and senior leaders on operationalising institutional racism, by adopting the definition proposed within this dissertation to facilitate progress. This support should be cascaded throughout institutions, particularly for personnel involved in decision-making roles and with direct contact with students. The guidance should clearly set out the responsibilities on governing bodies, senates and senior leaders in eradicating institutional racism and, where necessary, identify targets that enable the institution to advance race equality. To support this, HERAG should be tasked with monitoring the progress of the sector in tackling this issue, working in collaboration with Advance HE to provide a 'state of the nation' report on an annual basis.
- 19. Institutions must publish annually details of their workforce profile according to ethnicity, including across different grades (academic and professional and support), identifying any concerns arising from that data, how they are increasing their ethnic representation, how they are advancing equality of outcome and lowering attrition, etc. Issues identified through these analyses should inform the creation of appropriate action plans and the setting of SMART equality objectives.
- 8.2.5 Explore the presence of an ethnic penalty faced by black staff in UK higher education

There is no doubt that the statistical data relied upon within this research (Advance HE, 2020) indicates that black staff employed in UK higher education institutions experience differential outcomes when compared to staff of all other ethnic groups in terms of their representation in the workforce at all levels, their retention, and their pay and that this affects both academic and professional and support staff. The findings of this research show that there are a range

_

²² Also known as Equality Impact Assessment and is a process that allows organisations to demonstrate they are meeting the requirements of the PSED and how they pay due regard to the impact a policy, service or practice has on different groups of people. This should be undertaken before and during policy, service or practice development and whenever policies, service and practices are being reviewed and updated.

of workplace issues faced by black staff that are not experienced to the same extent by their peers from other ethnic groups and this indicates that institutional structures and the way those structures are produced and reproduced by dominant groups contribute to these differential outcomes and indicates the presence of an ethnic penalty to black staff.

In speaking with participants, there was a genuine feeling that a racial hierarchy operated within institutions, and this was evident in the absence of black staff within senior levels and is supported through sector demographics (Advance HE, 2020). It also cannot be ignored that black women face additional barriers where 'glass ceilings' and 'sticky floors' may inhibit opportunities for progression, and these outcomes are fuelled by racial and gendered stereotyping, creating differential outcomes and ethnic penalties for black women, which mean that they tend to occupy junior positions, are more likely to feel isolated, not just because they may be the only black person in a team but may also be the only woman. Black women will also be more likely to receive lower pay than their white female counterparts as a consequence of gender segregation in roles (again sector demographics show a high proportion of black staff in administrative roles), which result in lower lifetime income.

Participants have provided moving and sometimes concerning accounts of their experiences, and although there have been some similarities with staff of other ethnic groups those accounts highlighted the weighty differences that negatively impact black staff, and which strongly supports the assertion made in this research that black staff suffer an ethnic penalty that is not present to the same extent for other racially minoritised groups. This assertion is clearly supported by the secondary statistical data (UCEA, 2018; Advance HE, 2020), which this study has relied upon.

This research provides evidence that black staff face a psychological penalty concerning the barriers experienced in discussing matters of race for fear of being further ostracised, excluded or that is career limiting. These barriers are so significant that black staff choose not to report (EHRC, 2019) and will often tolerate ongoing inappropriate behaviours and attitudes, which have a significant detrimental impact on their mental health and wellbeing, their relationships with colleagues and peers, or they will choose to leave their institution all together (ECU, 2015). In addition to this, the psychological burden carried by black staff by virtue of their own identity places a distinct psychological penalty on this cohort of staff in navigating white-dominated academic spaces (see discussion in Chapter Six).

As a result of these findings the following actions are recommended:

20. Institutions must move away from viewing ethnic minority staff as a homogenous group so that the complexities and nuances of individual and specific ethnic group experiences can be considered more effectively. There is sufficient longitudinal trend data that shows the continued under-representation of black staff, particularly at senior level, higher attrition and lower pay than white staff. It is essential that institutions use this data to benchmark their own performance against the sector and implement programmes of activity to address these differential outcomes.

8.3 Situating the self as practitioner-researcher

This research has rightly focused on the subjects most touched by the topic of race and racism within UK higher education and how this impacts the outcomes of black staff. I have used Layder's (1993) research map to inform the context, setting, situated activity and the self in relation to the lived experiences of this cohort. However, there is no denying that throughout this research journey, there has been a significant impact on my 'self' during this time.

In hindsight, I embarked on this exploration naively thinking that I would remain detached from the topic in the interests of remaining neutral and objective. This research and my conversations with the participants, who I am grateful to for sharing their stories and accounts of life in UK higher education, has affected me profoundly and I have never been detached from this research. In fact, it has been all-consuming. Not just as a practitioner, or researcher, but as a human. Cunliffe (2003) asserts that reflexivity 'unsettles' as we constantly construct meaning and social realities as we interact with others and discuss our experiences. Cunliffe continues to argue that in taking a radical reflexive approach that we must go further than questioning the truths of others to question how we as researchers and practitioners make truth claims and construct meaning. These statements reflect the personal and professional journey I have taken throughout this research journey.

This study and my interactions with participants have deconstructed what I thought I knew and informed the reconstruction of the way I view the world, the way I see others, has increased my sensitivity to hidden forms of racism in the workplace and my ability to critique

organisational systems and processes. The social constructionist approach (Burr, 2015) I used to approach this research has also been applied in reflecting on my own practice as a diversity professional and informed who I am, how I interact with and create new realities with others. This learning, albeit painful at times, has made me a better practitioner in seeking more effective workplace equality solutions and enabled me to share that knowledge to equip and empower colleagues to do the same. This experience has provided a significant contribution to my practice. However, I completely underestimated the psychological impact this work would have, not in relation to the work required to complete a doctoral research programme, but in internalising the lived experiences of those who shared their stories with me. As a result, guilt, shame, anger, frustration, mistrust, suspicion have been the range of emotions and feelings that have travelled with me throughout this time. Despite this, I believe that the extent of my feelings and emotions are only a fraction of those experienced by some of the research participants. From an ethical perspective I battled the inclusion of any personal account, resulting in this reflection featuring at the end of this dissertation, as this research is not about me as a white woman practitioner-researcher, who does not need to think about how my colour or ethnic background affects my lived experiences in the UK. This research is directly for those who participated and indirectly for those who are affected by the issues I have explored.

I have stated that this work has made me a better practitioner, and it has. It has allowed me to build the skills, knowledge, and competence to tackle complex workplace challenges in relation to race. The impact has been almost immediate, particularly through the data gathering stage of this research as it enabled me to listen, hear, reflect, and respond through practice. As a result, I have implemented several changes that I have recommended in this chapter, albeit in another part of the public sector. What I have learned is that black and minority ethnic staff must have a voice in their organisation, be empowered to act, and know that they can be heard without prejudice or fear of reprisals. Talk of race and racism must be legitimised within higher education organisations if they wish to grow and effectively tackle workplace race inequalities. Organisations must listen, hear, reflect, and respond together with their staff to co-create solutions that can build trust and confidence – something that we take for granted in our employee-employer relationship.

The final section of this chapter will conclude the dissertation with a summary of how this research makes a contribution to knowledge and practice and identify where further research might be required to address the areas which could not be covered through this study.

8.4 Conclusion

This research has considered the perceptions and experiences of racism, the inconsistent and unfair implementation of policy and practice, and the under-representation of black staff at all levels of the institution. It is evident that together with the perceptions of the existence of institutional racism, that they are all connected. The significant under-representation of black staff in UK higher education further exacerbates the problems that affect this staff cohort, and this is particularly acute within senior decision-making roles. This research provides a significant contribution in terms of the lived experience of black academic *and* black professional and support staff, who have been largely absent from this debate through previously published literature. This research provides a voice for those colleagues affected by the issues discussed in this dissertation, many of whom feel silenced, intimidated and marginalised within their institutions.

In terms of practice, this research is an original contribution to practice in business administration and has brought together the views and experiences of other equality and diversity practitioners in the field and provided noteworthy insight into the challenging environments that practitioners find themselves in. Their low status within the institutional hierarchy and their organisational positioning, create difficult and stressful workplace conditions, with limited resources and often with little senior support or sponsorship. If UK higher education institutions are serious about equality, diversity and inclusion for staff and advancing the sector's position in relation to race equality, more value must be placed on the important role that these practitioners play in changing institutional culture. Institutions must provide these practitioners with the necessary support and sponsorship that is required to be valued practitioners within these complex environments.

Despite my own ethnic identity being different to most of the participants interviewed during the data gathering phase of the research, those participants gave honest, frank and often disturbing accounts of their experiences that provided uncomfortable food for thought to me personally and professionally. Throughout the research journey, I have personally felt responsible for my race in considering participant accounts and this has affected me profoundly. I have reflected on the ways that I may have contributed to others' negative experiences in the workplace, and how I may not have done enough for those who sought my advice in the past. This research journey has contributed to my knowledge and practice in a way that my occupational activities could never have achieved in isolation and has informed my practice throughout my time in higher education and the roles I have held since leaving the sector. This journey has increased my own awareness about the level of privilege I hold because I am white, and as a result I am using my white privilege to actively challenge the role we all play in supporting the structures that disproportionately disadvantage black people and uphold the systems of racism. Those structures exist in the organisational policies and practices that we rely on, and they are being used in a way that have created and maintain differential outcomes for black staff in UK higher education.

My research journey has consistently informed my practice, not solely in the way that I am better able to critically analyse information, but in the depth of knowledge I have gained from the generous contributions made by the participants of this study surrounding the topic of race. I am an active advocate of race equality in my current role leading a team of specialists in a large and complex public sector organisation. I have become a subject matter expert around race and ethnicity and share my knowledge across the organisation to increase awareness of race, ethnicity, racism, microaggressions, institutional racism, white privilege and white fragility and how the latter can perpetuate the former. In doing so, I am driving efforts to create an anti-racist organisation through learning interventions and other activities that supports leaders and peers to advance race equality.

The death of George Floyd and ensuing outrage reminded me of the aftermath of Stephen Lawrence. One equality and diversity practitioner stated that there needed to be another incident like this again before something would happen. This shocked and disappointed me at the time, years before it would happen again. This time it was recorded, and this time the world was watching. I hope this will be the event to bring an end to racial injustice and inequities. I also hope that this research has done justice to those who contributed by sharing

their feelings, opinions, and experiences. Some of these were clearly painful and will have made those participants feel exposed and vulnerable.

This dissertation's aim is that the evidence provided following the analysis and interpretation of participant perceptions and experiences is used to inform institutional strategy, policy, decision-making, policy implementation and practice to make a positive impact on the lives of black staff in UK higher education.

References

Acker, J. (2000) *Gendered contradictions in organizational equity projects*, Organization, Vol. 7(4), pp. 625-632.

Acker, J. (2006) *Inequality regimes: Gender, Class and Race in Organizations*, Gender and Society, Vol.20, No.4, pp.441-464.

Acker, J. (2012) *Gendered organizations and intersectionality: problems and possibilities*, Equality, diversity and inclusion an international journal, Vol. 31, no. 3, pp. 214-224.

Adams, R. (2017) *British universities employ no black academics in top roles, figures show*, London: The Guardian, https://www.theguardian.com/education/2017/jan/19/british-universities-employ-no-black-academics-in-top-roles-figures-show (accessed 04/09/2020).

Advance HE (2019), Equality in higher education: staff statistical report 2019, London https://www.advance-he.ac.uk/knowledge-hub/equality-higher-education-statistical-report-2019 (accessed 04/09/2020).

Advance HE (2020) *About ECU's Athena SWAN charter*, https://www.ecu.ac.uk/equality-charters/athena-swan/about-athena-swan/ (accessed 04/09/2020).

Agócs, C. (1997) *Institutionalized Resistance to Organizational Change: Denial, Inaction and Repression,* Journal of Business Ethics Vol. 16, pp. 917–931.

Ahmed, S. (2012) *On being included: Racism and diversity in institutional life*, London: Duke University Press.

Allport, G.W. (1954) The nature of prejudice, Addison-Wesley.

Andrews, T. (2012) *What is social constructionism?* Grounded Theory Review: An international journal, http://groundedtheoryreview.com/2012/06/01/what-is-social-constructionism/ (accessed 04/09/2020).

Apinall, P.J. (2009) *The Future of Ethnicity Classifications*, Journal of Ethnic and Migration Studies, 35:9, pp. 1417-1435.

Archer, L, (2003) *Race, Masculinity and Schooling: Muslim Boys and Education*, Buckingham: Open University Press.

Arday, J. (2018) Understanding race and educational leadership in higher education: Exploring the Black and ethnic minority (BME) experience. Management in Education, 32(4), pp. 192–200.

Atkinson, W. (2015) Class, Cambridge: Polity Press.

Augoustinos, M. and Every, D. (2010) *Accusations and denials of racism: Managing moral accountability in public discourse*, Discourse and Society, Vol. 21(3), pp. 251-256.

Barkan, E. (1992) The retreat of scientific racism: Changing concepts of race in Britain and the United States between the world wars, Cambridge: Cambridge University Press.

Bassett-Jones, N. (2005) *The Paradox of Diversity Management, Creativity and Innovation*, Diversity Management, Creativity and Innovation, Vol. 14 No.2, pp. 169-175.

Batty, D. (2019) *UK universities condemned for failure to tackle racism* https://www.theguardian.com/education/2019/jul/05/uk-universities-condemned-for-failure-to-tackle-racism (accessed 04/09/2020).

Batty, D. (2020) *Cambridge may drop BAME mentoring of white academics*, https://www.theguardian.com/education/2020/mar/14/cambridge-may-drop-bame-mentoring-of-white-academics (accessed 16/03/2020).

Batty, D. (2020) *Universities criticised for 'tokenistic' support for Black Lives Matter*, The Guardian https://www.theguardian.com/education/2020/jul/06/universities-criticised-for-tokenistic-support-for-black-lives-matter (accessed 09/11/2020).

Baumeister, R.F. and Leary, M.R. (1995) *The Need to Belong: Desire for Interpersonal Attachments as a Fundamental Human Motivation*, Psychological Bulletin, Vol. 117, no. 3, pp. 497-529.

BBC (2016) 1965: New UK race law 'not tough enough' http://news.bbc.co.uk/onthisday/hi/dates/stories/december/8/newsid 4457000/4457112.stm (accessed 04/09/2020).

BBC (2020) *George Floyd: What happened in the final moments of his life* (16 July 2020) https://www.bbc.co.uk/news/world-us-canada-52861726 (accessed 09/11/2020).

BBC (2021) Race report: UN experts say conclusions could 'fuel racism', https://www.bbc.co.uk/news/uk-politics-56800763 (accessed 25/04/2021).

Bell, D.A. (1980) *Brown v. Board of Education and the Interest-Convergence Dilemma*, Harvard Law Review, Vol. 93, No. 3, pp. 518-533.

Berger, P.L. and Luckmann, T. (1991) *The Social Construction of Reality: A treatise in the sociology of knowledge*, Penguin Books.

Bernasconi, R. (2001), Who invented the concept of Race? Kant's role in the Enlightenment construction of race, in Race, ed. Bernasconi, R. Oxford: Blackwell.

Bhopal, K. (2018) White privilege: The myth of a post-racial society, Bristol: Policy Press.

Bhopal, K. and Brown, H. (2016) *Black and minority ethnic leaders: Support networks and strategies for success in higher education*, London: Leadership Foundation for Higher Education.

Bhopal, K. and Pitkin, C. (2018) *Investigating higher education institutions and their views on the Race Equality Charter*, London: University and College Union (UCU) https://www.ucu.org.uk/media/9526/Investigating-higher-education-institutions-and-their-views-on-the-Race-Equality-Charter/pdf/Race_Equality_Charter_Kalwant_Bhopal_Clare_Pitkin.pdf (accessed 04/09/2020).

Bhopal, R. (2007) *The beautiful skull and Blumenbach's errors: the birth of the scientific concept of race*, BMJ, 2007 Dec 22; 335(7633) pp.1308-1309.

https://www.ncbi.nlm.nih.gov/pmc/articles/PMC2151154/ (accessed 04/09/2020).

Blackaby, D. and Frank, J. (2000) *Ethnic and other minority representation in UK academic economics*, Economic Journal 110(464), pp. 293–311.

Blum, L. (2002), "I'm not a racist, but...": The Moral Quandary of Race, New York: Cornell University Press.

Bonilla-Silva, E. (2002) *The linguistics of color blind racism: How to talk nasty about blacks without sounding 'racist'*, Critical Sociology, Vol. 28, 1(2), pp. 41-64.

Bonilla-Silva, E. (2015) *The structure of racism in color-blind, "post-racial" America*, American Behavioral Scientist, Vol. 59(11), pp1358-1376.

Bourdieu, P. (1977) Outline of a theory of practice, Cambridge: Cambridge University Press.

Bourdieu, P. (1984) *Distinction: A social critique of the judgement of taste*, Cambridge: Harvard University Press.

Bourdieu, P. (1984) Homo Academicus, English Edition trans. Collier, P. (1988), Cambridge: Polity Press.

Bourdieu, P. (1986) *The forms of capital* In Richardson, J., *Handbook of Theory and Research for the Sociology of Education*, Westport, CT: Greenwood, pp. 241–258.

Bourdieu, P. (1987) What Makes a Social Class? On the Theoretical and Practical Existence of Groups, Berkeley Journal of Sociology, 32, pp. 1-17.

Bourdieu, P. (1990) The logic of practice, Cambridge: Polity Press.

Bourdieu, P. (1991) Language and symbolic power, Cambridge: Polity Press.

Bourdieu, P. (1998) *Practical Reason: On the theory of action*, Cambridge: Polity Press.

Bourdieu, P. (2001a) Masculine domination, Oxford: Polity.

Bourdieu, P. (2001b) Television, European Review, Vol. 9, No. 3, pp. 245-256.

Bourdieu, P. and Wacquant, L.J.D. (1992) An invitation to reflexive sociology, Cambridge: Polity Press.

British Library (2020) Campaign for Abolition,

http://www.bl.uk/learning/histcitizen/campaignforabolition/abolitionbackground/abolitionintro.ht ml (accessed 04/09/2020).

Brown, J. (2018) *An early history of British race relations legislation*, House of Commons Briefing Paper Number 8360, 9 July 2018, London: House of Commons Library.

Brown, R. (1995) Racism and immigration in Britain, International Socialism Journal, Issue 68.

Bryman, A. and Bell, E. (2011) Business research methods. 3rd ed. Oxford: Oxford University Press.

Bryman, A. and Cassell, C. (2006) *The researcher interview: a reflexive perspective*, Qualitative Research in Organizations and Management: An International Journal, Vol. 1, No. 1, pp.41-55.

Burr, V. (2015) Social Constructionism, 3rd Edition, Florence: Taylor and Frances.

Cabinet Office (2018) Race Disparity Audit, London: Race Disparity Unit https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attachment_data/file/686071/Revised_RDA_report_March_2018.pdf (accessed 02/12/2020).

Cabinet Office (2019) *Ethnic diversity of public sector workforces*, London: Race Disparity Unit https://www.ethnicity-facts-figures.service.gov.uk/summaries/public-sector-workforces (accessed 04/09/2020).

Cabinet Office (2020) *Written Ministerial Statement on Unconscious Bias Training*, London. https://www.gov.uk/government/news/written-ministerial-statement-on-unconscious-bias-training (accessed 02/01/2021).

Campbell, A.F. (2016). *How to Get Americans to Talk About Race*, The Atlantic. www.theatlantic.com/politics/archive/2016/09/getting-people-to-talk-about-race/498944 (accessed 04/09/2020).

Carmichael, S. and Hamilton, C.V. (1967), *Black Power: The politics of liberation in America* in Macpherson (1999) *The Stephen Lawrence Inquiry*, London: The Stationery Office.

Carter, J, Fenton, S and Modood, T (1999) *Ethnicity and Employment in Higher Education,* London: Policy Studies Institute.

Caspari, R. (2003) *Types to Populations: A Century of Race, Physical Anthropology, and the American Anthropological Association*, American Anthropologist, Vol. 105, No. 1, Special Issue: Biological Anthropology: Historical Perspectives on Current Issues, Disciplinary Connections, and Future Directions (Mar. 2003), pp. 65-76.

Centre on Dynamics of Ethnicity (CoDE) (2014) *How are ethnic inequalities in education changing?* University of Manchester.

Chartered Institute of Personnel and Development (CIPD) (2017) *Addressing the barriers to BAME employee career progression to the top*, London: CIPD, https://www.cipd.co.uk/Images/addressing-the-barriers-to-BAME-employee-career-progression-to-the-top-tcm18-33336.pdf (accessed 04/09/2020).

Chen, J.M. and Hamilton, D.L. (2015) *The importance of social acceptance*, Personality and Social Psychology Bulletin, Vol. 41(4), pp. 586-598.

Coates, T. (2015) *Between the world and me,* The Text Publishing Company, Melbourne, Victoria, Australia.

Cole, M. (2009) *Critical Race Theory comes to the UK: A Marxist response*, Ethnicities, Vol. 9(2), 246-284.

Collinson, D. (2016) *Dialectics of leadership*, Human Relations (New York), Vol. 58, Issue 11, pp. 1419-1442.

Commission on Race and Ethnic Disparities (2021) *Commission on Race and Ethnic Disparities: The Report*, https://www.gov.uk/government/organisations/commission-on-race-and-ethnic-disparities (accessed 20/04/2021).

Commission on University Career Opportunity (CUCO) (1997) A report on Policies and Practices on Equal Opportunities in Employment and Colleges in Higher Education, London: CUCO.

Compton, M.E. and Meier, K.J. (2016) *Managing social capital and diversity for performance in public organizations*, Public Administration, Vol.94, No. 3, pp.609-629.

Connell-Smith, A. and Hubble, S. (2018) *Widening participation strategy in higher education in England*, House of Commons Briefing Paper Number 8204, 24 January 2018, London.

Coughlan, S. (2020) *'Unconscious bias training' to be scrapped by ministers*, London: BBC. https://www.bbc.co.uk/news/education-55309923 (accessed 02/01/2021).

Cox, T.H., Lobel, S.A. & McLeod, P.L. (1991) *Effects of Ethnic Group Cultural Differences on Cooperative and Competitive Behavior On a Group Task*, Academy of Management Journal, Vol. 34, no. 4, pp. 827-847.

Crawley, R. (2007) *Talking it out: Political correctness as resistance to anti-racism, Equal Opportunities International*, Vol. 26, No. 5, pp. 497-506.

Crenshaw, K. (1989) Demarginalizing the intersection of race and sex: A Black feminist critique of antidiscrimination doctrine, feminist theory and antiracist politics, University of Chicago Legal Forum, pp. 139-168.

Crotty, M. (1998) *The Foundations of Social Research: Meaning and Perspective in the Research Process*, London: SAGE.

Cunliffe, A.L. (2003) *Reflexive Inquiry in Organizational Research: Questions and Possibilities, Human relations (New York),* vol. 56, no. 8, pp. 983-1003.

Cunliffe, A.L. (2008) *Orientations to Social Constructionism: Relationally Responsive Social Constructionism and its Implications for Knowledge and Learning*, Management Learning, Vol. 39(2), pp.123-139.

Daly, J., Willis, K., Small, R., Green, J., Welch, N., Kealy, M. and Hughes, E. (2007) *A hierarchy of evidence for assessing qualitative health research*, Journal of Clinical Epidemiology 60, pp.43-49.

D'Ancona, M. (2018) *Let's be honest about what's really driving Brexit: bigotry*, The Guardian https://www.theguardian.com/commentisfree/2018/dec/02/honest-brexit-bigotry-ugly-chapter-history (accessed 04/09/2020).

Davies, J. (2020) Now even Sir Winston Churchill's legacy is reviewed by the Imperial War Museum: Bosses examine his views on 'sensitive topics' in wake of Black Lives Matter protests, Daily Mail https://www.dailymail.co.uk/news/article-8925305/Winston-Churchills-legacy-reviewed-Imperial-War-Museum-wake-Black-Lives-Matter-movement.html (accessed 09/11/2020).

Davies, R. (2020) Firms are waking up to race issues, but critics say hashtags aren't enough, The Guardian, https://www.theguardian.com/business/2020/jun/20/companies-wake-up-to-race-issues-black-lives-matter-protests (accessed 21/11/2020).

DeCuir-Gunby, J.T. and Gunby, N.W. (2016) *Racial micro-aggressions in the workplace: A critical analysis of the experiences of African-American educators*, Urban Education, Vol. 51(4), pp.390-414.

Deem, R. (1998) 'New Managerialism' and Higher Education: the management of performances and cultures in universities in the United Kingdom, International Studies in Sociology of Education, Vol. 8, No. 1, pp.47-70.

Deem, R. (2007) Managing a meritocracy or an equitable organisation? Senior managers' and employees' views about equal opportunities policies in UK universities, Journal of Education Policy, Vol. 22(6), pp. 615-636.

Deem, R. and Morley, L. (2006) *Diversity in the academy? Staff perceptions of equality policies in six contemporary higher education institutions*, Policy futures in education, Vol. 4, No. 2, pp. 185-202.

Delgado, R. & Stefancic, J. (Eds.) (2000) *Critical Race Theory: the cutting edge*. 2nd Edition, Philadelphia: Temple University Press.

Delgado, R. and Stefancic, J. (2007) *Critical Race Theory and Criminal Justice*, Humanity and Society, Vol. 31, pp. 133-145.

Department for Business Innovation and Skills (2013) *The Business Case for Equality and Diversity: a survey of the academic literature*, London: BIS Occasional Paper No.4.

DiAngelo, R. (2018) White Fragility: Why it's so hard for white people to talk about racism, London: Penguin Books.

Dickens, L. (1994) *The business case for women's equality: Is the carrot better than the stick?* Employee Relations, Vol. 16(8), pp. 5-18.

Dickens, L. (1999) *Beyond the business case: A three-pronged approach to equality action*, Human Resource Management Journal, Vol. 9, No. 1 pp. 919.

Dirlik, A. (2008) *Race talk, race and contemporary racism*, PMLA, Vol. 123, No. 5, Special Topic: Comparative Racialization (Oct. 2008), pp. 1363-1379.

Dudman, J. (2018) *A decade of cuts has set back diversity in the public sector*, The Guardian, https://www.theguardian.com/society/2018/sep/18/decade-cuts-set-back-diversity-public-sector (accessed 04/09/2020).

Durrani, S. (2002) *Combating racial discrimination in British public libraries: the role of the Quality Leaders Project*, Library Management, Vol. 23 Issue: 1/2, pp.23-52.

Eddo-Lodge, R. (2018) Why I'm no longer talking to white people about race, London: Bloomsbury.

Edwards, T. (2020) *Black Lives Matter: 'Backlash has been divisive like Brexit'*, BBC, London. https://www.bbc.co.uk/news/uk-wales-53503288

Encyclopaedia Britannica (2021) *Scientific classifications of race*, https://www.britannica.com/topic/race-human/Scientific-classifications-of-race (accessed 13/07/2021).

Essed, P. (1991) *Understanding Everyday Racism: An Interdisciplinary Theory*, Newbury Park, London, New Delhi, SAGE Publications.

Equality Act 2010, c.15, United Kingdom https://www.legislation.gov.uk/ukpga/2010/15 (accessed 04/09/2020).

Equality and Human Rights Commission (2012) *Publishing equality information: Commitment, engagement and transparency: Assessment of public authorities' implementation of the specific duty to publish equality information,* London: EHRC https://equalityhumanrights.com/en/advice-and-guidance/our-research-equality-duty (accessed 03/12/2020).

Equality and Human Rights Commission (2013) *Assessment of the publication of equality objectives by English public authorities*, London: EHRC https://equalityhumanrights.com/en/publication-download/assessment-publication-equality-objectives-english-public-authorities (accessed 03/12/2020).

Equality and Human Rights Commission (2014), *The Essential Guide to the Public Sector Equality Duty:* England (and Non-Devolved Public Authorities in Scotland and Wales), London: EHRC.

Equality and Human Rights Commission (2014) *Good equality practice for employers: equality policies, equality training and monitoring: Equality Act 2010 Guidance for employers*, Vol. 7 of 7, London, EHRC.

Equality and Human Rights Commission (2019) *Tackling racial harassment: Universities challenged*, London: EHRC. https://equalityhumanrights.com/sites/default/files/tackling-racial-harassment-universities-challenged.pdf (accessed 04/09/2020).

Equality Challenge Unit (2009) The experience of black and minority ethnic staff working in higher education: Literature Review, London: ECU.

Equality Challenge Unit (2011) *The experience of black and minority ethnic staff in higher education in England,* London http://www.ecu.ac.uk/wp-content/uploads/external/experience-of-bme-staff-in-he-final-report.pdf (accessed 04/09/2020).

Equality Challenge Unit (2015) Academic flight: how to encourage black and minority ethnic academics to stay in UK higher education, London: ECU.

Equality Challenge Unit (2016) ECU's Race Equality Charter Awards Handbook January 2016, London: ECU.

Equality Challenge Unit (2018) About ECU's Athena SWAN charter, https://www.ecu.ac.uk/equality-charters/athena-swan/about-athena-swan/ (accessed 04/09/2020).

Equality Challenge Unit (2018), *Strategic Plan 2017-2021*, https://www.ecu.ac.uk/who-we-are/equality-challenge-unit-strategic-plan-2017-2021/ (accessed 04/09/2020).

Equality Challenge Unit/Higher Education Academy (2011), *Improving the degree attainment of Black and minority ethnic students*, London http://www.ecu.ac.uk/wp-content/uploads/external/improving-degree-attainment-bme.pdf (accessed 04/09/2020).

Ethnos (2006), *The Decline of Britishness: A Research Study*, Commission for Racial Equality, London http://www.ethnos.co.uk/pdfs/10 decline of britishness.pdf (accessed 09/05/2021).

Fanon, F. (1967) Black skin white masks, New York: Grove Press Inc.

Ferner, A., Almond, P. and Colling, T. (2005) *Institutional theory and the cross-national transfer of employment policy: The case for 'workforce diversity' in US multinationals*, Journal of International Business Studies, Vol.6, No. 3, pp. 304-321.

Flick, U. (2014) The SAGE Handbook of Qualitative Data Analysis, London: SAGE Publications.

Fox, S. and Stallworth, L.E. (2005) *Racial/ethnic bullying: Exploring links between bullying and racism in the US workplace*, Journal of Vocational Behavior 66, pp. 438–456.

Fryer, P. (1984) Staying Power: The History of Black People in Britain, London: Pluto Press.

Gabbatiss, J. (2017) *Brexit strongly linked to xenophobia*, The Independent https://www.independent.co.uk/news/science/brexit-prejudice-scientists-link-foreigners-immigrants-racism-xenophobia-leave-eu-a8078586.html (accessed 04/09/2020).

Gair, S. (2012) Feeling Their Stories: Contemplating Empathy, Insider/Outsider Positionings, and Enriching Qualitative Research, Qualitative Health Research, Vol. 22, no. 1, pp. 134-143.

Galletta, A. (2013) Mastering the Semi-Structured Interview and Beyond: From Research Design to Analysis and Publication, New York: New York University Press.

Garner, R (2014) More than half of BME university lecturers and staff have suffered racial abuse at work, The Independent, London. http://www.independent.co.uk/student/news/more-than-half-of-bme-university-lecturers-and-staff-have-suffered-racial-abuse-at-work-9244689.html (accessed 04/09/2020).

Gamst, G. C., Liang, C. T. & Der-Karabetian, A. (2011) *Handbook of multicultural measures*, Thousand Oaks, CA: SAGE Publications Ltd.

Gaventa, J. (2003) *Power after Lukes: An overview of theories of power since Lukes and their application to development*, Brighton: Participation Group, Institute of Development Studies.

Geare, A., Edgar, F., McAndrew, I., Harney, B., Cafferkey, K. and Dundon, T. (2014) *Exploring the ideological undercurrents of HRM: workplace values and beliefs in Ireland and New Zealand*, International Journal of Human Resource Management, Vol. 25, No. 16, pp. 2275-2294.

Gibney, E (2013) *Race discrimination in academia 'has not improved' over past 20 years*, Time Higher Education, London https://www.timeshighereducation.com/news/race-discrimination-in-academia-has-not-improved-over-past-20-years/2003102.article (accessed 04/09/2020).

Giddens, A. (1979) *Central problems in social theory: action, structure and contradiction in social analysis,* London: Macmillan.

Giddens, A. (1984), The constitution of society: Outline of the theory of structuration, Polity, Cambridge.

Giddens, A. and Sutton, P.W. (2017) Essential Concepts in Sociology, 2nd Ed., Cambridge: Polity.

Gilbert, J.A., & Ivancevich, J.M. (2000) *Valuing diversity: A tale of two organizations* Academy of Management Executive, Vol. 14(1), pp. 93–105.

Gillard, D. (2011) *Education in England: a brief history* <u>www.educationengland.org.uk/history</u> (accessed 04/09/2020).

Gillborn, D. (2002) Education and institutional racism, London: Institute of Education Press.

Gillborn, D. (2005) *Education policy as an act of white supremacy: whiteness, critical race theory and education reform,* Journal of Education Policy, Vol. 20, No. 4, pp. 485–505.

Gillborn, D. (2006) *Critical Race Theory and Education: Racism and anti-racism in educational theory and praxis,* Discourse: studies in the cultural politics of education Vol. 27, No. 1, March 2006, pp. 11-32.

Gillborn, D. (2008) Racism and education: Coincidence or conspiracy, Abingdon: Routledge.

Gilroy, P. (2008) There Ain't No Black in the Union Jack, Abingdon: Routledge.

Golec de Zavala, A.G., Guerra, R. & Simão, C. (2017) The relationship between the Brexit vote and individual predictors of prejudice: Collective narcissism, right wing authoritarianism, social dominance orientation, Frontiers in Psychology, Vol. 8, pp. 2023.

Gossett, T. (1997) [1963]. *Race: The History of an Idea in America*, New York, Oxford: Oxford University Press. p. 418.

Graves, J. L. (2001) *The Emperor's New Clothes: Biological Theories of Race at the Millennium*, Rutgers University Press.

Green, J.P. (1976) All men are created equal: some reflections on the character of the American revolution: an inaugural lecture, Oxford: Clarendon Press.

Greene, A. & Kirton, G. (2009), *Diversity Management in the UK: Organizational and Stakeholder Experiences*, Taylor & Francis, Hoboken.

Greener, I. (2002) *Agency, social theory and social policy*, Critical social policy, Vol. 22, no. 4, pp. 688-705.

Hall, R. (2017) 'I've been mistaken for the coffee lady': experiences of black female academics, London: The Guardian https://www.theguardian.com/higher-education-network/2017/nov/16/black-female-academics-ive-been-mistaken-for-the-coffee-lady (accessed 04/09/2020).

Harries, B. (2014) We need to talk about race, Sociology, Vol. 48(6), pp. 1107-1122.

Harrison, F. (1995) *The Persistent Power of "Race" in the Cultural and Political Economy of Racism*, Annual Review of Anthropology, Vol.24, pp. 47-74.

Hay, C. and Wincott, D. (1998) *Structure, agency and historical institutionalism*, Political Studies, Vol. 46, No. 5, pp. 951-957.

Healy, G., Bradley, H. & Forson, C. (2011), *Intersectional Sensibilities in Analysing Inequality Regimes in Public Sector Organizations*, Gender, work, and organization, Vol. 18, no. 5, pp. 467-487.

Healy, G., Kirton, G. and Noon, M. (2011) *Inequalities, intersectionality and equality and diversity initiatives: The conundrums and challenges of researching equality, inequalities and diversity* in *Equality, Inequalities and Diversity: Contemporary Challenges and Strategies*, (Eds. Healy, Kirton and Noon), Basingstoke: Palgrave.

Healy, G. and Okielome, F. (2007) *Equality and diversity actors: A challenge to traditional industrial relations*, Equal Opportunities International, Vol. 26(1), pp.44-65.

Heath, A. and Cheung, S.Y. (2006) *Ethnic penalties in the labour market: Employers and discrimination*, research Report No. 341, Leeds: Department of Work and Pensions.

Helms, J.E., Jernigan, M. and Mascher, J. (2005) *The meaning of race in psychology and how to change it: A methodological approach*, Vol. 60(1), pp. 27-36.

Hendricks, M. (2000) *Surveying 'race' in Shakespeare*, in *Shakespeare and Race*, eds. Alexander, C.M.S. and Wells, S. Cambridge: Cambridge University Press.

Henry, P.J. and Sears, D.O. (2002) *The Symbolic Racism 2000 Scale*, Political Psychology, Vol.23(2) pp. 253-283.

Herrnstein, R.J. and Murray, C. (1994), *The bell curve: intelligence and class structure in American life,* London; New York, N.Y: Free Press.

HESA (2020) *HESA Collections*, https://www.hesa.ac.uk/collection/c19025/index (accessed 04/05/2020).

Higher Education Academy (2008), *Ethnicity, Gender and Degree Attainment Project: Final Report*, London http://www.ecu.ac.uk/wp-content/uploads/external/ethnicity-gender-and-degree-attainment-project-final-report.pdf (accessed 04/09/2020).

Higher Education Funding Council for England (2016), *HEFCE Equality and diversity statement and objectives 2016-17*, London. (archived)

https://webarchive.nationalarchives.gov.uk/20180405132008/http://www.hefce.ac.uk/workprovide/ed/object/ (accessed 04/09/2020).

Higher Education Funding Council for England (HEFCE) (2018), *Policy guide: Equality and diversity*, London, (archived)

https://webarchive.nationalarchives.gov.uk/20180405122625/http://www.hefce.ac.uk/workprovide/ed/#d.en.90786 (accessed 04/09/2020).

Hirsch, A. (2018) Brit(ish): On race, identity and belonging, London: Penguin Books.

Holstein, J.A. and Gubrium, J.F. (1997) The Active Interview, London: SAGE.

Home Office (2018) *Hate crime England and Wales 2017/18: Statistical bulletin 20/18*, London. https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attachment_data/file/748598/hate-crime-1718-hosb2018.pdf (accessed 04/09/2020).

Hoque, K. and Noon, M. (2004) *Equal Opportunities Policy and Practice in Britain: Evaluating the 'Empty Shell' Hypothesis*, Work, Employment and Society, Volume 18, No. 3, pp. 481-506.

Horvat, E.M. and Antonio, A.L. (1999) "Hey, Those Shoes Are Out of Uniform": African American Girls in an Elite High School and the Importance of Habitus, Anthropology and Education Quarterly, Vol. 30, No. 3, pp. 317–42.

Howard, N. (2020) What changes have British universities actually made since BLM, The Tab https://thetab.com/uk/2020/10/29/investigation-what-changes-have-british-universities-actually-made-since-blm-178008 (accessed 09/11/2020).

Huppatz, K. (2009) Reworking Bourdieu's 'Capital': Feminine and Female Capitals in the Field of Paid Caring Work, Sociology, Volume 43(1), pp. 45–66.

Husband, C, (1982) 'Race' in Britain: Continuity and change, London: Hutchinson University Library.

Ibarra, H. (1993) *Personal Networks of Women and Minorities in Management: A Conceptual Framework*, The Academy of Management Review, Vol. 18, no. 1, pp. 56-87.

Ibarra, H. (1995) *Race, Opportunity, and Diversity of Social Circles in Managerial Networks*, Academy of Management Journal, Vol. 38, no. 3, pp. 673-703.

Institute for Public Policy Research, (1997) Ed. Franklin, J. Equality, London: IPPR.

Institute of Employment Studies (2005) *Equal opportunities and diversity for staff in higher education:* non-disclosure and hidden discrimination in higher education. Project 2. Report to HEFCE, SHEFC, HEFCW. Institute of Employment Studies, Brighton.

Ivancevich, J.M and Gilbert, J.A. (2000) *Diversity management: Time for a new approach*, Public Personnel Management, Vol.29(1), pp. 75-92.

Janssens, M. & Zanoni, P. (2014) *Alternative diversity management: Organizational practices fostering ethnic equality at work*, Scandinavian journal of management, Vol. 30, no. 3, pp. 317-331.

Jeffries, S. (2014) *Britain's most racist election: the story of Smethwick 50 years on,* The Guardian https://www.theguardian.com/world/2014/oct/15/britains-most-racist-election-smethwick-50-years-on (accessed on 04/09/2020).

Jenkins, R. (1997) Rethinking ethnicity: Arguments and Explorations, London: Sage Publications Ltd.

Joas, H. and Knobl, W. (2011) *Between structuralism and theory of practice: The cultural sociology of Pierre Bourdieu*, in Susen, S. & Turner, B.S. *The Legacy of Pierre Bourdieu: Critical Essays*, Anthem Press, London.

Johnson, N.F. MacDonald, D.C. and Brabazon, T.M. (2008) *Rage against the machine? Symbolic violence in e-learning supported tertiary education*, E-learning Vol.5(3).

Johnson, P. and Duberley, J. (2000) Understanding management research. London: Sage Publications.

Johnson, P. and Duberley, J. (2003) *Reflexivity in management research*, Journal of Management Studies, 40(5), pp.1279-1303.

Johnston, W.B. and Packer, A.E. (1987) Workforce 2000: Work and workers for the 21st century, Indianapolis: Hudson Institute.

Joint Negotiating Commission for Higher Education Staff (JNCHES) and ECU (2003) *Partnership for Equality: Action for Higher Education*, https://www.ucu.org.uk/media/1924/Partnership-for-equality- <a href="https://www.ucu.org.uk/m

Jones, C. (2006) *Falling between the cracks: what diversity means for black women in higher education,* Policy Futures in Education, Volume 4, No. 2, pp. 145-159.

Jones, P. and Bradbury, L. (2017) Introducing Social Theory, Newark: Polity Press.

Jordan, W.D. (1974) The White Man's Burden, Oxford: Oxford University Press.

Jordan, W.D. (1982) First impressions: Initial English confrontations with Africans, in Race in Britain: Continuity and change, ed. Husband, C. London: Hutchinson University Library.

Joseph Rowntree Foundation (2019) *Working-age employment rate by ethnicity* https://www.jrf.org.uk/data/working-age-employment-rate-ethnicity (accessed 04/09/2020).

Kaler, J. (2000) *Diversity, Equality, Morality* in Noon, M and Ogbonna, E. (eds.) *Equality, Diversity and Disadvantage in Employment*, Basingstoke: Palgrave Macmillan, pp. 51-64.

Kamoche, K., Kannan, S. & Siebers, L.Q. (2014) *Knowledge-Sharing, Control, Compliance and Symbolic Violence*, Organization studies, Vol. 35, no. 7, pp. 989-1012.

Kandola, B. (2018) Racism at work: The danger of indifference, Oxford: Pearn Kandola Publishing.

Kandola, R. and Fullerton, J. (1998) *Diversity in Action: Managing the mosaic*, Chartered Institute of Personnel and Development (CIPD), London.

Kandola, R.S., Milner, D., Banerji, N.A. and Wood, R. (1991) *Equal opportunities can damage your health*, Oxford: Pearn Kandola Downs.

Kanter, R.M. (1993) Men and Women of the Corporation New York: Basic Books.

Kelley, N., Khan, O. and Sharrock, S. (2017) *Racial prejudice in Britain today*, London: NatCen/Runnymede.

Kelly, E. and Dobbin. F. (1998) <u>How Affirmative Action Became Diversity Management: Employer</u> Response to Antidiscrimination Law, 1961-1996. American Behavioral Scientist 41 (7):960-984.

Kendi, I.X. (2017) *Stamped from the beginning: The definitive history of racist ideas in America*, London: Penguin.

Kirchberg, V. (2007) *Cultural consumption analysis: Beyond structure and agency*, Cultural Sociology, Vol. 1, No. 1, pp. 15-135.

Kim, K., Ormiston, M.E., Easterbrook, M.J. & Vignoles, V.L. (2017) *Ethnic dissimilarity predicts belonging motive frustration and reduced organizational attachment*, Group Processes & Intergroup Relations, Vol. 22, no. 1, pp. 116-138.

King, A. (2016;2000) *Thinking with Bourdieu against Bourdieu: A 'Practical' Critique of the Habitus*, Sociological theory, Vol. 18, no. 3, pp. 417-433.

Knox, R. (1850), *The Races of Men: A Fragment*, Philadelphia: Lea and Blanchard http://hdl.handle.net/2027/hvd.32044004820817 (accessed 04/09/2020).

Kolluri, S. (2019) *Reconsidering Organizational Habitus in Schools: One Neighborhood, Two Distinct Approaches to Advanced Placement*, Harvard educational review, Vol. 89, no. 1, pp. 109-131.

Ladson-Billings, G. (1998) *Just what is critical race theory and what's it doing in a nice field like education?*, International Journal of Qualitative Studies in Education, Vol. 11(1), pp. 7-24.

Law, I, Phillips, D and Turney, L (eds.) (2004) *Institutional racism in higher education* Stoke on Trent: Trentham Books.

Layder, D. (1993) New strategies in social research: An introduction and guide, Polity Press.

Leiter, M. (2012) *Analysing and Theorizing the Dynamics of Workplace Incivility Crisis*, Netherlands: Springer.

Library of Congress, *Creating the United States, Creating the Declaration of Independence*, https://www.loc.gov/exhibits/creating-the-united-states/interactives/declaration-of-independence/equal/index.html (accessed 13/07/2021).

Liff, S. and Dickens, L. (2000) *Ethics and equality: reconciling false dilemmas* in *Ethical Issues in Contemporary Human Resource Management*, Winstanley, D. and Woodall, J. (Eds) (2000) Basingstoke: Macmillan.

Lingayah, S., Khan, O. and McIntosh, K. (2018) *Making the case for racial equality: The potential and limits of 'framing'*, London: Runnymede Trust.

Littlefield, A. Lieberman, L. and Reynolds, L.T. (1982) *Redefining Race: The potential demise in physical anthropology*, Current Anthropology, Vol 23(6), pp. 641-655.

Local Government Association, *Equality Standard for Local Government*, London https://www.local.gov.uk/our-support/guidance-and-resources/equality-frameworks/equality-frameworks-local-government (accessed 04/09/2020).

Lorimer, D.A. (1996), *Race, science and culture: historical continuities and discontinuities, 1850-1914*, in *The Victorians and Race*, ed. S. West, Aldershot: Scolar Press.

Macpherson, W. (1999) *The Stephen Lawrence Inquiry*, London: The Stationery Office, https://www.gov.uk/government/uploads/system/uploads/attachment_data/file/277111/4262.pdf (accessed 04/09/2020).

Malik, K., (1996). *The meaning of race: Race, history and culture in western society.* Basingstoke: Macmillan, p. 91.

Mander, M.S. (1987) *Bourdieu, the sociology of culture and cultural studies: A critique*, European Journal of Communication, Vol. 2, pp. 427-453.

Mandla v Dowell-Lee [1982] UKHL 7.

Mapedzahama, V. and Kwansah-Aidoo, K. (2017) *Blackness as Burden? The Lived Experience of Black Africans in Australia*, SAGE Open. doi: 10.1177/2158244017720483.

Marx, K. and Engels, F. (1848) *The Communist Manifesto* (originally The Manifesto of the Communist Party) https://www.marxists.org/archive/marx/works/1848/communist-manifesto/index.htm (accessed 28/12/2020).

Maslow, A.H. (1943) A theory of human motivation, Psychological Review, Vol. 50, no. 4, pp. 370-396.

Mason, D. (2000), Race and ethnicity in modern Britain, Oxford: Oxford University Press.

Mazur, B. (2010) *Cultural Diversity in Organisational Theory and Practice*, Journal of Intercultural Management, Vol. 2, No. 2, pp. 5–15.

McAuley, J., Duberley, J. and Johnson, P. (2013) *Organization theory: challenges and perspectives*, 2nd ed. Harlow: Pearson Education.

McColgan, A. (2015) *Litigating the Public Sector Equality Duty: The Story So Far*, Oxford Journal of Legal Studies, Volume 35, Issue 3, pp. 453–485.

McConahay, J. B. (1986). *Modern racism, ambivalence, and the Modern Racism Scale* in J. F. Dovidio & S. L. Gaertner (Eds.), Prejudice, discrimination and racism (pp. 91-126). New York: Academic Press.

McCulloch, A. (2020) *Global businesses embrace Black Lives Matter movement*, Personnel Today, https://www.personneltoday.com/hr/global-businesses-embrace-black-lives-matter-movement/ (accessed 21/11/2020).

McDonough, P. M. (1998) *Structuring college opportunities: A cross-case analysis of organizational cultures, climates, and habiti,* In C. A. Torres & T. R. Mitchell (Eds.), Sociology of education: Emerging perspectives (pp. 181-210). Albany: State University of New York Press.

McDonough, P. (2016;2006) *Habitus and the practice of public service, Work, employment and society,* Vol. 20, no. 4, pp. 629-647.

McGrandle, J. (2017) *Understanding Diversity Management in the Public Sector: A Case for Contingency Theory*, International Journal of Public Administration, 40:6, pp. 526-537.

McGregor-Smith, R. (2017) *Race in the workplace: The McGregor-Smith Review*, London: Department for Business, Energy and Industrial Strategy

https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attachment_data/file/594336/race-in-workplace-mcgregor-smith-review.pdf (accessed 04/09/2020).

McGuigan, J. (2014) *The Neoliberal Self*, Journal of Current Cultural Research, Culture Unbound, Vol.6, pp.223-240.

McIntosh, P. (1997) White privilege and male privilege: a personal account of coming to see correspondences through work in Women's Studies. In R. Delgado & J. Stefancic (Eds.) Critical White Studies: Looking behind the mirror. Philadelphia: Temple University Press, pp. 291-99.

McKinsey and Company (2020) *Women in the Workplace 2020*, https://www.mckinsey.com/featured-insights/diversity-and-inclusion/women-in-the-workplace# (accessed 02/12/2020).

McLachlan, C.J. and Garcia, R.J., (2015) *Philosophy in practice? Doctoral struggles with ontology and subjectivity in qualitative interviewing*, Management Learning, 46(2), pp.195-210.

McLeod, P. L., Lobel, S. A., & Cox, T. H. (1996) *Ethnic Diversity and Creativity in Small Groups*, Small Group Research, 27(2), 248–264.

Mellor, D., Bynon, G., Maller, J., Cleary, F., Hamilton, A. and Watson, L. (2001) *The perception of racism in ambiguous scenarios*, Journal of ethnic and migration studies, Vol. 27(3), pp. 473-488.

Metcalf, H, Rolfe, H., Stevens, P. and Weale, M. (2005) *Recruitment and Retention of Academic Staff in Higher Education*, Nottingham: National Institute of Economic and Social Research.

Migration Observatory (2020) *Briefing - UK Public Opinion toward Immigration: Overall Attitudes and Level of concern*, University of Oxford, https://migrationobservatory.ox.ac.uk/resources/briefings/uk-public-opinion-toward-immigration-overall-attitudes-and-level-of-concern/ (accessed 04/09/2020).

Modood, T., Berthoud, R., Lakey, J., Nazroo, J., Smith, P., Virdee, S. and Beishon, S. (1997) *Ethnic Minorities in Britain: Diversity and Disadvantage*. London: Policy Studies Institute.

Modood, T. Berthoud, R. and Nazroo, J. (2002) 'Race', Racism and Ethnicity: A response to Ken Smith, Sociology, Vol. 36(2), pp. 419-427.

Mohamed, Z., Newton, J.M. & McKenna, L. (2014), *Belongingness in the workplace: a study of Malaysian nurses' experiences: Nurses and sense of belonging in the workplace*, International nursing review, Vol. 61, no. 1, pp. 124-130.

Mohdin, A. (2021) *UK universities are institutionally racist, says leading vice-chancellor*, The Guardian, https://amp-theguardian-

com.cdn.ampproject.org/c/s/amp.theguardian.com/education/2021/apr/28/uk-universities-are-institutionally-racist-says-leading-vice-chancellor (accessed 08/05/2021).

Monbiot, G. (2016) *Neoliberalism – the ideology at the root of all our problems*, London: The Guardian https://www.theguardian.com/books/2016/apr/15/neoliberalism-ideology-problem-george-monbiot

Montague, A. (1972) Statement on Race, Oxford: Oxford University Press.

Moore, Jerry D. (2009). "Franz Boas: Culture in Context". Visions of Culture: An Introduction to Anthropological Theories and Theorists, Walnut Creek, California: Altamira. pp. 33–46.

Moore, K.S. (2005) What's Class Got to Do With It? Community Development and Racial Identity, Journal of Urban Affairs, Vol. 27, Issue 4, pp. 437-451.

Morrison, K. (2005) *Structuration theory, habitus and complexity theory: elective affinities or old wine in new bottles?*, British Journal of Sociology of Education, Vol. 26, no. 3, pp. 311-326.

Murugesu, J.A. (2019) *Ethnic minority academics get less UK research funding*, New Scientist https://www.newscientist.com/article/2222994-ethnic-minority-academics-get-less-uk-research-funding/ (accessed 09/11/2019).

National Archives, Demobilisation in Britain, 1918-20

http://www.nationalarchives.gov.uk/pathways/firstworldwar/spotlights/demobilisation.htm (accessed 27/03/2018).

National Archives, *Cabinet Conclusion 2, Racial Discrimination*, 21st December 1967 http://filestore.nationalarchives.gov.uk/pdfs/small/cab-128-42-cc-67-74-74.pdf (accessed 04/09/2020).

Nelson, T.D. (2009) *Handbook of Prejudice, Stereotyping, and Discrimination,* Taylor & Francis, Hoboken.

New York Times (2021) *Derek Chauvin Trial: Chauvin found guilty of murdering George Floyd,* https://www.nytimes.com/live/2021/04/20/us/derek-chauvin-verdict-george-floyd (accessed 25/04/2021).

Ng, E.S.W. (2008) Why organisations choose to manage diversity? Toward a leadership-based theoretical framework, Human Resource Development Review, Vol. 7, No. 1, pp. 58-78.

NHS England, Workforce Race Equality Standard (2018) https://www.england.nhs.uk/about/equality/equality-hub/equal

https://www.england.nhs.uk/about/equality/equality-hub/equality-standard/ (accessed 04/09/2020).

Noon, M. (2007) *The fatal flaws of diversity and the business case for ethnic minorities*, Work, Employment and Society, 21(4), pp. 773–784.

Noon, M. (2010) *The shackled runner: time to rethink positive discrimination?* Work, Employment and Society, Vol 24, Issue 4, pp. 728-739.

Noon, M. (2017) *Pointless diversity training: Unconscious bias, new racism and agency,* Work, employment and society, pp. 1-12.

Noon, M. (2018) *Pointless diversity training: Unconscious bias, new racism and agency*, Work, Employment and Society, Vol. 32(1), pp. 198-209.

Noon, M., Healy, G., Forson, C. and Okielome, F. (2013) *The Equality Effects of the 'Hyper-Formalization'* of selection, British Journal of Management, Vol. 24, pp. 333-346.

Offermann, L. R., Basford, T.E., Graebner, R., Jaffer, S., De Graaf, S.B. and Kaminsky, S.E. (2014), *See No Evil: Color Blindness and Perceptions of Subtle Racial Discrimination in the Workplace*, Cultural Diversity and Ethnic Minority Psychology, Vol. 20, No. 4, pp. 499–507.

Office of the High Commissioner Human Rights (OHCHR) (2021), Statement: *UN experts condemn UK Commission on Race and Ethnic Disparities report*,

https://www.ohchr.org/EN/NewsEvents/Pages/DisplayNews.aspx?NewsID=27004&LangID=E (accessed 25/04/2021).

Office for National Statistics (2011), *Census for England and Wales 2011*, http://www.ons.gov.uk/ons/rel/census/2011-census/key-statistics-for-local-authorities-in-england-and-wales.html#tab---Ethnic-group (accessed 04/09/2020).

Office for National Statistics (2019) *Labour market status by ethnicity* https://www.ons.gov.uk/employmentandlabourmarket/peopleinwork/employmentandemployeetypes/datasets/labourmarketstatusbyethnicgroupa09 (accessed 04/09/2020).

Office for Students (undated) *Equality and diversity objectives for the Office for Students 2018-2022*, https://www.officeforstudents.org.uk/media/0d30cb3f-2e49-4541-ad57-f5823a4d3893/ofs-e-and-d-statement-and-objectives.pdf (accessed 04/09/2020).

Oloyede, F.D., Christoffersen, A. and Cornish, T. (2021) *Race Equality Charter Review*, London: Advance HE https://s3.eu-west-2.amazonaws.com/assets.creode.advancehe-document-manager/documents/advance-he/AdvHE Race%20Equality%20Charter%20Review 1615534497.pdf (accessed 11/07/2021).

Olusoga, D. (2017) Black and British: a forgotten history, London: Pan Books.

Oswick, C. (2011) The social construction of diversity, equality and inclusion: An exploration of academic and public discourses, in Equality, Inequalities and Diversity: Contemporary Challenges and Strategies, (Eds. Healy, Kirton and Noon), Basingstoke: Palgrave.

Oxford English Dictionary (2021)

https://www.oed.com/view/Entry/91004?redirectedFrom=identity#eid (accessed 12/07/2021).

Özbilgin, M. and Tatli, A. (2005) Book Review Essay: *Understanding Bourdieu's contribution to organization and management studies*, The Academy of Management Review, Vol. 30, No. 4, pp. 855-869.

Özbilgin, M. and Tatli, A. (2011) *Mapping out the field of equality and diversity: Rise of individualism and voluntarism*, Human Relations, Vol. 64, No. 9, pp. 1229-1253.

Özbilgin, M., Tatli, A., Ipek, G. & Sameer, M. (2016) *Four approaches to accounting for diversity in global organisations*, Critical perspectives on accounting, Vol. 35, pp. 88-99.

Parker Review Commission (2020) Ethnic Diversity Enriching Business Leadership: An update report from The Parker Review, London: EY https://assets.ey.com/content/dam/ey-sites/ey-com/en_uk/news/2020/02/ey-parker-review-2020-report-final.pdf (accessed 02/12/2020).

Patton, M.Q. (2015) *Qualitative research & evaluation methods: integrating theory and practice,* Fourth edn, California: SAGE Publications.

Payne, J.C. (2015) *Investigating the Role of Cultural Capital and Organisational Habitus in Architectural Education: A Case Study Approach*, International Journal of Art & Design Education 34(1): 9-24.

Pearce, S. (2019) 'It was the small things': Using the concept of racial microaggressions as a tool for talking to new teachers about racism, Teaching and Teachers Education, Vol. 79, pp.83-92.

Percival, R. (2020) Winston Churchill's legacy review over BLM claims leaves supporters 'speechless with rage', Express https://www.express.co.uk/news/uk/1357614/winston-churchill-legacy-imperial-war-museum-black-lives-matter (accessed 09/11/2020).

Personnel Today (2020) *Civil Service unconscious bias training to be scrapped*, London. https://www.personneltoday.com/hr/civil-service-unconscious-bias-training-to-be-scrapped/ (accessed 02/01/2021).

Phillips, D. (2012) *Unsettling Spaces: Higher Education Institutions in the United Kingdom*, in As the World Turns: Implications of Global Shifts in Higher Education for Theory, Research and Practice, 03/2012, Emerald Group Publishing Limited, pp. 421-432.

Philpot, T. (1999) (Ed.), *Political Correctness and Social Work*, Choice in Welfare No. 54, IEA Health and Welfare Unit, London.

Pilkington, A. (2011) *Institutional racism in the academy: A case study*, Stoke on Trent: Trentham Books.

Pilkington, A. (2013) *The interacting dynamics of institutional racism in higher education*. Race, Ethnicity and Education. 16(2), pp. 225-245.

Public Health England (2020) Beyond the data: Understanding the impact of COVID-19 on BAME groups, Public Health England: London

https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attachment_data/file/892376/COVID_stakeholder_engagement_synthesis_beyond_the_data.pdf (accessed 21/11/2020).

Randle, K. Forson, C. and Calveley, M. (2014) *Towards a Bourdeusian analysis of the social composition of the UK film and television workforce*, Work, Employment and Society, Vol. 29(4), pp. 590-606.

Reay, D. (2004) 'It's all becoming a habitus': beyond the habitual use of habitus in educational research, British Journal of Sociology of Education, Vol. 25, No. 4, pp. 431-444.

Regents of the University of California v Bakke [1978], US Supreme Court, Legal Information Institute, https://www.law.cornell.edu/supremecourt/text/438/265 (accessed 24/01/2022)

Riley, R., Metcalf, H. and Forth, J. (2013) *The business case for equal opportunities*, Industrial Relations Journal, Vol. 44, No. 3, pp. 216-239.

Rollock, N. (2011) *Unspoken rules of engagement: Navigating racial microaggressions in the academic terrain,* International Journal of Qualitative Studies in Education, pp. 1-16.

Rollock, N. (2012) *The invisibility of race: Intersectional reflections of the liminal space of alterity*, Race, Ethnicity and Education, Vol. 15(1), pp. 65-84.

Rollock, N. (2019) *Staying Power: The career experiences and strategies of UK Black female professors,* University and College Union, London.

Rollock, N. and Gillborn, D. (2011) *Critical Race Theory (CRT)*, British Educational Research Association online resource. https://www.bera.ac.uk/researchers-resources/publications/critical-race-theory-crt (accessed 04/09/2020).

Rowe, M.P. (1990), Barriers to equality: The power of subtle discrimination to maintain unequal opportunity, Employee Responsibilities and Rights Journal, 3, pp. 153–163.

Saeed, A. Rae, E. Neil, R. Connell-Hall, V. and Munro, F. (2019) *To BAME or not to BAME: the problem with racial terminology in the civil service*, Civil Service World https://www.civilserviceworld.com/news/article/to-bame-or-not-to-bame-the-problem-with-racial-terminology-in-the-civil-service (accessed 04/09/2020).

Samson, J. (2005), *Race and Empire*, Harlow: Pearson Education.

Sandford, S. (2018) Kant, race, and natural history, Philosophy & Social Criticism, 44(9), pp. 950–977.

Sandhu, R. (2018) *Should BAME be ditched as a term for black, Asian and minority ethnic people?* https://www.bbc.co.uk/news/uk-politics-43831279 (accessed 04/09/2020).

Sayer, A. (1992) Method in social research: A realist approach, 2nd Edition, London: Routledge.

Scarman, L. (1981) *The Brixton Disorders 10–12 April 1981* (The Scarman Report), Cmnd 8427. London: HMSO.

Schwabenland, C. and Tomlinson, F. (2015) *Shadows and light: Diversity management as phantasmagoria*, Human Relations, 68(12), pp. 1913–1936.

Searle, J.R. (1995) The construction of social reality, New York: Free Press.

Sears, D.O. and Henry, P.J. (2003) *The origins of symbolic racism*, Journal of Personality and Social Psychology, Vol. 85, No. 2, pp. 259–275.

Shelton, J.E. (2008) *The investment in blackness hypothesis: Toward Greater Understanding of Who Teaches What during Racial Socialization*, Du Bois Review, Vol. 5, no. 2, pp. 235-257.

Shepherd, J. (2011) *14,000 British professors – but only 50 are black*, The Guardian, London https://www.theguardian.com/education/2011/may/27/only-50-black-british-professors (accessed 04/09/2020).

Shultz, D.L. (1993) *To Reclaim a Legacy of Diversity: Analyzing the "Political Correctness" Debates in Higher Education*, National Council for Research on Women, New York, NY.

Sian, K. (2017) *Being black in a white world: Understanding racism in British universities*, Papeles del CEIC, Vol. 2017, no. 2, pp. 1-26.

Sian, K. (2019) Extent of institutional racism in British universities revealed through hidden stories, The Conversation (27/06/2019) https://theconversation.com/extent-of-institutional-racism-in-british-universities-revealed-through-hidden-stories-118097 (accessed 04/09/2020).

Singh, H. (2014) Conference Report – *Critical philosophy of race: here and now*, Radical Philosophy 187, Sept/Oct 2014.

Skopeliti, C. (2020) *UK supermarkets unite after Sainsbury's advert prompts racist backlash*, London: The Guardian, https://www.theguardian.com/world/2020/nov/28/uk-supermarkets-unite-after-sainsburys-advert-prompts-racist-backlash)accessed 03/12/2020).

Smith, K. (2002) Some critical observations on the use of the concept of 'ethnicity' in Modood et al., Ethnic Minorities in Britain, Sociology, Vol. 36(2), pp. 399-417.

Smith, V. (Eds) (2013) Sociology of Work: An Encyclopaedia, London: Sage.

Smith, J.A. Flower, P. and Larkin, M. (2009), *Interpretative Phenomenological Analysis: Theory, Method and Research*, Qualitative research in psychology, Vol. 6, no. 4, pp. 346-347.

Smulowitz, S., Becerra, M. & Mayo, M. (2019) *Racial diversity and its asymmetry within and across hierarchical levels: The effects on financial performance*, Human relations (New York), Vol. 72, no. 10, pp. 1671-1696.

Solanke, I. (2018) *The stigma of being Black in Britain*, Identities: Global Studies in Culture and Power, Vol. 25, Issue 1: Symposium: Stuart Hall's Familiar Stranger: A life between two islands, pp. 49-54.

Solomos, J. (1999) *Social Research and the Stephen Lawrence Inquiry,* Sociological Research Online, vol. 4, no. 1, http://www.socresonline.org.uk/4/lawrence/solomos.html (accessed 04/09/2020).

Souhami, A. (2014) *Institutional racism and police reform: an empirical critique*, Policing & Society, 2014, Vol. 24, No. 1, p. 121.

Stainback, K. & Irvin, M. (2012) Workplace racial composition, perceived discrimination, and organizational attachment, Social Science Research, Vol. 41, no. 3, pp. 657-670.

Stanford Encyclopaedia of Philosophy, *Equality* http://plato.stanford.edu/entries/equality/#DefCon (accessed 04/09/2020).

Steinbugler, A. (2015) I'm black and I'll always be that way: black identities through the lens of interracial intimacy, Ethnic and Racial Studies, 38(10), pp. 1690-1706.

Stone, A. (2017) Hegel and Colonialism, Hegel Bulletin, pp. 1-24.

Sue, D.W., Capodilupo, C.M., Torino, G.C., Bucceri, J.M., Holder, A.M.B., Nadal, K.L. & Esquilin, M. (2007), *Racial Microaggressions in Everyday Life: Implications for Clinical Practice*, American Psychologist, vol. 62, no. 4, pp. 271-286.

Sue, D.W., Capodilupo, C.M. and Holder, A.M.B. (2008) *Racial microaggressions in the life experience of Black Americans*, Professional Psychology: Research and Practice, Vol. 39, No. 3, pp. 329-336.

Swan, A. (2009) *Putting words in our mouths: Diversity training as heteroglossic organisational spaces,* In Özbilgin, M. (ed.) *Equality, Diversity and Inclusion at Work,* Cheltenham: Edward Elgar, pp. 308-321.

Syed, J. & Özbilgin, M. (2009), *A relational framework for international transfer of diversity management practices*, The International Journal of Human Resource Management, Vol. 20, No. 12, pp. 2435-2453.

Tatli, A. (2011) A Multi-layered Exploration of the Diversity Management Field: Diversity Discourses, Practices and Practitioners in the UK, British journal of management, Vol. 22, no. 2, pp. 238-253.

Tatli, A. and Alasia, S. (2011) *Resources and constraints of diversity and equality officers: Theoretical and practitioner reflections*, in *Equality, Inequalities and Diversity: Contemporary Challenges and Strategies*, (Eds. Healy, Kirton and Noon), Basingstoke: Palgrave.

Tatli, A., Nicolopoulou, K., Özbilgin, M., Karatas-Ozkan, M. & Öztürk, M.B. (2015), *Questioning impact: interconnection between extra-organizational resources and agency of equality and diversity officers*, The International Journal of Human Resource Management, vol. 26, no. 9, pp. 1243-1258.

Tatli, A. & Özbilgin, M.F. (2009) *Understanding diversity managers' role in organizational change: Towards a conceptual framework*, Canadian Journal of Administrative Sciences, Vol. 26, no. 3, pp. 244-258.

Thapor-Björket, S. Samelius, L. and Sanghera, G.S. (2016) *Exploring symbolic violence in the everyday: misrecognition, condescension, consent and complicity,* Feminist Review, 112, pp.144-162.

The Conversation (2021) (Kaur, R. and Hague, G.M.) *Race commission report: the rights and wrongs*, https://theconversation.com/race-commission-report-the-rights-and-wrongs-158316 (accessed 25/04/2021).

The National Committee of Inquiry into Higher Education (Dearing Committee) (1997), London: Stationery Office.

The Week (2020) Academics claim political correctness has gone mad: New survey reveals fears for freedom of expression at universities https://www.theweek.co.uk/105685/academics-claim-political-correctness-has-gone-mad (accessed 17/11/2020)

Thomas, D.R. (2016) A General Inductive Approach for Analyzing Qualitative Evaluation Data, The American journal of evaluation, Vol. 27, no. 2, pp. 237-246.

Times Higher Education (2020) We are at a crucial moment to address racial justice in higher education: An open letter from academics, university staff and students to sector leaders (6 July 2020) https://www.timeshighereducation.com/letters/we-are-crucial-moment-address-racial-justice-highereducation (accessed 09/11/2020).

Treaty on the Functioning of the European Union (TFEU) 2009, Article 157, https://eur-lex.europa.eu/LexUriServ/LexUriServ.do?uri=CELEX:12012E/TXT:EN:PDF (accessed 04/09/2020).

Trenerry, B and Paradies, Y. (2012) *Organizational Assessment: An Overlooked Approach to Managing Diversity and Addressing Racism in The Workplace*, Journal of Diversity Management, Vol. 7:1, pp. 11-26.

Turner, C. (2019) *Top universities will need to accept fewer middle class students to meet diversity targets, regulator admits*, The Telegraph, https://www.telegraph.co.uk/education/2019/02/21/top-universities-will-need-accept-fewer-middle-class-students/ (accessed 04/09/2020).

Turney, L., Law, I. and Phillips, D. (2002) *Institutional racism in higher education. Building the anti-racist university: a toolkit*. Centre for Ethnicity and Racism Studies, University of Leeds.

UNESCO, (1969) Four statements on the race question, Paris: UNESCO http://unesdoc.unesco.org/images/0012/001229/122962eo.pdf (accessed 04/09/2020).

University and College Union (2016) *The experiences of Black and Minority Ethnic staff in further and higher education*, London https://www.ucu.org.uk/media/7861/The-experiences-of-black-and-minority-ethnic-staff-in-further-and-higher-education-Feb-16/pdf/BME_survey_report_Feb161.pdf (accessed 04/09/2020).

University and College Union (2017) *Witness: The lived experiences of UCU Black Members*, London https://www.ucu.org.uk/media/8476/Witness---voice-of-UCU-black-members-Feb-17/pdf/WITNESS - Voice of UCU Black Members - Feb 2017.pdf (accessed 04/09/2020).

Universities and Colleges Employers Association (UCEA) (2018) Caught at the crossroads: An intersectional approach to gender and ethnicity pay gaps, London: UCEA, https://uceastorage.blob.core.windows.net/ucea/download.cfm/docid/caught at the crossroads.p df (accessed 04/09/2020).

Universities and Colleges Employers Association (UCEA) (2019) *Higher Education Workforce Report 2019*, London: UCEA, https://www.ucea.ac.uk/library/publications/he-workforce-report-2019/ (accessed 04/09/2020).

Universities UK, (2017) Report of the review group on UK higher education sector agencies 'Bell Review', London http://www.universitiesuk.ac.uk/policy-and-analysis/reports/Pages/report-of-the-review-group-on-uk-higher-education-sector-agencies.aspx (accessed 04/09/2020).

Universities UK (2020) *Tackling racial harassment in higher education: Executive Summary*, UUK, https://universitiesuk.ac.uk/policy-and-analysis/reports/Documents/2020/tackling-racial-harassment-in-higher-education-exec.pdf (accessed 09/05/2021).

Van Buren III, H.J. (2020) *The value of including employees: a pluralist perspective on sustainable HRM*, Employee Relations, Vol. ahead-of-print No. ahead-of-print. https://doi-org.ezproxy.herts.ac.uk/10.1108/ER-01-2019-0041

Van Dijk, T.A. (2002) *Discourse and racism*, in Goldberg, D and Solomos, J. (Eds.) *The Blackwell Companion to Racial and Ethnic Studies*, Oxford: Blackwell, pp. 145-159.

Visweswaran, K. (1998) *Race and the Culture of Anthropology*, American Anthropologist, 100(1), pp. 70-83.

Wall, T. (2020) *The day Bristol dumped its slave trader in the docks and a nation began to search its soul*, The Observer, London. https://www.theguardian.com/uk-news/2020/jun/14/the-day-bristol-dumped-its-hated-slave-trader-in-the-docks-and-a-nation-began-to-search-its-soul

Walvin, J. (1982) *Black caricature: the roots of racialism*, in *Race in Britain: Continuity and change*, ed. Husband, C. London: Hutchinson University Library.

Waterson, J. (2020) Ofcom dismisses complaints over BLM dance on Britain's Got Talent, The Guardian, London. https://www.theguardian.com/tv-and-radio/2020/sep/17/ofcom-dismisses-complaints-over-blm-dance-on-britains-got-talent

Weaver, M. (2018) *Hate crime surge linked to Brexit and 2017 terrorist attacks*, The Guardian, London. https://www.theguardian.com/society/2018/oct/16/hate-crime-brexit-terrorist-attacks-england-wales (accessed 04/09/2020).

Webb, J., Schirato, T. and Danaher, G. (2004) *Understanding Bourdieu*, London: Sage.

Whitchurch, C. (2006) Who do they think they are? The changing identities of professional administrators and managers in UK higher education, Journal of Higher Education Policy and Management, Vol. 28, No. 2, pp. 159-171.

Wiggins-Romesburg, C.A. & Githens, R.P. (2018) *The Psychology of Diversity Resistance and Integration*, Human Resource Development Review, Vol. 17, no. 2, pp. 179-198.

Wight, C. (2003) *The Agent–Structure Problem and Institutional Racism*, Political Studies, Vol.51, pp. 706–721.

Williams, J, Cocking, J. and Davies, L. (1989) *Words or Deeds? A review of equal opportunities policies in higher education*, London: Commission for Racial Equality.

Wilson, K.Y. (2010) An analysis of bias in supervisor narrative comments in performance appraisal, Human Relations, Vol. 63(2), pp. 1903-1933.

Wilson, K. Y. and Jones, R. G. (2008) *Reducing Job-Irrelevant Bias in Performance Appraisals: Compliance and beyond, Journal of General Management*, Vol. 34(2), pp. 57–70.

Winant, H. (2000) Race and Race Theory, Annual Review of Sociology, Vol. 26, pp. 169-185.

Wright, J. (2020) Bristol girls' school named in honour of slave trader Edward Colston rebrands itself Montpelier High School...169 years after being built with his money, Daily Mail (Mail Online)

https://www.dailymail.co.uk/news/article-8921833/Black-Lives-Matter-Bristol-school-named-honour-slave-trader-Edward-Colston-rebrands-itself.html (accessed 09/11/2020).

Yemm, R. (2019) *Immigration, race and local media: Smethwick and the 1964 general election*, Contemporary British History, Vol. 33, no. 1, pp. 98-122.

Zamudio, M.M. and Rios, F. (2006) *Traditional to liberal racism: Living racism in the everyday*, Sociological Perspectives, Vol. 49, No. 4, pp. 483-501.

Zuberi, T. (2013) *Racial Domination and the Evolution of Racial Classification*, in *Race*, Gender, Sexuality and Social Class: Dimensions of Inequality, ed. S.J Ferguson, London: Sage Publications.

Appendix 1 Office for Students (OfS) Equality Objectives 2018-2022

Objective 1 – The OfS will dvelop, implement and consult on our equality and diversity objectives, evidence base, impact assessments and action plan to ensure successful implementation of our PSED.

Objective 2 – The OfS will conduct and publish rigorous and influential analysis, research and insight into the E&D (including the protected characteristics and socioeconomic disadvantage) issues across the student lifecycle.

Objective 3 – The OfS will challenge the sector to significantly reduce gaps in access, success and progression for students from all backgrounds and identities and across all disciplines.

Objective 4 – The OfS will work to address the risk of some students not receiving a high-quality higher education experience.

Objective 5 – The OfS will work to reduce the risk that some students are prevented from maximising their outcomes through their higher education experience and therefore not maximise their potential in terms of employment or further study.

Objective 6 – Fostering inclusive leadership and an inclusive and open culture.

Objective 7 – Supporting staff to build diversity and inclusion into their work.

Objective 8 – Behaving as an inclusive employer which attracts and retains the widest pool of talent where all staff have the opportunity to unlock their potential.

(Office for Students, undated).

Appendix 2 Advance HE Race Equality Charter principles

- Racial inequalities are a significant issue within higher education. Racial inequalities are not necessarily overt, isolated incidents. Racism is an everyday facet of UK society and racial inequalities manifest themselves in everyday situations, processes and behaviours.
- UK higher education cannot reach its full potential unless it can benefit from the talents of the whole population and until individuals from all ethnic backgrounds can benefit equally from the opportunities it affords.
- In developing solutions to racial inequalities, it is important that they are aimed at achieving long-term institutional culture change, avoiding a deficit model where solutions are aimed at changing the individual.
- Minority ethnic staff and students are not a homogenous group. People from different ethnic backgrounds have different experiences of and outcomes from/within higher education, and that complexity needs to be considered in analysing data and developing actions.
- All individuals have multiple identities, and the intersection of those different identities should be considered wherever possible (Advance HE, 2020).

Appendix 3 Call for expression of interest

On 13 April 2016 at 13:40, Rodriguez, Min < m.rodriguez@herts.ac.uk> wrote: APOLOGIES FOR CROSS POSTING

Please forward to anyone you think might be interested.

Doctoral research project: Race equality and black staff in UK higher education institutions: An exploration of barriers to progress.

Call for expressions of interest to take part in this study.

Protocol number: cBUS/PGR/UH/02322

A diverse workforce is crucial to innovation and creativity in organisations (Özbiligin and Tatli, 2011), yet evidence continues to show that Black and Minority Ethnic (BME) staff are under-represented in UK higher education (ECU, 2015). This is despite the year-on-year increase in the ethnic diversity of the student population in higher education. Since 2003/4 the proportion of BME students has increased from 14.9% to 20.2% in 2013/14. In contrast, the proportion of BME staff in higher education institutions has only increased from 4.8% in 2003/4 to 6.7% in 2013/14 (ECU,2015).

The slow diversification of the higher education workforce has attracted studies and media attention highlighting the continued recruitment of BME staff at lower levels of the academy, yet the outcomes differ according to different ethnic groups (Singh and Kwahli, 2015; ECU, 2015). Latest statistics show that of UK black academic staff, 4.5% are professors, compared to 11.2% of all white academic staff (ECU, 2015). In relation to pay, 19.7% of white academic staff earned in the top academic pay spine range of £57,032 and above compared to only 8.9% of black UK academic staff in this pay spine.

This study aims to understand the barriers to progress of race equality in higher education and more specifically how this continues to adversely affect black staff. I would particularly welcome expressions of interest to participate in the research from Black staff working in higher education (professional/technical and academic), as well as equality and diversity practitioners, HR managers, senior managers (with line management responsibility) and trade union representatives of any ethnic background with experience in developing or implementing race equality policies.

Involvement in this study will be through confidential one-to-one interviews and will begin from May 2016 through to July 2017. If you are interested in taking part in this important study, please contact Min Rodriguez m.rodriguez@herts.ac.uk for more details.

Best wishes

Min Rodriguez

Head of Equality
Equality Office
University of Hertfordshire
LB161, College Lane
Hatfield
Hertfordshire
AL10 9AB

Question What attracted you to higher education?	Prompt Was diversity a key attraction? Were you inspired in some way?	Reason for question Icebreaker - To establish whether the interviewee made a conscious decision to enter HE or whether it was the role.
Do you feel a sense of belonging to your institution?	Do you feel part of your institutional community?	Literature has suggested a lack of belonging and I want to know if this is a live issue for a range of black staff.
Do you think the terminology used 'race/ethnicity' helps or hinders the advancement of equality of opportunity in this area?	Any suggestions how this could change in the future?	Lots of debate about terminology and I want to establish whether this is relevant in practice.
What do you understand racism to be/look like in today's institutions?	Have you experienced it/ or witnessed it happen whilst in HE?	Literature suggests that racism is not always overt and has shape-shifted over time. In practice I have dealt with staff that have complained of bullying, harassment and discrimination, but are not explicit that the treatment is linked to race/ethnicity. I want to understand whether this is due to overall understanding of racism or if there is a fear behind claims of this type.
How committed is your own institution in dealing with matters of race/ethnicity?	Does it engage in any activity to promote race/ethnicity? Have they been effective? Why do think this is?	I want to understand whether in practice race equality policies are seen as 'empty shell' or if there is activity to reinforce the commitment of the institution.
=5Does the way you feel about how your institution deals with race/ethnicity affect your sense of belonging to it or mindful of your identity within the institution?	Are you conscious of your own ethnic identity in the workplace?	I'd like to understand whether issues in the workplace affect the individual's sense of belonging and/or whether they are mindful that their identity affects their place in the institution.

Are you aware of the term 'institutional racism'? Interviewee will be shown a card with the definition of this, taken from the Macpherson report.	Do you think institutional racism exists in your institution based on this definition?	This aspect has not been addressed explicitly in HE. I want to understand people's perception about this and whether it affects their level of engagement at work.
What impact has the leadership team had on the advancement of race equality in your institution?	Do you feel there has been sufficient resource? Do you feel engaged with the process?	I would like to get a sense of how much institutional leadership are involved in advancing race equality and whether it is effective in directing implementation and in engaging black staff.
Who in your institution is responsible for implementing race equality policies/initiatives?	What activity has taken place and how often? Is this effective and are staff engaged? Who do you think should lead on race equality?	I wish to understand where the responsibility lies in the institution and what works and what doesn't in order to affect future practice.
Do you think the issues around low representation, career progression/development of Black staff are taken seriously within your institution?	What initiatives are taking place to address it? Are these initiatives effective?	I would like to build an understanding of the perceptions of black staff in relation to their own representation and career enhancement within the sector and what might be done about it.
What are your views on the value of BME Staff Networks?	Are you part of this network? What do you/your institution get from this network? Does your institution support it or obtain feedback from it?	Staff networks are difficult to set up and maintain and often are stand-alone entities that do not feed into institutional policy yet there is an appetite for them. I want to understand why people do/don't use them and how they might be used better to advance race equality.
What do you think has/hasn't worked in advancing race equality in higher education?	If you could change some of those aspects what would they be?	In the interest of changing black staff experience and the way institutions address race equality — I want to understand what has/hasn't worked and gain an opinion to what could change.

Question Tell me about your role and how you are involved with race equality in higher education?	Prompt	Reason for question Icebreaker To establish the key aspects of their role and the relevance to race equality
To what extent do you feel the terms 'race/ethnicity' help or hinder the advancement of equality of opportunity in this area?	Any suggestions how this could change in the future?	Lots of debate about terminology and I want to establish how practitioners feel this might help work in this area.
What is your understanding of what racism looks like in your/today's institutions?	Have you experienced it/ or witnessed it happen whilst in HE? What actions/initiatives have you taken or been involved in to address it?	Literature suggests that racism is not always overt and has shape-shifted over time. In practice I have dealt with staff that have complained of bullying, harassment and discrimination, but are not explicit that the treatment is linked to race/ethnicity. I want to understand whether this is due to overall understanding of racism or if there is a fear behind claims of this type.
How confident are you in dealing with matters of race?	How might knowledge and confidence levels be approached amongst managers and practitioners in the sector?	I want to understand the level of knowledge and confidence about race in higher education.
As an advisor/manager in the sector, what opportunities or challenges do you face in dealing with matters of race/ethnicity? Locally or across the sector.	Do you experience difficulties in advancing the agenda? Are staff engaged with race?	It is often difficult to engage dominant groups in discussions about race, however, are there things we can learn from practitioner/manager experiences?
How committed do you think your own organisation is to dealing with matters of race/ethnicity?	Does it engage in any activity to promote race/ethnicity? Have they been effective? Why do think this is? How does this transfer to your sector stakeholders? (non-HEI participant)	I want to understand whether in practice race equality policies are seen as 'empty shell' or if there is activity to reinforce the commitment of the institution.
Are you aware of the term 'institutional racism'?	Do you think institutional racism exists in higher education based on this definition?	This aspect has not been addressed explicitly in HE. I want to understand people's perception about this and

Interviewee will be shown a card with the definition of this, taken from the Macpherson report.	Do you think institutional racism exists in your organisation based on this definition?	whether it affects their level of engagement at work.
Do you feel that there is sufficient focus/resource in the area of race equality in the HE sector?	Do you feel there has been sufficient resource? Do you feel engaged with the process?	I would like to get a sense of how much institutional leadership are involved in advancing race equality and whether it is effective in directing implementation and in engaging black staff.
Do you think the issues around low representation, career progression/development of Black staff are taken seriously within your institution?	What initiatives are taking place to address it? Are these initiatives effective?	I would like to build an understanding of the perceptions of black staff in relation to their own representation and career enhancement within the sector and what might be done about it.
What are your views on the value of BME Staff Networks?	Are you part of this network? What do you/your institution get from this network? Does your institution support it or obtain feedback from it?	Staff networks are difficult to set up and maintain and often are stand-alone entities that do not feed into institutional policy yet there is an appetite for them. I want to understand why people do/don't use them and how they might be used better to advance race equality.
What do you think has/hasn't worked in advancing race equality in higher education?	If you could change some of those aspects what would they be?	In the interest of changing black staff experience and the way institutions address race equality — I want to understand what has/hasn't worked and gain an opinion to what could change.